Technische Universität München TUM School of Computation, Information and Technology



Duality Methods for Dynamic Portfolio Optimization with Constraints

Michel Kschonnek

Vollständiger Abdruck der von der TUM School of Computation, Information and Technology

der Technischen Universität München zur Erlangung eines

Doktors der Naturwissenschaften (Dr. rer. nat.)

genehmigten Dissertation.

Vorsitz: Prof. Dr. Elisabeth Ullmann

Prüfer*innen der Dissertation:

- 1. Prof. Dr. Rudi Zagst
- 2. Prof. Dr. Marcos Escobar-Anel
- 3. Prof. Dr. Gaurav Khemka

Die Dissertation wurde am 26.06.2023 bei der Technischen Universität München eingereicht und durch die TUM School of Computation, Information and Technology am 06.09.2023 angenommen.

Abstract

This thesis studies the dynamic portfolio optimization problem of an investor who seeks to maximize his expected utility from terminal wealth, trades continuously in time, and whose portfolio allocation is restricted by constraints. This thesis aims to provide new solution methods that lead to closed-form characterizations for the optimal terminal wealth and optimal portfolio process for realistic settings with sophisticated financial market models and constraints that have previously not been discussed extensively in the existing academic literature. In this spirit, we developed a dual methodology that provides several equivalent dual optimality conditions which can be used to solve the portfolio optimization problem under simultaneous constraints on the investor's relative portfolio allocation and terminal wealth for general financial market models. We illustrate the usefulness of this methodology in a Black-Scholes model with convex cone allocation constraints, pointwise bounds on terminal wealth, and simultaneous Value-at-Risk or expected shortfall constraints by characterizing the investor's optimal terminal wealth up to two deterministic constants. Furthermore, by viewing one of the dual optimality conditions from a stochastic control perspective, we provide a condition under which the exponentially affine separability structure of the investor's value function in stochastic factor models is retained when allocation constraints are introduced. By verifying this condition in financial market models with stochastic short rate or stochastic volatility, we can ensure that the optimal constrained portfolio process remains deterministic and can be determined by solving an associated Riccati differential equation. We verify this condition and derive a closed-form solution for the optimal allocation-constrained portfolio in Heston's stochastic volatility model and make a surprising observation: The optimal constrained portfolio in a Heston market is not always equal to the 'naïve' capped portfolio, which caps off the optimal unconstrained portfolio at the boundaries of the constraints. Thus, our results suggest that allocation constraints have a fundamentally different impact on the optimal portfolio in financial markets with stochastic volatility compared to markets with constant volatility. In extensive numerical experiments, we demonstrate that following the naïve capped portfolio causes high annual wealth-equivalent losses for risk-averse investors in volatile markets, which can be experienced during a financial crisis.

Acknowledgements

First and foremost, I would like to express my immense gratitude to my supervisor Rudi Zagst, who gave me the opportunity to pursue a doctoral degree at the chair of mathematical finance at TUM. Through his detailed feedback and our countless research discussions, both during our recurring jour fixe as well as during spontaneous conversations and phone calls, Rudi has greatly shaped my doctoral research which ultimately culminates in this dissertation. His long-term vision, leadership and detailed feedback have been instrumental in the success of this dissertation. Moreover, Rudi has always encouraged me to get involved in exciting projects with experts and leaders from different fields at Zobel Values, the TUM Think Tank 'Post-Corona Economy', Bayerische Versorgungskammer and Allianz Global Investors. While working in multiple such projects has been demanding at times, each project still offered vast learning opportunities that I am incredibly grateful for.

Furthermore, I would like to greatly thank Marcos Escobar-Anel, the co-supervisor and mentor of my doctoral project. Just like Rudi, Marcos' ideas and insights have greatly influenced my research. It is truly impressive how well Marcos' and Rudi's supervision style complement each other and have consistently lead to fruitful research collaborations over many years. Moreover, Marcos has welcomed and supported me during my visits at Western University as a guest researcher. Not only do my two visits to Western constitute the most academically productive periods of my doctoral research, but they were also characterized by many insightful discussion with Marcos about mathematics, sports and life in general.

Moreover, I am deeply grateful to my colleagues at the chair of mathematical finance at TUM, many of whom I would consider friends rather than colleagues at this point. I will very fondly remember the stimulating, controversial and often extremely fun discussions we have had. In particular, I would like to thank Ben Spies and Dominik de Witte who have been amazing support by sharing opinions, feedback and advice on all kinds of matters and were always available for a quick chat if necessary.

Finally, I would like to thank my friends and family for their steady support and encouragement during the entirety of my university studies. This of course warrants a very special thanks to my parents for their unconditional love and backing as well as to my long-term, long-distance girlfriend Janice for her inexhaustible trust, patience and emotional support.

> Michel Kschonnek June 25, 2023

Contents

1	Intro	oduction												
	1.1	Motivation & Objective												
	1.2	Structure of the Thesis												
	1.3	Contributions												
2	Mathematical Preliminaries													
	2.1	Convex Analysis												
		2.1.1 Basic Notions from Convex Analysis												
		2.1.2 Concave Optimization												
	2.2	Ordinary Differential Equations												
	2.3	The Continuous-Time Portfolio Optimization Problem												
		2.3.1 Problem Formulation												
		2.3.2 Classic Methods for Unconstrained Portfolio Optimization												
3	Con	Constraints on Allocation and Wealth in Complete Financial Market Models 37												
	3.1	Introduction												
	3.2	Setting												
	3.3	Generalized Martingale Approach												
	3.4	Auxiliary Markets with Bounds on Terminal Wealth												
	3.5	The Optimal Auxiliary Market												
		3.5.1 Optimization Problem Associated with Condition D												
		3.5.2 Convex Cone Constraints												
	36	3.3.3 Examples												
	0.0													
4	Con	straints on Allocation in Stochastic Factor Models 57												
	4.1	Introduction												
	4.2	The Dual HIPI DDF												
	4.5	1 Ine Dual HJDI F DE												
		4.3.2 Exponential Affine Separability 65												
	44	Examples 65												
	1.1	4 4 1 Black-Scholes Model 66												
		4.4.2 Multi-Factor Stochastic Covariance of CIR-Type												
		4.4.3 Multi-Factor Short Rate of OU-Type												
	4.5	Conclusion												
5	Con	straints on Allocation in Heston's Stochastic Volatility Model 73												
5	5.1	Introduction												
	5.2	Heston's Stochastic Volatility Model												
		5.2.1 Solution to HJB PDE												
		5.2.2 Verification Theorem												
		5.2.3 Comparison to Unconstrained Portfolio												
		-												

Contents

5.3 Implications for Related Models											•			 83						
		5.3.1	PCSV Me	odel .													•			 84
		5.3.2	Inverse V	olatility	y Co	nstra	ints										•			 85
	5.4	Numer	ical Studie	es													•			 87
		5.4.1	Optimal 0	Constra	ained	l Mei	ton 1	Port	folio	$\pi =$	= Ca	ap (*	π_M	α, β	3).		•			 88
		5.4.2	Capped ()ptima	l Une	const	raine	ed H	esto	n Po	ortf	olio	$\pi =$	= Ca	ър (7	τ_u, η	α, β	3)		 90
	5.5	Conclu	sion														•			 92
6	Con	clusion																		95
Α	Proc	ofs																		98
	A.1	Proofs	Chapter 2	2													•			 98
	A.2	Proofs	Chapter 3	5													•			 113
	A.3	Proofs	Chapter 4	L													•			 139
	A.4	Proofs	Chapter 5)				• •									•		•	 152
Bil	bliogr	aphy																		167

1 Introduction

1.1 Motivation & Objective

Expected utility theory has been widely used within the realm of decision theory to determine and discuss how people should behave in uncertain environments. As such, it has been widely applied in mathematical finance and in particular in the context of dynamic portfolio optimization. Such applications can be roughly divided by their underlying model for the financial market into discrete time (see e.g. [68] and [79]) and continuous time (see e.g. [64], [65]). While discrete time models can in principle perfectly reflect how prices of exchange-traded assets change (after all, physical limitations ensure that order routing and processing at exchanges occur in discrete time), they lack analytical tractability and closed-form solutions for all but the most simple return distributions. For this reason, the analysis of sophisticated and realistic discrete time models may require complex or computation heavy numerical schemes. In contrast, applications of continuous time models yield tractable closed-form solutions in a variety of contexts and are widely employed in the mathematical finance literature and in the financial industry, despite the fact that they require the heuristic assumption of continuous trading.

In this spirit, this thesis considers several variations of a dynamic portfolio optimization problem for an investor who aims to maximize his expected utility and trades continuously in time. As suggested by the thesis title, we additionally consider an aspect which extremely relevant for investors trading in the financial markets: constraints, especially allocation constraints.

In most practical applications investors need to abide by allocation constraints either due to regulatory requirements or due to client preferences. [85] and [49] first presented a duality approach for portfolio optimization problems with constraints on short-selling and trading of individual assets. Their duality approach was later generalized for general convex allocation constraints in [17], who derived a dual optimal control problem which seeks the least favorable market coefficients among a suitable set of 'dual' stochastic processes. For CRRA-utility functions, [17] determined the optimal constrained portfolio process in closed-form up to a deterministic minimizer of a real convex optimization problem. [89] considered a similar setting in a one-dimensional Black Scholes market, but did not employ any duality techniques. Rather, the author characterized the value function as the unique viscosity solution of the associated HJB PDE and gave a semi-explicit expression of the optimal portfolio allocation in terms of the value function. From both [17] and [89], one can easily see that the optimal constrained portfolio allocation for an investor with a CRRA utility function in the Black-Scholes model is equal to the unconstrained optimal solution if the constraints are satisfied, and is otherwise capped at the boundary of the constraint. Ever since, allocation constraints have been integrated into portfolio optimization problems in a myriad of ways (see e.g. [16], [71], [60], [66], [6], [70], [23] and [24]). However, only in rare cases were closed-form formulae or solution methods presented which do not require extensive Monte-Carlo simulation. The most notable exceptions to this observation are financial market models and utility functions which result in a myopic optimal portfolio process. This includes the logarithmic utility function for general diffusion models as well as the power utility function for stochastic factor models where the stochastic factor only describes systematic risk (of which the Black-Scholes model is a special case), see [17] and [71]. In other cases, solution to the allocation constrained portfolio optimization problem need to be estimated by sophisticated numerical schemes (e.g. [19], [92]).

1 Introduction

Even though the continuous time dynamic portfolio optimization problem under allocation constraints has been extensively studied in the existing literature, we make the argument that there is a lack of tractable closed-form solutions for many realistic settings and therefore, the main upside of continuous time over discrete time financial market models is hardly utilized in the existing literature. Hence, the objective of this thesis is to extend the existing literature on constrained portfolio optimization and provide new methodologies which yield new closed-form solutions to existing problems or provide otherwise new insights about them. Specifically, we

- extend the auxiliary framework of [17] to include simultaneous constraints on relative portfolio allocation and terminal wealth (Chapter 3),
- develop a novel solution approach for allocation constrained portfolio optimization problems in incomplete financial markets with market dynamics influenced by an external stochastic factor (Chapter 4),
- and apply the aforementioned approach to derive closed form solution formulae for the optimal portfolio in Heston's stochastic volatility model (Chapter 5).

1.2 Structure of the Thesis

The structure of this thesis is as follows: In Chapter 2 we begin by collecting a selection of useful results from different fields. As this thesis focusses on portfolio optimization problems with (allocation) constraints, we pay special attention to convex duality theory and some of its applications to constrained optimization in Section 2.1. Moreover, we formally define the overarching setting of this thesis and give an overview of two classic methods for (unconstrained) portfolio optimization in continuous time: The martingale method and the stochastic control approach.

In Chapter 3, we consider a portfolio optimization problem with simultaneous constraints on portfolio allocation and terminal wealth. We propose a generalized martingale method which is applicable under the presence of constraints on terminal wealth in complete financial markets. Afterwards, we proceed to integrate this methodology into the auxiliary market framework of [17] and derive a set of equivalent dual optimality conditions for portfolio optimization problems with simultaneous constraints on portfolio allocation and terminal wealth. The optimality conditions correspond to different duality approaches generated by the allocation constraints or the wealth constraints. If the allocation constraints are a convex cone and risky asset prices follow a geometric Brownian motion, then we can derive closed-form expressions for the optimal terminal wealth as a function of an adjusted pricing kernel and a deterministic Lagrange multiplier. This is demonstrated for constraints on Value-at-Risk and expected shortfall.

In Chapter 4, we study a portfolio optimization problem in an incomplete financial market where the risky asset dynamics depend on stochastic factors and the relative portfolio allocation is constrained to lie within a given convex set. We employ fundamental duality results from real constrained optimization to formally derive a dual representation of the associated HJB PDE. Using this representation, we provide a condition on the market dynamics and the allocation constraints, which ensures that the solution to the HJB PDE is exponentially affine and separable. This condition is used to derive an explicit expression for the optimal allocationconstrained portfolio up to a deterministic minimizer and the solution to a system of Riccati ODEs in a market with CIR volatility and in a market with multi-factor OU short rate.

In Chapter 5, we specifically consider a portfolio optimization problem with convex constraints in Heston's stochastic volatility model. Naturally, this setting is a special case of Chapter 5 and we may thus apply the previously developed duality methods developed to obtain a closed-form

1 Introduction

expression for the optimal portfolio allocation. In doing so, we observe that allocation constraints impact the optimal constrained portfolio allocation in a fundamentally different way in Heston's stochastic volatility model than in the Black Scholes model. In particular, the optimal constrained portfolio may be different from the naive 'capped' portfolio, which caps off the optimal unconstrained portfolio at the boundaries of the constraints. Despite this difference, we illustrate by way of a numerical analysis that in most realistic scenarios the capped portfolio leads to slim annual wealth equivalent losses compared to the optimal constrained portfolio. During a financial crisis, however, a capped solution might lead to compelling annual wealth equivalent losses.

Chapter 6 summarizes the main findings of this thesis and gives an outlook on potential future research.

All detailed mathematical derivations and proofs have been moved to Appendix A to improve readability.

1.3 Contributions

This thesis is based on and parts of it have been quoted verbatim from the following research articles:

Chapter 3:

[34] Marcos Escobar-Anel, Michel Kschonnek and Rudi Zagst. 'Portfolio Optimization: Not Necessarily Concave Utility and Constraints on Wealth and Allocation'. In: *Mathematical Methods of Operations Research* 95 (2022), pp. 101–140.

Chapter 4:

[33] Marcos Escobar-Anel, Michel Kschonnek and Rudi Zagst. 'Portfolio Optimization with Allocation Constraints and Stochastic Factor Market Dynamics'. 2023. arXiv: 2303.09835 [q.fin.PM]

Chapter 5:

[32] Marcos Escobar-Anel, Michel Kschonnek and Rudi Zagst. 'Mind the Cap! – Constrained Portfolio Optimisation in Heston's Stochastic Volatility Model'. 2023. arXiv: 2306.11158 [q.fin.PM]

As part of the elite doctoral program 'TopMath', an earlier version of [34] has been accredited as M.Sc. thesis of the author of this thesis. However, the inclusion of expected value constraints and the discussion of the associated dual optimization problems in Chapter 3 constitute an innovation which has not been published as part of [34].

Further, the author of this thesis has co-authored the following publications, which are not part of this thesis:

- [35] Marcos Escobar-Anel, Yevhen Havrylenko, Michel Kschonnek and Rudi Zagst. 'Decrease of capital guarantees in life insurance products: Can reinsurance stop it?'. In: *Insurance: Mathematics and Economics* 105 (2022), pp. 14–40.
- [58] Michel Kschonnek, Iryna Dobrovolska, Ulrike Protzer and Rudi Zagst. 'COVIX–An Index Allowing for the Assessment of the Pandemic Situation Based on Infections and Hospitalisation Data'. In: Applied Sciences 13.7 (2023).

2.1 Convex Analysis

2.1.1 Basic Notions from Convex Analysis

The majority of concepts and results presented in this section can be found in [76] and [29]. Throughout this section we consider a general vector space V over \mathbb{R} with a scalar product $\langle \cdot, \cdot \rangle : V \times V \to \mathbb{R}$ and induced norm $\|\cdot\| : V \to \mathbb{R}$. Note that all results that hold for V also hold for its natural extension with the real line $V_{\mathbb{R}} := V \times \mathbb{R}$ and the scalar product $\langle \cdot, \cdot \rangle_{\mathbb{R}} : V_{\mathbb{R}}^2 \to \mathbb{R}, \langle (x, \alpha), (y, \beta) \rangle_{\mathbb{R}} = \langle x, y \rangle + \alpha \beta$. An easy illustrating example would be \mathbb{R}^d for some $d \in \mathbb{N}$ with the standard scalar product $\langle x, y \rangle = x'y$ and the Euclidean distance $\|x\| = \sqrt{x'x}$ as norm. Further, we define the extended real numbers as $\mathbb{R} := \mathbb{R} \cup \{\infty, -\infty\}$.

Definition 2.1.1 (Convex Sets). A set $K \subset V$ is called convex if for all $x, y \in K$ every point on the line segment connecting x and y is again in K, i.e., $\forall \lambda \in (0, 1)$ we have

$$\lambda x + (1 - \lambda)y \in K.$$

The convex hull co K of a set $K \subset V$ is the smallest convex set containing K, i.e.,

$$co \ K = \bigcap_{\substack{M \ convex\\K \subset M}} M.$$

Definition 2.1.2 (Open and Closed Sets). A set $K \subset V$ is open, if for all $x \in K$, there exists $\epsilon > 0$ such that

 $||x - y|| \le \epsilon \quad \Rightarrow y \in K, \quad \forall y \in V.$

A set $K \subset V$ is closed, if its complement $V \setminus K$ is open.

The closure cl K of a set $K \subset V$ is the smallest closed set containing K, i.e.,

$$cl \ K = \bigcap_{\substack{M \ closed\\ K \subset M}} M.$$

Clearly, any $K \subset V$ is closed if and only if $cl \ K = K$. Similarly, K is convex if and only if $co \ K = K$. Moreover, if K is indeed closed and convex, then there exists a more concise characterization of K in terms of closed half spaces.

Definition 2.1.3 (Closed Half Space). For given $(y, \beta) \in V_{\mathbb{R}}$, the set

$$H(y,\beta) = \left\{ x \in V \mid \langle x, y \rangle \le \beta \right\}$$

is called a closed half space.

Lemma 2.1.4 (Closed Convex Hull as Intersection of Half Spaces). Let $K \subset V$ be arbitrary. Then,

$$cl \ co \ K = \bigcap_{\substack{K \subset H(y,\beta) \\ (y,\beta) \in V_{\mathbb{R}}}} H(y,\beta).$$

In particular, K is closed and convex (closed convex) if and only if

$$K = \bigcap_{\substack{K \subset H(y,\beta) \\ (y,\beta) \in V_{\mathbb{R}}}} H(y,\beta).$$

Since we will at a later point almost exclusively be considering concave optimization problems (maximization) instead of convex optimization problems (minimization) we introduce the following classic tools for the analysis of concave and convex functions (epigraphs, hypographs, conjugates, etc.) for concave functions only.

Definition 2.1.5 (Domain in Concave Sense and Hypograph). For a function $f: V \to \overline{\mathbb{R}}$, we define its effective domain dom_f $\subset V$ and its hypograph $H_f \subset V_{\mathbb{R}}$ as

$$dom_f = \left\{ x \in V \mid f(x) > -\infty \right\}$$
$$H_f = \left\{ (x, \alpha) \in V_{\mathbb{R}} \mid f(x) \ge \alpha \right\}.$$

Geometrically, one may think of the hypograph of a function f as the closure of the area below the graph of a function f. It is easy to see that the hypograph of an affine function $h: V \to \mathbb{R}$ defines a half space in $V_{\mathbb{R}}$. Note that the hypograph characterizes a function uniquely and every function $f: V \to \mathbb{R}$ can be reconstructed from its hypograph. We frequently switch between the classic representation of a function f and the representation of a function f through its hypograph. More precisely, given a hypograph $H_f \subset V_{\mathbb{R}}$, we can recover f by setting

$$f(x) := \sup \left\{ \alpha \mid (x, \alpha) \in H_f \right\},\$$

where we make use of the convention $\sup \emptyset = -\infty$ and $\sup \mathbb{R} = \infty$.

Definition 2.1.6 (Convex & Concave Functions). A function $f : V \to \mathbb{R}$ is concave if its hypograph H_f is convex. If additionally there exists an $\hat{x} \in V$ such that $f(\hat{x}) > -\infty$ and $f(x) < \infty \quad \forall x \in V$, then f is called a proper concave function. If -f is concave, then f is called a convex function.

Remark 2.1.7. Analogously to Definition 2.1.6, we also refer to non-concave functions $f : V \to \overline{\mathbb{R}}$ as proper if $f(x) < \infty$ for all $x \in V$ and $f(\hat{x}) > -\infty$ for some $\hat{x} \in V$.



Figure 2.1: Illustration of the hypograph H_f for a function f which is concave (Figure 2.1(a)) and non-concave (Figure 2.1(b)).

Definition 2.1.6 is slightly more abstract than usual, but has the advantage that we can classify functions taking values only in $\{-\infty, \infty\}$ as concave (or non-concave) without encountering the undefined operations $-\infty + \infty$ or $\infty - \infty$. If we restrict ourselves to functions $f: V \to \overline{\mathbb{R}}$ with $f(x) < \infty \ \forall x \in V$, then we recover the usual definition of concavity.

Lemma 2.1.8. Consider a proper function $f: V \to \overline{\mathbb{R}}$. Then, f is concave if and only if it satisfies

$$f(\lambda x + (1 - \lambda)y) \ge \lambda f(x) + (1 - \lambda)f(y) \quad \forall x, y \in V, \lambda \in (0, 1).$$

$$(2.1)$$

Definition 2.1.9 (Usc Functions). A function $f: V \to \overline{\mathbb{R}}$ is upper semi-continuous ('usc') if

for any
$$\alpha \in \mathbb{R}$$
, $M_f(\alpha) = \{x \in V \mid f(x) \ge \alpha\}$ is closed in V. (2.2)

An equivalent characterization is

$$\limsup_{x' \to x} f(x') \le f(x) \quad \forall x \in V.$$
(2.3)

Analogous to the convex hull *co* K and closure *cl* K of a set $K \subset V$, we may define similar notions for functions $f: V \to \overline{\mathbb{R}}$.

Definition 2.1.10 (Concave Hull, Usc Hull and Closure). Consider a function $f: V \to \mathbb{R}$.

- (i) The concave hull co $f: V \to \overline{\mathbb{R}}$ is the smallest concave function greater or equal than f.
- (ii) The usc hull usc $f: V \to \overline{\mathbb{R}}$ is the smallest usc function greater or equal than f.
- (iii) The closure $cl \ f: V \to \overline{\mathbb{R}}$ is defined as

$$cl \ f(x) = \begin{cases} usc \ f(x), & \forall x \in V, \ if \ usc \ f(x) < \infty \ \forall x \in V \\ \infty, & \forall x \in V, \ if \ usc \ f(x) = \infty \ for \ some \ x \in V \end{cases}$$

If usc f (resp. usc co f) is proper, then $cl f \equiv usc f$ (resp. $cl co f \equiv usc co f$). This technical construction allows us to characterize usc co f by the closed convex hull of its hypograph and cl co f as the minimum of affine functions h larger or equal than f. This illustrates the strong link between the concave hull, usc hull and closure of a function f and the convex hull and closure of its hypograph H_f .

Lemma 2.1.11. Consider a proper function $f: V \to \overline{\mathbb{R}}$. Then the following holds:

- (i) f is use if and only if H_f is closed in $V_{\mathbb{R}}$
- $(ii) \quad cl \ H_f = H_{usc \ f}$ $(iii) \quad co \ H_f = H_{co \ f}$ $(iv) \quad H_{usc \ co \ f} = cl \ co \ H_f = \left(cl \ co \ dom_f \times \mathbb{R}\right) \cap \left(\bigcap_{\substack{h: V \to \mathbb{R} \ affine \\ f \le h}} H_h\right)$ $(v) \quad cl \ co \ f(x) = \inf_{z \in V} \left(\sup_{\hat{x} \in V} \left(f(\hat{x}) \langle z, \hat{x} \rangle\right) + \langle z, x \rangle\right)$

Especially statement (iv) and (v) of Lemma 2.1.11 and their proof are quite remarkable. Not only is illustrated that it is equivalent to compute the usc hull and concave hull usc co f of a function f directly or computing the closure of the concave hull cl co H_f of H_f and deriving usc co f from it, but it connects these two equivalent operations to a min-max-problem for f.

The final statements of Lemma 2.1.11 motivate the definition of the (concave) conjugate and bi-conjugate of a given function $f: V \to \overline{\mathbb{R}}$:

Definition 2.1.12 (Concave Conjugate). Consider a function $f : V \to \overline{\mathbb{R}}$. We define the conjugate $f^* : V \to \overline{\mathbb{R}}$ and the bi-conjugate $f^{**} : V \to \overline{\mathbb{R}}$ as

• $f^*(y) = \sup_{x \in V} (f(x) - \langle y, x \rangle)$ for $y \in V$ • $f^{**}(x) = \inf_{y \in V} (\langle y, x \rangle + f^*(y))$ for $x \in V$.

Going back to the proof of Lemma 2.1.11 (iv), for a given slope $z \in V$, the conjugate $f^*(z)$ can be identified as the minimal intercept β_z such that the affine function $h_z : V \to \mathbb{R}$, $h_z(x) = \beta_z + \langle x, z \rangle$ is larger or equal than f, i.e., such that the hypograph H_h is contained in the half space H_{h_z} . Finding z_x such that $h_{z_x}(x) \ge f(x)$ is minimized, we take the bi-conjugate $f^{**}(x)$ of f. This is in turn equivalent to taking the intersection of all half spaces H_{h_z} , which contain H_f .

Remark 2.1.13. Let $x \in V$ be arbitrary but fixed. Then, so-called "weak duality" holds, i.e.,

$$f^{**}(x) = \inf_{y \in V} \left(\langle y, x \rangle + \underbrace{f^{*}(y)}_{\geq f(x) - \langle y, x \rangle} \right)$$

$$\geq \inf_{y \in V} \left(\langle y, x \rangle + f(x) - \langle y, x \rangle \right) = \inf_{y \in V} \left(f(x) \right) = f(x)$$

Remark 2.1.14. In the convex analysis literature (see e.g. equation (3.25) in [76]) it is customary to define the conjugate of a function $f: V \to \overline{\mathbb{R}}$ in the concave sense as

$$f^*(y) = -\sup_{x \in V} \left(f(x) - \langle y, x \rangle \right) = \inf_{x \in V} \left(\langle y, x \rangle - f(x) \right).$$

In contrast to our Definition 2.1.12, this has the satisfying consequence that the bi-conjugate of f is obtained from taking the conjugate of f^* , i.e., $f^{**} = (f^*)^*$.

According to our definition we have the (slightly) less elegant version $f^{**} = -(-f^*)^*$. However, in the mathematical finance literature, in particular in a portfolio optimization context, Definition 2.1.12 is more prevalent. Since, we are going to apply the above theory in a portfolio optimization context, we decided to stick to the corresponding convention. We summarize some useful properties of the conjugate and bi-conjugate in Lemma 2.1.15:

Lemma 2.1.15. Consider a proper function $f: V \to \overline{\mathbb{R}}$. The following holds:

- (i) $-f^*$ and f^{**} are use and concave
- (ii) f is use and concave if and only if

$$f(x) = f^{**}(x) = cl \ co \ f(x) \quad \forall x \in V.$$

Example 2.1.16 (Indicator Functions of Convex Sets). Consider a non-empty closed convex set $K \subset V$ and define its indicator function $\mathcal{X}_K : V \to \overline{\mathbb{R}}$ as

$$\mathcal{X}_K(x) = \begin{cases} 0, & \text{if } x \in K \\ -\infty, & \text{if } x \notin K. \end{cases}$$

This indicator function is intentionally chosen differently from the standard indicator function $\mathbb{1}_{K}(x)$, which takes values in $\{0,1\}$, depending on whether $x \in K$ or not. Regardless, the above concept comes in handy, when considering concave optimization problems that are constrained over convex sets K. The concave conjugate of \mathcal{X}_{K} is given as

$$\begin{aligned} \mathcal{X}_{K}^{*}(y) &= \sup_{x \in V} \left(\mathcal{X}_{K}(x) - \langle y, x \rangle \right) = \sup_{x \in K} \left(- \langle y, x \rangle \right) \\ &= -\inf_{x \in K} \left(\langle y, x \rangle \right) =: \delta_{K}(y). \end{aligned}$$

The function δ_K is called the support function of K and plays an important role in constrained concave optimization problems. The domain of δ_K is the barrier cone

$$X_K = dom_{\delta_K} = \{ y \in V \mid \delta_K(y) < \infty \}.$$

The support function δ_K offers a natural characterization of closed convex sets in a dual sense. As K is a closed convex set, the indicator function \mathcal{X}_K is use and concave. Thus, by Lemma 2.1.15, we have $\mathcal{X}_K = \mathcal{X}_K^{**}$. This yields

$$x \in K \Leftrightarrow \mathcal{X}_{K}(x) \geq 0$$

$$\Leftrightarrow \mathcal{X}_{K}^{**}(x) \geq 0$$

$$\Leftrightarrow \inf_{y \in V} \left(\delta_{K}(y) + \langle x, y \rangle \right) \geq 0$$

$$\Leftrightarrow \delta_{K}(y) + \langle x, y \rangle \geq 0 \ \forall y \in V$$

$$\Leftrightarrow \delta_{K}(y) + \langle x, y \rangle \geq 0 \ \forall y \in X_{K}$$
(2.4)

Due to its significance for the remainder of the thesis, we summarize the key properties of the support function δ_K below.

Lemma 2.1.17. Let $K \subset V$ be a non-empty closed, convex set. Then,

(i) δ_K is positive homogeneous of order 1:

$$\delta_K(\alpha y) = \alpha \delta_K(y) \quad \forall \alpha \ge 0, \ \forall y \in V.$$

(ii) δ_K is sub-additive:

$$\delta_K(y_1 + y_2) \le \delta_K(y_1) + \delta_K(y_2) \quad \forall y_1, y_2 \in V.$$

(iii)

$$\begin{aligned} x \in K & \Leftrightarrow & 0 \leq \delta_K(y) + \langle x, y \rangle \quad \forall y \in X_K \\ & \Leftrightarrow & 0 \leq \delta_K(y) + \langle x, y \rangle \quad \forall y \in X_K \text{ with } \langle y, y \rangle \leq 1 \end{aligned}$$

(iv) $\delta_K(y) = 0 \quad \forall y \in X_K \iff K \text{ is a convex cone.}$

2.1.2 Concave Optimization

In the following, we consider a proper function $f: V \to \overline{\mathbb{R}}$ and the primal optimization problem (\mathbf{P}) of the form

$$(\mathbf{P}) \left\{ \Phi_P = \sup_{x \in V} f(x).$$
(2.5)

The representation (2.5) is sufficiently general so that classic examples of constraints can be incorporated into this formulation. Similar to a portfolio optimization context, we consider two different types of constraints:

- (i) Constraints explicitly defined on $x \in V$, i.e., we require that $x \in K$ for a subset $K \subset V$.
- (ii) Constraints implicitly defined through a function $g: V \to H$, i.e., we require that $g(x) \in K$ for a subset $K \subset H$ of another vector space H over \mathbb{R} .

Assuming we aim to maximize a proper function $f: V \to \mathbb{R}$ subject to the constraint (i). Then, we can rewrite this constrained maximization over f as an unconstrained maximization problem over a suitable function \tilde{f} by considering $\mathcal{X}_K: V \to \{-\infty, 0\}$ as defined in Example 2.1.16 and defining

$$\sup_{x \in K} f(x) = \sup_{x \in V} \left(\underbrace{f(x) + \mathcal{X}_K(x)}_{=:\tilde{f}(x)} \right) = \sup_{x \in V} \tilde{f}(x).$$
(2.6)

Note that \mathcal{X}_K is use and concave if and only if K is closed convex. Hence, if f is use and concave, \tilde{f} remains use and concave if K is closed convex. Similarly, if we aim to maximize f subject to constraint (ii), then we may use the same notation as above, where $\mathcal{X}_K : H \to \{-\infty, 0\}$ now takes arguments $u \in H$, and write

$$\sup_{x \in V, \ g(x) \in K} f(x) = \sup_{x \in V} \left(\underbrace{f(x) + \mathcal{X}_K(g(x))}_{=:\tilde{f}(x)} \right) = \sup_{x \in V} \tilde{f}(x).$$
(2.7)

Clearly, we could have done the same transformation as in (2.6) by defining the set

$$\tilde{K} := \{ x \in V \mid g(x) \in K \}$$

and maximizing over $x \in \tilde{K}$. However, we will later see that the distinction between (2.6) and (2.7) can still be useful. Again, \tilde{f} is use concave if f is use concave and \tilde{K} is closed convex.

We now follow the framework presented in [76] to define a dual optimization problem (**D**) corresponding to the primal optimization problem (**P**). For this purpose consider another vector space H over \mathbb{R} with scalar product $\langle \cdot, \cdot \rangle$ and assume that there exists a function

$$F: V \times H \to \mathbb{R}$$
, with $F(x, \bar{u}) = f(x)$ for some $\bar{u} \in H$.

One may think of F as a perturbation of f or as a function, which expresses the dependence of F on the parameter choice $\bar{u} \in H$. Clearly, the choice of F for a given f is not unique, but can be chosen freely, although there may exist a *natural* choice for F. Regardless, the following results hold for any choice of F, but their usefulness is heavily dependent on the specific choice of F.

Definition 2.1.18. For any $u \in H$, we define the perturbed problem $(\mathbf{P}_{\mathbf{u}})$ as

$$(\mathbf{P}_{\mathbf{u}}) \left\{ \Phi(u) = \sup_{x \in V} F(x, u).$$
(2.8)

In particular, we clearly have $\Phi(\bar{u}) = \Phi_P$.

Definition 2.1.19. We define the Lagrangian of F as $L: V \times H \to \overline{\mathbb{R}}$ as

$$L(x,\lambda) := \sup_{u \in H} \left(F(x,u) - \langle u, \lambda \rangle \right) + \langle \bar{u}, \lambda \rangle.$$

Moreover, we define the dual problem (\mathbf{D}) as

$$(\mathbf{D}) \begin{cases} \Psi_D &= \inf_{\lambda \in H} G(\lambda) \\ G(\lambda) &= \sup_{x \in V} L(x, \lambda). \end{cases}$$
(2.9)

Example 2.1.20. Consider a proper function $f : V \to \mathbb{R}$, which we would like to optimize over the closed convex set $K \subset V$. We may equivalently optimize over the function $\tilde{f}(x) := f(x) + \mathcal{X}_K(x)$ instead (as seen in (2.6)). We assume that there exists at least one feasible $x \in V$ with $\tilde{f}(x) > -\infty$, i.e., \tilde{f} is proper, too.

Choosing H = V, we define a perturbation function $F: V \times V \to \mathbb{R}$ by setting

$$F(x,u) = f(x) + \mathcal{X}_K(x+u) = f(x) + \mathcal{X}_{K-u}(x).$$

As hinted by the last equality, this can be regarded as a perturbation to the constraint set K through an affine shift K - u. In particular for $\bar{u} := 0$, we clearly have $F(x, \bar{u}) = f(x)$ for $x \in K$. For fixed $(x, \lambda) \in V^2$, the Lagrangian L can be calculated as

$$\begin{split} L(x,\lambda) &= \sup_{u \in H} \left(F(x,u) - \langle u, \lambda \rangle \right) + \langle \bar{u}, \lambda \rangle \\ &= \sup_{u \in V} \left(f(x) + \mathcal{X}_K(x+u) - \langle u, \lambda \rangle \right) + \underbrace{\langle 0, \lambda \rangle}_{=0} \\ &= f(x) + \sup_{u \in V} \left(\mathcal{X}_K(x+u) - \langle u, \lambda \rangle \right) \\ &= f(x) + \sup_{u \in V} \left(\mathcal{X}_K(\underbrace{x+u}_{=v \in V}) - \underbrace{\langle x+u}_{=v \in V}, \lambda \rangle + \langle x, \lambda \rangle \right) \\ &= f(x) + \langle x, \lambda \rangle + \sup_{v \in V} \left(\mathcal{X}_K(v) - \langle v, \lambda \rangle \right) \\ &= f(x) + \langle x, \lambda \rangle + \delta_K(\lambda). \end{split}$$

The objective function G of the dual problem (\mathbf{D}) is thus given by

$$G(\lambda) = \sup_{x \in V} f(x) + \langle x, \lambda \rangle + \delta_K(\lambda),$$

which is the value function of an unconstrained optimization over the objective function $f(x) + \langle x, \lambda \rangle + \delta_K(\lambda)$, where the additional term $\langle x, \lambda \rangle + \delta_K(\lambda)$ is non-negative whenever $x \in K$ (see Lemma 2.1.17). In this sense, this adjusted objective function implicitly rewards $x \in V$ which abide by the constraint $x \in K$. In particular, we have $G(\lambda) \ge \Phi_P$ for all $\lambda \in H = V$. Correspondingly, the dual problem (**D**) in this context reads as

$$(\mathbf{D}) \begin{cases} \Psi_D &= \inf_{\lambda \in V} G(\lambda) \\ G(\lambda) &= \sup_{x \in V} \left(f(x) + \langle x, \lambda \rangle \right) + \delta_K(\lambda), \end{cases}$$
(2.10)

i.e., the additional reward gained by the removal of the constraint K and the additional term $\langle x, \lambda \rangle + \delta_K(\lambda)$ is minimized.

Example 2.1.21. Consider functions $f : V \to \mathbb{R}$, $g : V \to H$. We would like to maximize f over all $x \in V$ such that $g(x) \in K$, for a closed convex set $K \subset H$. By setting

$$\tilde{f}(x) = f(x) + \mathcal{X}_K(g(x))$$

we obtain our usual representation of (**P**). By defining the perturbed objective function $F : V \times H \to \mathbb{R}$ as

$$F(x, u) = f(x) + \mathcal{X}_K(g(x) + u),$$

we obtain the corresponding Lagrangian L (using the same calculations as in Example 2.1.20) as

$$L(x,\lambda) = f(x) + \langle g(x), \lambda \rangle + \delta_K(\lambda).$$

It is important to note that in this particular example it is possible that $H \neq V$ and hence $\lambda \in H$ may be of an entirely different form than $x \in V$. The dual optimization problem (**D**) is thus defined as

$$(\mathbf{D}) \begin{cases} \Psi_D &= \inf_{\lambda \in H} G(\lambda) \\ G(\lambda) &= \sup_{x \in V} \left(f(x) + \langle g(x), \lambda \rangle \right) + \delta_K(\lambda). \end{cases}$$
(2.11)

Lemma 2.1.22. The Lagrangian $L: V \times H \to \overline{\mathbb{R}}$ has the following properties:

- (i) If $F(x, \cdot) : H \to \overline{\mathbb{R}}$ is proper for a given $x \in V$, then $-L(x, \cdot) : H \to \overline{\mathbb{R}}$ is use and concave
- (ii) If $F(x, \cdot) : H \to \overline{\mathbb{R}}$ is proper, use and concave for a given $x \in V$, then

$$\inf_{\lambda \in H} L(x, \lambda) = F(x, \bar{u}) = f(x)$$

(iii) If $F(x, \cdot) : H \to \overline{\mathbb{R}}$ is proper, use and concave $\forall x \in V$, then the primal problem (**P**) can equivalently be written as

$$(\mathbf{P}) \begin{cases} \Phi_P = \sup_{x \in V} \inf_{\lambda \in H} L(x, \lambda) \end{cases}$$
(2.12)

Theorem 2.1.23. The following duality relation holds between the optimization problems (\mathbf{P}) and (\mathbf{D}) :

$$\Phi(\bar{u}) = \sup_{x \in V} f(x) = \Phi_P \le \Psi_D = \inf_{\lambda \in H} \sup_{x \in V} L(x, \lambda) = cl \ co \ \Phi(\bar{u}).$$

By virtue of Theorem 2.1.23, we have successfully derived a duality representation between (**P**) and (**D**). The optimal value of both optimization problems coincides if and only if the concave closure of Φ coincides with Φ at the parameter choice $u = \bar{u}$, i.e.,

$$\Phi_P = \Psi_D \quad \Leftrightarrow \quad \Phi(\bar{u}) = cl \ co \ \Phi(\bar{u}).$$

The difference $\Psi_D - \Phi_P$ is also referred to as duality gap. If $\Phi_P = \Psi_D$, we say the duality gap is zero. If the dependence of F on the perturbation (or parameter) u is u.s.c and concave, then according to (*iii*) in Lemma 2.1.22, we view (**P**) as a max-min problem and (**D**) as a min-max problem over the same Lagrangian L. When both (**P**) and (**D**) obtain the optimal value at an $(x^*, \lambda^*) \in V \times H$, we call (x^*, λ^*) a saddle-point. **Definition 2.1.24.** $(x^*, \lambda^*) \in V \times H$ is called a saddle-point of L, if

$$L(x,\lambda^*) \le L(x^*,\lambda^*) \le L(x^*,\lambda) \quad \forall (x,\lambda) \in V \times H.$$

Remark 2.1.25. In the current abstract context, when $(x^*, \lambda^*) \in V \times H$ is a saddle-point we also say that (x^*, λ^*) satisfies the so-called Karush-Kuhn-Tucker condition ('KKT condition').

Corollary 2.1.26. If $(x^*, \lambda^*) \in V \times H$ is a saddle-point of L, then λ^* is optimal for the dual optimization problem (**D**).

If additionally $F(x^*, \cdot)$ is proper, use and concave in u, then x^* is optimal for the primal optimization problem (**P**).

In summary, we obtain the following equivalent statements:

Theorem 2.1.27. Let $F(x^*, \cdot)$ be proper, use and concave in u for a specific $x^* \in V$. Consider the following statements:

(i) $\Phi_P = \Psi_D$

(ii) $\Phi(\bar{u}) = cl \ co \ \Phi(\bar{u})$

(iii) x^* is optimal for (**P**), λ^* is optimal for (**D**) and $\Phi_P = \Psi_D$

(iv) (x^*, λ^*) is a saddle-point of the Lagrangian L.

Then, $(i) \Leftrightarrow (ii)$ and $(iii) \Leftrightarrow (iv)$.

Theorem 2.1.27 does not provide direct properties of F, which ensure that either of the statements (i) - (iv) are satisfied. However, it is possible to derive several properties for F, which guarantee that statement (ii) in Theorem 2.1.27 is satisfied and therefore the duality gap must be zero. The so-called Slater's condition is one prominent example.

Theorem 2.1.28 (Slater's condition). Assume F is concave jointly in (x, u) and there exists an $\hat{x} \in V$ such that $F(\hat{x}, \cdot)$ is bounded below on a neighbourhood of \bar{u} . Then, Φ is concave and continuous (hence usc) at \bar{u} . In particular, we then have

$$\Phi(\bar{u}) = cl \ co \ \Phi(\bar{u}).$$

Throughout this thesis, we apply the duality theory developed in this section for two different applications: constrained optimization over $V = H = \mathbb{R}^d$ and constrained optimization over $V = H = L_Q^2$ on some complete probability space (Ω, \mathcal{F}, Q) . We continue by developing useful duality results for these applications below.

Constrained Optimization over \mathbb{R}^d : Consider a real-valued usc function $f : \mathbb{R}^d \to \mathbb{R}$ and a non-empty closed convex set $K \subset \mathbb{R}^d$. Then, $V = H = \mathbb{R}^d$ and the standard scalar product $\langle x, y \rangle = x'y$ for $x, y \in \mathbb{R}^d$ form a Hilbert space on which we can apply the previously developed duality techniques. Consider the primal optimization problem

$$(\mathbf{P}) \begin{cases} \Phi_P &= \sup_{x \in \mathbb{R}^d} f(x) \\ \text{s.t.} & x \in K. \end{cases}$$
(2.13)

This setting is a special case of Example 2.1.20 throughout this Section. In particular, we can obtain the Lagrangian of (\mathbf{P}) directly from Example 2.1.20.

Lemma 2.1.29. Consider (P) as defined in (2.13) and define the associated Lagrangian as $L: \mathbb{R}^d \times \mathbb{R}^d \to \overline{\mathbb{R}}$ with

$$L(x,\lambda) = f(x) + x'\lambda + \delta_K(\lambda).$$

If (x^*, λ^*) is a saddle point of L, then x^* is optimal for (**P**).

Lemma 2.1.29 leads to the following dual optimization problem (\mathbf{D}) corresponding to (\mathbf{P}) from (2.13).

$$(\mathbf{D})\left\{\Psi_D = \inf_{\lambda \in \mathbb{R}^d} \left(\sup_{x \in \mathbb{R}^d} \left(f(x) + x'\lambda \right) + \delta_K(\lambda) \right).\right.$$

Similarly, we can derive the KKT conditions for (\mathbf{P}) and (\mathbf{D}) .

Corollary 2.1.30 (KKT Conditions). Consider (**P**) as defined in (2.13). $(x^*, \lambda^*) \in \mathbb{R}^d \times \mathbb{R}^d$ is a saddle point of L if and only if

(i)
$$x^* \text{ maximizes } f(x) + \langle x, \lambda^* \rangle + \delta_K(\lambda^*) \text{ over } x \in \mathbb{R}^d,$$

(ii) $(x^*)'\lambda^* + \delta_K(\lambda^*) = 0,$
(iii) $x^* \in K.$
(2.14)

Theorem 2.1.31. Consider (**P**) as defined in (2.13) and let $\lambda^* \in \mathbb{R}^d$ be optimal for (**D**). If there exists a function $x^* : \mathbb{R}^d \to \mathbb{R}^d$ such that

$$\sup_{x \in \mathbb{R}^d} L(x, \lambda) = L(x^*(\lambda), \lambda) \quad \forall \lambda \in X_K$$

and $x^*(\lambda)'\Delta\lambda$ is use at $\lambda = \lambda^*$ for all $\Delta\lambda \in X_K$, then $(x^*(\lambda^*), \lambda^*)$ satisfies (2.14). In particular, $x^*(\lambda^*)$ is optimal for (**P**) and $\Phi_P = \Psi_D$.

Corollary 2.1.32. Consider (\mathbf{P}) as defined in (2.13) and let

$$f(x) = -x'Ax + b'x + c_{x}$$

for a symmetric positive definite matrix $A \in \mathbb{R}^{d \times d}$, $b \in \mathbb{R}^d$ and $c \in \mathbb{R}$. Then, there exists a unique optimal λ^* for (**D**), the requirements of Theorem 2.1.31 are satisfied and in particular

$$\sup_{x \in K} f(x) = \inf_{\lambda \in \mathbb{R}^d} \sup_{x \in \mathbb{R}^d} L(x, \lambda) = \inf_{\lambda \in \mathbb{R}^d} \left(\sup_{x \in \mathbb{R}^d} \left(f(x) + x'\lambda \right) + \delta_K(\lambda) \right).$$

Constrained Optimization over L_Q^2 : Consider a complete probability space (Ω, \mathcal{F}, Q) and let L_Q^2 be the space of \mathcal{F} -measurable random variables X with $\mathbb{E}[X^2] < \infty$. Then, $V = L_Q^2$ with $\langle X, Y \rangle = \mathbb{E}[X \cdot Y]$ is a Hilbert space on which we can employ the previously developed duality theory. Further, consider a proper usc function $U : \mathbb{R} \to \overline{\mathbb{R}}$, a function $g = (g_1, ..., g_n) : \mathbb{R}^n \to \overline{\mathbb{R}}$ such that each g_i is usc and proper, and a non-empty closed convex set $K \subset \mathbb{R}^n$. Note that g is allowed to be random, as long as its values are \mathcal{F}_T -measurable. Then, we define the fully constrained optimization problem over L_Q^2 as

$$(\mathbf{P}) \begin{cases} \Phi_P &= \sup_{D \in L^2_Q} \mathbb{E}\left[U(D)\right] \\ \text{s.t.} & \mathbb{E}\left[g(D)\right] \in K. \end{cases}$$
(2.15)

Lemma 2.1.33. Consider (**P**) as defined in (2.15). Define the associated Lagrangian L: $L^2_Q \times \mathbb{R}^n \to \overline{\mathbb{R}}$ as

$$L(D, y) = \mathbb{E}\left[U(D)\right] + y' \mathbb{E}\left[g(D)\right] + \delta_K(y).$$
(2.16)

If (D^*, y^*) is a saddle point of L, then D^* is optimal for (\mathbf{P}) .

Lemma 2.1.33 leads to the following dual optimization problem (\mathbf{D}) corresponding to (\mathbf{P}) from (2.15).

$$(\mathbf{D})\left\{\Psi_D = \inf_{y \in \mathbb{R}^n} \left(\sup_{D \in L^2_Q} \left(\mathbb{E}\left[U(D)\right] + y'\mathbb{E}\left[g(D)\right]\right) + \delta_K(y)\right).$$
(2.17)

Definition 2.1.34. In the setting of (**P**) as defined in (2.15), we define the generalized conjugate U_g^* and its maximizing argument \mathcal{I}_g as

$$U_g^* : \mathbb{R}^n \to \bar{\mathbb{R}}, \quad U_g^*(y) = \sup_{x \in \mathbb{R}} \left(U(x) + y'g(x) \right)$$

$$\mathcal{I}_g : \mathbb{R}^n \to \bar{\mathbb{R}}, \quad \mathcal{I}_g(y) = \inf \left\{ x \in \mathbb{R} \mid U_g^*(y) \le U(x) + y'g(x) \right\}$$

$$(2.18)$$

Lemma 2.1.35. Consider (\mathbf{P}) as defined in (2.15) and (\mathbf{D}) as in (2.17). Then,

$$\Psi_D \le \inf_{y \in \mathbb{R}^n} \left(\mathbb{E} \left[U_g^*(y) \right] + \delta_K(y) \right), \tag{2.19}$$

with equality if the infimum is attained at $y^* \in \mathbb{R}^n$ such that $\mathcal{I}_g(y^*) \in L^2_Q$.

Note that it is not totally unreasonable that the infimum in 2.19 is attained, as $U_g^*(y) + \delta_K(y)$ is convex in y (see Lemmas 2.1.17 and Lemma 2.1.38). We may use the expression from Lemma 2.1.35 to again derive the KKT conditions for (**P**), respectively (**D**).

Corollary 2.1.36 (KKT Conditions). Consider (**P**) as in (2.15). Let $y^* \in \mathbb{R}^n$ be determined such that $\mathcal{I}_g(y^*) \in L^2_Q$,

$$(y^*)' \mathbb{E}\left[g(\mathcal{I}_g(y^*))\right] + \delta_K\left(y^*\right) = 0 \quad and \quad \mathbb{E}\left[g(\mathcal{I}_g(y^*))\right] \in K.$$
(2.20)

Then, $(\mathcal{I}_g(y^*), y^*)$ is a saddle-point of the Lagrangian L. In particular, $\mathcal{I}_g(y^*)$ is optimal for (\mathbf{P}) , y^* is optimal for (\mathbf{D}) and $\Phi_P = \Psi_D$.

Moreover, if the maximizing argument \mathcal{I}_g satisfies an upper semi-continuity assumption, we can show that any optimal solution $y^* \in \mathbb{R}^n$ to (**D**) satisfies (2.20). Under this assumption, any $y^* \in \mathbb{R}^n$ which is optimal for (**D**) directly defines a saddle-point of L and an optimal solution to (**P**) as shown in Corollary 2.1.36. Here, the maximizing argument \mathcal{I}_g takes the same role as x^* in Theorem 2.1.31.

Theorem 2.1.37. Consider (P) as in (2.15) and let $y^* \in \mathbb{R}^n$ attain the infimum in (2.19). If $\mathcal{I}_g(y^*) \in L^2_Q$ and

$$y \to \mathbb{E}\left[g(\mathcal{I}_g(y))\right]' \Delta y$$

is use at $y = y^*$ for all $\Delta y \in X_K$, then y^* satisfies (2.20). In particular, $\mathcal{I}_g(y^*)$ is optimal for (**P**) and $\Phi_P = \Psi_D$.

We summarize a number of useful properties of the generalized conjugate and its maximizing argument for use in later chapters.

Lemma 2.1.38.

- (i) If g_i is non-increasing and $g_i(x) \leq 0$ for all $x \in dom_U$, then $U_g^*(y)$ and $\mathcal{I}_g(y)$ are non-increasing in y_i .
- (ii) Define $\hat{g} = (g_1, \dots, g_{i-1}, \hat{g}_i, g_{i+1}, \dots, g_n)'$ for a given function $\hat{g}_i : \mathbb{R} \to \overline{\mathbb{R}}$. If g_i, \hat{g}_i are non-increasing, $g_i(x) \leq \hat{g}_i(x)$ for all $x \in dom_U$ and $y_i \geq 0$, then

$$U_q^*(y) \le U_{\hat{q}}^*(y)$$
 and $\mathcal{I}_g(y) \le \mathcal{I}_{\hat{g}}(y)$.

(iii) U_a^* is a convex function.

2.2 Ordinary Differential Equations

In this section, selected results from the theory of ordinary differential equations are summarized.

Lemma 2.2.1.

(i) Let $f : \mathbb{R} \to \mathbb{R}$ be a convex function. Then, for every $x_0 \in \mathbb{R}$, $\epsilon > 0$, there exists an L > 0 such that if $|x - x_0| < \epsilon$, $|y - x_0| < \epsilon$, then

$$|f(x) - f(y)| < L|x - y|.$$

In particular, f is locally Lipschitz continuous.

(ii) Consider an interval $I \subset \mathbb{R}$ and let $f_1, f_2 : I \to \mathbb{R}$ be Lipschitz continuous on I. Then, $f := min(f_1, f_2)$ is Lipschitz continuous on I.

Theorem 2.2.2. Let $f : \mathbb{R} \to \mathbb{R}$ be locally Lipschitz continuous on \mathbb{R} and $\tau_0, B_0 \in \mathbb{R}$ be constants. Consider the ordinary differential equation

$$B'(\tau) = f(B(\tau)), \quad B(\tau_0) = B_0.$$
 (2.21)

Then there exists an $\epsilon > 0$ such that the ODE (2.21) has a unique solution $B(\tau)$ for $\tau \in (\tau_0 - \epsilon, \tau_0 + \epsilon)$.

Lemma 2.2.3. Consider the setting of Theorem 2.2.2 and an open interval $I \subset \mathbb{R}$ with $\tau_0 \in I$, and let $B : I \to \mathbb{R}$ satisfy ODE (2.21). Then, B is a monotone function in τ . If $f(B_0) \neq 0$, then B is strictly monotone in τ . Moreover, if $f(B_0) = 0$, then B is constant in τ .

Lemma 2.2.4. Consider a terminal time point T > 0, real coefficients B_0 , r_0 , r_1 , r_2 and the following Riccati-ODE

$$B'(\tau) = -r_0 + r_1 B(\tau) + \frac{1}{2} r_2 B(\tau)^2, \quad B(0) = B_0,$$
(2.22)

where $r_2 \neq 0$, $r_1^2 + 2r_0r_2 > 0$ and define $r_3 = \sqrt{r_1^2 + 2r_0r_2}$.

(i) The function

$$B(\tau) = \frac{2r_2r_3B_0 + (e^{r_3\tau} - 1)(r_1 + r_3)(r_1 + r_2B_0 - r_3)}{2r_2r_3 - r_2(e^{r_3\tau} - 1)(r_1 + r_2B_0 - r_3)}$$
(2.23)

is the unique solution of equation (2.22) on its maximal interval $[0, t_+(B_0))$ with life-time $t_+(B_0) > 0$. Moreover, for any $T \in [0, t_+(B_0))$ we have

$$\int_{0}^{T} B(\tau) d\tau = \frac{2}{r_2} \ln \left(\frac{2r_3 e^{\frac{r_3 - B_0}{2}T}}{r_3 \left(e^{r_3 T} + 1 \right) - r_1 \left(e^{r_3 T} - 1 \right) - r_2 \left(e^{r_3 T} - 1 \right) B_0} \right).$$
(2.24)

(ii) The life time $t_{+}(B_0)$ of B (in the sense of Lemma 10.1 in [38]) is given as

$$t_{+}(B_{0}) = \begin{cases} \frac{1}{r_{3}} \ln \left(\frac{r_{1} + r_{2}B_{0} + r_{3}}{r_{1} + r_{2}B_{0} - r_{3}} \right), & \text{if } r_{1} + r_{2}B_{0} - r_{3} > 0\\ \infty, & \text{if } r_{1} + r_{2}B_{0} - r_{3} \le 0. \end{cases}$$

Corollary 2.2.5. Consider the setting of Lemma 2.2.4, let B be as in (2.23) and let $\hat{B} \in \mathbb{R}$ be given. If

$$\tau_{\hat{B}} := \frac{1}{r_3} \ln \left(\frac{2r_2 r_3 \left(\hat{B} - B_0 \right) + \left(r_1 + r_2 B_0 - r_3 \right) \left(r_1 + \hat{B} r_2 + r_3 \right)}{\left(r_1 + r_2 B_0 - r_3 \right) \left(r_1 + \hat{B} r_2 + r_3 \right)} \right) \le t_+(B_0)$$

then $B(\tau_{\hat{B}}) = \hat{B}$.

2.3 The Continuous-Time Portfolio Optimization Problem

2.3.1 Problem Formulation

Throughout this thesis we consider different variations of a continuous-time portfolio optimization problem with finite investment horizon T > 0. The underlying setting assumes a complete, filtered probability space $(\Omega, \mathcal{F}_T, \mathbb{F} = (\mathcal{F}_t)_{t \in [0,T]}, Q)$, where the filtration \mathbb{F} is generated by the independent *m*-dimensional Wiener process $W^z = (W^z(t))_{t \in [0,T]}$ and *d*-dimensional Wiener process $\hat{W} = (\hat{W}(t))_{t \in [0,T]}$. The set of all \mathcal{F}_T -measurable random variables will be denoted by \mathbf{X} and \mathbf{X}_+ denotes the restriction of \mathbf{X} to *Q*-a.s. non-negative random variables. Moreover, we consider an $\mathbb{R}^{m \times d}$ -valued stochastic process ρ , which is assumed to be progressively measurable with respect to the natural filtration of W^z and whose columns ρ_i , i = 1, ..., d satisfy

$$\|\rho_i(t,\omega)\| \leq 1$$
 $\mathcal{L}[0,T] \otimes Q$ – a.e..¹

This construction allows us to define another *d*-dimensional \mathbb{F} -Wiener process $W = (W_1, ..., W_d)'$ through the equations

$$W_i(t) = \rho_i(t)' W^z(t) + \sqrt{1 - \|\rho_i(t)\|^2} \hat{W}_i(t), \quad i = 1, ..., d.$$

The Wiener process W is defined in such way that it correlates with W^z , since its covariation with W^z satisfies

$$d\langle W_i^z, W_j \rangle_t = \rho_{ij}(t) dt \quad \mathcal{L}[0, T] \otimes Q - \text{a.e.}.$$

We employ this setting to define a market model \mathcal{M} with d risky assets (e.g. stocks) $P = (P_1, ..., P_d)'$ and a risk-free asset (e.g. a bank account) P_0 , which satisfy $P_0(0) = P_1(0) = ... = P_d(0) = 1$ and evolve according to the dynamics

$$dP(t) = \operatorname{diag} \left(P(t)\right) \left(\mu(t)dt + \Sigma(t)dW(t)\right)$$

$$dP_0(t) = P_0(t)r(t)dt, \qquad (2.25)$$

where we used diag $(P(t)) \in \mathbb{R}^{d \times d}$ to denote the diagonal matrix with diagonal entries $P(t) \in \mathbb{R}^d$. The volatility matrix is assumed to be $\mathcal{L}[0,T] \otimes Q$ -a.e. non-singular and thereby has a welldefined inverse.² Moreover, the market coefficients, i.e., the risk-free interest rate r, the mean rate of return μ as well as the volatility matrix Σ and its inverse Σ^{-1} with columns $\Sigma_1^{-1}, ..., \Sigma_d^{-1}$, are assumed to be progressively measurable processes w.r.t. \mathbb{F} such that

$$\sup_{t \in [0,T]} \left\{ \max\left(|r(t)|, \|\mu(t)\|^2, \|\Sigma(t)\|^2, \|\Sigma(t)^{-1}\|^2 \right) \right\} < \infty \quad Q\text{-a.s.},$$
(2.26)

where $\|\cdot\|$ denotes the standard Euclidean norm on \mathbb{R}^d and its induced matrix norm on $\mathbb{R}^{d \times d}$. This ensures that the SDEs in (2.25) have well-defined solutions. While the Wiener process W is the diffusion of the risky asset prices, the market coefficients may depend also on the Wiener process W^z . Unless W and W^z are perfectly correlated, this implies that not all randomness in the financial market \mathcal{M} can be hedged by trading in the risk-free and risky assets.

Within the financial market \mathcal{M} , we consider a single investor with initial wealth $v_0 > 0$ at time t = 0, who trades continuously in time. The investor's wealth process is assumed to satisfy $V^{v_0,\pi}(0) = v_0$ and the SDE

$$dV^{v_0,\pi}(t) = V^{v_0,\pi}(t) \Big([r(t) + (\mu(t) - r(t)\mathbb{1})'\pi(t)] dt + \pi(t)'\Sigma(t)dW(t) \Big).$$
(2.27)

The *d*-dimensional portfolio process π is chosen by the investor and determines the fraction of wealth $\pi_i(t)$ that is allocated to the risky asset P_i at time *t*, while the remaining fraction $1 - \sum_{i=1}^d \pi_i(t)$ is allocated to the risk-free asset. Note that $1 - \sum_{i=1}^d \pi_i(t)$ may be negative, in which case the investor goes short the risk-free asset, or more intuitively, borrows from the bank account. To ensure that the investor allocates his wealth solely based on past price developments and to ensure that (2.27) is well-defined, we restrict the admissible portfolio processes π to

$$\|\Sigma(t)'x\|^2 \ge \xi \|x\|^2, \quad \forall x \in \mathbb{R}^d, \quad \mathcal{L}[0,T] \otimes Q - a.e.,$$

 $^{{}^{1}\}mathcal{L}[0,T]$ denotes the Lebesgue-measure on [0,T], $\omega \in \Omega$ is used to denote the elements of the sample space Ω and $\|\cdot\|$ denotes the standard Euclidean norm.

²For this assumption it is for example sufficient if $\Sigma(t)\Sigma(t)'$ is strongly positive definite $\mathcal{L}[0,T] \otimes Q - a.e.$, i.e., if

for some constant $\xi > 0$.

$$\Lambda = \left\{ \pi = \left((\pi_1(t), ..., \pi_d(t))' \right)_{t \in [0,T]} \text{ progr. measurable } \middle| \mathbb{E} \left[(V^{v_0, \pi}(T))^2 \right] < \infty, \\ \int_0^T \|\pi(t)\|^2 \, dt < \infty \ Q - a.s. \right\}$$
(2.28)

In particular, if $\pi \in \Lambda$, then the investor's terminal wealth has finite variance and it is straightforward to show that the unique solution to (2.27) is given by

$$V^{v_0,\pi}(t) = v_0 \exp\left(\int_0^t r(s) + (\mu(s) - r(s)\mathbb{1})'\pi(s) - \frac{1}{2} \left\|\Sigma(s)'\pi(s)\right\|^2 ds + \int_0^t \pi(s)'\Sigma(s)dW(s)\right).$$

We assume that the investor aims to choose his portfolio process $\pi \in \Lambda$ to maximise his utility derived from terminal wealth $V^{v_0,\pi}(T)$ at time T. Thus, we choose an appropriate class of utility functions which can capture the investor's risk preferences in a flexible way. For this purpose we define the class of utility functions \mathcal{U} , which contains all functions $U : \mathbb{R} \to \mathbb{R} \cup \{\infty, -\infty\} =: \mathbb{R}$ such that

> (i) U is proper and use and $dom_U = [c, \infty)$ for some $c \ge 0$ (ii) U is strictly increasing on dom_U (iii) $U(x) = -\infty \quad \forall x < 0$ (iv) $\lim_{x \to \infty} \frac{U(x)}{x} = 0.$ (2.29)

If U is finite on $[0, \infty)$, differentiable and strictly increasing (as in [17]), then (iv) is equivalent to the condition $\lim_{x\to\infty} U'(x) = 0$. We set $U(\infty) = \lim_{x\to\infty} U(x)$. It is fairly common to restrict this analysis to concave utility functions only ('less risk is preferred to more'), and we will do so throughout the ensuing chapters when appropriate. However, for a general formulation in the context of constraints on allocation and terminal wealth, such restrictions may hinder the exposition. We define the unconstrained portfolio optimization problem for our investor as

$$(\mathbf{P^{unc}}) \left\{ \Phi^{unc}(v_0) = \sup_{\pi \in \Lambda} \mathbb{E} \left[U \left(V^{v_0, \pi}(T) \right) \right].$$

The unconstrained portfolio optimization problem $(\mathbf{P^{unc}})$ plays an important role in this thesis, as the addition of constraints on portfolio allocation or terminal wealth are a natural generalization of $(\mathbf{P^{unc}})$. In particular, any method used to solve a constrained portfolio optimization problem is to some extent a generalization of a method for solving $(\mathbf{P^{unc}})$. In this thesis we consider two different definitions of constraints:

- (i) Constraints on the portfolio allocation π .
- (ii) Constraints on the terminal wealth $V^{v_0,\pi}(T)$.

Naturally, any constraint on the portfolio allocation automatically induces a constraint on the terminal wealth and vice versa. However, during this thesis we aim to treat these constraints as separate, since they are defined in completely different ways. The allocation constraints considered in this thesis are defined by considering a convex set $K \subset \mathbb{R}^d$ and requiring that the investor's portfolio process π takes values only in K, i.e.,

$$\pi(t) \in K \quad \mathcal{L}[0,T] \otimes Q$$
-a.e..

Classic examples for such constraints would be $K = [0, \infty)^d$ (no-shortselling), $K = 0^{d_1} \times \mathbb{R}^{d-d_1}$ for a $1 \le d_1 \le d$ (non-traded assets) or $K = \bigotimes_{i=1}^d [\alpha_i, \beta_i]$ for some constants $\alpha_i < \beta_i, i = 1, ..., d$

(interval constraints).

In contrast, terminal wealth constraints are formulated directly as restrictions on the admissible terminal wealths $V^{v_0,\pi}(T)$. A specific example would be pointwise bounds $B_1, B_2 \in \mathbb{R}$ with the requirement that

$$B_1 \le V^{v_0,\pi}(T) \le B_2 \quad Q-\text{a.s.}.$$
 (2.30)

Such bounds could correspond to a minimum performance guarantee that the investor needs to fulfil or a maximal performance limit which is supposed to discourage excessive risk-taking. Alternatively, we consider expected-value constraints for a non-increasing real function g: $(0, \infty) \rightarrow \mathbb{R}$ such that

$$\mathbb{E}\left[g\left(V^{v_0,\pi}(T)\right)\right] \le 0.$$
(2.31)

Letting $g(x) = \mathbb{1}_{[0,B_{VaR}]}(x) - \epsilon$ we obtain a Value-at-Risk ('VaR') constraint with lower bound B_{VaR} and tolerance level $\epsilon \geq 0$. Similarly, $g(x) = (B_{ES} - x)\mathbb{1}_{[0,B_{ES}]}(x) - \epsilon$ yields an expected shortfall ('ES') constraint with lower bound B_{ES} and tolerance level $\epsilon \geq 0$.

Both types of constraints lead to a restriction $\Lambda' \subset \Lambda$ of the set of admissible portfolio processes. For such a general constraint set $\Lambda' \subset \Lambda$, we define the general constrained portfolio optimization problem (**P**) as

$$(\mathbf{P}) \begin{cases} \Phi(v_0) &= \sup_{\pi \in \Lambda'} \mathbb{E} \left[U \left(V^{v_0, \pi}(T) \right) \right] \end{cases}$$

We impose the standing assumption throughout this thesis that both, the constrained portfolio optimization problem (\mathbf{P}) and the unconstrained portfolio optimization problem (\mathbf{P}^{unc}) are feasible and yield finite, well-defined expected utility:

Assumption 2.3.1 (Standing Assumption).

$$-\infty < \Phi(v_0) \le \Phi^{unc}(v_0) < \infty \quad \forall v_0 > 0.$$

(**P**) is the most general portfolio optimization problem considered in this thesis. Unsurprisingly, we are unable to find explicit solutions or useful characterizations of solutions for (**P**) in this general form. However, if we impose more specific assumptions on the market parameters, the utility function and the constraints, then we can be more successful. In this sense, each of the following chapters works with specific assumptions which constitute a special case of (**P**) as defined in this section. Each of the following chapters, aims to utilize different duality techniques from convex analysis, which we developed in Section 2.1 to obtain solutions to (**P**).

2.3.2 Classic Methods for Unconstrained Portfolio Optimization

In this section we provide a short review of the two most common approaches which are used to solve the unconstrained portfolio optimization problem (\mathbf{P}^{unc}) for selected models and utility functions: The stochastic control approach and the martingale method. This review will help us to understand the limitation and challenges of each approach. Moreover, this review will allow us to highlight the extensions that are required in later Chapters, when we are facing constraints on the portfolio allocation and terminal wealth.

Stochastic Control Approach: The unconstrained portfolio optimization problem ($\mathbf{P^{unc}}$) can be regarded as a stochastic control problem, with control process π , controlled process $V^{v_0,\pi}$, and maximization objective $\mathbb{E}[U(V^{v_0,\pi}(T))]$. This is the approach followed in the seminal works [64] and [65], where the continuous time portfolio optimization problem was originally introduced.

If the wealth process (i.e., the 'state of the investor') and the market coefficients (i.e., the 'state of the market \mathcal{M} ' can be jointly represented as Markov process $X = (X(t))_{t \in [0,T]}$, then principles from dynamic programming are applicable. This can be assured by restricting the analysis to stochastic factor models, i.e., assuming that

• there exists a stochastic factor $z = (z(t))_{t \in [0,T]}$, which takes values in \mathbb{R}^m and satisfies the SDE

$$dz(t) = \mu^{z}(t, z(t))dt + \Sigma^{z}(t, z(t))dW^{z}(t), \quad z(0) = z_{0} \in \mathbb{R}^{m},$$

for some constant $z_0 \in \mathbb{R}^m$ and deterministic functions $\mu^z : [0,T] \times \mathbb{R}^m \to \mathbb{R}^m$ and $\Sigma^z : [0,T] \times \mathbb{R}^m \to \mathbb{R}^{m \times m}$.

- the market coefficients r(t), $\mu(t)$, $\Sigma(t)$, and $\rho(t)$ are deterministic function of the current time and value of the stochastic factor (t, z(t)).
- the current portfolio allocation is given as a deterministic function of time, current wealth and value of the stochastic factor $(t, V^{v_0,\pi}(t), z(t))$, i.e., there exists a deterministic function $\underline{\pi} : [0, T] \times (0, \infty) \times \mathbb{R}^m \to \mathbb{R}^d$ such that

$$\pi(t) = \pi(t, V^{v_0, \pi}(t), z(t)).$$

In such cases, we say that the portfolio process π is defined in 'feedback-form' and $\underline{\pi}$ is the corresponding feedback control. In particular, we may define a wealth process directly through a feedback control. For a given feedback control $\underline{\pi}$, we may then define the state process $X^{\pi} = (V^{v_0,\pi}, z)'$ and one can show that X^{π} satisfies an SDE of the form

$$dX^{\pi}(t) = \mu^{X^{\pi}}(t, X^{\pi}(t))dt + \Sigma^{X^{\pi}}(t, X^{\pi}(t))dW(t) + \Sigma^{X^{\pi}, z}(t, X^{\pi}(t))dW^{z}(t),$$

and deterministic functions $\mu^{X^{\pi}}, \Sigma^{X^{\pi}}, \Sigma^{X^{\pi},z}$. Then, we can define the dynamic version of the unconstrained portfolio optimization problem (**P**^{unc}) as

$$(\mathbf{P_{t,x}^{unc}}) \begin{cases} \Phi^{unc}(t,x) &= \sup_{\pi \in \Lambda_f} \mathbb{E} \left[U\left(V^{v_0,\pi}(T) \right) \left| X^{\pi}(t) = x \right] \\ \Lambda_f &= \left\{ \pi \in \Lambda \mid \pi(s) = \pi(s, V^{v_0,\pi}(s), z(s)) = \pi(s, X^{\pi}(s)) \quad \forall s \in [t,T] \\ & \text{and a feedback control } \underline{\pi} \right\}. \end{cases}$$

If the feedback control $\underline{\pi}^*$ defines an optimal portfolio process π^* for $(\mathbf{P}_{\mathbf{t},\mathbf{x}}^{\mathbf{unc}})$ and each $(t,x) \in [0,T] \times (0,\infty) \times \mathbb{R}^m$, then the optimality of π^* and Markovity of X^{π^*} ensure that

$$\Phi^{unc}(t, X^{\pi^*}(t)) = \sup_{\pi \in \Lambda_f} \mathbb{E} \Big[U(V^{v_0, \pi}(T)) \Big| X^{\pi}(t) = X^{\pi^*}(t) \Big]$$
$$\stackrel{\pi^* \text{ optimal}}{=} \mathbb{E} \Big[U(V^{v_0, \pi^*}(T)) \Big| X^{\pi^*}(t) \Big]$$
$$\stackrel{\text{Markovity } X^{\pi^*}}{=} \mathbb{E} \Big[U(V^{v_0, \pi^*}(T)) \Big| \mathcal{F}_t \Big].$$

Hence, it is straightforward to see that $\Phi^{unc}(t, X^{\pi^*}(t))$ is a martingale. Similarly, we have for any suboptimal portfolio process $\hat{\pi} \in \Lambda^3$ and $s \leq t \leq T$:

$$\mathbb{E}\left[\Phi^{unc}(t, X^{\hat{\pi}}(t)) \middle| \mathcal{F}_s\right] = \mathbb{E}\left[\mathbb{E}\left[U(V^{v_0, \pi^*}(T)) \middle| X^{\pi^*}(t) = X^{\hat{\pi}}(t)\right] \middle| \mathcal{F}_s\right]$$

³Note that the subsequent argument does not even require that the suboptimal portfolio process $\hat{\pi}$ is defined in feedback-form.

$$\begin{split} \stackrel{\text{Markovity } X^{\pi^*}}{=} & \mathbb{E}\left[\mathbb{E}\left[U(V^{v_0,\pi^*}(T))\Big|X^{\pi^*}(t) = X^{\hat{\pi}}(t), \ X^{\pi^*}(s) = X^{\hat{\pi}}(s)\right]\Big|\mathcal{F}_s\right] \\ \stackrel{\hat{\pi} \text{ suboptimal}}{\leq} & \mathbb{E}\left[\mathbb{E}\left[U(V^{v_0,\pi^*}(T))\Big|X^{\pi^*}(s) = X^{\hat{\pi}}(s)\right]\Big|\mathcal{F}_s\right] \\ &= & \mathbb{E}\left[\Phi(s,X^{\hat{\pi}}(s))\Big|\mathcal{F}_s\right] \\ &= & \Phi^{unc}(s,X^{\hat{\pi}}(s)), \end{split}$$

i.e., $\Phi^{unc}(t, X^{\hat{\pi}}(t))$ is a supermartingale. Therefore, under some regularity conditions (see Chapter III, Theorem 16 in [74]), the Doob-Meyer decomposition suggests that $\Phi^{unc}(t, X^{\hat{\pi}}(t))$ can be uniquely decomposed into the sum of a non-increasing, non-positive process A(t) and a local martingale M(t). If we additionally assume that Φ^{unc} is once continuously differentiable in $t \in [0, T]$ and twice continuously differentiable in $x \in (0, \infty) \times \mathbb{R}^m$, then we can define the operator $\mathcal{H}^{\pi}[\Phi^{unc}]^4$ as (omitting the arguments (t, x) for readability)

$$\mathcal{H}^{\pi}[\Phi^{unc}] = \Phi_t^{unc} + (\mu^{X^{\pi}})' \nabla_x \Phi^{unc} + \frac{1}{2} \operatorname{Trace} \left[\begin{pmatrix} \Sigma^{X^{\pi}} & \rho' \\ \rho & \Sigma^{X^{\pi},z} \end{pmatrix} \begin{pmatrix} \Sigma^{X^{\pi}} & \rho' \\ \rho & \Sigma^{X^{\pi},z} \end{pmatrix}' \nabla_x^2 \Phi^{unc} \right]$$

Then, an application of Itô's Lemma yields

$$d\Phi^{unc}(t, X^{\hat{\pi}}(t)) = \underbrace{\mathcal{H}^{\pi}[\Phi^{unc}](t, X^{\hat{\pi}}(t))dt}_{=dA(t)} + \underbrace{(\dots)dW(t) + (\dots)dW^{z}(t)}_{=dM(t)}.$$

As A(t) is non-positive and non-increasing for any $\pi \in \Lambda$, the Doob-Meyer decomposition implies that $\mathcal{H}^{\pi}[\Phi^{unc}](t, X^{\hat{\pi}}(t)) \leq 0 \ Q \times \mathcal{L}[0, T]$ -a.e., with equality for the optimal $\pi = \pi^* \in \Lambda_f$. Thus, we obtain a characterization of $\Phi^{unc}(t, x)$ and the optimal feedback control $\pi^*(t, x)$ for all $(t, x) \in [0, T) \times (0, \infty) \times \mathbb{R}^m$ through a PDE, the so-called Hamilton-Jacobi-Bellman PDE ('HJB PDE'):

$$\mathcal{H}^{\pi}[\Phi^{unc}](t,x) = 0 \quad \text{and} \quad \underline{\pi}^*(t,x) \in \operatorname*{argmax}_{\pi \in \mathbb{R}^d} \mathcal{H}^{\pi}[\Phi^{unc}](t,x), \tag{2.32}$$

with terminal condition $\Phi^{unc}(T, x) = \Phi(t, v, z) = U(v)$ for all $x \in (0, \infty) \times \mathbb{R}^m$. We have arrived at the HJB PDE (2.32) under the assumption that an optimal feedback control exists and the corresponding value function Φ^{unc} is smooth. In contrast, the stochastic control approach begins by determining a solution to the HJB PDE (2.32) and then proves a 'verification theorem' which ensures that the obtained PDE-solution is indeed the value function Φ^{unc} . While this approach is valid for general (Markovian) market models \mathcal{M} (i.e., m > 0 and deterministic functions $\mu^z, \Sigma^z, \rho, r, \mu$, and Σ), actually obtaining an explicit solution to the HJB PDE as well as proving a verification theorem is extremely challenging for complex models. In practice, this limits the usefulness of the stochastic control approach to models and utility functions where additional information about the functional form Φ^{unc} is known or can be guessed, such as CRRA utility functions or exponentially affine models.

Martingale Method: The martingale method goes back to [73], [47] and [15] and is closely linked to risk-neutral pricing in continuous time. The central idea underlying the martingale method is to split the portfolio optimization problem (\mathbf{P}^{unc}) from an optimization over all admissible portfolio processes into a 'static' optimization problem over all attainable terminal wealths and a subsequent 'hedging' problem, which seeks to determine the portfolio process that replicates the optimal terminal wealth for the static optimization problem.

⁴Throughout this thesis, this differential operator is mainly used within the context of Hamilton-Jacobi-Bellman PDEs and is therefore denoted as ' \mathcal{H} '.

If the underlying financial market \mathcal{M} is complete (i.e., all reasonable \mathcal{F}_T -measurable payoffs can be represented as a terminal wealth $V^{v_0,\pi}(T)$ for some $v_0 > 0$ and $\pi \in \Lambda$), then the static optimization problem will simplify considerably and allow a convenient characterization of the optimal terminal wealth for ($\mathbf{P}^{\mathbf{unc}}$). To make the notion of market completeness more precise, we define the market price of risk γ and pricing kernel \tilde{Z} as

$$\begin{split} \gamma(t) &:= \Sigma(t)^{-1} \big(\mu(t) - r(t) \mathbb{1} \big), \\ \tilde{Z}(t) &:= \exp\left(-\int_0^t r(s) + \frac{1}{2} \| \gamma(s) \|^2 ds - \int_0^t \gamma(s)' dW(s) \right) \end{split}$$

and consider a Lemma which we will revisit in Chapter 3:

Lemma (Market Completeness, cf. Lemma 3.3.1). Let m = 0 and consider $D \in \mathbf{X}_+$ with $\mathbb{E}[D^2] < \infty$ and $0 < \mathbb{E}[\tilde{Z}(T)D] =: v_0 < \infty$. Then, there exists a $\pi \in \Lambda$ so that $V^{v_0,\pi}(T) = D$ Q-a.s. and

$$V^{v_0,\pi}(t) = \mathbb{E}\left[D\frac{\tilde{Z}(T)}{\tilde{Z}(t)} \mid \mathcal{F}_t\right].$$
(2.33)

Therefore, if m = 0, i.e., all uncertainty in \mathcal{M} is generated by the diffusion of the risky asset's log returns (dW(t)), then we can consider the following equivalent static representation of $(\mathbf{P^{unc}})$:

$$(\mathbf{P^{unc}}) \begin{cases} \Phi^{unc}(v_0) &= \sup_{D \in C(v_0)} \mathbb{E}\left[U(D)\right] \\ C(v_0) &= \left\{ V^{v_0,\pi}(T) \mid \pi \in \Lambda \right\} \\ &\stackrel{(*)}{=} \left\{ D \in L^2_Q \mid \mathbb{E}[\tilde{Z}(T)D] = v_0 \right\}. \end{cases}$$

where we used the market completeness and the fact that $U(x) = -\infty$ for all x < 0 in (*). This is a constrained optimization problem over L_Q^2 , as considered in Section 2.1.2 and it can be treated with the same duality techniques. In particular, if we set $g(x) = -\tilde{Z}(T)x$ and $K = \{-v_0\}$, then $\delta_K(y_0) = y_0v_0$ and the associated dual optimization problem is

$$(\mathbf{D^{unc}})\left\{\Psi_{D^{unc}} = \inf_{y_0 \in \mathbb{R}} \mathbb{E}[U_g^*(y)] + y_0 v_0.\right.$$

Hence, if there exists a $y_0^* \in \mathbb{R}$ such that the KKT conditions of Corollary 2.1.36 are satisfied, i.e., $\mathcal{I}_g(y_0^*) \in L^2_Q$ and

$$0 = y_0^* \mathbb{E}\left[g\left(\mathcal{I}_g(y_0^*)\right)\right] + \underbrace{\delta_K(y_0^*)}_{=y_0^*v_0} = y_0^* \left(\mathbb{E}\left[-\mathcal{I}_g(y_0^*)\tilde{Z}(T)\right] + v_0\right) \& \mathbb{E}\left[g\left(\mathcal{I}_g(y_0^*)\right)\right] \in K$$

$$\Leftrightarrow \quad \mathbb{E}\left[\mathcal{I}_g(y_0^*)\tilde{Z}(T)\right] = v_0, \tag{2.34}$$

then $D^* = \mathcal{I}_g(y_0^*)$ is the optimal terminal wealth for $(\mathbf{P^{unc}})$ and thus a solution to the static optimization problem. Under mild regularity conditions on the utility function and the market coefficients, the existence of a y_0^* satisfying (2.34) can be guaranteed (see e.g. Theorem 7.4 in [17] or Lemma 3.1 in [34]) and can be determined numerically (as \mathcal{I}_g is non-increasing in y_0 by Lemma 2.1.38, such numerical procedures are even reasonably reliable). Note that we arrived at this characterization of the optimal terminal wealth for $(\mathbf{P^{unc}})$ without any major structural assumptions on the utility function U, the market coefficients or the optimal portfolio process π^* .

In contrast, solving the hedging problem, i.e., determining an explicit representation for the optimal portfolio π^* such that $V^{v_0,\pi^*}(T) = \mathcal{I}_g(y_0^*)$, is no longer feasible in this general setting.

While the market completeness ensures the existence of a replicating portfolio for $\mathcal{I}_g(y_0^*)$, it does not offer a direct way of constructing this portfolio. However, we do obtain a characterization of the optimal wealth process through a conditional expectation by virtue of (2.33):

$$\tilde{Z}(t)V^{v_0,\pi^*}(t) = \mathbb{E}\left[\mathcal{I}_g(y_0^*)\tilde{Z}(T)\middle|\mathcal{F}_t\right].$$
(2.35)

The right side of (2.35) is trivially a martingale and thus the left side is also a martingale. Hence, the left side is an Itô process with zero drift and SDE

$$d\Big(\tilde{Z}(t)V^{v_0,\pi^*}(t)\Big) \stackrel{\text{Itô's product rule}}{=} \tilde{Z}(t)V^{v_0,\pi^*}(t)\Big(\pi^*(t)'\Sigma(t)-\gamma(t)'\Big)dW(t).$$

Thus, if we can obtain an SDE for the right side of (2.35), then we can determine π^* by matching the diffusions on the left and right. For example, this can be achieved by using the Feynman-Kac representation, which establishes another link to partial differential equations:

Theorem 2.3.2. Let $n \in \mathbb{N}$, \overline{W} be an n-dimensional Wiener process and X be the unique solution to the stochastic differential equation $X(0) = x \in \mathbb{R}^n$,

$$dX(t) = \mu^X(t, X(s))dt + \Sigma^X(t, X(t))d\bar{W}(t),$$

for deterministic functions $\mu^X : [0,T] \times \mathbb{R}^n \to \mathbb{R}^n$, $\Sigma^X : [0,T] \times \mathbb{R}^n \to \mathbb{R}^{n \times n}$. For a function $u : [0,T] \times \mathbb{R}^n \to \mathbb{R}$, which is continuously differentiable in [0,T] and twice continuously differentiable in x, we define the operator \mathcal{H}^X applied to u as (again omitting arguments (t,x) for readability)

$$\mathcal{H}^{X}[u] = u_{t} + (\mu^{X})' \nabla_{x} u + Trace\left(\Sigma^{X} \left(\Sigma^{X}\right)' \nabla_{x}^{2} u\right).$$

If $u: [0,T] \times \mathbb{R}^n \to \mathbb{R}$ is a continuous solution to the PDE

$$\mathcal{H}^{X}[u](t,x) = 0, \quad u(T,x) = f(x) \quad \forall t \in [0,T) \forall x \in \mathbb{R}^{n},$$
(2.36)

for a continuous terminal condition $f : \mathbb{R}^m \to [0, \infty)$ and

• there exists a constant K > 0 such that for all $x, y \in \mathbb{R}^n$

$$\begin{aligned} \left\| \mu^{X}(t,x) - \mu^{X}(t,y) \right\| + \left\| \Sigma^{X}(t,x) - \Sigma^{X}(t,y) \right\| &\leq K \left\| x - y \right\|, \\ \left\| \mu^{X}(t,x) \right\|^{2} + \left\| \Sigma^{X}(t,x) \right\|^{2} &\leq K^{2} \left(1 + \|x\|^{2} \right). \end{aligned}$$

• there exist constants M > 0, $\epsilon > 1$ such that for all $x \in \mathbb{R}^n$

$$\max_{0 \le t \le T} |u(t, x)| \le M \left(1 + \|x\|^{2\epsilon} \right),$$

then u satisfies the Feynman-Kac representation

$$u(t,x) = \mathbb{E}\Big[f(X(T))\Big| X(t) = x\Big].$$

The PDE (2.36) is also referred to as 'Cauchy problem' or 'Feynman-Kac PDE'. If now \tilde{Z} is a Markov process⁵, then we can aim to apply Theorem 2.3.2 with $X = \tilde{Z}$ and terminal condition

$$f(z) = \left(\mathcal{I}_g(y_0^*) \tilde{Z}(T) \right) \Big|_{\tilde{Z}(T)=z}$$

⁵The ensuing arguments can easily be generalized if \tilde{Z} is a component of a multi-dimensional Markov process X (similar to the relation between $V^{v_0,\pi}$ and X^{π} in the stochastic control approach).

$$= \left(\inf \left\{ x \in \mathbb{R} \mid \sup_{\hat{x} \in \mathbb{R}} \left(U(\hat{x}) - y_0^* z \hat{x} \right) \le U(x) - y_0^* z x \right\} \right) z.$$

As the stochasticity of g (and U_g^* , \mathcal{I}_g) only depends on the value of $\tilde{Z}(t)$, the terminal condition f above is a deterministic function in z. If u is a solution to the Feynman-Kac PDE (2.36) with terminal condition f such that the requirements of Theorem 2.3.2 are satisfied, then (2.35) yields for any $t \in [0, T]$:

$$\begin{aligned} v_{0} + \int_{0}^{t} \tilde{Z}(s) V^{v_{0},\pi^{*}}(s) \Big(\pi^{*}(s)'\Sigma(s) - \gamma(s)'\Big) dW(s) &= \tilde{Z}(t) V^{v_{0},\pi^{*}}(t) \\ \stackrel{(2.35)}{=} & \mathbb{E} \left[\mathcal{I}_{g}(y_{0}^{*}) \tilde{Z}(T) \Big| \mathcal{F}_{t} \right] \\ \stackrel{\text{Markovity of } \tilde{Z}}{=} & \mathbb{E} \left[f(\tilde{Z}(T)) \Big| \tilde{Z}(t) \right] \\ \stackrel{T. 2.3.2}{=} & u(t, \tilde{Z}(t)) \\ \stackrel{\text{Itô}}{=} & \underbrace{u(0, \tilde{Z}(0))}_{v_{0}} + \underbrace{\int_{0}^{t} (...) dt}_{=0} - \int_{0}^{t} \tilde{Z}(s) u_{z}(s, \tilde{Z}(s)) \gamma(s)' dW(s) \\ \\ &= & v_{0} - \int_{0}^{t} \tilde{Z}(s) u_{z}(s, \tilde{Z}(s)) \gamma(s)' dW(s). \end{aligned}$$

As this equation holds for all $t \in [0, T]$, we conclude that

$$\pi^{*}(t) = \underbrace{\frac{1}{V^{v_{0},\pi^{*}}(t)} \left(V^{v_{0},\pi^{*}}(t) - u_{z}(t,\tilde{Z}(t)) \right)}_{=:\alpha^{*}(t,V^{v_{0},\pi^{*}}(t),\tilde{Z}(t))} \underbrace{\left(\Sigma(t)\Sigma(t)' \right)^{-1} \left(\mu(t) - r(t)\mathbb{1} \right)}_{=:\pi_{M}(t)} \\ = \alpha^{*}(t,V^{v_{0},\pi^{*}}(t),\tilde{Z}(t))\pi_{M}(t)$$

Note that this expression for π^* demonstrates that the optimal portfolio process satisfies a version of a two fund theorem: For any utility function $U \in \mathcal{U}$, such that the optimal π^* can be determined using the Feynman-Kac-representation, we can decompose π^* as a linear combination of the risk-free portfolio $\pi_0 \equiv 0$ and the Merton portfolio π_M (for risk aversion 1) with stochastic weighting factor α^* . In particular, the optimal portfolio for all such utility functions have the same relative allocation between risky assets in \mathcal{M} and only their exposure to the risky assets is influenced by the utility function.⁶

Discussion: The stochastic control approach is applicable for general Markovian market models, without requiring the completeness of the financial market. In particular, the stochastic control approach can be applied in incomplete markets, such as Heston's stochastic volatility model ([56]), whereas the martingale method cannot be applied directly. However, actually obtaining a closed-form expression for the optimal portfolio process requires the solution to the associated HJB PDE, for which explicit solutions are only known for selected combinations of market models and utility functions (e.g. for exponentially affine models and CRRA utility functions as considered in [61]).

The martingale method is only directly applicable in complete financial markets, i.e. when all uncertainty in \mathcal{M} is generated by the Wiener process W, which is the diffusion of the risky asset's log returns. Therefore, it is not directly applicable to most stochastic volatility models

⁶A similar statement can be derived via the stochastic control approach if we explicitly write out $\mu^{X^{\pi}}$, $\Sigma^{X^{\pi}}$, $\Sigma^{X^{\pi},z}$ and determine an explicit expression for the (optimal) portfolio which maximizes the characteristic operator $\mathcal{H}^{\pi}[\Phi^{unc}]$.

unless the market is completed through the addition of (fictitious) volatility-dependent assets (as in [78]). However, unlike the stochastic control approach, the martingale method yields a characterization of the optimal terminal wealth $V^{v_0,\pi^*}(t) = \mathcal{I}_g(y_0^*)$ for ($\mathbf{P^{unc}}$), which is explicit up to $y_0^* \in \mathbb{R}$ – the solution to the budget equation (2.34). An explicit expression for the corresponding optimal portfolio π^* is more challenging to obtain and requires either the Markovity of the pricing kernel and a solution to the associated Feynman-Kac PDE or an explicit representation of the expectation in (2.35) as a stochastic integral. Therefore, explicit expressions for π^* have again only been derived for selected utility functions and market models.

Hence, the martingale method is ideally suited for characterizing the portfolio optimization problem and its associated optimal terminal wealth for a large, general class of utility functions and models - as long as the financial market is complete. Determining an explicit expression for the optimal portfolio process requires the solution to an associated partial differential equation in both the stochastic control approach and the martingale method. However, as the stochastic control approach can be directly applied to incomplete markets, it is the natural choice for this objective.

Following this logic, we extend the classic martingale method in Chapter 3, where we characterize the optimal terminal wealth of portfolio optimization problem (\mathbf{P}) with simultaneous constraints on terminal wealth and allocation for general utility functions and complete financial markets. In contrast, we extend the classic stochastic control approach in Chapter 4, where we determine (semi-)explicit expressions for the optimal portfolio process to a portfolio optimization problem (\mathbf{P}) with constraints on the portfolio allocation in (incomplete) stochastic factor models. A special case of these models, Heston's stochastic volatility model, is discussed in great detail in Chapter 5.
3 Constraints on Allocation and Wealth in Complete Financial Market Models

3.1 Introduction

In this chapter, we consider a finite-horizon portfolio optimization problem for an expected utility maximizing investor who trades in continuous time. The investor's portfolio choice is restricted by convex allocation constraints as well as pointwise bounds and expected value constraints on terminal wealth. These constraints are the main difference of our setting in comparison to the classic problem originally introduced by [64]. We give a short overview of the related literature on wealth-constrained portfolio optimization:

In a portfolio optimization context, the two most commonly considered constraints on terminal wealth are either pointwise bounds on terminal wealth or expected value constraints. As suggested by their name, pointwise bounds require that the terminal wealth at the end of the investment horizon lies within a (possibly random) real interval $[B_1, B_2] \subset \mathbb{R}$. The natural choice of a lower bound $(0 \le B_1 < \infty, B_2 = \infty)$, or 'minimum performance constraint', was first considered and analyzed by [82], who assumed classic Black-Scholes market dynamics and derived closed-form expressions for the optimal portfolio under the condition that the investor derives utility from a power utility function and the lower bound B_1 is deterministic. Later, [52] considered a similar problem and presented a two-step approach for portfolio optimization problems with lower bound on terminal wealth in complete financial market models. The approach is based on using a share of the investor's initial wealth to replicate the lower bound and then use the remaining share of the initial wealth to maximize the expected utility surplus over the lower bound. Despite being unconventional, [25] considered a pointwise upper bound on terminal wealth $(B_2 < \infty)$ and illustrated that such upper bounds on terminal wealth increase the quantiles of the terminal wealth distribution, i.e., decrease the likelihood of poor performance of the portfolio. This is especially relevant for investments in a retirement plan.

On the other hand, expected value constraints impose a restriction on the expected value of a function $g(V^{v_0,\pi}(T))$ of the investors terminal wealth. As such, these type of constraints allow for 'softer' restrictions than strict pointwise bounds on terminal wealth. In [55], the authors propose a dual approach for general convex expected value constraints and showcase the developed approach on a mean-variance optimization problem. [4] uses a similar duality approach to solve portfolio optimization problems with Value-at-Risk ('VaR') or expected shortfall ('ES') constraints. The authors of [57] effectively consider the same problem setting as [4] with VaR and ES constraints, but use a novel methodology based on dynamic programming to solve these optimization problems. Expected value constraints based on more general risk measures were considered in [72], [67], [11] and [12].

Up until this point only few papers considered simultaneous constraints on portfolio allocation and terminal wealth. [3] extends the auxiliary market framework of [17] to include random lower bounds on terminal wealth and proves a set of analogue equivalent optimality conditions. However, the author failed to explicitly determine neither the optimal relative portfolio process nor the optimal terminal wealth for any example within his setting. [24] and [35] consider simultaneous VaR constraints and convex cone constraints in a Black-Scholes financial market, but their approaches do not directly extend to general constraints and models. A different type of simultaneous constraints was considered by [36]. The authors describe the Solvency II capital requirements as a wealth-dependent allocation constraint and transform the associated portfolio optimization problem into an equivalent problem with constant allocation constraints, which falls within the scope of [17]. None of the aforementioned papers, aside from [3], aimed to provide a general solution framework for wealth and allocation constrained portfolio optimization problems. However, [3] did not provide extensions going beyond strict lower bounds on terminal wealth, which is only one possible type of wealth constraints.

The contribution of this chapter to the existing literature is threefold:

- We present a generalized version of the martingale approach which is applicable for general expected value constraints and pointwise bounds on terminal wealth. The approach is based on a duality ansatz that can be applied in general complete financial markets.
- We integrate the generalized martingale approach into the auxiliary market framework of [17] to derive a set of equivalent optimality conditions, which can be used to solve portfolio optimization problems with simultaneous constraints on allocation and wealth. This is a generalization of the framework based on the 'capped Legendre-Fenchel transform', which was presented in [34].
- We illustrate the utility of our methodology in a Black-Scholes market by explicitly characterizing the optimal terminal wealth if the allocation constraints are a convex cone, the pointwise bounds are deterministic and the expected value constraint is either a VaR constraint or an ES constraint.

The remainder of this chapter is structured as follows: The financial market model, the portfolio optimization problem and the standing assumptions are introduced in Section 3.2. Afterwards, in Section 3.3, we generalize the well-known martingale approach to financial markets with constraints on terminal wealth. In a similar way, we generalize the auxiliary market framework of [17] by including constraints on terminal wealth in Section 3.4. Lastly, we propose a methodology for determining the optimal auxiliary market in a Black-Scholes setting in Section 3.5 and Section 3.6 concludes this chapter.

3.2 Setting

In this chapter we consider an investor who simultaneously faces constraints on his allocation and his terminal wealth. The constraints on terminal wealth can be either pointwise bounds (as considered in [34]), expected value constraints or both. In this regard, this chapter extends the work of [34]. To simplify the presentation we restrict the market setting defined in Section 2.3 by assuming that all randomness in \mathcal{M} is generated by W, i.e., m = 0. This has the advantage that the considered financial market \mathcal{M} is complete and we can derive a useful, equivalent characterization of (**P**) as an optimization over random variables $D \in L^2_Q$. Market completeness is a key ingredient which will be frequently used in the duality arguments in the subsequent sections. In this regard, the restriction to m = 0 in this chapter, can also be seen as a restriction to complete financial market \mathcal{M} . Note however, completeness of \mathcal{M} is not a strict prerequisite for this chapter, but rather another instrument to facilitate the exposition. Just as in [49], in the case of an incomplete financial market \mathcal{M} , one can add additional fictitious assets to complete the market and use the theory developed in Section 3.4 to rule out investments into these assets through an allocation constraint. For a non-empty closed convex set $K \subset \mathbb{R}^d$, \mathcal{F}_T -measurable random variables $0 \leq B_1 < B_2$, and a non-increasing real function $g: (0, \infty) \to \mathbb{R}$, we define the set of admissible constrained portfolio processes as

$$\Lambda' := \Lambda(v_0, K, B_1, B_2, g) := \left\{ \pi \in \Lambda \mid \pi(t) \in K, \quad B_1 \le V^{v_0, \pi}(T) \le B_2, \quad \mathbb{E}\left[g(V^{v_0, \pi}(T))\right] \le 0 \right\},$$
(3.1)

where the requirement ' $\pi \in K$ ' is meant to hold $\mathcal{L}[0,T] \otimes Q$ -a.e. and ' $B_1 \leq V^{v_0,\pi}(T) \leq B_2$ ' is meant to hold Q-a.s.. To ensure that the constraints are well-posed, i.e., $\Lambda' \neq \emptyset$, we need to make some additional technical assumptions which amount to requiring that the risk-free portfolio process $\pi \equiv 0$ is admissible for (**P**), but not trivially optimal for (**P**). Specifically, we assume that

$$0 \in K, \tag{3.2}$$

$$B_1 < v_0 P_0(T) < B_2, (3.3)$$

and
$$\mathbb{E}[g(v_0 P_0(T))] \le 0.$$
 (3.4)

Finally, this ensures that we can reasonably consider the following, fully constrained portfolio optimization problem

$$(\mathbf{P}) \begin{cases} \Phi(v_0) &= \sup_{\pi \in \Lambda(v_0, K, B_1, B_2, g)} \mathbb{E}\left[U\left(V^{v_0, \pi}(T)\right)\right]. \end{cases}$$

Clearly, we can rewrite (\mathbf{P}) directly as a maximization over all attainable terminal wealths:

$$(\mathbf{P}) \begin{cases} \Phi(v_0) &= \sup_{D \in C(v_0, K, B_1, B_2, g)} \mathbb{E}[U(D)] \\ C(v_0, K, B_1, B_2, g) &= \{ V^{v_0, \pi}(T) \mid \pi \in \Lambda(v_0, K, B_1, B_2, g) \} \end{cases}$$

This may seem trivial at first, but since $C(v_0, K, B_1, B_2, g)$ can later be simplified substantially (depending on the choice of K, B_1 , B_2 and g), it is more convenient to write (**P**) this way.

We attempt to solve (**P**) by following a similar approach as the authors of [34]: First, we introduce a suitable class of dual processes λ which parametrize the 'auxiliary' financial markets \mathcal{M}_{λ} . The wealth process of an investor trading continuously in \mathcal{M}_{λ} is adapted in such a way that it is advantageous to only choose portfolio processes which take values in K. In \mathcal{M}_{λ} , we consider a portfolio optimization problem (**P**_{λ}), without allocation constraint but with the same wealth constraints as in \mathcal{M} . Using a generalized martingale approach, which is based on the duality techniques introduced in Section 2.1.1, we can characterize the optimal terminal wealth for (**P**_{λ}) in \mathcal{M}_{λ} and establish a duality relation between the optimal portfolio process π^* for the original portfolio optimization problem (**P**) and a certain minimizing $\lambda^* \in \mathcal{D}$. A schematic overview of this approach is illustrated in Figure 3.1.



Figure 3.1: Schematic illustration of the solution approach employed throughout Chapter 3. V^{v_0,π_λ} denotes the wealth process of an investor trading in the auxiliary market \mathcal{M}_λ and will be formally defined in Section 3.4.

The standing assumptions for this chapter are summarized below:

Assumption 3.2.1 (Standing Assumptions Chapter 3:). m = 0, (3.1), (3.2), (3.3), and (3.4).

3.3 Generalized Martingale Approach

In this section we present a generalized martingale approach, which can be applied in complete financial markets. This approach can be used to characterize the optimal terminal wealth for portfolio optimization problems with simultaneous pointwise and expected value constraints on terminal wealth. To this end, we consider the portfolio optimization problem (**P**) under the assumption that no allocation constraints are present (denoted as '(**P**₀)'). Under this assumption, we can prove that the financial market \mathcal{M} satisfies a notion of market completeness, which can be used to derive an equivalent representation of (**P**₀) as a constrained optimization problem over L_Q^2 . This representation falls within the setting of Section 2.1.2 and can be treated with the associated dual optimization problem (**D**₀^V).

Let us for now consider the allocation-unconstrained analogue (\mathbf{P}_0) of the portfolio optimization problem (\mathbf{P}) , i.e., we set $\Lambda(v_0, B_1, B_2, g) := \Lambda(v_0, \mathbb{R}^d, B_1, B_2, g)$ and define

$$(\mathbf{P}_0) \left\{ \Phi_0(v_0) = \sup_{\pi \in \Lambda(v_0, \mathbb{R}^d, B_1, B_2, g)} \mathbb{E} \left[U \left(V^{v_0, \pi}(T) \right) \right]. \right.$$

As suggested earlier, we can equivalently write (\mathbf{P}_0) as an optimization of $\mathbb{E}[U(D)]$ over all $D \in L^2_Q$ such that there exists a $\pi \in \Lambda(v_0, B_1, B_2, g)$ with $D = V^{v_0, \pi}(T)$. Although this may seem trivial at first glance, we can use the notion of market completeness to develop a useful equivalent representation of (\mathbf{P}_0) . To make this notion more precise, recall the definition of the

market price of risk γ , the corresponding exponential local martingale Z and the pricing Kernel \tilde{Z} from Section 2.3.2:

$$\begin{split} \gamma(t) &:= \Sigma(t)^{-1} \big(\mu(t) - r(t) \mathbb{1} \big), \\ Z(t) &:= \exp\left(-\frac{1}{2} \int_0^t \|\gamma(s)\|^2 ds - \int_0^t \gamma(s)' dW(s) \right), \\ \tilde{Z}(t) &:= \frac{Z(t)}{P_0(t)} = \exp\left(-\int_0^t r(s) + \frac{1}{2} \|\gamma(s)\|^2 ds - \int_0^t \gamma(s)' dW(s) \right) \end{split}$$

for $t \in [0, T]$. Note that Z and \tilde{Z} satisfy the SDEs

$$dZ(t) = -Z(t)\gamma(t)'dW(t)$$

$$d\tilde{Z}(t) = -\tilde{Z}(t)(r(t)dt + \gamma(t)'dW(t))$$

In particular, Z is a positive local martingale and thus a supermartingale. If Z satisfies Novikov's condition, then it is indeed a martingale. In our setting, \mathcal{M} satisfies the following notion of market completeness.

Lemma 3.3.1 (Market Completeness). Consider $D \in \mathbf{X}_+$ with $\mathbb{E}[D^2] < \infty$ and $0 < \mathbb{E}[\tilde{Z}(T)D] =: v_0 < \infty$. Then, there exists a $\pi \in \Lambda$ so that $V^{v_0,\pi}(T) = D$ Q-a.s. and

$$V^{v_0,\pi}(t) = \mathbb{E}\left[D\frac{\tilde{Z}(T)}{\tilde{Z}(t)} \mid \mathcal{F}_t\right].$$

Lemma 3.3.1 implies that any non-negative payoff $D \in L^2_Q$ can be written as a payoff $V^{v_0,\pi}(T)$ for a $\pi \in \Lambda'$ if and only if D satisfies the so-called 'budget equation' $\mathbb{E}[\tilde{Z}(T)D] = v_0$. Noting that $U(x) = -\infty$ for all x < 0, this allows us to rewrite (\mathbf{P}_0) as an equivalent constrained optimization problem over $D \in L^2_Q$:

$$(\mathbf{P_0}) \begin{cases} \Phi(v_0) &= \sup_{D \in C(v_0, B_1, B_2, g)} \mathbb{E}[U(D)] \\ C(v_0, B_1, B_2, g) &= \{ D \in L_Q^2 \mid \mathbb{E}[\tilde{Z}(T)D] = v_0, \ B_1 \le D \le B_2, \ \mathbb{E}[g(D)] \le 0 \}. \end{cases}$$

Changing the optimization variable from admissible portfolio processes π to admissible terminal wealths $D = V^{v_0,\pi}(T)$ is a common technique in dynamic portfolio optimization (see e.g. [73], [47] and [17]) and is usually referred to as 'martingale approach'. If an optimal terminal payoff D^* for ($\mathbf{P_0}$) has been determined, then the completeness of \mathcal{M} ensures that there exists an optimal portfolio process π^* such that $V^{v_0,\pi^*}(T) = D^*$ holds Q-a.s.. Despite this theoretical guarantee, as explained in Section 2.3.2, it is challenging to determine explicit expressions for π^* . For this reason, we exclusively focus on characterizing the optimal terminal wealth for ($\mathbf{P_0}$) and do not explicitly determine the corresponding optimal portfolio process π^* .

We proceed in a similar way as [34] and incorporate the pointwise constraint into an adjusted utility function and adjusted expected value constraint

$$U(x; B_1, B_2) := \begin{cases} -\infty, & x < B_1 \\ U(x), & B_1 \le x \le B_2 \\ U(B_2), & B_2 < x \end{cases} \text{ and } g(x; B_2) = \begin{cases} g(x), & x \le B_2 \\ g(B_2), & B_2 < x \end{cases}.$$

Lemma 3.3.2. Consider the optimization problem

$$(\tilde{\mathbf{P}}_{\mathbf{0}}) \begin{cases} \tilde{\Phi}_{0}(v_{0}) &= \sup_{D \in C(v_{0}, g(\cdot; B_{2}))} \mathbb{E}\left[U(D; B_{1}, B_{2})\right] \\ C(v_{0}, g) &= C(v_{0}, 0, \infty, g(\cdot; B_{2})) \\ &= \left\{D \in L_{Q}^{2} \mid \mathbb{E}[\tilde{Z}(T)D] = v_{0}, \ \mathbb{E}[g(D; B_{2})] \leq 0\right\}. \end{cases}$$

Then, $\tilde{\Phi}_0(v_0) = \Phi_0(v_0)$. In particular, D^* is optimal for $(\tilde{\mathbf{P}}_0)$ if and only if D^* is optimal for (\mathbf{P}_0) .

The optimization problem $(\tilde{\mathbf{P}}_0)$ is a constrained optimization problem over L_Q^2 , as considered in Section 2.1.2, with objective function $U(\cdot; B_1, B_2)$ and 2-dimensional expected value constraints

$$\left(\mathbb{E}[\tilde{Z}(T)D], \mathbb{E}[g(D;B_2)]\right)' \in \{v_0\} \times (-\infty, 0] \quad \Leftrightarrow \quad \left(\mathbb{E}[-\tilde{Z}(T)D], \mathbb{E}[-g(D;B_2)]\right)' \in \{-v_0\} \times [0,\infty)$$

Hence, we may define

$$g_0 : \mathbb{R}^2 \to \mathbb{R}, \quad g_0(x) = \begin{pmatrix} -\tilde{Z}(T)x\\ -g(x;B_2) \end{pmatrix} \quad \text{and} \quad K_V := \{-v_0\} \times [0,\infty)$$

and use the associated dual optimization problem and generalized conjugate

$$U_{g_{0}}^{*}(y; B_{1}, B_{2}) \stackrel{y = (y_{0}, y_{1})'}{=} \sup_{x \in \mathbb{R}} \left(U(x; B_{1}, B_{2}) - y_{0} \tilde{Z}(T) x - y_{1} g(x; B_{2}) \right)$$

$$\stackrel{\text{if } y \in [0, \infty)^{2}}{=} \sup_{B_{1} \le x \le B_{2}} \left(U(x) - y_{0} \tilde{Z}(T) x - y_{1} g(x) \right)$$

of $U(\cdot; B_1, B_2)$ to characterize the optimal terminal payoff D_0^* for $(\tilde{\mathbf{P}}_0)$ (and (\mathbf{P}_0)). We denote the maximizing argument corresponding to $U_{g_0}^*(y; B_1, B_2)$ as $\mathcal{I}_{g_0}(y; B_1, B_2)$. Note that we specifically used the 'negated' version of the constraints in $(\tilde{\mathbf{P}}_0)$ (i.e., we consider g_0 instead of $-g_0$), as we would otherwise obtain a definition of the generalized conjugate and its associated dual optimization problem, which is uncommon for our portfolio optimization context (despite being perfectly equivalent).

Theorem 3.3.3. Let $y^* = (y_0^*, y_1^*)' \in [0, \infty)^2$ be optimal for the dual minimization problem

$$(\mathbf{D_0^V}) \left\{ \Psi_{D_0^V} = \inf_{y = (y_0, y_1)' \in [0, \infty)^2} \left(\mathbb{E} \left[U_{g_0}^*(y; B_1, B_2) \right] + y_0 v_0 \right). \right.$$

If $\mathcal{I}_{g_0}(y^*; B_1, B_2) \in L^2_Q$ and

 $y \to \mathbb{E}\left[g_0(\mathcal{I}_{g_0}(y; B_1, B_2))\right]' \Delta y$

is use at $y = y^*$ for all $\Delta y \in \mathbb{R} \times [0, \infty)$, then $\mathcal{I}_{g_0}(y^*; B_1, B_2)$ is optimal for (\mathbf{P}_0) and $y_1^* \mathbb{E}\left[g\left(\mathcal{I}_{g_0}(y^*; B_1, B_2)\right)\right] = 0.$

The minimization problem $(\mathbf{D}_0^{\mathbf{V}})$ can be regarded as the dual optimization problem of (\mathbf{P}_0) in \mathcal{M} , where the duality is induced by the remaining wealth constraints (i.e., the budget equation and the expected-value constraints g.) This is reflected in the proof of Theorem 3.3.3, which heavily relies on the duality results from Section 2.1.2. A common way of proving the existence of an optimal $y^* = (y_0^*, y_1^*)'$ for $(\mathbf{D}_0^{\mathbf{V}})$ is to show that there exists a y^* which satisfies the associated KKT-conditions, i.e., satisfies

$$\mathbb{E}[\mathcal{I}_{g_0}(y^*; B_1, B_2)\tilde{Z}(T)] = v_0 \quad \text{and} \quad y_1^* \mathbb{E}\left[g\left(\mathcal{I}_{g_0}(y^*; B_1, B_2)\right)\right] = 0.$$

One could easily generalize the present setting to consider *n*-dimensional expected-value constraints $g_i : \mathbb{R} \to \mathbb{R}$ with $\mathbb{E}[g_i(x)] \leq 0$ for all i = 1, ..., n. This would in turn yield an n + 1dimensional function $g_0(x) = (\tilde{Z}(T)x, g_1(x; B_2), ..., g_n(x; B_2))'$ and is not investigated throughout this chapter to improve the presentation.

Remark 3.3.4. The upper-semi continuity assumption on $\mathbb{E}\left[g_0(\mathcal{I}_{g_0}(y; B_1, B_2))\right]' \Delta y$ in 3.3.3 may seem quite abstract, but we refrain from providing general conditions under which it is satisfied. If there is no additional restriction on the admissible terminal payoffs due to g, e.g., if $g \equiv 0$, then we can disregard the argument y_1 and this assumption reduces to a continuity assumption on

$$y_0 \to \mathbb{E}\left[\mathcal{I}_{g_0}(y_0; B_1, B_2)\tilde{Z}(T)\right]$$

As \mathcal{I}_{g_0} is non-increasing in y_0 (Lemma 2.1.38), the above function mapping can only be continuous at $y_0 = y_0^* > 0$, if there exists a $0 < \hat{y}_0 < y_0^*$ such that

$$\mathbb{E}\left[\mathcal{I}_{g_0}(\hat{y}_0; B_1, B_2)\tilde{Z}(T)\right] < \infty,$$

as otherwise

$$\infty = \limsup_{y_0 \uparrow y_0^*} E\left[\mathcal{I}_{g_0}(y_0; B_1, B_2)\tilde{Z}(T)\right] > v_0 = \mathbb{E}\left[\mathcal{I}_{g_0}(y_0^*; B_1, B_2)\tilde{Z}(T)\right].$$

3.4 Auxiliary Markets with Bounds on Terminal Wealth

The generalized martingale approach from Section 3.3 is a dual optimization approach which can be used to solve wealth constrained portfolio optimization problems. In this section we develop a duality approach which can be used to solve portfolio optimization problems with simultaneous constraints on terminal wealth and portfolio allocation, and is thus applicable to (**P**). We proceed by first introducing the concept of auxiliary markets (see [17]) and then ensure the feasibility of the wealth constrained portfolio optimization problem by integrating the pointwise bounds on terminal wealth into the utility function (as in [34]). Further, we prove a set of equivalent optimality conditions, which yield two different dual optimization problems, (**D**^K) and (**D**^{K,V}), associated with (**P**).

Consider the following set of \mathbb{R}^d -valued dual processes \mathcal{D} :

$$\mathcal{D} := \left\{ \lambda = \left((\lambda_1(t), \dots, \lambda_d(t))' \right)_{t \in [0,T]} \text{ prog. measurable} \right| \\ \int_0^T \|\lambda(t)\|^2 dt < \infty \text{ } Q\text{-a.s.}, \ \int_0^T \delta_K(\lambda(t)) dt < \infty \text{ } Q\text{-a.s.} \right\}.$$

For any $\lambda \in \mathcal{D}$, the boundedness conditions guarantee that $\lambda(t)$ takes finite values $\mathcal{L}[0,T] \otimes Q$ a.e. and Q-a.s. only takes values in X_K . For a given $\lambda \in \mathcal{D}$, we define the auxiliary market \mathcal{M}_{λ} as the asset universe with d risky assets $P^{\lambda} = (P_1^{\lambda}, ..., P_d^{\lambda})$ and one risk-free asset P_0^{λ} , which satisfy $P_0^{\lambda}(0) = P_1^{\lambda}(0) = ... = P_d^{\lambda}(0) = 1$ and evolve according to the dynamics

$$dP^{\lambda}(t) = \operatorname{diag}\left(P^{\lambda}(t)\right)\left(\mu(t) + \lambda(t) + \delta_{K}(\lambda(t))\mathbb{1}dt + \Sigma(t)dW(t)\right)$$
$$dP_{0}^{\lambda}(t) = P_{0}^{\lambda}(t)\left(r(t) + \delta_{K}(\lambda(t))\right)dt.$$

The risk-free asset P_0^{λ} still represents the same risk-free asset and the assets P_i^{λ} represent the same risky assets from the original market \mathcal{M} , but with changed drift coefficients. As a

consequence, the wealth process of an investor trading according to the relative portfolio process π in the market \mathcal{M}_{λ} satisfies an SDE with adjusted drift coefficients:

$$dV_{\lambda}^{v_{0},\pi}(t) = V_{\lambda}^{v_{0},\pi}(t) \left([r(t) + \delta_{K}(\lambda(t)) + (\mu(t) + \lambda(t) - r(t)\mathbb{1})'\pi(t)] dt + \pi(t)'\Sigma(t)dW(t) \right) = V_{\lambda}^{v_{0},\pi}(t) \left([r(t) + (\mu(t) - r(t)\mathbb{1})'\pi(t)] + \pi(t)'\Sigma(t)dW(t) \right) + \underbrace{V_{\lambda}^{v_{0},\pi}(t) [\delta_{K}(\lambda(t)) + \lambda(t)'\pi(t)]}_{(*)} dt.$$
(3.5)

The wealth process $V_{\lambda^*}^{v_0,\pi}$ satisfies the same SDE as in the original market \mathcal{M} apart from the additional drift term (*). Moreover, we recover $\mathcal{M} = \mathcal{M}_0$ for $\lambda \equiv 0$. Due to Lemma 2.1.17, the additional drift (*) is non-negative as long $\pi(t) \in K$. Hence, any portfolio process $\pi \in \Lambda(v_0, K, B_1, B_2, g)$, which is admissible for (**P**), yields a wealth process $V_{\lambda}^{v_0,\pi}$ in \mathcal{M}_{λ} which is larger or equal than the corresponding wealth process $V^{v_0,\pi}$ in \mathcal{M} . This can be regarded as the central motivation behind the construction of \mathcal{M}_{λ} and naturally leads to an optimization problem which is dual to (**P**) with respect to the allocation constraints K. For this purpose, analogous to Section 3.3, we consider the allocation unconstrained, wealth-constrained portfolio optimization problem

$$(\mathbf{P}_{\lambda}) \begin{cases} \Phi_{\lambda}(v_0) &= \sup_{\pi \in \Lambda_{\lambda}(v_0, B_1, B_2, g)} \mathbb{E}\left[U\left(V_{\lambda}^{v_0, \pi}(T)\right)\right] \\ \Lambda_{\lambda}(v_0, B_1, B_2, g) &= \left\{\pi \in \Lambda \mid B_1 \le V_{\lambda}^{v_0, \pi}(T) \le B_2, \ \mathbb{E}\left[g(V_{\lambda}^{v_0, \pi}(T))\right] \le 0\right\}. \end{cases}$$

 (\mathbf{P}_{λ}) is the same optimization problem as $(\mathbf{P}_{\mathbf{0}})$, but formulated in the financial market \mathcal{M}_{λ} , rather than \mathcal{M} . In \mathcal{M}_{λ} , the market price of risk γ_{λ} , the corresponding exponential local martingale Z_{λ} and pricing kernel \tilde{Z}_{λ} are given as

$$\begin{split} \gamma_{\lambda}(t) &= \Sigma(t)^{-1}(\mu(t) - r(t)\mathbb{1} + \lambda(t)) = \gamma(t) + \Sigma(t)^{-1}\lambda(t), \\ Z_{\lambda}(t) &:= \exp\left(-\frac{1}{2}\int_{0}^{t} \|\gamma_{\lambda}(s)\|^{2}ds - \int_{0}^{t} \gamma_{\lambda}(s)'dW(s)\right), \\ \tilde{Z}_{\lambda}(t) &:= \frac{Z_{\lambda}(t)}{P_{0}^{\lambda}(t)} = \exp\left(-\int_{0}^{t} r_{\lambda}(s) + \frac{1}{2}\|\gamma_{\lambda}(s)\|^{2}ds - \int_{0}^{t} \gamma_{\lambda}(s)'dW(s)\right), \end{split}$$

for $t \in [0, T]$. Again, Z_{λ} and \tilde{Z}_{λ} satisfy the SDEs

$$dZ_{\lambda}(t) = -Z_{\lambda}(t)\gamma_{\lambda}(t)'dW(t)$$

$$d\tilde{Z}_{\lambda}(t) = -\tilde{Z}_{\lambda}(t)(r_{\lambda}(t)dt + \gamma_{\lambda}(t)'dW(t)).$$

As before, the market coefficients in \mathcal{M}_{λ} need not be uniformly bounded and thus it is not clear if the local martingale Z_{λ} is indeed a true martingale. However, since Z_{λ} is non-negative, it is a supermartingale. The adjusted drift coefficients and the remaining market coefficients in any auxiliary \mathcal{M}_{λ} satisfy the same boundedness assumption (2.26) as in the original market \mathcal{M} . Thus, we can use the same notion of market completeness to rewrite (\mathbf{P}_{λ}) as an optimization over terminal payoffs.

Lemma 3.4.1 (Market Completeness \mathcal{M}_{λ}). Consider $\lambda \in \mathcal{D}$, $D \in \mathbf{X}_{+}$ with $\mathbb{E}[D^{2}] < \infty$ and $0 < \mathbb{E}[\tilde{Z}_{\lambda}(T)D] =: v_{0} < \infty$. Then, there exists a $\pi \in \Lambda$ so that $V_{\lambda}^{v_{0},\pi}(T) = D$ Q-a.s. and

$$V_{\lambda}^{v_0,\pi}(t) = \mathbb{E}\left[D\frac{\tilde{Z}_{\lambda}(T)}{\tilde{Z}_{\lambda}(t)} \mid \mathcal{F}_t\right].$$

Lemma 3.4.1 directly leads to the equivalent formulation of (\mathbf{P}_{λ}) as

$$(\mathbf{P}_{\lambda}) \begin{cases} \Phi_{\lambda}(v_{0}) &= \sup_{D \in C_{\lambda}(v_{0}, B_{1}, B_{2}, g)} \mathbb{E}[U(D)] \\ C(v_{0}, B_{1}, B_{2}, g) &= \{D \in L_{Q}^{2} \mid \mathbb{E}[\tilde{Z}_{\lambda}(T)D] = v_{0}, B_{1} \leq D \leq B_{2}, \mathbb{E}[g(D)] \leq 0 \}. \end{cases}$$

As the bounds B_1 and B_2 not only constrain the downside of the portfolio value, but also its upside, an increase of the terminal wealth in an auxiliary market \mathcal{M}_{λ} (due to the added positive drift in (3.5)) may lead to a violation of the terminal wealth constraints and may even lead to (\mathbf{P}_{λ}) being infeasible. We circumvent this issue by incorporating the pointwise bounds on Dinto the utility function $U(\cdot; B_1, B_2)$, analogously to 3.3.

Lemma 3.4.2. Let $\lambda \in \mathcal{D}$, consider the optimization problem

$$(\tilde{\mathbf{P}}_{\lambda}) \begin{cases} \Phi_{\lambda}(v_0) &= \sup_{D \in C_{\lambda}(v_0, g(\cdot; B_2))} \mathbb{E}\left[U(D; B_1, B_2)\right] \\ C_{\lambda}(v_0, g(\cdot; B_2)) &= C_{\lambda}(v_0, 0, \infty, g(\cdot; B_2)) \\ &= \left\{D \in L_Q^2 \mid \mathbb{E}[\tilde{Z}_{\lambda}(T)D] = v_0, \ \mathbb{E}[g(D; B_2)] \le 0\right\} \end{cases}$$

and define $v_{\lambda}(B_2) := \mathbb{E}[B_2 \tilde{Z}_{\lambda}(T)].$

- (i) If $v_0 \leq v_{\lambda}(B_2)$, then $\tilde{\Phi}_{\lambda}(v_0) = \Phi_{\lambda}(v_0)$ and D_{λ}^* is optimal for $(\tilde{\mathbf{P}}_{\lambda})$ if and only if D_{λ}^* is optimal for (\mathbf{P}_{λ}) .
- (ii) If $v_0 > v_\lambda(B_2)$, then $D^*_\lambda = \frac{v_0}{v_\lambda(B_2)}B_2$ is optimal for $\tilde{\Phi}_\lambda(v_0)$ and $\tilde{\Phi}_\lambda(v_0) = \mathbb{E}[U(B_2)]$.

In contrast to (\mathbf{P}_0) and $(\mathbf{\tilde{P}}_0)$ from Section 3.3, the changed market coefficients in \mathcal{M}_{λ} imply that (\mathbf{P}_{λ}) and $(\mathbf{\tilde{P}}_{\lambda})$ are only equivalent if $v_0 \leq v_{\lambda}(B_2)$. It is easy to see that (\mathbf{P}_{λ}) is even infeasible if $v_0 > v_{\lambda}(B_2)$ and thus we use $(\mathbf{\tilde{P}}_{\lambda})$ to develop the duality relation with respect to the allocation constraints K. Due to the non-negativity of the drift (*) in (3.5), we know that for any $\lambda \in \mathcal{D}$ and $\pi \in \Lambda(v_0, K, B_1, B_2, g)$, $V^{v_0, \pi}(T) \leq V_{\lambda}^{v_0, \pi}(T)$ and $D = V_{\lambda}^{v_0, \pi}(T) \in C_{\lambda}(v_0, g)$. Hence, Lemma 3.4.2 yields the weak duality relation

$$\Phi(v_0) = \sup_{\pi \in \Lambda(v_0, K, B_1, B_2, g)} \mathbb{E}[U(V^{v_0, \pi}(T))]$$

$$\leq \sup_{\pi \in \Lambda(v_0, K, B_1, B_2, g)} \mathbb{E}[U(V_{\lambda}^{v_0, \pi}(T))]$$

$$\leq \sup_{D \in C_{\lambda}(v_0, g)} \mathbb{E}[U(D; B_1, B_2)] = \tilde{\Phi}_{\lambda}(v_0) \quad \forall \lambda \in \mathcal{D}.$$
(3.6)

To further develop this notion of duality, we aim to find a $\lambda^* \in \mathcal{D}$ which achieves equality in the above relation, i.e $\Phi(v_0) = \tilde{\Phi}_{\lambda^*}(v_0)$, and the optimal portfolio processes for (**P**) and ($\tilde{\mathbf{P}}_{\lambda^*}$) coincide. This is the case if λ^* satisfies a similar slackness condition as Condition (B) from [17] (resp. (\tilde{B}) from [34]).

Lemma 3.4.3. Let $\lambda^* \in \mathcal{D}$, $\pi_{\lambda^*} \in \Lambda_{\lambda^*}$ such that $D^*_{\lambda} = V^{v_0, \pi_{\lambda^*}}_{\lambda^*}$ is optimal for $(\tilde{\mathbf{P}}_{\lambda^*})$. If further

$$v_0 \le v_{\lambda^*}(B_2), \quad \pi_{\lambda^*} \in K \quad and \ \delta_K(\lambda^*(t)) + \pi_{\lambda^*}(t)'\lambda^*(t) = 0 \quad \mathcal{L}[0,T] \otimes Q - a.e., \tag{3.7}$$

then π_{λ^*} is admissible and optimal for the primal problem (**P**) and $\Phi(v_0) = \tilde{\Phi}_{\lambda^*}(v_0)$.

Equation (3.7) can be seen as an analogue of the KKT condition (Corollary 2.1.30) for allocation constrained portfolio optimization problems. In the same way, Lemma 3.4.3 only provides a sufficient condition under which the optimal portfolio for (**P**) in \mathcal{M} and (**P**_{λ^*}) in \mathcal{M}_{λ^*} coincide, but does not provide a constructive way of finding the 'correct' $\lambda^* \in \mathcal{D}$. However, we realize if $\lambda^* \in \mathcal{D}$ satisfies the requirements of Lemma 3.4.3, then it minimizes $\tilde{\Phi}_{\lambda}(v_0)$ over $\lambda \in \mathcal{D}$ and therefore can be regarded as the solution to a dual minimization problem

$$(\mathbf{D}^{\mathbf{K}}) \left\{ \Psi_{D^{K}} = \inf_{\lambda \in \mathcal{D}} \tilde{\Phi}_{\lambda}(v_{0}),$$
(3.8)

where the duality is now induced by the allocation constraints K. As a matter of fact, under some regularity conditions, the converse statement is also true and (3.7) is satisfied if $\lambda = \lambda^*$ is optimal for $(\mathbf{D}^{\mathbf{K}})$. To prove this statement formally, we start by characterizing the optimal terminal wealth for $(\tilde{\mathbf{P}}_{\lambda})$ by following the same methodology as in Section 3.3. For a given $\lambda \in \mathcal{D}$, we define

$$g_{\lambda} : \mathbb{R}^2 \to \mathbb{R}, \quad g_{\lambda}(x) = \begin{pmatrix} -Z_{\lambda}(T)x \\ -g(x; B_2) \end{pmatrix}$$

and again make use of the dual optimization problem associated with $(\tilde{\mathbf{P}}_{\lambda})$ and its expected value constraints.

Theorem 3.4.4. Let $\lambda \in \mathcal{D}$, $v_0 \leq v_{\lambda}(B_2)$, $y^* = (y_0^*, y_1^*)' \in [0, \infty)^2$ be optimal for the dual minimization problem

$$\left(\mathbf{D}_{\lambda}^{\mathbf{V}}\right) \left\{ \Psi_{D_{\lambda}^{V}} = \inf_{y = (y_{0}, y_{1})' \in [0, \infty)^{2}} \left(\mathbb{E} \left[U_{g_{\lambda}}^{*}(y; B_{1}, B_{2}) \right] + y_{0} v_{0} \right).$$

If $\mathcal{I}_{g_{\lambda}}(y^*; B_1, B_2) \in L^2_Q$ and

$$y \to \mathbb{E}\left[g_{\lambda}(\mathcal{I}_{g_{\lambda}}(y; B_1, B_2))\right]' \Delta y$$

is use at $y = y^*$ for all $\Delta y \in \mathbb{R} \times [0, \infty)$, then $\mathcal{I}_{g_{\lambda}}(y^*; B_1, B_2)$ is optimal for (\mathbf{P}_{λ}) and $y_1^* \mathbb{E} \left[\mathcal{I}_{g_{\lambda}}(y^*; B_1, B_2) \right] = 0.$

Combining the dual optimization problems $(\mathbf{D}^{\mathbf{K}})$ and $(\mathbf{D}^{\mathbf{V}}_{\lambda})$ yields a joint duality relation between the allocation constraints *and* wealth constraints

$$\Phi(v_0) \stackrel{(\mathbf{D}^{\mathbf{K}})}{\leq} \inf_{\lambda \in \mathcal{D}} \tilde{\Phi}_{\lambda}(v_0) \stackrel{(\mathbf{D}^{\mathbf{V}})}{\leq} \inf_{\lambda \in \mathcal{D}} \left(\inf_{y=(y_0,y_1)' \in [0,\infty)^2} \left(\mathbb{E} \left[U_{g_{\lambda}}^*(y;B_1,B_2) \right] + y_0 v_0 \right) \right)$$
(3.9)

and naturally leads to the joint dual optimization problem

$$(\mathbf{D}^{\mathbf{K},\mathbf{V}}) \begin{cases} \Psi_D^{K,V} &= \inf_{\lambda \in \mathcal{D}} \left(\inf_{y=(y_0,y_1)' \in [0,\infty)^2} \left(\mathbb{E} \left[U_{g_\lambda}^*(y;B_1,B_2) \right] + y_0 v_0 \right) \right) \\ &= \inf_{y=(y_0,y_1)' \in [0,\infty)^2} \left(\inf_{\lambda \in \mathcal{D}} \left(\mathbb{E} \left[U_{g_\lambda}^*(y;B_1,B_2) \right] \right) + y_0 v_0 \right) \end{cases}$$

 $(\mathbf{D}^{\mathbf{K},\mathbf{V}})$ is a dual with respect to (\mathbf{P}) , where the duality is induced jointly by the allocation constraints (K) and the wealth constraints (budget equation, B_1 , B_2 and g). Moreover, as we can change the order of minimization in $(\mathbf{D}^{\mathbf{K},\mathbf{V}})$, we know that if $(\lambda^*, y^*) \in \mathcal{D} \times [0, \infty)^2$ is optimal for $(\mathbf{D}^{\mathbf{K},\mathbf{V}})$, then y^* is optimal for $(\mathbf{D}^{\mathbf{V}}_{\lambda^*})$ and λ^* minimizes $\mathbb{E}\left[U_{g_{\lambda}}^*(y^*; B_1, B_2)\right]$. The latter minimization plays an important role in Section 3.5.

To ensure that the optimal values (λ^*, y^*) for the joint dual $(\mathbf{D}^{\mathbf{K}, \mathbf{V}})$ actually yield an optimal solution to the primal jointly constrained portfolio optimization problem (\mathbf{P}) (i.e., $\Phi(v_0) = \Psi_{D^{K,V}}$), it is necessary that the allocation unconstrained optimization problem $(\mathbf{\tilde{P}}_{\lambda^*})$ in \mathcal{M}_{λ^*} is solvable by the dual $(\mathbf{D}_{\lambda^*}^{\mathbf{V}})$ (i.e., (λ^*, y^*) satisfies the requirements of Theorem 3.4.4). To this end, we define the subset of dual processes $\mathcal{D}' \subset \mathcal{D}$ such that every $\lambda \in \mathcal{D}'$ satisfies

- $v_0 \leq v_\lambda(B_2)$,
- there exist $(y_0^*, y_1^*)' \in [0, \infty)^2$ which is optimal for $(\mathbf{D}_{\lambda}^{\mathbf{V}})$,
- $\mathcal{I}_{g_{\lambda}}(y^*; B_1, B_2) \in L^2_Q$
- $y \to \mathbb{E}\left[g_{\lambda}(\mathcal{I}_{g_{\lambda}}(y; B_1, B_2))\right]' \Delta y$ is use at $y = y^*$ for all $\Delta y \in \mathbb{R} \times [0, \infty)$.

For $\lambda^* \in \mathcal{D}'$, we can now define a set of equivalent conditions which ensure that the optimal portfolios for (\mathbf{P}_{λ^*}) and (\mathbf{P}) coincide.

Optimality Conditions for (**P**): For $\pi^* \in \Lambda(v_0, K, B_1, B_2, g), \lambda^* \in \mathcal{D}'$ with optimal $y^* \in [0, \infty)^2$ for $(\mathbf{D}_{\lambda^*}^{\mathbf{V}})$, we define the following conditions:

 $(\tilde{A}) \ \forall \pi \in \Lambda(v_0, K, B_1, B_2, g)$ we have

$$\mathbb{E}[U(V^{v_0,\pi}(T))] \le \mathbb{E}[U(V^{v_0,\pi^*}(T))].$$

(B) The optimal portfolio process π_{λ^*} for (\mathbf{P}_{λ^*}) satisfies:

$$\pi_{\lambda^*} \in K$$
 and $[\delta_K(\lambda^*(t)) + \pi_{\lambda^*}(t)'\lambda^*(t)] = 0$ $\mathcal{L}[0,T] \otimes Q - a.e.$

 $(\tilde{C}) \ \forall \lambda \in \mathcal{D}$ we have

$$\tilde{\Phi}_{\lambda^*}(v_0) \le \tilde{\Phi}_{\lambda}(v_0)$$

 $(\tilde{D}) \ \forall \lambda \in \mathcal{D}$ we have

$$\mathbb{E}\left[U_{a_{\lambda}}^{*}(y^{*}; B_{1}, B_{2})\right] \leq \mathbb{E}\left[U_{a_{\lambda}}^{*}(y^{*}; B_{1}, B_{2})\right]$$

 $(\tilde{E}) \ \forall \lambda \in \mathcal{D}$ we have

$$\mathbb{E}\left[\mathcal{I}_{g_{\lambda^*}}(y^*; B_1, B_2)\tilde{Z}_{\lambda}(T)\right] \le v_0.$$

Theorem 3.4.5. Let $\lambda^* \in \mathcal{D}'$. Then, Conditions (\tilde{B}) , (\tilde{C}) , (\tilde{D}) and (\tilde{E}) are equivalent and imply (\tilde{A}) with $\pi^* := \pi_{\lambda^*}$.

Due to Lemma 2.1.17, (iii), it is sufficient in the proof of $(\tilde{D}) \Rightarrow (\tilde{B})$ to consider only $\nu \in \mathcal{D}$ with $\|\nu(t)\| \leq 1 \mathcal{L}[0,T] \otimes Q - a.e.$ and $\nu(t) = -\lambda^*(t)/\max(1, \|\lambda^*(t)\|)$. Even though this does not affect the proof of Theorem 3.4.5 in any meaningful way, it has the satisfying consequence that any local minimizer λ^* of Conditions (\tilde{C}) or (\tilde{D}) is indeed a global minimizer over the whole space \mathcal{D} and satisfies Condition (\tilde{B}) . This will be useful in a verification theorem in Section 3.5.

Corollary 3.4.6. Let $\lambda^* \in \mathcal{D}'$. If Condition (\tilde{B}) , (\tilde{C}) , (\tilde{D}) or (\tilde{E}) is satisfied for all $\lambda \in \mathcal{D}$ with $\|\lambda^*(t) - \lambda(t)\| \leq 1 \mathcal{L}[0,T] \otimes Q - a.e.$, then (\tilde{B}) , (\tilde{C}) , (\tilde{D}) and (\tilde{E}) are satisfied for all $\lambda \in \mathcal{D}$. In particular, $\pi^* := \pi_{\lambda^*}$ is optimal for (**P**).

The optimality conditions $(\tilde{B}) - (\tilde{E})$ offer alternative ways to find and verify the optimality of a portfolio process π^* for the fully constrained portfolio optimization problem (**P**) in \mathcal{M} . The central underlying assumption is that we can find a different market with adjusted market coefficients \mathcal{M}_{λ^*} , where the optimal portfolio π_{λ^*} for the allocation unconstrained problem ($\tilde{\mathbf{P}}_{\lambda^*}$) coincides with π^* . According to Condition (\tilde{B}), π^* and π_{λ^*} coincide if the wealth processes $V^{v_0,\pi_{\lambda^*}}$ in \mathcal{M} and $V_{\lambda^*}^{v_0,\pi_{\lambda^*}}$ in \mathcal{M}_{λ^*} are equal. Hence, the change in market coefficients from the original \mathcal{M} to \mathcal{M}_{λ^*} must not have any impact on the portfolio performance of π^* . Following Condition (\tilde{C}) , we additionally know that \mathcal{M}_{λ^*} yields the least expected utility under all \mathcal{M}_{λ} , $\lambda \in \mathcal{D}$, if the investor follows an optimal strategy. In this sense, \mathcal{M}_{λ^*} has the least favourable market coefficients from the investor's perspective. Condition (\tilde{D}) is in fact just a dual reformulation of Condition (\tilde{C}) , where the duality is now induced not by the allocation constraints K, but by the expectation constraint g on terminal wealth and the budget equation. As we will see in Section 3.5, Condition (\tilde{D}) proves to be particularly useful in explicitly determining λ^* and π^* . Lastly, Condition (\tilde{E}) states that there exists no market \mathcal{M}_{λ} , where hedging the optimal terminal wealth $D^*_{\lambda^*} := V^{v_0,\pi_{\lambda^*}}_{\lambda^*}(T) = \mathcal{I}_{g_{\lambda^*}}(y^*; B_1, B_2)$ for $(\tilde{\mathbf{P}}_{\lambda^*})$ is more expensive than in \mathcal{M}_{λ^*} . Again, the market coefficients of \mathcal{M}_{λ^*} can be regarded as least favourable for the investor. This is a special case of more general results about hedging contingent claims under allocation constraints, which is discussed in great detail in [18].

3.5 The Optimal Auxiliary Market

In this section we illustrate how one can make use of the equivalent optimality conditions derived in the previous section to solve the fully constrained portfolio optimization problem (**P**) in a Black-Scholes market. We first aim to identify the minimizing λ^* for every $y^* \in [0, \infty)^2$ in Condition (\tilde{D}) in Section 3.4, by considering an associated HJB PDE. If $U_{g_{\lambda}}^*$ satisfies a polynomial growth condition for deterministic $\lambda \in \mathcal{D}$ and the allocation constraint K is a convex cone, then the minimizing $\lambda^* \in \mathcal{D}$ is deterministic, independent of y^* and can be determined as the optimizer of a deterministic convex optimization problem. This is shown in Section 3.5.2.

Throughout the whole of Section 3.5 we make the following additional assumptions about the market coefficients and bounds on terminal wealth:

Assumption 3.5.1 (Standing Assumption Section 3.5). The market coefficients r, μ and Σ as well as the bounds on terminal wealth $0 \leq B_1 < v_0 e^{rT} < B_2 \leq \infty$ are constants and δ_K is continuous on X_K .

Note however that a generalization to deterministic and continuous r(t), $\mu(t)$ and $\Sigma(t)$ is straightforward. To include the case $B_1 = 0$ or $B_2 = \infty$ one has to impose additional growth assumptions on $\mathcal{I}_{g_{\lambda^*}}$ for the (unknown!) $\lambda^* \in \mathcal{D}$ which satisfies condition (\tilde{D}). Assumption 3.5.1 allows the use of the ensuing dynamic programming techniques, which lead to closed-form solutions to the primal, fully constrained portfolio optimization problem (\mathbf{P}) for convex cone allocation constraints K.

3.5.1 Optimization Problem Associated with Condition D

By interchanging the order of minimization in $(\mathbf{D}^{\mathbf{K},\mathbf{V}})$ or by considering Condition (\tilde{D}) , we have seen that the 'optimal' λ^* , for which the optimal portfolios for (\mathbf{P}) and $(\tilde{\mathbf{P}}_{\lambda^*})$ coincide, is the solution to the 'inner' optimization problem

$$(\mathbf{D}_{\mathbf{y}})\left\{\Psi_{D_{y}}=\inf_{\lambda\in\mathcal{D}}\mathbb{E}\left[U_{g_{\lambda}}^{*}(y;B_{1},B_{2})\right],\right.$$

for the specific choice $y = y^*$, where y^* is optimal for $(\mathbf{D}_{\lambda^*}^{\mathbf{V}})$. In particular, y^* already depends on λ^* . Hence, while the optimizer $\lambda^*(y^*)$ for $(\mathbf{D}_{\mathbf{y}^*})$ is optimal in the sense of Condition (\tilde{D}) , this may not be the case for general $\lambda^*(y)$ and $y \in [0, \infty)^2$. Irrespectively, a characterization of the minimizer $\lambda^*(y)$ for $(\mathbf{D}_{\mathbf{y}})$ for arbitrary $y \in [0, \infty)^2$ may still yield valuable insights for the solution of the primal portfolio optimization problem (**P**). Recall that the generalized conjugate $U_{g_{\lambda}}^{*}(y; B_{1}, B_{2})$ only depends on $\lambda \in \mathcal{D}$ through the terminal value $\tilde{Z}_{\lambda}(T) \in (0, \infty)$. Hence, by conditioning on this value, we obtain a deterministic function

$$U_{g_{\lambda}}^{*}(y; B_{1}, B_{2})\Big|_{\tilde{Z}_{\lambda}(T)=z} = \sup_{B_{1} \le x \le B_{2}} \left(U(x) - y' \begin{pmatrix} xz\\g(x) \end{pmatrix} \right) =: U_{\hat{g}}^{*}(y; B_{1}, B_{2}, z),$$

with $\hat{g}(x) = \hat{g}(x;z) := (-xz;g(x;B_2))'$. Therefore, we can view $(\mathbf{D}_{\mathbf{y}})$ as a stochastic control problem with state process \tilde{Z}_{λ} , control process $\lambda \in \mathcal{D}$ and minimization objective $U_{\hat{g}}^*(y;B_1,B_2,\tilde{Z}_{\lambda}(T))$. The HJB equation associated with $(\mathbf{D}_{\mathbf{y}})$ is then

$$0 = G_t(t, z; y) - rzG_z(t, z; y) + \inf_{x \in X_K} \left(-\delta_K(x)zG_z(t, z; y) + \frac{1}{2} \|\gamma + \Sigma^{-1}x\|^2 z^2 G_{zz}(t, z; y) \right)$$
(3.10)
$$G(T, z; y) = U_{\hat{g}}^*(y; B_1, B_2, z).$$

For the remainder of this section, we focus on the HJB equation (3.10) and show that its solution, provided it satisfies some regularity conditions, induces a solution to the (inner) dual optimization problem $(\mathbf{D}_{\mathbf{y}})$. Assuming that G solves (3.10) and is strictly decreasing and convex in z, there exists a minimizer $\lambda^*(t, z; y)$, which attains the infimum in (3.10). By slightly rewriting the PDE, one can see that $\lambda^*(t, z; y)$ actually minimizes

$$\lambda^{*}(t,z;y) = \underset{x \in X_{K}}{\operatorname{argmin}} \left(\frac{z^{2}}{2} \|\gamma + \Sigma^{-1}x\|^{2} G_{zz}(t,z;y) - \delta_{K}(x) z G_{z}(t,z;y) \right)$$

$$\overset{G_{z} < 0}{=} \underset{x \in X_{K}}{\operatorname{argmin}} \left(\underbrace{-\frac{1}{2} \frac{z G_{zz}(t,z;y)}{G_{z}(t,z;y)}}_{=:\operatorname{RRA}(t,z;y) \geq 0} \|\gamma + \Sigma^{-1}x\|^{2} + \delta_{K}(x) \right)$$
(3.11)

$$= \operatorname{argmin}_{x \in X_K} \Big(\operatorname{RRA}(t, z; y) \| \gamma + \Sigma^{-1} x \|^2 + \delta_K(x) \Big).$$
(3.12)

This means that the (non-negative) relative risk aversion $\operatorname{RRA}(t, z; y)$ of G with respect to z serves as a weighting factor in the minimization between the non-negative components $\|\gamma + \Sigma^{-1}x\|^2$ and $\delta_K(x)$.

Lemma 3.5.2. Let G = G(t, z; y) be continuously differentiable in $t \in [0, T)$, twice continuously differentiable in $z \in (0, \infty)$, be convex and strictly decreasing in z and a solution to the HJB equation (3.10). Then there exists a corresponding minimizing argument $\lambda^*(t, z; y)$ (as in (3.11)), which is uniformly bounded in (t, z; y).

Using this observation and assuming polynomial growth, convexity and monotonicity conditions in z, we are able to locally prove a verification theorem for the HJB equation (3.10). Having the minimization property locally is sufficient for our purposes as noted in Corollary 3.4.6.

Theorem 3.5.3 (Verification Theorem).

Let Assumption 3.5.1 hold, let $y \in [0, \infty)^2$ be fixed. Further, let G = G(t, z; y) be continuously differentiable in $t \in [0, T)$, twice continuously differentiable in $z \in (0, \infty)$, be a solution to the HJB equation (3.10), be convex, strictly decreasing and satisfy the polynomial growth condition

$$G(t, z; y) \leq C(z^{-\alpha} + z^{\alpha}), \quad \text{for some } \alpha > 0, \ C > 0.$$

Further, let

$$\lambda^{*}(t,z;y) := \underset{x \in X_{K}}{\operatorname{argmin}} \left(\frac{z^{2}}{2} \| \gamma + \Sigma^{-1} x \|^{2} G_{zz}(t,z;y) - \delta_{K}(x) z G_{z}(t,z;y) \right),$$
(3.13)

be uniformly bounded in (t, z; y). Define the stochastic process $\lambda^* = (\lambda^*(t))_{t \in [0,T]}$ in feedback-form as

$$\lambda^*(t) = \lambda^*(t, \tilde{Z}_{\lambda^*}(t); y).$$

Then, $\forall \lambda \in \mathcal{D}$ with $\|\lambda(s) - \lambda^*(s)\| \leq 1$ and $(t, z) \in [0, T] \times (0, \infty)$:

$$G(t, z; y) \le \mathbb{E}[U_{g_{\lambda}}^*(y; B_1, B_2) | Z_{\lambda}(t) = z]$$

and

$$G(t, z; y) = \mathbb{E}[U_{g_{\lambda^*}}^*(y; B_1, B_2) | Z_{\lambda^*}(t) = z].$$

Note that Theorem 3.5.3 does not provide verification for the fully constrained portfolio optimization problem (**P**), but only for the inner dual optimization problem (**D**_y). We still need to determine the corresponding 'outer' minimizer y^* and show that the process λ^* is indeed an element of \mathcal{D}' .

3.5.2 Convex Cone Constraints

In this section we solve the HJB equation (3.10) and determine the pointwise minimizer $\lambda^*(t, z; y)$ (3.13) under the assumption that K is a convex cone and the generalized conjugate satisfies a polynomial growth condition in z. If $|U(B_1)|$ and $|U(B_2)|$ are finite, then this growth condition is always satisfied. We then use the verification theorem from Section 3.5.1 to link the pointwise minimizer $\lambda^*(t, z; y)$ to optimality Condition (\tilde{D}) and finally characterize the optimal terminal wealth for (**P**) in the original market \mathcal{M} as the optimal terminal wealth for $\tilde{\mathbf{P}}_{\lambda^*}$ in the auxiliary market \mathcal{M}_{λ^*} .

As per Lemma 2.1.17, (iv), the allocation constraints K form a convex cone if and only if $\delta_K(x) = 0$ for all $x \in X_K$. In this special case the HJB equation (3.10) simplifies to

$$0 = G_t(t,z;y) - rzG_z(t,z;y) + \frac{z^2}{2} \inf_{x \in X_K} \left(\|\gamma + \Sigma^{-1}x\|^2 G_{zz}(t,z;y) \right)$$
(3.14)
$$G(T,z;y) = U_{\hat{g}}^*(y;B_1, B_2, z).$$

If G is convex, the infimum is attained by the pointwise minimizer

$$\lambda^*(t, z; y) := \lambda^* := \underset{x \in X_K}{\operatorname{argmin}} \|\gamma + \Sigma^{-1} x\|^2,$$

which is independent of (t, z; y). Further, the HJB PDE reduces to a linear PDE, which can be solved through a transformation to the well-studied heat equation (see e.g. [6]). For this purpose, recall the following result about the heat equation:

Lemma 3.5.4. Consider a real function $f : \mathbb{R} \to \mathbb{R}$ for which exist constants C_0 , α_0 such that

$$|f(u)| \le C_0 e^{\alpha_0 u^2} \quad \forall u \in \mathbb{R}.$$
(3.15)

Then, for all $0 < T < \frac{1}{4\alpha_0}$ the function $F: (0,T] \times \mathbb{R} \to \mathbb{R}$ defined by

$$F(\tau, u) = \frac{1}{\sqrt{4\pi\tau}} \int_{\mathbb{R}} e^{-\frac{(u-x)^2}{4\tau}} f(x) dx = \frac{1}{\sqrt{4\pi\tau}} \int_{\mathbb{R}} e^{-\frac{x^2}{4\tau}} f(u-x) dx,$$

is in $C^{(1,2)}((0,T]\times\mathbb{R})$ and is a solution to the heat equation

$$F_{\tau}(\tau, u) = F_{uu}(\tau, u) \quad \forall (\tau, u) \in (0, T] \times \mathbb{R}$$

$$F(0, u) := \lim_{\tau \downarrow 0} F(\tau, u) = f(u) \qquad \text{for almost all } u \in \mathbb{R}.$$

$$(3.16)$$

Lemma 3.5.5. Let $y \in [0,\infty)^2$ and let $U^*_{\hat{g}}(y; B_1, B_2, z)$ satisfy a polynomial growth condition in z:

$$|U_{\hat{g}}^{*}(y; B_{1}, B_{2}, z)| \le C(z^{-\alpha} + z^{\alpha}), \quad \forall z \in (0, \infty)$$
(3.17)

and some constants $C, \alpha > 0$. If K is a convex cone, then

$$G(t,z;y) := \frac{1}{\sqrt{4\pi(T-t)}} \int_{\mathbb{R}} e^{-\frac{x^2}{4(T-t)}} U_{\hat{g}}^* \left(y; B_1, B_2, zy e^{-(r+\frac{1}{2} \|\gamma_{\lambda^*}\|^2)(T-t) - \frac{\|\gamma_{\lambda^*}\|}{\sqrt{2}}x}\right) dx$$

is continuously differentiable in t, twice continuously differentiable in z, convex in z, strictly decreasing in z and satisfies the HJB equation (3.10) with

$$\lambda^*(t,z;y) := \lambda^* := \underset{x \in X_K}{\operatorname{argmin}} \|\gamma + \Sigma^{-1}x\|^2 \quad and \quad \gamma_{\lambda^*} := \gamma + \Sigma^{-1}\lambda^*.$$

Further, there exists a constant \tilde{C} such that G satisfies the polynomial growth condition¹

$$|G(t,z;y)| \le \tilde{C} \left(z^{-\alpha} + z^{\alpha} \right) \quad \forall z \in (0,\infty).$$
(3.18)

It is important to emphasize that the previous techniques heavily relied on K being a convex cone (hence $\delta_K(\lambda^*) = 0$) as this simplifies the HJB equation (3.10) to a linear PDE. For more general allocation constraints with $\delta_K(\lambda^*) \neq 0$, the PDE may become non-linear and extremely difficult to solve.

The polynomial growth condition (3.17) is designed in such a way that it is satisfied if $U(x) - y_1g(x)$ is a power utility function. In particular, if U and -g are bounded from above and below by a power utility function, then we can show that (3.17) is satisfied. Further, if B_2 , $U(B_2)$ and $g(B_2)$ are finite, then the growth condition is satisfied trivially.

Lemma 3.5.6. Consider $U \in \mathcal{U}$, $(y_0, y_1)' \in [0, \infty)^2$ and let Assumption 3.5.1 be satisfied. If one of the conditions

(i) There exist constants $C_- > 0$, $C_+ > 0$ and $b_- < 0$, $b_+ > 0$ such that

$$C_{-}\left(\frac{1}{b_{-}}x^{b_{-}}-1\right) \le \min\left(U(x), -g(x)\right) \text{ and } \max\left(U(x), -g(x)\right) \le C_{+}\left(\frac{1}{b_{+}}x^{b_{+}}+1\right).$$

(*ii*) $C = |U(B_2)| + y_0(B_1 + B_2) + y_1|g(B_2)| < \infty$

is satisfied, then $U_{\hat{a}}^*$ satisfies the growth condition (3.17).

Corollary 3.5.7. Let K be a convex cone, let Assumption 3.5.1 hold and either (i) or (ii) from Lemma 3.5.6 be satisfied. If

$$\lambda^* := \underset{x \in X_K}{\operatorname{argmin}} \|\gamma + \Sigma^{-1} x\|^2 \tag{3.19}$$

is an element of \mathcal{D}' , then λ^* satisfies condition (\tilde{D}) and the optimal portfolio for the wealthconstrained portfolio optimization problem $(\tilde{\mathbf{P}}_{\lambda^*})$ is optimal for the fully constrained portfolio optimization problem (**P**). In particular, if y^* is optimal for ($\mathbf{D}_{\lambda^*}^{\mathbf{V}}$), then the optimal terminal wealth for (**P**) is $D^* = \mathcal{I}_{g_{\lambda^*}}(y^*; B_1, B_2)$.

¹Note that the constants C, \tilde{C} and α are allowed to depend on y, B_1 and B_2 .

Remark 3.5.8. If K is a convex cone, we have $\delta_K(x) = 0$ for all $x \in X_K$ and thus

$$V^{v_0,0}(T) = v_0 P_0(T) = v_0 P_0^{\lambda}(T) = V_{\lambda}^{v_0,0}(T) \quad \forall \lambda \in \mathcal{D}.$$

According to Assumption 3.2.1, this implies

$$v_{\lambda}(B_2) = \mathbb{E}[B_2 \tilde{Z}_{\lambda}(T)] > \mathbb{E}[V^{v_0,0}(T)\tilde{Z}_{\lambda}(T)] = \mathbb{E}[V_{\lambda}^{v_0,0}(T)\tilde{Z}_{\lambda}(T)] = v_0.$$

Moreover, if $B_2 < \infty$ under Assumption 3.5.1, the maximizing arguments $\mathcal{I}_{g_{\lambda}}(y; B_1, B_2)$ is bounded from above (by $B_2 < \infty$) Q-a.s. for all $\lambda \in \mathcal{D}$. Thus,

$$\mathbb{E}\left[\mathcal{I}_{g_{\lambda}}(y; B_1, B_2)^2\right] < \infty, \quad and \quad \mathbb{E}\left[\mathcal{I}_{g_{\lambda}}(y; B_1, B_2)\tilde{Z}_{\lambda}(T)\right] < \infty \quad \forall \lambda \in \mathcal{D}.$$

Therefore, λ^* (as in Corollary 3.5.7) is in \mathcal{D}' if there exists a minimizer y^* for $(\mathbf{D}_{\lambda^*}^{\mathbf{V}})$ and $\mathbb{E}[g_{\lambda^*}(\mathcal{I}_{g_{\lambda^*}}(y; B_1, B_2))]' \Delta y$ is use at $y = y^*$ for all $\Delta y \in \mathbb{R} \times [0, \infty)$.

3.5.3 Examples

In this section we consider examples where we can characterize the optimal terminal wealth for (**P**) explicitly, i.e. up to the deterministic minimizers $(y_0^*, y_1^*) \in [0, \infty)^2$ and $\lambda^* \in X_K$. For this purpose we restrict our analysis to power utility functions of the form

$$U(x) = \begin{cases} \frac{1}{b}x^{b}, & x > 0\\ \liminf_{x \downarrow 0} \frac{1}{b}x^{b}, & x = 0\\ -\infty, & x < 0 \end{cases} \text{ with } b \in (-\infty, 1) \setminus \{0\}.$$
(3.20)

This decision is motivated partially by (i) in Lemma 3.5.6, but it is also necessary to ensure the analytical tractability of the maximizing argument $\mathcal{I}_{g_{\lambda^*}}$. We begin by disregarding the expected value constraints, i.e., setting $g \equiv 0$, which leaves us in the setting of [34]. We then increase the complexity by additionally considering Value-at-Risk ('VaR') constraints and expected shortfall ('ES') constraints as natural and relevant practical examples. Further, for λ^* as in (3.19) and $y_0 > 0$, we define the expression

$$\mathcal{I}(y_0) = \left(y_0 \tilde{Z}_{\lambda^*}(T)\right)^{\frac{1}{b-1}} = \operatorname*{argmax}_{x \ge 0} \left(\frac{1}{b} x^b - y_0 \tilde{Z}_{\lambda^*}(T) x\right).$$

If y_0 is chosen such that $\mathbb{E}[\mathcal{I}(y_0)\tilde{Z}_{\lambda^*}(T)] = v_0$, then $\mathcal{I}(y_0)$ is the optimal terminal wealth for (**P**) if there are no wealth constraints (i.e., $B_1 = 0$, $B_2 = \infty$ and $g \equiv 0$). It will be useful to express $\mathcal{I}_{g_{\lambda^*}}$ in terms of \mathcal{I} , when we add wealth constraints in the subsequent analysis.

We summarize the additional technical requirements for this Section in Assumption 3.5.9.

Assumption 3.5.9. $K \subset \mathbb{R}^d$ is a convex cone, U is a power utility function as in (3.20) and λ^* is defined as in (3.19).

No Expected Value Constraints: We begin by assuming that there are no expected value constraints, i.e., that $g \equiv 0$. In this simple context, $U(x) - y_0 \tilde{Z}_{\lambda^*}(T)x - y_1g(x)$ is a strictly concave function in x for all $(y_0, y_1)' \in [0, \infty)^2$ and therefore the maximizing argument $\mathcal{I}_{g_{\lambda^*}}(y; B_1, B_2)$ can be obtained by capping the global maximizing argument $\mathcal{I}(y_0)$ at the boundaries B_1 and B_2 .

Example 3.5.10.

Let Assumptions 3.5.1 and 3.5.9 hold. If $g \equiv 0$ and $y^* = (y_0^*, 0)' \in (0, \infty) \times [0, \infty)$ is optimal for $(\mathbf{D}_{\lambda^*}^{\mathbf{V}})$, then the optimal terminal wealth for (\mathbf{P}) is

$$D^* = \mathcal{I}_{g_{\lambda^*}}(y^*; B_1, B_2) = B_1 + \left(\mathcal{I}(y_0^*) - B_1\right)^+ - \left(\mathcal{I}(y_0^*) - B_2\right)^+ =: Cap(\mathcal{I}(y_0^*), B_1, B_2).$$



Figure 3.2: Illustration of the dependence of the optimal terminal wealth D^* on $Z_{\lambda^*}(T)$ without expected value constraints and for a given constant value of y_0^* (cf. Example 3.5.10).

Value-at-Risk Constraints: For constants $B_1 < B_{VaR} < B_2$ and $\epsilon > 0$ we define the VaR constraint as

$$g(x) = \mathbb{1}_{\{x < B_{VaR}\}} - \epsilon.$$
(3.21)

The VaR is a risk measure which is frequently used by practitioners in risk management and is deeply integrated into financial regulations (e.g. in Solvency II and Basel III). The VaR constraint restricts the probability that the investor's terminal wealth falls below B_{VaR} to a value of at most ϵ . Hence, the VaR constraint restricts the portfolio loss $L = -(V^{v_0,\pi}(T) - B)$ in comparison to a benchmark $B \in \mathbb{R}$ to at most $B - B_{VaR}$ in $(1 - \epsilon)\%$ of all scenarios. However, conditioned on a loss of more than $B - B_{VaR}$ occurring, the VaR constraint does not impose a restriction on the magnitude of this loss. Further, the VaR is not a convex risk measure (i.e. g is not convex) and thus it is possible that the convex combination of two terminal payoffs violates the VaR constraint, even though the individual payoffs satisfy the VaR constraint. This can be regarded as a punishment for diversification and thus poses another weakness of VaR.

Example 3.5.11.

Let Assumptions 3.5.1 and 3.5.9 hold. If g is a VaR constraint (as in (3.21)), $B_1 = 0, B_2 = \infty$, and $y^* = (y_0^*, y_1^*)' \in (0, \infty) \times [0, \infty)$ is optimal for $(\mathbf{D}_{\lambda^*}^{\mathbf{V}})$, then the optimal terminal wealth for (**P**) is

$$D^* = \mathcal{I}_{g_{\lambda^*}}(y^*) = \begin{cases} \mathcal{I}(y_0^*), & B_{VaR} < \mathcal{I}(y_0^*) \\ B_{VaR}, & \mathcal{I}(y_0^*) \le B_{VaR} \& \frac{1-b}{b} \mathcal{I}(y_0^*)^b + y_0^* \tilde{Z}_{\lambda^*}(T) B_{VaR} < \frac{1}{b} B_{VaR}^b + y_1^* \\ \mathcal{I}(y_0^*), & \mathcal{I}(y_0^*) \le B_{VaR} \& \frac{1-b}{b} \mathcal{I}(y_0^*)^b + y_0^* \tilde{Z}_{\lambda^*}(T) B_{VaR} \ge \frac{1}{b} B_{VaR}^b + y_1^* \end{cases}$$

As $U(x) - y_0 \tilde{Z}_{\lambda^*}(T)x - y_1 g(x)$ is no longer concave in x (note the discontinuity of g(x) at $x = B_{VaR}$), the maximizing argument $\mathcal{I}_{g_{\lambda^*}}(y; B_1, B_2)$ can no longer be obtained by capping $\mathcal{I}_{g_{\lambda^*}}(y)$ at the boundaries B_1 and B_2 .

Example 3.5.12.

Let Assumptions 3.5.1 and 3.5.9 hold. If g is a VaR constraint (as in (3.21)) and $y^* = (y_0^*, y_1^*)' \in (0, \infty) \times [0, \infty)$ is optimal for $(\mathbf{D}_{\lambda^*}^{\mathbf{V}})$, then the optimal terminal wealth for (\mathbf{P}) is

$$D^* = \mathcal{I}_{g_{\lambda^*}}(y^*; B_1, B_2) = \begin{cases} B_1, & \mathcal{I}_{g_{\lambda^*}}(y^*) < B_1 \& \tilde{Z}_{\lambda^*}(T) \ge \frac{\frac{1}{b}(B_{VaR}^b - B_1^b) + y_1^*}{y_0^*(B_{VaR} - B_1)} \\ B_{VaR}, & \mathcal{I}_{g_{\lambda^*}}(y^*) < B_1 \& \tilde{Z}_{\lambda^*}(T) < \frac{\frac{1}{b}(B_{VaR}^b - B_1^b) + y_1^*}{y_0^*(B_{VaR} - B_1)} \\ \mathcal{I}_{g_{\lambda^*}}(y^*), & B_1 \le \mathcal{I}_{g_{\lambda^*}}(y^*) \le B_2 \\ B_2, & B_2 < \mathcal{I}_{g_{\lambda^*}}(y^*). \end{cases}$$



Figure 3.3: Illustration of the dependence of the optimal terminal wealth D^* on $\tilde{Z}_{\lambda^*}(T)$ with VaR constraints and a given constant value of y^* (cf. Example 3.5.12). In Figure 3.3(a), B_{VaR} is large enough so that $\mathcal{I}_{g_{\lambda^*}}(y^*) < B_1$ already implies $\tilde{Z}_{\lambda^*}(T) \geq \frac{\frac{1}{b}(B_{VaR}^b - B_1^b) + y_1^*}{y_0^*(B_{VaR} - B_1)}$ and thus the pointwise constraints B_1 , B_2 just cap off $\mathcal{I}_{g_{\lambda^*}}(y^*)$. This is no longer the case in Figure 3.3(b), as B_{VaR} is substantially smaller.

Expected Shortfall Constraints: For constants $B_1 < B_{ES} < B_2$ and $\epsilon > 0$ we define the ES constraint as

$$g(x) = (B_{ES} - x) \mathbb{1}_{\{x < B_{ES}\}} - \epsilon.$$
(3.22)

The ES is an alternative risk measure which mitigates some of the main weaknesses of VaR. The ES constraint restricts the average portfolio loss in comparison to a benchmark $B_{ES} > 0$, if the loss is positive, to at most ϵ . Therefore, in contrast to the VaR constraint, the ES constraint also takes the magnitude of the loss into account, if it occurs. Further, the ES is a convex risk measure and thus always encourages diversification of payoffs. For these reasons, ES has been proposed as an alternative tool for financial risk management and its advantages as well as some of its drawbacks have been discussed by e.g. [2] [44] and [86].

Example 3.5.13.

Let Assumptions 3.5.1 and 3.5.9 hold. If g is an ES constraint (as in (3.22)), $B_1 = 0, B_2 = \infty$, and $y^* = (y_0^*, y_1^*)' \in (0, \infty) \times [0, \infty)$ is optimal for $(\mathbf{D}_{\lambda^*}^{\mathbf{V}})$, then the optimal terminal wealth for (**P**) is

$$D^* = \mathcal{I}_{g_{\lambda^*}}(y^*) = \begin{cases} \mathcal{I}(y_0^*), & B_{ES} < \mathcal{I}(y_0^*), \\ B_{ES}, & \mathcal{I}(y_0^*) \le B_{ES} \le \mathcal{I}\left(y_0^* - \frac{y_1^*}{\tilde{Z}_{\lambda^*}(T)}\right), \\ \mathcal{I}\left(y_0^* - \frac{y_1^*}{\tilde{Z}_{\lambda^*}(T)}\right), & \mathcal{I}\left(y_0^* - \frac{y_1^*}{\tilde{Z}_{\lambda^*}(T)}\right) < B_{ES}. \end{cases}$$

In contrast to the VaR constraint, $U(x) - y_0 \tilde{Z}_{\lambda^*}(T)x - y_1 g(x)$ remains strictly concave in x > 0for all $y = (y_0, y_1)' \in [0, \infty)^2$ and thus $\mathcal{I}_{g_{\lambda^*}}(y; B_1, B_2)$ can again be obtained by capping $\mathcal{I}_{g_{\lambda^*}}(y)$ at the boundaries B_1 and B_2 .

Example 3.5.14.

Let Assumptions 3.5.1 and 3.5.9 hold. If g is an ES constraint (as in (3.22)) and $y^* = (y_0^*, y_1^*)' \in (0, \infty) \times [0, \infty)$ is optimal for $(\mathbf{D}_{\lambda^*}^{\mathbf{V}})$, then the optimal terminal wealth for (\mathbf{P}) is

$$D^* = \mathcal{I}_{g_{\lambda^*}}(y^*; B_1, B_2) = Cap(\mathcal{I}_{g_{\lambda^*}}(y^*), B_1, B_2).$$



Figure 3.4: Illustration of the dependence of the optimal terminal wealth D^* on $\tilde{Z}_{\lambda^*}(T)$ with ES constraints and a given constant value of y^* (cf. Example 3.5.14).

3.6 Conclusion

In this chapter, we extended the duality framework of [17] and [34] to include simultaneous constraints on relative portfolio allocation and terminal wealth, where the constraints on wealth may be defined pointwise, in expected value or both. Similar to the previous work in [34], we were able to integrate the wealth constraints into the framework of [17] by defining a generalized concave conjugate of the utility function and using it to generalize the well-known martingale method for portfolio optimization in complete financial markets. Just as in [17], the developed framework allows us to disregard the allocation constraints if the drift coefficients in the financial market are adjusted suitably, i.e., if the 'optimal' auxiliary market \mathcal{M}_{λ^*} is considered. The optimal auxiliary market can be determined by a set of equivalent optimality conditions, which are induced by two distinct dual optimization problems which correspond to the allocation constraints and the terminal wealth constraints. The optimal terminal wealth for this fully constrained portfolio optimization problem can then be expressed as a deterministic function of the pointwise bounds B_1 , B_2 and the pricing kernel $Z_{\lambda^*}(T)$ in the optimal auxiliary market \mathcal{M}_{λ^*} . When assuming the dynamics of a Black-Scholes market, assuming that K is a convex cone and assuming some additional regularity conditions, then the adjustments to the market coefficients λ^* can be determined as the minimizer of a deterministic convex optimization problem. We illustrate our methodology for investors who face pointwise bounds on terminal wealth, Valueat-Risk constraints and expected shortfall constraints.

4 Constraints on Allocation in Stochastic Factor Models

4.1 Introduction

In this chapter, we consider a portfolio optimization problem of an investor who trades in continuous time and seeks to maximize his utility from terminal wealth at the end of a finite investment horizon. The investor is assumed to be risk-averse and his risk-preferences are modelled by a power utility function. Our problem setting differs from the classic problem formulated in [64] with respect to two main aspects: allocation constraints and market coefficients dependent on a stochastic factor. Since Section 1.1, already provided an overview over the relevant literature on allocation constraints, the following paragraph presents a brief overview of the relevant literature with respect to stochastic factor market dynamics:

Modelling stochastic market coefficients as a function of an additional stochastic factor is a natural extension to the classic Black-Scholes model which can capture some of the stylized facts observed in the financial market. One of the earliest discussions of such models in a portfolio optimization context was in [88], where the author was able to characterize the solution to the associated HJB equation for a power-utility function in terms of a linear parabolic PDE. Further, under the assumption of a global Lipschitz-condition on the market coefficients, a verification result was proven. However, explicit closed-form expressions for the optimal allocation were only given when the stochastic factor is completely uncorrelated with the financial market, i.e., when the optimal allocation is myopic. If the stochastic factor correlates with the financial market, closed-form expressions for the optimal allocation were recovered on individual occasions, e.g., in [20] and [53] for financial markets with stochastic short rate and in [56] for financial markets with stochastic volatility. These advances required the solvability of certain underlying Riccati ODEs. The seminal work of [61] unified these approaches by introducing a class of models where the asset returns have a quadratic dependence on the stochastic factor. Within such quadratic models, the author directly characterizes the HJB PDE as an exponentially quadratic function of the stochastic factor with coefficients determined by the solution to a system of Riccati ODEs. When reducing the framework of [61] to an affine dependence on the stochastic factor, the results are closely related to affine term structure models of [26]. This affine reduction proved to be particularly fruitful for portfolio optimization applications, see e.g. [46], [5] and [30]. In addition, an extensive overview of related literature is given in [90].

Both allocation constraints and stochastic factor market dynamics result in the investor not being able to replicate all measurable payoffs at the end of the investment horizon and therefore standard martingale techniques cannot be employed to characterize the value function of the optimization problem. Further, it is unclear if the Hamilton-Jacobi-Bellman PDE ('HJB' PDE) of the optimization problem admits a smooth solution due to pointwise constraints on the optimal relative portfolio allocation. Both aspects have only been studied simultaneously on rare occasions. [71] used a logarithmic transformation to characterize the solution to the constrained HJB PDE through the solution to a semi linear PDE. Assuming Lipschitz- and non-degeneracy conditions on the stochastic factor, the existence of a smooth solution to the transformed PDE can be guaranteed and closed-form expressions for the optimal portfolio can be given if the stochastic factor is uncorrelated with the financial market. In [21], semi-closed-form expressions are provided for the optimal allocation for a market with generalized Vasicek short rate and bounds on the portfolio allocation to a hedging instrument for interest-rate risk. [66] and [91] develop numerical schemes for allocation constrained portfolio optimization problems in jump-diffusion models, where the asset volatilities and jumps depend on an external stochastic factor.

Our contribution to this literature is threefold. First, we present an approach to constrained portfolio optimization that transforms the HJB PDE associated with the constrained portfolio optimization problem into an equivalent dual PDE, which is the HJBI PDE associated with a dual minimization problem akin to Condition (C) in [17] or Condition (\tilde{C}) from Chapter 3. However, unlike in [17], the validity of this method is not tied to the completeness of the underlying financial market and can thus be applied in a broader context. Secondly, in the spirit of [61], we derive a condition on the dynamics of the financial market and the allocation constraints, which ensures that the value function of the optimal investment problem is exponentially affine. Lastly, we provide expressions for the allocation constrained optimal allocation in a market with multi-factor stochastic volatility of CIR-Type and multi-factor short rate of OU-type. These expressions are explicit up to a deterministic minimizer and the solution of a system of Riccati ODEs, which leads to a non-myopic optimal allocation is generally non-myopic.

The remainder of this chapter is structured as follows: The financial market model, the portfolio optimization problem and the standing assumptions are introduced in Section 4.2. Afterwards, in Section 4.3, we use a result from real constrained optimization to show that the constrained HJB PDE associated with (\mathbf{P}) is equivalent to the Hamilton-Jacobi-Bellman-Isaacs PDE ('HJBI PDE') associated with a dual min-max problem and derive a condition under which the solution to both PDEs is exponentially affine. The versatility and use of the derived condition is illustrated in examples with deterministic market coefficients, stochastic volatility and stochastic short rate in Section 4.4. Finally, Section 4.5 concludes the chapter.

4.2 Setting

In this chapter, we consider a Markovian stochastic factor model, which is a similar setting as in the introduction to the stochastic control approach in Section 2.3.2. Specifically, we assume that there exist deterministic functions

$$\mu^{z}: [0,T] \times \mathbb{R}^{m} \to \mathbb{R}^{m}, \ \Sigma^{z}: [0,T] \times \mathbb{R}^{m} \to \mathbb{R}^{m \times m}, \ \rho: [0,T] \times \mathbb{R}^{m} \to \mathbb{R}^{m \times d}$$

$$r: [0,T] \times \mathbb{R} \to \mathbb{R}, \ \mu: [0,T] \times \mathbb{R}^{m} \to \mathbb{R}^{d} \text{ and } \Sigma: [0,T] \times \mathbb{R}^{m} \to \mathbb{R}^{d \times d},$$

$$(4.1)$$

and an m-dimensional stochastic factor z which satisfies the SDE

$$dz(t) = \mu^{z}(t, z(t))dt + \Sigma^{z}(t, z(t))'dW^{z}(t), \quad z(0) = z_{0} \in \mathbb{R}.$$
(4.2)

such that the market coefficients in \mathcal{M} are given as deterministic functions of (t, z(t)), i.e., we have $\mathcal{L}[0,T] \times Q$ -a.e.

$$\rho(t) = \rho(t, z(t)), \ r(t) = r(t, z(t)), \ \mu(t) = \mu(t, z(t)), \ \text{and} \ \Sigma(t) = \Sigma(t, z(t)).$$
(4.3)

In particular, the risk-free asset P_0 , the risky assets P and the Brownian motions W, W^z then satisfy

$$dP_0(t) = P_0(t) \cdot r(t, z(t))dt, \qquad P_0(0) = 1,$$

4 Constraints on Allocation in Stochastic Factor Models

$$dP(t) = \operatorname{diag}(P(t)) \cdot \left[\mu(t, z(t))dt + \Sigma(t, z(t))dW(t)\right], \quad P(0) = \mathbb{1} \in \mathbb{R}^d,$$

$$d\langle W_i^z, W_j \rangle_t = \rho_{ij}(t, z(t))dt, \quad \forall 1 \le i \le m, \ 1 \le j \le d.$$

We assume that the deterministic functions (4.3) are sufficiently regular such that the SDEs for z, P_0 and P admit a unique strong solution and fall within the market model of \mathcal{M} defined in Section 2.3. In particular, we require that Σ and the columns $\rho_1, ..., \rho_d$ of ρ satisfy

det
$$(\Sigma(t, z(t))) \neq 0$$
 and $\|\rho_i(t, z(t))\| \leq 1 \quad \forall i = 1, .., d$ hold $\mathcal{L}[0, T] \otimes Q$ - a.e.

As working within this setting becomes very notation-heavy, we only write μ^z , Σ^z , ρ , r, μ , Σ instead of $\mu^z(t, z(t))$, $\Sigma^z(t, z(t))$, $\rho(t, z(t))$, r(t, z(t)), $\mu(t, z(t))$, $\Sigma(t, z(t))$ if the context is unambiguous to improve the clarity of presentation. In this sense, the wealth process $V^{v_0,\pi}$ of an investor with initial wealth $v_0 > 0$ and trading in \mathcal{M} according to a *d*-dimensional relative portfolio process $\pi \in \Lambda$ satisfies the usual SDE

$$V^{v_0,\pi}(0) = v_0$$

$$dV^{v_0,\pi}(t) = V^{v_0,\pi}(t) \left(\left[r + (\mu - r\mathbf{1})'\pi(t) \right] dt + \pi(t)' \Sigma dW(t) \right)$$

and can be expressed in closed form as

$$V^{v_0,\pi}(t) = v_0 \exp\left(\int_0^t r + (\mu - r\mathbb{1})'\pi(s) - \frac{1}{2}\|\Sigma'\pi(s)\|^2 ds + \int_0^t \pi(s)'\Sigma dW(s)\right).$$

In this chapter we only consider allocation constraints , i.e., for a non-empty closed convex set $K \subset \mathbb{R}^d$, we define the set of admissible constrained portfolio processes as

$$\Lambda' := \Lambda(K) := \left\{ \pi \in \Lambda \mid \pi(t) \in K \ \mathcal{L}[0,T] \otimes Q - \text{a.e.} \right\}.$$
(4.4)

For a power utility function $U(v) = \frac{1}{b}v^b$ with b < 1 and $b \neq 0$ we then define the allocation constrained primal portfolio optimization problem as

$$(\mathbf{P}) \begin{cases} \Phi(v_0) &= \sup_{\pi \in \Lambda} \mathbb{E} \left[U(V^{v_0,\pi}(T)) \right], \\ \Lambda' &= \left\{ \pi(t) \in K \ \mathcal{L}[0,T] \otimes Q - \text{a.e.} \mid \pi \in \Lambda \right\}. \end{cases}$$

Since we consider a Markovian setting and did not assume that $m \neq 0$, we aim to solve (**P**) using classic methods from stochastic optimal control and thus need to determine a solution to the HJB PDE associated with (**P**). Even without the additional presence of allocation constraints, this is notoriously difficult in the general setting described above. For this reason, we devote the upcoming Section 4.3 to deriving an equivalent dual representation of the associated HJB PDE, which leads to a dual approach to solving the allocation constrained portfolio optimization problem (**P**). The dual representation is closely related to Condition (\tilde{C}) from Section 3.4.

The standing assumptions for this chapter are summarized below:

Assumption 4.2.1 (Standing Assumptions Chapter 4:). \mathcal{M} is a stochastic factor model ((4.2) and (4.3)), (4.4) and U is a power utility function.

4.3 The Dual HJBI PDE

We approach (**P**) using classic methods from stochastic optimal control and frequently work with so-called feedback controls $\pi \in \Lambda$, which are defined in feedback-form

$$\pi(t) = \underline{\pi}(t, V^{v_0, \pi}(t), z(t)),$$

4 Constraints on Allocation in Stochastic Factor Models

for a deterministic measurable function $\underline{\pi} : [0, T] \times (0, \infty) \times \mathbb{R}^m \to \mathbb{R}^d$ of the current state $(t, V^{v_0,\pi}(t), z(t))$ of the financial market \mathcal{M} . As these processes are uniquely defined through the function $\underline{\pi}$, we use the process π and the function $\underline{\pi}$ interchangeably. When a process is specifically defined in feedback-form, we follow the notation in [39] and denote this fact by a 'lower bar', i.e., $\underline{\pi}$. The set of admissible unconstrained Markovian controls is thus denoted by $\underline{\Lambda}$ and the wealth-process corresponding to a Markovian control $\underline{\pi}$ is denoted by $V^{v_0,\underline{\pi}}$. The remaining notation will carry over analogously. Further, we also introduce the generalized primal portfolio optimization problem ($\mathbf{P}^{(\mathbf{t},\mathbf{v},\mathbf{z})}$) as

$$(\mathbf{P}^{(\mathbf{t},\mathbf{v},\mathbf{z})}) \begin{cases} \Phi(t,v,z) &= \sup_{\pi \in \Lambda'(t)} \mathbb{E} \left[U(V^{v_0,\pi}(T)) \mid V^{v_0,\pi}(t) = v, \ z(t) = z \right] \\ \Lambda'(t) &= \left\{ \left(\pi(s) \right)_{s \in [t,T]} \mid \pi \in \Lambda_K \right\}. \end{cases}$$

Then, the Hamilton-Jacobi-Bellman equation ('HJB equation') associated with $(\mathbf{P}^{(\mathbf{t},\mathbf{v},\mathbf{z})})$ is given by

$$0 = \sup_{\pi \in K} \left\{ G_t + v \left[r + (\mu - r\mathbf{1})'\pi \right] G_v + \frac{1}{2} v^2 \|\Sigma'\pi\|^2 G_{vv} + (\mu^z)' (\nabla_z G) + v \left(\Sigma^z \rho \Sigma'\pi\right)' \nabla_z (G_v) + \frac{1}{2} \operatorname{Trace} \left[\Sigma^z (\Sigma^z)' \nabla_z^2 G\right] \right\}$$

$$= G_t + vrG_v + (\mu^z)' (\nabla_z G) + \frac{1}{2} \operatorname{Trace} \left[\Sigma^z (\Sigma^z)' \nabla_z^2 G\right] + v \sup_{\pi \in K} \left\{ (\mu - r\mathbf{1})'\pi G_v + (\Sigma^z \rho \Sigma'\pi)' \nabla_z (G_v) + \frac{1}{2} v \|\Sigma'\pi\|^2 G_{vv} \right\}$$
(4.5)
$$G(T, v, z) = U(v),$$

Any (sufficiently regular) solution G to (4.5) yields a candidate optimal Markovian control through the maximizing argument

$$\underline{\pi}^{*}(t,v,z) = \underset{\pi \in K}{\operatorname{argmax}} \Big\{ (\mu - r\mathbb{1})' \pi G_{v} + \left(\Sigma^{z} \rho \Sigma' \pi \right)' \nabla_{z} \left(G_{v} \right) + \frac{1}{2} v \| \Sigma' \pi \|^{2} G_{vv} \Big\}.$$
(4.6)

4.3.1 Dual Approach to Allocation Constrained Portfolio Optimization

In this subsection, we use Corollary 2.1.32 to derive an equivalent dual representation of the HJB equation (4.5), which can be regarded as the Hamilton-Jacobi-Bellman-Isaacs equation ('HJBI equation') associated with a dual optimization problem over a certain class of stochastic processes. This dual optimization problem as well as our solution approach to constrained portfolio optimization will closely resemble optimality Condition (\tilde{C}) from Section 3.4 (resp. Condition (C) from [17]). However, unlike in Chapter 3, we arrive at this approach applying duality arguments directly to the pointwise optimization on the level of the HJB PDE (4.5) rather than on the level of stochastic processes. This reduces the level of technicality involved and removes the necessity for market completeness and the solvability of the wealth-constrained dual problem ($\mathbf{D}_{\lambda^*}^{\mathbf{V}}$) as central underlying assumptions.

Lemma 4.3.1. (Dual HJBI PDE)

Let $G \in C^{(1,2,2)}([0,T] \times (0,\infty) \times \mathbb{R}^m)$ be strictly concave and strictly increasing in the second component v. Then, G is a solution to (4.5) if and only if G(T, v, z) = U(v) and

$$0 = G_t + vrG_v + (\mu^z)' (\nabla_z G) + \frac{1}{2} Trace \left[\Sigma^z (\Sigma^z)' \nabla_z^2 G \right]$$

4 Constraints on Allocation in Stochastic Factor Models

$$+ v \inf_{\lambda \in \mathbb{R}^{d}} \sup_{\pi \in \mathbb{R}^{d}} \left\{ \left[\delta_{K}(\lambda) + (\mu + \lambda - r\mathbb{1})'\pi \right] G_{v} + \left(\Sigma^{z}\rho\Sigma'\pi \right)' \nabla_{z} \left(G_{v} \right) + \frac{1}{2}v \|\Sigma'\pi\|^{2} G_{vv} \right\}$$

$$(4.7)$$

$$= G_{t} + vrG_{v} + (\mu^{z})' \left(\nabla_{z}G \right) + \frac{1}{2}Trace \left[\Sigma^{z} \left(\Sigma^{z} \right)' \nabla_{z}^{2}G \right]$$

$$+ v \inf_{\lambda \in \mathbb{R}^{d}} \left\{ \delta_{K}(\lambda)G_{v} - \frac{1}{2}\frac{1}{vG_{vv}} \|\Sigma^{-1} \left[\mu + \lambda - r\mathbb{1} \right] G_{v} + \left(\Sigma^{z}\rho \right)' \nabla_{z} \left(G_{v} \right) \|^{2} \right\}.$$

The dual PDE (4.7) is the HJBI PDE (see e.g. Section 4.2 in [45]) associated with the min-max stochastic control problem

$$\inf_{\lambda \in \mathcal{D}} \sup_{\pi \in \Lambda} \mathbb{E} \left[U \left(\underbrace{V^{v_0, \pi}(T) \cdot \exp\left(\int_0^T \lambda(t)' \pi(t) + \delta_K(\lambda(t)) dt\right)}_{=:V_{\lambda}^{v_0, \pi}(T)} \right) \right], \tag{4.8}$$

where the dual process $\lambda = (\lambda(t))_{0 \le t \le T}$ is taken from the set of progressively measurable processes (cf. Section 3.4)

$$\mathcal{D} = \left\{ \lambda = \left((\lambda_1(t), ..., \lambda_d(t))' \right)_{t \in [0,T]} \text{ prog. measurable} \right|$$
$$\sup_{0 \le t \le T} \|\lambda(t)\|^2 < \infty \text{ } Q\text{-a.s.}, \ \sup_{0 \le t \le T} \delta_K(\lambda(t)) < \infty \text{ } Q\text{-a.s.} \right\}.$$

Just as with portfolio processes, we refer to dual processes defined in feedback form $\lambda(t) = \underline{\lambda}(t, V_{\lambda}^{v_0,\pi}(t), z(t))$, for a deterministic measurable function $\underline{\lambda}$, as Markovian dual controls. Analogously, when specifically referring to Markovian dual controls, we write ' $\underline{\lambda}$ ' and collect all admissible Markovian controls in $\underline{\mathcal{P}}$.

Although intuitively appealing, the relationship between HJB(I) PDEs and the associated optimization problems still requires formal mathematical justification via verification theorems. Due to the generality of the setting considered in this chapter, we can only provide general verification theorems under additional assumptions on the candidate optimal controls π^* (and λ^*), the solution G to the HJB(I) PDE and the financial market \mathcal{M} . Verifying such conditions is typically only feasible in more narrowly focussed settings (see e.g. Corollary 4.4.3 and 4.4.10 in Section 4.4).

Here, we make the relation between the PDE (4.7) and the dual control problem (4.8) more precise, by proving a verification theorem, which relies on an additional uniform integrability condition (UI_{λ}) (compare to e.g. Definition 4.2 in [56]).

Condition (UI_{λ}). For given $n \in \mathbb{N}$, $t \in [0,T]$, $G \in C^{(1,2,2)}([0,T] \times (0,\infty) \times \mathbb{R}^m)$, $\lambda \in \mathcal{D}$, $\pi \in \Lambda$ we define the stopping time $\tau_{n,t}^{\lambda} = \min(T, \hat{\tau}_{n,t}^{\lambda})$, with

$$\hat{\tau}_{n,t}^{\lambda} = \inf \Big\{ t \le u \le T \Big| \int_{t}^{u} \big(V_{\lambda}^{v_{0},\pi}(s) \cdot \|\Sigma(s,z(s))'\pi(s)\| \cdot G_{v}(s,V_{\lambda}^{v_{0},\pi}(s),z(s)) \big)^{2} \, ds \ge n, \\ \int_{t}^{u} \| \left(\Sigma^{z}(s,z(s))' \, \nabla_{z}\left(G\right)\left(s,V_{\lambda}^{v_{0},\pi}(s),z(s)\right) \|^{2} \, ds \ge n \Big\}.$$

We say that G, π , and λ satisfy condition (UI_{λ}) if for every $t \in [0,T]$, the sequence

$$\left(G(\tau_{n,t}^{\lambda}, V_{\lambda}^{v_0, \pi}(\tau_{n,t}^{\lambda}), z(\tau_{n,t}^{\lambda}))\right)_{n \in \mathbb{N}}$$

is uniformly integrable.

Remark 4.3.2. According to Theorem 4.5.4 in [8], if G, π , and λ satisfy Condition (UI_{λ}), then we have for every $t \in [0,T]$ that $\tau_{n,t} \to T$ Q-a.s., as $n \to \infty$ and

$$\mathbb{E}\Big[G(T, V_{\lambda}^{v_0, \pi}(T), z(T)) \ \Big| \mathcal{F}_t\Big] = \mathbb{E}\Big[\lim_{n \to \infty} G(\tau_{n,t}, V_{\lambda}^{v_0, \pi}(\tau_{n,t}), z(\tau_{n,t})) \ \Big| \mathcal{F}_t\Big]$$
$$= \lim_{n \to \infty} \mathbb{E}\Big[G(\tau_{n,t}, V_{\lambda}^{v_0, \pi}(\tau_{n,t}), z(\tau_{n,t})) \ \Big| \mathcal{F}_t\Big]. \tag{4.9}$$

Lemma 4.3.3 (Verification Theorem Dual Control Problem).

Let $G \in C^{(1,2,2)}([0,T] \times (0,\infty) \times \mathbb{R}^m)$ be a solution to the dual HJB equation (4.7), be nonnegative, strictly concave and increasing in v. Let the feedback controls $\underline{\lambda}^*(t,v,z), \ \underline{\pi}^*(t,v,z)$ be such that for all $(t,v,z) \in [0,T] \times (0,\infty) \times \mathbb{R}^m$

$$\inf_{\lambda \in \mathbb{R}^d} \sup_{\pi \in \mathbb{R}^d} \left\{ \left[\delta_K(\lambda) + (\mu + \lambda - r\mathbb{1})'\pi \right] G_v + \left(\Sigma^z \rho \Sigma' \pi \right)' \nabla_z \left(G_v \right) + \frac{1}{2} v \|\Sigma' \pi\|^2 G_{vv} \right\} \\
= \left[\delta_K(\underline{\lambda}^*(t, v, z)) + (\mu + \underline{\lambda}^*(t, v, z) - r\mathbb{1})' \underline{\pi}^*(t, v, z) \right] G_v \\
+ \left(\Sigma^z \rho \Sigma' \underline{\pi}^*(t, v, z) \right)' \nabla_z \left(G_v \right) + \frac{1}{2} v \|\Sigma' \underline{\pi}^*(t, v, z)\|^2 G_{vv}.$$

Then the following holds $\forall (t, v, z) \in [0, T] \times (0, \infty) \times \mathbb{R}^m$:

(i) If $(\underline{\lambda}^*, \pi) \in \mathcal{D} \times \Lambda$ satisfy condition (UI_{λ}) , then

$$G(t, v, z) \ge \mathbb{E} \left[U(V_{\underline{\lambda}^*}^{v_0, \pi}(T)) \mid V_{\underline{\lambda}^*}^{v_0, \pi}(t) = v, \ z(t) = z \right].$$
(4.10)

(ii) If $(\lambda, \underline{\pi}^*) \in \mathcal{D} \times \underline{\Lambda}$ satisfy condition (UI_{λ}) , then

$$G(t, v, z) \le \mathbb{E} \left[U(V_{\lambda}^{v_0, \underline{\pi}^*}(T)) \mid V_{\lambda}^{v_0, \underline{\pi}^*}(t) = v, \ z(t) = z \right].$$
(4.11)

(iii) If $(\underline{\lambda}^*, \underline{\pi}^*) \in \mathcal{D} \times \underline{\Lambda}$ satisfy condition (UI_{λ}) , then

$$G(t, v, z) = \mathbb{E}\left[U(V_{\underline{\lambda}^*}^{v_0, \underline{\pi}^*}(T)) \mid V_{\underline{\lambda}^*}^{v_0, \underline{\pi}^*}(t) = v, \ z(t) = z\right].$$
(4.12)

Remark 4.3.4. If we restrict the minimization and maximization in the min-max optimization only to such $\lambda \in \mathcal{D}_{UI} \subset \mathcal{D}$, $\pi \in \Lambda_{UI} \subset \Lambda$ so that every pair (λ, π) , $(\underline{\lambda}^*, \pi)$, $(\lambda^*, \underline{\pi}^*)$ and $(\underline{\lambda}^*, \underline{\pi}^*)$ and the solution G to the dual HJB PDE (4.7) satisfy Condition (UI_{λ}) , then we directly obtain from Lemma 4.3.3

$$\begin{split} G(t,v,z) &= \mathbb{E} \left[U(V_{\underline{\lambda}^*}^{v_0,\underline{\pi}^*}(T)) \ \big| \ V_{\underline{\lambda}^*}^{v_0,\underline{\pi}^*}(t) = v, \ z(t) = z \right] \\ &= \inf_{\lambda \in \mathcal{D}_{UI}} \sup_{\pi \in \Lambda_{UI}} \mathbb{E} \left[U(V_{\lambda}^{v_0,\pi}(T)) \ \big| \ V_{\lambda}^{v_0,\pi}(t) = v, \ z(t) = z \right] \end{split}$$

(4.8) is the min-max control problem associated with Condition (\tilde{C}) from Section 3.4 in the absence of wealth constraints (resp. Condition (C) from [17]). However, we arrived at the same optimization problems by applying convex duality results from real constraints directly to the pointwise optimization at the level of the HJB PDE, whereas we applied martingale methods to the underlying stochastic processes in Chapter 3. For the setting of Chapter 3, were able to to prove that the optimal controls for the dual control problem (4.8) lead to an optimal portfolio process for the allocation constrained portfolio optimization problem (\mathbf{P}). In doing so, we used the so-called (generalized) concave conjugate of U to transform the dual control problem (4.8) to 'another' dual representation of (\mathbf{P}). These arguments heavily relied on the completeness of the underlying financial market (Assumption 3.2.1) as well as the solvability of the dual optimization problem ($\mathbf{D}_{\lambda^*}^{\mathbf{V}}$) associated with the wealth constraints and are thus not directly available to us in this chapter.

4.3.2 Exponential Affine Separability

In this section, we derive a condition under which the solution G to the dual HJB PDE (4.7) is of an exponentially affine and separable form, i.e.,

$$G(t, v, z) = \frac{1}{b} v^b \exp\left(A(T-t) + B(T-t)'z\right),$$
(4.13)

for some functions $A : [0,T] \to \mathbb{R}$, and $B : [0,T] \to \mathbb{R}^m$ with A(0) = 0 and B(0) = 0. In a setting without the presence of allocation constraints and time-independent market coefficients, [61] provides such a condition which can be directly verified for any given market coefficients (see equations (9)-(11) and (13)-(17) in [61]).¹ Under the presence of additional constraints on allocation, we need to adapt this condition suitably.

To this end, for any $(t, z, B) \in [0, T] \times \mathbb{R}^m \times \mathbb{R}^m$, we define $\hat{\lambda}^*(t, z, B)$ as the minimizing argument

$$\hat{\lambda}^{*}(t,z,B) = \underset{\lambda \in \mathbb{R}^{d}}{\operatorname{argmin}} \left\{ 2(1-b)\delta_{K}(\lambda) + \left\| \Sigma^{-1} \left(\mu - r\mathbb{1} + \lambda \right) + \left(\Sigma^{z}\rho \right)' B \right\|^{2} \right\}$$

$$= \underset{\lambda \in \mathbb{R}^{d}}{\operatorname{argmin}} \left\{ 2(1-b)\delta_{K}(\lambda) + 2\lambda' \left(\Sigma \cdot \Sigma' \right)^{-1} \left[\mu - r\mathbb{1} + \left(\Sigma^{z}\rho \Sigma' \right)' B \right] + \left\| \Sigma^{-1}\lambda \right\|^{2} \right\}.$$

$$(4.14)$$

Given $\hat{\lambda}^*$, we provide a condition which ensures that (4.13) holds. This can be achieved by considering the corresponding condition from [61] and augmenting the market coefficients by $\hat{\lambda}^*$.

Condition (EAS).

We say that Condition (EAS) is satisfied if for any $(t, z, B) \in [0, T] \times \mathbb{R}^m \times \mathbb{R}^m$ the market coefficients and the minimizer $\hat{\lambda}^*$ satisfy

$$\begin{split} \mu^{z}(t,z) &= k_{0}(t) + k_{1}(t)z\\ \Sigma^{z}(t,z)\Sigma^{z}(t,z)' &= h_{0}(t) + h_{1}(t)[z]\\ \Sigma^{z}(t,z)\rho(t,z) \left(\Sigma^{z}(t,z)\rho(t,z)\right)' - \Sigma^{z}(t,z)\Sigma^{z}(t,z)' &= l_{0}(t) + l_{1}(t)[z]\\ r(t,z) + \delta_{K}(\hat{\lambda}^{*}(t,z,B)) &= p_{0}(t,B) + p_{1}(t,B)'z\\ \left\|\Sigma^{-1}(t,z) \left(\mu(t,z) + \hat{\lambda}^{*}(t,z,B) - r(t,z)\mathbb{1}\right)\right\|^{2} &= q_{0}(t,B) + q_{1}(t,B)'z\\ \Sigma^{z}(t,z)\rho(t,z)\Sigma^{-1}(t,z) \left(\mu(t,z) + \hat{\lambda}^{*}(t,z,B) - r(t,z)\mathbb{1}\right) = g_{0}(t,B) + g_{1}(t,B)z, \end{split}$$

for some functions such that $p_0(t, B)$, $q_0(t, B) \in \mathbb{R}$, $k_0(t)$, $p_1(t, B)$, $q_1(t, B)$, $g_0(t, B) \in \mathbb{R}^m$ as well as $k_1(t)$, $h_0(t)$, $l_0(t)$, $g_1(t, B) \in \mathbb{R}^{m \times m}$ and the functions $h_1(t)[\cdot]$, $l_1(t)[\cdot] : \mathbb{R}^m \to \mathbb{R}^{m \times m}$ are linear² for every fixed $(t, B) \in [0, T] \times \mathbb{R}^m$.

¹In fact, the result of [61] even includes the more general case of an exponentially quadratic separation. The approach we present below can be extended to include quadratic separation in a natural manner. However, such an extension would complicate the involved notation and thus diminish the presentation of the core concepts involved. Moreover, we were not able to construct realistic working examples that require quadratic separation in an allocation constrained setting. Hence, we restrict our analysis in this chapter to exponentially affine separation.

²The functions $h_1[\cdot]$ and $l_1[\cdot]$ are three-dimensional tensors, which are a generalization of vectors and matrices to higher dimensions. In our context, we may think of h_1 and l_1 as being represented by matrices, whose entries $(h_1)_{ij}$ and $(l_1)_{ij}$ are \mathbb{R}^m -valued. Upon being evaluated at a $z \in \mathbb{R}^m$, each entry of $h_1[z]$ and $l_1[z]$ is obtained by computing the scalar product $z'(h_1)_{ij}$ and $z'(l_1)_{ij}$. Hence, for any $x, y \in \mathbb{R}^m$ and applying the rules of ordinary vector-matrix multiplication, the products $x'h_1[\cdot]y$ and $x'l_1[\cdot]y$ are vectors in \mathbb{R}^m . In particular, $x'h_1[z]y = z'(x'h_1[\cdot]y)$ and $x'l_1[z]y = z'(x'l_1[\cdot]y)$ for any $x, y, z \in \mathbb{R}^m$.

Provided that Condition (EAS) is satisfied, we can characterize the exponents A and B in (4.13) through the system of $ODEs^3$

$$A_{\tau}(\tau) = bp_0 \left(T - \tau, B(\tau)\right) + k_0 (T - \tau)' B(\tau) + \frac{1}{2} B(\tau) h_0 (T - \tau) B(\tau) + \frac{1}{2} \frac{b}{1 - b} \Big[q_0 \left(T - \tau, B(\tau)\right) + 2g_0 \left(T - \tau, B(\tau)\right)' B(\tau) + B(\tau) \left(l_0 (T - \tau) + h_0 (T - \tau) \right) B(\tau) \Big]$$
(4.15)

$$B_{\tau}(\tau) = bp_{1}(T - \tau, B(\tau)) + k_{1}'(T - \tau)B(\tau) + \frac{1}{2}B(\tau)'h_{1}(T - \tau)[\cdot]B(\tau) + \frac{1}{2}\frac{b}{1 - b}\Big[q_{1}(T - \tau, B(\tau)) + 2g_{1}(T - \tau, B(\tau))B(\tau) + B(\tau)'(l_{1}(T - \tau)[\cdot] + h_{1}(T - \tau)[\cdot])B(\tau)\Big].$$
(4.16)

Transferred to the context of Chapter 3, while disregarding the wealth constraints, Condition (EAS) effectively ensures that the allocation unconstrained optimization problem (\mathbf{P}_{λ^*}) in \mathcal{M}_{λ^*} for the optimal dual process $\lambda^* = \hat{\lambda}^*(t, z(t), B(T - t))$ can be solved via the methodology presented in [61]. This idea is depicted in Table 4.1. From this perspective, it is not surprising that we can obtain a solution to the primal HJB PDE (4.5) under Condition (EAS).

Theorem 4.3.5. Let Condition (EAS) be satisfied and let A, B be solutions to the ODEs (4.15) and (4.16) with initial condition A(0) = 0, B(0) = 0. Then,

$$G(t, v, z) = \frac{1}{b}v^b \exp\left(A(T-t) + B(T-t)'z\right)$$

is a solution to the primal HJB PDE (4.5) and the dual HJBI PDE (4.7).

Remark 4.3.6. Although we can by no means provide explicit solutions to the ODEs (4.15) and (4.16) in general, at least the local existence of a solution is guaranteed by the existence theorems of Peano (and Picard-Lindelöf) if their respective right-hand sides are continuous (Lipschitz-continuous). In particular, we can then obtain an approximate solution to the dual HJBI PDE (4.7) for small $\tau = T - t$ by approximating A and B, by e.g. the Euler method.

	No Allocation Constraints $(K = \mathbb{R}^d)$	Allocation Constraints $(K \subset \mathbb{R}^d)$
Black-Scholes Model	Merton, 1971 ([65]): $\pi^* = \frac{1}{1-b} (\Sigma \Sigma')^{-1} (\mu - r\mathbb{1})$	Cvitanic & Karatzas, 1992 ([17]) $\pi^* = \frac{1}{1-b} (\Sigma \Sigma')^{-1} (\mu - r\mathbb{1} + \lambda^*),$ where λ^* is obtained from dual problem $\Rightarrow \pi^*$ optimal for $K = \mathbb{R}^d$ in \mathcal{M}_{λ^*}
Stochastic Factor Model	Liu, 2006 ([61]): $\pi^* = \frac{1}{1-b} (\Sigma\Sigma')^{-1} (\mu - r\mathbb{1} + (\Sigma^z \rho \Sigma')' B(T-t)),$ if Condition (EAS) is satisfied with $\lambda^* = 0$ & B solves associated ODE (4.16)	$ \begin{array}{l} \textbf{Theorem 4.3.5:} \\ \pi^* = \frac{1}{1-b} \left(\Sigma \Sigma' \right)^{-1} \left(\mu - r \mathbb{1} + \lambda^* + \left(\Sigma^2 \rho \Sigma' \right)' B(T-t) \right), \\ \text{where } \lambda^* \text{ is obtained from dual problem,} \\ \text{market coefficients in } \mathcal{M}_{\lambda^*} \text{ satisfy Condition (EAS)} \\ \& B \text{ solves associated ODE (4.16)} \end{array} $

Table 4.1: A schematic illustration of how Theorem 4.3.5 relates the existing results of [65], [61] and [17].

³Here, A_{τ} and B_{τ} denote the derivatives of A and B with respect to $\tau \in [0, T]$.

If Condition (EAS) is satisfied, then we can extend the verification approach used in [13] to formally verify the optimality of the obtained candidate optimal portfolio π^* . Unlike in Lemma 4.3.3, we only need to assume that π^* satisfies a uniform integrability condition and not make any assumption about other portfolios $\pi \in \Lambda_K$. This is possible because we can exploit the additional knowledge that G is exponentially affine due to Condition (EAS).

Theorem 4.3.7 (Verification Theorem Primal Problem).

Let Condition (EAS) be satisfied, let A and B be solutions to the ODEs (4.15) and (4.16) with initial condition A(0) = 0 and B(0) = 0 and let G be defined as in (4.13). Define $\underline{\lambda}^*(t, v, z) := \hat{\lambda}^*(t, z, B(T-t))$ and

$$\underline{\pi}^{*}(t,v,z) := \frac{1}{1-b} \left(\Sigma \Sigma' \right)^{-1} \left[\mu + \underline{\lambda}^{*}(t,v,z) - r \mathbb{1} + \left(\Sigma^{z} \rho \Sigma' \right)' B(T-t) \right].$$
(4.17)

If $G, \pi^*, \lambda \equiv 0$ satisfy Condition (UI_{λ}), then

$$G(t, v, z) = \mathbb{E}\left[U(V^{v_0, \underline{\pi}^*}(T)) \mid V^{v_0, \underline{\pi}^*}(t) = v, \ z(t) = z\right]$$
(4.18)

$$\geq \mathbb{E}\left[U(V^{v_0,\pi}(T)) \mid V^{v_0,\pi}(t) = v, \ z(t) = z\right] \quad \forall \pi \in \Lambda_K(t).$$
(4.19)

In particular, $G(t, v, z) = \Phi(t, v, z)$, for all $(t, v, z) = [0, T] \times (0, \infty) \times \mathbb{R}^m$ and $\underline{\pi}^*$ is optimal for (**P**).

Remark 4.3.8. If Condition (EAS) is satisfied, then we can alternatively use Theorem 4.3.5, compute the derivative of G and insert them into equation (4.6) to rewrite π^* as

$$\underline{\pi}^*(t,v,z) = \underset{\pi \in K}{\operatorname{argmax}} \left([\mu - r\mathbb{1}]'\pi + (1-b)\frac{1}{2}\pi'\Sigma\Sigma'\pi + B(T-t)'\Sigma^z\rho\Sigma'\pi \right).$$

In other words, we may equivalently write the optimal constrained feedback control π^* from (4.17) as the solution to a constrained pointwise optimization problem.

4.4 Examples

We consider three different choices of models for which Condition (EAS) can be verified and an explicit expression for the ODEs (4.15) and (4.16) can be derived. Throughout the examples, we always follow the same steps in chronological order:

- (i) Define the underlying financial market model, by choosing the market coefficients μ^z , Σ^z , ρ , r, μ and Σ .
- (ii) Derive an explicit representation of the minimizer $\hat{\lambda}^*$ of (4.14) in the given market.
- (iii) Verify that Condition (EAS) is satisfied for the given market.
- (iv) Derive an explicit representation for the ODEs (4.15), (4.16) and the candidate optimal portfolio $\underline{\pi}^*$ in terms of the market coefficients and $\hat{\lambda}^*$.
- (v) If possible: Formally verify the optimality of $\underline{\pi}^*$ for (**P**) by proving that $\underline{\pi}^*$, G (as in (4.13)) and $\lambda \equiv 0$ satisfy Condition (UI_{λ}).

4.4.1 Black-Scholes Model

First, we consider a *d*-dimensional Black-Scholes model \mathcal{M}_{BS} with time-dependent coefficients, which is exactly the setting of Section 15 in [17] (and a canonical generalization of Section 3.5.3 with $g \equiv 0$, $B_1 = 0$ and $B_2 = \infty$).

Definition 4.4.1 (\mathcal{M}_{BS}).

Let m = 1 and $d \in \mathbb{N}^4$ Consider continuous functions $r: [0,T] \to \mathbb{R}$, $\mu: [0,T] \to \mathbb{R}^d$, and $\Sigma: [0,T] \to \mathbb{R}^{d \times d}$ such that the inverse $\Sigma(t)^{-1}$ exists for all $t \in [0,T]$. Then, the d-dimensional Black-Scholes market \mathcal{M}_{BS} is defined by the market coefficients

$$z_0 = \mu^z(t, z) = \Sigma^z(t, z) = \rho(t, z) = 0,$$

and $r(t, z) = r(t), \quad \mu(t, z) = \mu(t), \quad \Sigma(t, z) = \Sigma(t)$

We can directly apply the duality theory developed in Section 4.3 to verify Condition (EAS) and obtain a solution to the HJBI PDE (4.7) in \mathcal{M}_{BS} .

Lemma 4.4.2 (Dual ODEs in \mathcal{M}_{BS}).

Consider the financial market \mathcal{M}_{BS} . Then Condition (EAS) is satisfied. Let

$$\lambda^*(t) = \underset{\lambda \in \mathbb{R}^d}{\operatorname{argmin}} \left\{ 2(1-b)\delta_K(\lambda) + \left\| \Sigma(t)^{-1}(\mu(t) - r(t)\mathbb{1} + \lambda) \right\|^2 \right\}$$
(4.20)

and $A: [0,T] \to \mathbb{R}$ satisfy A(0) = 0 and

$$A_{\tau}(\tau) = br(T-\tau) + \frac{1}{2} \frac{b}{1-b} \inf_{\lambda \in \mathbb{R}^d} \left\{ 2(1-b)\delta_K(\lambda) + \left\| \Sigma(T-\tau)^{-1} (\mu(T-\tau) - r(T-\tau)\mathbb{1} + \lambda) \right\|^2 \right\}$$

Then,

$$G(t, v, z) = \frac{1}{b}v^b \exp(A(T - t))$$

is a solution to the dual HJBI PDE (4.7) and the corresponding candidate optimal portfolio is

$$\underline{\pi}^{*}(t,v,z) = \frac{1}{1-b} \left(\Sigma(t)\Sigma(t)' \right)^{-1} \left(\mu(t) - r(t)\mathbb{1} + \lambda^{*}(t) \right).$$

Unsurprisingly, the candidate optimal portfolio process $\underline{\pi}^*$ proposed by Lemma 4.4.2 is the same as that obtained in Example 15.2 by [17] via the auxiliary markets methodology. Moreover, due to the simplicity of this set-up, we can even formally verify the optimality of $\underline{\pi}^*$ by showing that Condition (UI_{λ}) is satisfied.

Corollary 4.4.3. Consider the financial market \mathcal{M}_{BS} . Then, G, π^* as in Lemma 4.4.2 and $\lambda \equiv 0$ satisfy Condition (UI_{λ}) . In particular, π^* is optimal for (**P**).

Remark 4.4.4. The optimal unconstrained portfolio in \mathcal{M}_{BS} is given by the Merton portfolio

$$\pi_M(t) = \frac{1}{1-b} \left(\Sigma(t) \Sigma(t)' \right)^{-1} \left(\mu(t) - r(t) \mathbb{1} \right).$$

⁴We could equivalently consider m = 0 and completely disregard the stochastic factor z and its drift, diffusion and correlation coefficients in the definition of the Black-Scholes model \mathcal{M}_{BS} .

Following the argument in Remark 4.3.8, we realize that the candidate optimal constrained portfolio $\underline{\pi}^*$ in \mathcal{M}_{BS} can be obtained for arbitrary constraints $K \subset \mathbb{R}^d$ through the application of the (pointwise) projection $\mathcal{P}_K^{BS} : \Lambda \to \Lambda'$ defined through

$$\begin{split} \pi^*(t,v,z) &= \pi^* = \operatorname*{argmax}_{\pi \in K} \left(\underbrace{[\mu - r\mathbf{1}]'}_{=(1-b)\pi'_M \Sigma \Sigma'} \pi + (1-b) \frac{1}{2} \pi' \Sigma \Sigma' \pi + \underbrace{B(T-t)'}_{=0} \Sigma^z \rho \Sigma' \pi \right) \\ &= \operatorname*{argmax}_{\pi \in K} \left((1-b) \pi'_M \Sigma \Sigma' \pi - (1-b) \frac{1}{2} \pi' \Sigma \Sigma' \pi \right) \\ &= \operatorname*{argmin}_{\pi \in K} \left\| \Sigma'(\pi - \pi_M) \right\|^2 \\ &= \operatorname*{argmin}_{\pi \in K} \left\| \Sigma(t)'(\pi(t) - \pi_M(t)) \right\|^2 =: \mathcal{P}_K^{BS}[\pi_M](t). \end{split}$$

4.4.2 Multi-Factor Stochastic Covariance of CIR-Type

Next, we consider a financial market model with a stochastic covariance matrix, which depends on m independent CIR processes. More specifically, we assume that the covariance matrix $\Sigma(t, z)$ is a block-diagonal matrix, whose diagonal blocks Σ_i are scaled proportionally to the *i*-th CIR process z_i . One may think of the underlying financial market \mathcal{M}_{CIR} as consisting of risky assets from m unrelated asset classes, where the covariances within each asset class are driven by one of m independent stochastic (CIR) risk factors. Special cases of this model are the Heston model ([42]) for m = d = 1 and the PCSV model with independent assets ([30]) for $m = d \in \mathbb{N}$ and $d_i = 1$ for all i = 1, ..., m.⁵

For notational convenience in the following discussion, we introduce the element-wise product between any two real vectors x, y of identical dimension as $x \odot y$.

Definition 4.4.5 (\mathcal{M}_{CIR}).

Let $m, d, d_1, ..., d_m \in \mathbb{N}$ such that $m \leq d$ and $\sum_{i=1}^m d_i = d$. Consider constants $\kappa, \theta, \sigma \in (0, \infty)^d$ such that

$$2\kappa_i \theta_i > \sigma_i^2 \quad \forall i = 1, ..., m.$$

$$(4.21)$$

Moreover, let $r \in \mathbb{R}$ and $\rho_i \in (-1, 1)^{d_i}$, $\eta_i \in \mathbb{R}^{d_i}$, and non-singular $\Sigma_i \in \mathbb{R}^{d_i \times d_i}$ be given for i = 1, ..., m. Then, the d-dimensional market \mathcal{M}_{CIR} with m-factor volatility of CIR-type is defined by the market coefficients

$$\mu^{z}(t,z) = \kappa \odot (\theta - z), \ \Sigma^{z}(t,z) = \begin{pmatrix} \sigma_{1}\sqrt{z_{1}} & 0 \\ & \ddots & \\ 0 & \sigma_{m}\sqrt{z_{m}} \end{pmatrix}, \ \rho(t,z) = \begin{pmatrix} \rho_{1}' & 0 \\ & \ddots & \\ 0 & \rho_{m}' \end{pmatrix} \in \mathbb{R}^{m \times d},$$
$$r(t,z) = r, \ \mu(t,z) = r(t,z)\mathbb{1} + \begin{pmatrix} \eta_{1}z_{1} \\ \vdots \\ \eta_{m}z_{m} \end{pmatrix}, \ \Sigma(t,z) = \begin{pmatrix} \Sigma_{1}\sqrt{z_{1}} & 0 \\ & \ddots & \\ 0 & \Sigma_{m}\sqrt{z_{m}} \end{pmatrix}.$$

⁵We will later see in Lemma 4.4.6 that Condition (EAS) is only satisfied in \mathcal{M}_{CIR} if the structure of the allocation constraints allows for a convenient separability in (4.14). In the definition of \mathcal{M}_{CIR} , we have intentionally limited the covariance Σ of risky assets to be of block-diagonal structure to facilitate the presentation of this fact. However, we can in principle also choose more complex models for Σ , such as the general PCSV model, and adjust the allocation constraints accordingly without changing the underlying theory in a significant way.

In \mathcal{M}_{CIR} , the minimization (4.14) can be equivalently rewritten as⁶

$$\operatorname{argmin}_{\substack{\lambda = (\lambda_1, \dots, \lambda_m)'\\\lambda_i \in \mathbb{R}^{d_i}}} \left\{ 2(1-b)\delta_K(\lambda) + \sum_{i=1}^m \left(2\left(\Sigma_i^{-1}\lambda_i\right)' \left(\Sigma_i^{-1}\eta_i + \sigma_i B_i \rho_i\right) + \left\|\Sigma_i^{-1}\lambda_i\right\|^2 z_i \right) \right\}.$$

However, as the underlying financial market model \mathcal{M}_{CIR} consists of m independent asset classes, it is natural to assume a certain independence with respect to the allocation constraints, too. This independence can be expressed in mathematical terms by assuming that K can be written as the Cartesian product of m constraints $K_1, ..., K_m$ on the individual asset classes.

Lemma 4.4.6 (Dual ODEs in \mathcal{M}_{CIR}). Consider the financial market \mathcal{M}_{CIR} . If $K = X_{i=1}^m K_i$ with $K_i \subset \mathbb{R}^{d_i}$ closed convex and nonempty interior for every i = 1, ..., m, then Condition (EAS) is satisfied. Let

$$\lambda^*(t, z, B) := \begin{pmatrix} \lambda_1^*(B_1)z_1\\ \vdots\\ \lambda_m^*(B_m)z_m \end{pmatrix},$$

where

$$\lambda_{i}^{*}(B_{i}) = \underset{\lambda_{i} \in \mathbb{R}^{d_{i}}}{\operatorname{argmin}} \left\{ 2(1-b)\delta_{K_{i}}\left(\lambda_{i}\right) + \left\|\Sigma_{i}^{-1}\left(\eta_{i}+\lambda_{i}\right) + \sigma_{i}B_{i}\rho_{i}\right\|^{2} \right\}$$

and $A: [0,T] \to \mathbb{R}, B: [0,T] \to \mathbb{R}^m$ satisfy A(0) = 0, B(0) = 0 and

$$A_{\tau}(\tau) = br + (\kappa \odot \theta)' B(\tau)$$

$$(B_{\tau})_{i}(\tau) = -\kappa_{i}B_{i}(\tau) + \frac{1}{2}\sigma_{i}^{2}(B_{i}(\tau))^{2}$$

$$+ \frac{1}{2}\frac{b}{1-b}\inf_{\lambda_{i}\in\mathbb{R}^{d_{i}}}\left\{2(1-b)\delta_{K_{i}}(\lambda_{i}) + \left\|\Sigma_{i}^{-1}(\eta_{i}+\lambda_{i}) + \sigma_{i}B_{i}\rho_{i}\right\|^{2}\right\}.$$

Then,

$$G(t, v, z) = \frac{1}{b}v^b \exp(A(T-t) + B(T-t)'z)$$

is a solution to the dual HJBI PDE (4.7) and the corresponding candidate optimal portfolio is

$$\underline{\pi}^{*}(t, v, z) = \begin{pmatrix} \pi_{1}^{*}(B_{1}(T-t)) \\ \vdots \\ \pi_{m}^{*}(B_{m}(T-t)) \end{pmatrix},$$

with

$$\pi_i^*(B_i(T-t)) = \frac{1}{1-b} \left(\sum_i \sum_i' \right)^{-1} \left(\eta_i + \lambda_i^*(B_i(T-t)) + \sigma_i B_i(T-t) \sum_i \rho_i \right).$$

The ODEs for B_i in Lemma 4.4.6 do not admit a general closed-form solution, as the right-hand side of the ODE still depends on a convex minimization problem. Note however, as long as all minimizer $\lambda_i^*(B_i)$ are continuous in B_i (e.g. if each K_i are compact sets), then the right-hand side of each ODE for B_i is continuous in B_i and therefore admits at least a local solution.

Due to this lack of an explicit representation for B, we also lack an explicit representation for π^* . In addition, the quadratic variations of $\ln (V^{v_0,\pi^*})$ and z are stochastic and we can thus no longer follow the approach from Corollary 4.4.3 to formally verify the optimality of π^* . This topic will be investigated thoroughly in Chapter 5.

⁶Compare to the derivation of (A.53) in the proof of the subsequent Lemma 4.4.6 for details.

4.4.3 Multi-Factor Short Rate of OU-Type

Lastly, we consider a financial market \mathcal{M}_{OU} with a stochastic short rate r, which is driven by an *m*-dimensional Ornstein-Uhlenbeck process and d = m zero-coupon bonds with maturities $T_1, ..., T_m > T$ as primary traded assets (similar models were discussed e.g. in Section 7.3 in [59] for derivatives pricing, in [87] for economic scenario generation as well as in [61] and [80] in a portfolio optimization context). For this purpose, we define μ^z and Σ^z as

$$\mu^{z}(t,z) = \kappa \odot \left[\theta - z\right], \quad \Sigma^{z}(t,z) = \sigma, \tag{4.22}$$

for arbitrary constants $\kappa \in (0, \infty)^m$, $\theta \in \mathbb{R}^m$, and a non-singular matrix $\sigma \in \mathbb{R}^{m \times m}$. For two weights $w_0 \in \mathbb{R}$, $w_1 \in \mathbb{R}^m$ we then define the short rate r through

$$r(t,z) = w_0 + w_1'z.$$

In particular, the Q-dynamics of the short rate are given as

$$dr(t, z(t)) = d(w_0 + w'_1 z(t)) = w'_1 dz(t) = w'_1 (\kappa \odot [\theta - z(t)]) dt + w'_1 \sigma dW^z(t)$$

To formally define \mathcal{M}_{OU} , we still need to explicitly determine the dynamics of the traded zerocoupon bonds. We determine these dynamics via risk-neutral pricing. Assuming a constant market price of risk $\eta \in \mathbb{R}^m$, we can define the equivalent martingale measure \tilde{Q} through its Radon-Nikodym derivative

$$\frac{d\tilde{Q}}{dQ} = \exp\left(-\frac{1}{2}\|\eta\|^2 T - \int_0^T \eta' dW^z(t)\right).$$

Then, according to Girsanov's theorem, there exists a \tilde{Q} -Wiener process \tilde{W}^z such that the \tilde{Q} -dynamics of the short rate are given as

$$dr(t, z(t)) = \left(\kappa \odot \left[\theta - z(t)\right] - w_1' \sigma \eta\right) dt + w_1' \sigma d\tilde{W}^z(t).$$

Moreover, we can now make use of risk-neutral pricing to determine the arbitrage-free prices of zero-coupon bonds with different maturities. The financial market \mathcal{M}_{OU} belongs to the group of affine factor models (without stochastic volatility) studied in [26]. Hence, there exist suitable deterministic continuously differentiable functions $a: (0, \infty) \to \mathbb{R}, b: (0, \infty) \to \mathbb{R}^m$ such that the price of a zero-coupon bond with maturity T_i at time $t \in [0, T]$ is given by

$$P(t,T_i) = \mathbb{E}_{\tilde{Q}}\left[\exp\left(-\int_t^{T_i} r(s,z(s))ds\right) \left|\mathcal{F}_t\right] = \exp\left(a(T_i-t) + b(T_i-t)'z(t)\right).$$
(4.23)

By applying Itô's formula and noting that the discounted price processes $(P(t,T_i)/P_0(t))_{t\in[0,T]}$ are martingales with respect to \tilde{Q} , we see that the Q-dynamics of $(P(t,T_i))_{t\in[0,T]}$ are

$$dP(t,T_i) = P(t,T_i) \left(\left[r(t,z(t)) + b(T_i-t)'\sigma\eta \right] dt + b(T_i-t)'\sigma dW^z(t) \right).$$

These zero-coupon bonds $P(t, T_1), ..., P(t, T_m)$ constitute the primary traded assets of the financial market \mathcal{M}_{OU} which is formally defined below.

Definition 4.4.7 (\mathcal{M}_{OU}).

Let $m = d \in \mathbb{N}$. Consider constants $w_0 \in \mathbb{R}$, $\kappa \in (0, \infty)^m$, $w_1, \theta, \eta \in \mathbb{R}^m$, a non-singular matrix

 $\sigma \in \mathbb{R}^{m \times m}$, maturities $\hat{T} = (T_1, ..., T_m)' \in (T, \infty)^m$ and a continuously differentiable function $b: (0, \infty) \to \mathbb{R}^m$ such that the matrix

$$b(t; \hat{T}) = (b(T_1 - t), ..., b(T_m - t)) \in \mathbb{R}^{m \times m}$$

has an inverse $b(t;\hat{T})^{-1}$ for all $t \in [0,T]$. Then, the m-dimensional Bond market \mathcal{M}_{OU} with OU short rate is defined by the market coefficients

$$\mu^{z}(t,z) = \kappa \odot (\theta - z), \quad \Sigma^{z}(t,z) = \sigma, \quad \rho(t,z) = I_{m}$$

$$r(t,z) = w_{0} + w_{1}'z, \quad \mu(t,z) = r(t,z)\mathbb{1} + b(t;\hat{T})'\sigma\eta, \quad \Sigma(t,z) = b(t;\hat{T})'\sigma.$$

Despite the stochastic short rate, the market \mathcal{M}_{OU} is surprisingly tractable. Specifically, none of the terms involved in the minimization (4.14) are dependent on the stochastic factor z, which results in a time-dependent but deterministic candidate optimal portfolio π^* .

Lemma 4.4.8 (Dual ODEs in \mathcal{M}_{OU}). Consider the financial market \mathcal{M}_{OU} . Then Condition (EAS) is satisfied. Let

$$\lambda^*(t,B) = \underset{\lambda \in \mathbb{R}^d}{\operatorname{argmin}} \left\{ 2(1-b)\delta_K(\lambda) + \left\| \eta + \left(b(t;\hat{T})'\sigma \right)^{-1}\lambda + \sigma'B \right\|^2 \right\}$$

and $A: [0,T] \to \mathbb{R}, B: [0,T] \to \mathbb{R}^m$ satisfy A(0) = 0, B(0) = 0 and

$$A_{\tau}(\tau) = bw_{0} + (\kappa \odot \theta)' B(\tau) + \frac{1}{2} \|\sigma' B(\tau)\|^{2} + \frac{1}{2} \frac{b}{1-b} \inf_{\lambda \in \mathbb{R}^{d}} \left\{ 2(1-b)\delta_{K}(\lambda) + \left\| \eta + \left(b(T-\tau;\hat{T})'\sigma \right)^{-1} \lambda + \sigma' B \right\|^{2} \right\} \\ B_{\tau}(\tau) = bw_{1} - \kappa \odot B(\tau).$$
(4.24)

Then,

$$G(t, v, z) = \frac{1}{b}v^{b} \exp(A(T - t) + B(T - t)'z)$$

is a solution to the dual HJBI PDE (4.7) and the corresponding candidate optimal portfolio $\underline{\pi}^*$ (as in (4.17)) is given as

$$\underline{\pi}^{*}(t,v,z) = \frac{1}{1-b} \left(\sigma' b(t;\hat{T}) \right)^{-1} \left(\eta + \left(b(t;\hat{T})' \sigma \right)^{-1} \lambda^{*}(t,B(T-t)) + \sigma' B(T-t) \right).$$

Remark 4.4.9. The solution to the ODE (4.24) is known in closed-form (see e.g. equations (6) and (7) in Chapter 1, Section §2 of [83]) and is given by $B(\tau) = (B_1(\tau), ..., B_m(\tau))'$ with

$$B_{i}(\tau) = (w_{1})_{i} b e^{-\kappa_{i}\tau} \int_{0}^{\tau} e^{\kappa_{i}s} ds = \frac{(w_{1})_{i} b}{\kappa_{i}} \left(1 - e^{-\kappa_{i}\tau}\right).$$

Unlike in \mathcal{M}_{CIR} , the quadratic variations of $\ln (V^{v_0, \pi^*})$ and z are deterministic and bounded in \mathcal{M}_{OU} . Therefore, it is straightforward to adapt the proof of Corollary 4.4.3 to formally verify the optimality of π^* for (**P**).

Corollary 4.4.10. Consider the financial market \mathcal{M}_{OU} . Then, $G, \underline{\pi}^*$ as in Lemma 4.4.8 and $\lambda \equiv 0$ satisfy Condition (UI_{λ}) . In particular, $\underline{\pi}^*$ is optimal for (**P**).

Remark 4.4.11. Other than in \mathcal{M}_{CIR} , the ODE for B in \mathcal{M}_{OU} does not depend on the constraint K (cf. Lemma 4.4.6 and Lemma 4.4.8). Therefore, setting $K = \mathbb{R}^d$ and $\lambda^* \equiv 0$ yields the optimal unconstrained portfolio in \mathcal{M}_{OU} as

$$\pi_{OU}(t) = \frac{1}{1-b} \left(\sigma' b(t; \hat{T}) \right)^{-1} \left(\eta + \sigma' B(T-t) \right).$$

Similarly, we can replicate the arguments in Remarks 4.3.8 and 4.4.4 to see that the optimal constrained portfolio in \mathcal{M}_{OU} can be written as

$$\underline{\pi}^{*}(t,v,z) = \underset{\pi \in K}{\operatorname{argmin}} \left\| \underbrace{\sigma'b(t;\hat{T})}_{=:\Sigma(t)'}(\pi(t) - \pi_{OU}(t)) \right\|^{2} = \underset{\pi \in K}{\operatorname{argmin}} \left\| \Sigma(t)'(\pi(t) - \pi_{OU}(t)) \right\|^{2} = \mathcal{P}_{K}^{BS}[\pi_{OU}](t).$$

In other words, allocation constraints have exactly the same impact on the optimal allocation in \mathcal{M}_{OU} as in \mathcal{M}_{BS} .

4.5 Conclusion

In this chapter, we examined a portfolio optimization problem in a financial market where asset dynamics depend on a stochastic factor. In the spirit of [61], we were able to derive Condition (EAS) which guarantees that the solution to the HJB PDE for an allocation constrained portfolio optimization problem is exponentially affine and separable in wealth and the stochastic factor. We were able to use Condition (EAS) to characterize the optimal allocation constrained portfolio up to the solution of a deterministic optimization problem and the solution of Riccati ODEs in a market with stochastic volatility of CIR-type and in a market with stochastic short rate of OU-type. Special examples of these models include the Heston model, the PCSV model and the Vasicek model. We derived a formal verification result for the market with stochastic short rate and a general verification result up to a uniform integrability condition for arbitrary markets satisfying Condition (EAS). The proposed methodology is general enough to derive and study the optimal allocation constrained portfolios for several financial markets with dynamics depending on a stochastic factor.
5 Constraints on Allocation in Heston's Stochastic Volatility Model

5.1 Introduction

with respect to stochastic volatility:

Despite its widespread use in both academia and the financial industry, it has been welldocumented in the mathematical finance literature that a variety of properties of financial time series, so-called stylized facts, are not captured by the Black-Scholes model (see e.g. [14], [63], [84]). A major point of criticism is the constant volatility of modelled log returns, whereas empirical returns appear to be time-dependent and random ([81]). The stochastic volatility model proposed by [42] aims to avoid this gap by modelling the volatility of log returns as a Cox-Ingersoll-Ross process (CIR process). While the Heston model was originally proposed in the context of option pricing, its analytical tractability has led to insightful applications in continuous-time portfolio optimization (see e.g. [56]), which is the subject of this chapter. Specifically, we consider the portfolio optimization problem of an investor trading continuously in a financial market consisting of one risk-free asset and one risky asset with stochastic Heston volatility. The investor seeks to maximise his expected utility derived from terminal wealth at a finite time point T > 0 under the condition that his portfolio allocation π abides by given convex allocation constraints K. The considered optimization problem involves two major facets, which differ from the original portfolio optimization setting of [65]: stochasticity of the volatility of risky asset log returns ('stochastic volatility') and the presence of convex allocation constraints. Since Section 1.1, already provided an overview over the relevant literature on al-

location constraints, the following paragraph presents a brief overview of the relevant literature

Our continuous time set-up can be traced back to the seminal work of [65], who used dynamic programming methods to derive explicit solutions to the unconstrained dynamic portfolio optimization problem for an investor with HARA utility function in a Black-Scholes model with constant volatility. Generalizations for different utility functions and more complex financial markets were achieved by employing martingale methods ([73] and [47]), which rely heavily on the assumption that all contingent claims in the financial market are replicable. However, this assumption is not generally satisfied for financial markets with stochastic volatility, which means that martingale methods are not directly applicable, unless the financial market is artificially completed by the addition of fictitious, volatility-dependent assets (see e.g. [62], [9], [28], [30], and [10]). Without such completion, the solvability of the portfolio optimization problem in financial markets with stochastic volatility is often directly linked to the solvability of the associated HJB PDE.¹ Solutions to such portfolio optimization problems were first characterized in terms of viscosity solutions to the associated HJB PDE for multi-factor models in [88], and explicit closed-form solutions for Heston's stochastic volatility model were first derived and formally verified in [56]. Subsequently, [61] derived explicit solution formulae for an optimal consumption and portfolio allocation problem in a general multi-factor model, where the factor is a quadratic diffusion. [46] used the notion of an opportunity process and semi-martingale

¹Note that obtaining and formally verifying the optimality of a candidate portfolio process requires more than just a solution to the associated HJB PDE, as pointed out by [54].

characteristics to develop an approach which leads to closed-form solutions for the optimal portfolio process in a range of exponentially affine stochastic volatility models, including the Heston model and the jump model of [40]. The PCSV model and the Wishart process were considered as multi-dimensional extensions of the Heston model for multidimensional asset universes with stochastic correlation in [30] and [5]. An extensive overview of related papers in the field of dynamic portfolio optimization in stochastic factor models can be found in [90].

In this chapter, we make a threefold contribution to this literature:

- We complete a gap in Chapter 4 (resp. [33]) by deriving the first explicit, closed-form expression for the optimal portfolio allocation π^* in Heston's stochastic volatility model under the presence of convex allocation constraints. This optimality is verified formally in a verification theorem.
- We show that the optimal portfolio allocation π^* may be different from the capped optimal unconstrained portfolio allocation π_u for Heston's stochastic volatility model. In particular, we prove an equivalent characterization which describes when these two portfolios are different. We conduct a numerical study to show that this difference is slim for calm market scenarios, but can lead to significant annual wealth equivalent losses during turbulent market scenarios.
- We extend these results from the one-dimensional Heston's stochastic volatility model to a multi-dimensional PCSV model with constraints on the exposures to individual stochastic market factors and to generalized financial markets with inverse volatility constraints.

The remainder of this chapter is structured as follows: In Section 5.2, we introduce and solve the constrained portfolio optimization problem (**P**) in Heston's stochastic volatility model. Specifically, we derive a solution to the HJB PDE associated with (**P**) and the candidate optimal portfolio π^* in Section 5.2.1, verify its optimality formally by proving a verification theorem in Section 5.2.2 and discuss its relation to the optimal unconstrained portfolio π_u in Section 5.2.3. In Section 5.3, we consider a generalized financial market model which depends on a multi-dimensional CIR process and derive the optimal portfolio in the PCSV model under exposure constraints (Section 5.3.1) and in general financial markets with inverse volatility constraints (Section 5.3.2). In Section 5.4, we illustrate our theoretical results for Heston's stochastic volatility model in a numerical analysis, where we analyze the wealth equivalent loss of the optimal constrained portfolio for a Black Scholes model (Section 5.4.1) and the capped optimal unconstrained portfolio for Heston's stochastic volatility model (Section 5.4.2). Section 5.5 concludes this chapter.

5.2 Heston's Stochastic Volatility Model

In this chapter, we consider several special cases of the stochastic factor model from Chapter 5. In particular, the standing assumptions for this chapter are the same as for Chapter 5:

Assumption 5.2.1 (Standing Assumptions Chapter 5:). \mathcal{M} is a stochastic factor model ((4.2) and (4.3)), (4.4) and U is a power utility function.

The first such special case is Heston stochastic volatility model \mathcal{M}_H , which is defined by considering the stochastic factor model (4.3) and choosing m = d = 1 as well as the deterministic functions

$$\mu^{z}(t,z) = \kappa(\theta - z), \ \Sigma^{z}(t,z) = \sigma\sqrt{z}, \ \rho(t,z) = \rho,$$

$$r(t,z) = r, \ \mu(t,z) = r(t,z) + \eta z, \ \Sigma(t,z) = \sqrt{z}$$

where the coefficients $r, \eta, \kappa, \theta, \sigma, z_0$ are assumed to be positive constants, and $\rho \in (-1, 1)$. Further, it is assumed that Feller's condition holds for the parameters of z, i.e., $2\kappa\theta > \sigma^2$, and, therefore, z(t) is guaranteed to take only positive values with probability 1 (see [41]). Clearly, the financial market \mathcal{M}_H is a special, one-dimensional case of \mathcal{M}_{CIR} from Definition 4.4.5.

The resulting financial market \mathcal{M}_H consists of one risk-free asset P_0 and one risky asset P_1 , which satisfy $P_0(0) = P_1(0) = 1$ and follow the dynamics

$$dP_0(t) = P_0(t)rdt,$$

$$dP_1(t) = P_1(t)(r + \eta \cdot z(t))dt + \sqrt{z(t)}dW(t),$$

where $z(0) = z_0 > 0$ and

$$dz(t) = \kappa(\theta - z(t))dt + \sigma\sqrt{z(t)}dW^{z}(t).$$

As a consequence, the wealth process $V^{v_0,\pi}$ of an investor trading in \mathcal{M}_H according to a relative portfolio process $\pi \in \Lambda$ and initial wealth $v_0 > 0$ satisfies the SDE

$$dV^{v_0,\pi}(t) = V^{v_0,\pi}(t) \Big([r + \eta z(t)\pi(t)] dt + \pi(t)\sqrt{z(t)} dW(t) \Big)$$

and it is straightforward to show that $V^{v_0,\pi}(t)$ can be expressed in closed-form as

$$V^{v_0,\pi}(t) = v_0 \exp\Big(\int_0^t r + \eta z(s)\pi(s) - \frac{1}{2}z(s)\pi(s)^2 ds + \int_0^t \pi(s)\sqrt{z(s)}dW(s)\Big).$$
(5.1)

For a non-empty closed convex set $K \subset \mathbb{R}$ and power utility function $U(v) = \frac{1}{b}v^b$ with b < 1and $b \neq 0$, we consider the portfolio optimization problem

$$(\mathbf{P}) \begin{cases} \Phi(v_0) &= \sup_{\pi \in \Lambda'} \mathbb{E} \left[U(V^{v_0,\pi}(T)) \right] \\ \Lambda' &= \left\{ \pi \in \Lambda \mid \pi(t) \in K \ \mathcal{L}[0,T] \otimes Q - \text{a.e.} \right\} \end{cases}$$

As the considered financial market contains only one risky asset, the set of allocation constraints K is a subset of the real numbers \mathbb{R} . However, in this one-dimensional setting, any such closed convex set $K \subset \overline{\mathbb{R}}$ with non-empty interior can be expressed as an interval of the form²

$$K = [\alpha, \beta], \quad \text{with} \quad -\infty \le \alpha < \beta \le \infty.$$
(5.2)

We make substantial use of this fact in the subsequent analysis.

5.2.1 Solution to HJB PDE

Following the arguments from Section 4.3, the HJB equation associated with $(\mathbf{P}^{(t,v,z)})$ is given by

$$0 = \sup_{\pi \in K} \left(G_t + v(r + \eta z \pi) G_v + \kappa (\theta - z) G_z + \frac{1}{2} v^2 z \pi^2 G_{vv} + \rho v z \pi \sigma G_{vz} + \frac{1}{2} \sigma^2 z G_{zz} \right)$$
(5.3)

G(T, v, z) = U(v),

²As any $\pi \in \Lambda$ can only take finite values $\mathcal{L}[0,T] \otimes Q$ -a.e., we do not need to distinguish between $(-\infty,\beta]$ and $[-\infty,\beta]$ or $[\alpha,\infty)$ and $[\alpha,\infty]$ for any $-\infty \leq \alpha,\beta \leq \infty$.

Any (sufficiently regular) solution G to (5.3) naturally yields a candidate optimal portfolio π^* to $(\mathbf{P^{(t,v,z)}})$ (and therefore (\mathbf{P})) as the maximizing argument of (5.3). In Chapter 4 we characterized G as an exponentially affine function, whose exponents satisfy certain Riccati ODEs.

Lemma 5.2.2. Let A and B be solutions to the system of ODEs

$$A'(\tau) = br + \kappa \theta B(\tau), \tag{5.4}$$

$$B'(\tau) = -\kappa B(\tau) + \frac{1}{2}\sigma^2 (B(\tau))^2 + \frac{1}{2}\frac{b}{1-b}\inf_{\lambda \in \mathbb{R}} \left(2(1-b)\delta_K(\lambda) + (\eta + \lambda + \sigma\rho B(\tau))^2 \right), \quad (5.5)$$

with initial condition A(0) = B(0) = 0. Then, $G(t, v, z) = \frac{1}{b}v^b \exp(A(T-t) + B(T-t)z)$ is a solution to (5.3).

Given the minimizing argument and the solution B from (5.5), a candidate optimal portfolio π^* for (**P**) is known. If we define the sequence of stopping times $\tau_{n,t}$ as $\tau_{n,t} = \min(T, \hat{\tau}_{n,t})$, with

$$\begin{aligned} \hat{\tau}_{n,t} &= \inf \left\{ t \le u \le T \ \Big| \ \int_t^u \left(b \cdot \sqrt{z(s)} \cdot \pi(s) \cdot G(s, V^{v_0, \pi^*}(s), z(s)) \right)^2 ds \ge n, \\ &\int_t^u \left(\sigma \sqrt{z(s)} \cdot B(T-s) \cdot G(s, V^{v_0, \pi^*}(s), z(s)) \right)^2 ds \ge n \right\}, \end{aligned}$$

then we can give a uniform integrability condition which guarantees that the candidate optimal portfolio π^* is indeed optimal for (**P**).

Lemma 5.2.3. Consider A, B and G from Lemma 5.2.2. Moreover, define

$$\lambda^*(B) = \underset{\lambda \in \mathbb{R}}{\operatorname{argmin}} \left\{ 2(1-b)\delta_K(\lambda) + (\eta + \lambda + \sigma\rho B)^2 \right\}$$
(5.6)

$$\pi^*(t) = \frac{1}{1-b} \Big(\eta + \lambda^* (B(T-t)) + \sigma \rho B(T-t) \Big).$$
(5.7)

If $(G(\tau_{n,t}, V^{v_0,\pi^*}(\tau_{n,t}), z(\tau_{n,t})))_{n\in\mathbb{N}}$ is uniformly integrable for every $t \in [0,T]$, then π^* is optimal for (**P**) and $\Phi(t,z,v) = G(t,v,z)$ for all $(t,v,z) \in [0,T] \times (0,\infty) \times (0,\infty)$.

Lemma 5.2.2 and Lemma 5.2.3 naturally lead to a three-step procedure for finding the optimal portfolio π^* for (**P**):

- (i) Determine the minimizing argument λ^* in (5.6).
- (ii) Determine the solution B to ODE (5.5) and thereby the candidate optimal portfolio π^* from (5.7).
- (iii) Verify that π^* satisfies the uniform integrability condition from Lemma 5.2.3.

In the following, we complete these steps consecutively and complete the results from Section 4.4 by providing a fully closed-form solution for the optimal allocation-constrained portfolio in Heston's stochastic volatility model via steps (i) and (ii) and formally verifying its optimality in step (iii).

Since K is an interval, as specified in (5.2), the ODE (5.5) can be written as a composition of three Riccati ODEs - each with constant coefficients.

Lemma 5.2.4. Define $B_{-} = \frac{(1-b)\alpha - \eta}{\sigma}$ and $B_{+} = \frac{(1-b)\beta - \eta}{\sigma}$. Then, the minimizing argument λ^* , as in (5.6), is given as

$$\lambda^{*}(B) = [(1-b)\alpha - (\eta + \sigma\rho B)] \mathbb{1}_{\{\rho B < B_{-}\}} + [(1-b)\beta - (\eta + \sigma\rho B)] \mathbb{1}_{\{\rho B > B_{+}\}}.$$
 (5.8)

Moreover, $B(\tau)$ is a solution to (5.5) if and only if B(0) = 0 and

$$B'(\tau) = \left(-\underbrace{\frac{1}{2}b\alpha((1-b)\alpha-2\eta)}_{=:r_{0}^{-}} + \underbrace{(b\sigma\rho\alpha-\kappa)}_{=:r_{1}^{-}}B(\tau) + \frac{1}{2}\underbrace{\sigma^{2}}_{=:r_{2}^{-}}(B(\tau))^{2}\right)\mathbb{1}_{\{\rho B(\tau) < B_{-}\}} \\ + \left(-\underbrace{\frac{-b}{2(1-b)}\eta^{2}}_{=:r_{0}} + \underbrace{(\frac{b}{1-b}\eta\sigma\rho-\kappa)}_{=:r_{1}}B(\tau) + \frac{1}{2}\underbrace{\sigma^{2}(1+\frac{b}{1-b}\rho^{2})}_{=:r_{2}}(B(\tau))^{2}\right)\mathbb{1}_{\{B_{-} \le \rho B(\tau) \le B_{+}\}} \\ + \left(-\underbrace{\frac{1}{2}b\beta((1-b)\beta-2\eta)}_{=:r_{0}^{+}} + \underbrace{(b\sigma\rho\beta-\kappa)}_{=:r_{1}^{+}}B(\tau) + \frac{1}{2}\underbrace{\sigma^{2}}_{=:r_{2}^{+}}(B(\tau))^{2}\right)\mathbb{1}_{\{B_{+} < \rho B(\tau)\}}$$
(5.9)
$$= \left(-r_{0}^{-}+r_{1}^{-}B(\tau) + \frac{1}{2}r_{2}^{-} + (B(\tau))^{2}\right)\mathbb{1}_{\{\rho B(\tau) < B_{-}\}} \\ + \left(-r_{0}^{+}+r_{1}B(\tau) + \frac{1}{2}r_{2}(B(\tau))^{2}\right)\mathbb{1}_{\{B_{-} \le \rho B(\tau) \le B_{+}\}} \\ + \left(-r_{0}^{+}+r_{1}^{+}B(\tau) + \frac{1}{2}r_{2}^{+}(B(\tau))^{2}\right)\mathbb{1}_{\{B_{+} < \rho B(\tau)\}}.$$

Remark 5.2.5. By restricting the minimization in ODE (5.5) from $\lambda \in \mathbb{R}$ to one of the three optimal values $\lambda \in \{(1-b)\alpha - (\eta + \sigma\rho B), (1-b)\beta - (\eta + \sigma\rho B), 0\}$ (cf. (5.8)), we may use (5.9) to write

$$B'(\tau) = -\kappa B(\tau) + \frac{1}{2}\sigma^2 (B(\tau))^2 + \frac{1}{2}\frac{b}{1-b}\inf_{\lambda\in\mathbb{R}} \left(2(1-b)\delta_K(\lambda) + (\eta+\lambda+\sigma\rho B(\tau))^2\right)$$

= min $\left(-r_0^- + r_1^- B(\tau) + \frac{1}{2}r_2^- (B(\tau))^2, -r_0 + r_1 B(\tau) + \frac{1}{2}r_2 (B(\tau))^2, -r_0^+ + r_1^+ B(\tau) + \frac{1}{2}r_2^+ (B(\tau))^2\right)$
=: $f(B(\tau)).$

The coefficients r_2^- , r_2 and r_2^+ are non-negative, and therefore f is the minimum of three convex functions. As real convex functions are locally Lipschitz continuous and Lipschitz continuity is preserved when taking the minimum over a finite number of functions, f is locally Lipschitz continuous too. Hence, by the existence and uniqueness theorem of Picard-Lindelöf, there exists a unique solution B to (5.5) for small $\tau > 0$. Moreover, as f does not depend on τ , the ODE for B is autonomous and its solution B is either constant (if f(0) = 0) or strictly monotone in τ (if $f(0) \neq 0$). Analogous arguments can be used to conclude the (strict) monotonicity of $B_u(\tau)$ from Corollary 5.2.13 in Section 5.2.3.

Remark 5.2.6. Note that if B is a solution to (5.5), then Lemma 5.2.3 and Lemma 5.2.4 imply

$$\pi^{*}(t) = \frac{\eta + \lambda^{*}(B(T-t)) + \sigma\rho B(T-t)}{1-b} = \begin{cases} \alpha, & \rho B(T-t) < B_{-} \\ \frac{\eta + \sigma\rho B(T-t)}{1-b}, & B_{-} \le \rho B(T-t) \le B_{+} \\ \beta, & B_{+} < \rho B(T-t). \end{cases}$$

Therefore, the zones $Z_{-} = (-\infty, B_{-})$, $Z_{0} = [B_{-}, B_{+}]$ and $Z_{+} = (B_{+}, \infty)$ determine whether the allocation constraint $K = [\alpha, \beta]$ is enforced for the candidate optimal portfolio process π^{*} from Lemma 5.2.3. Moreover, we may define $\hat{\pi}^*(t) := \frac{1}{1-b} \left(\eta + \sigma \rho B(T-t)\right)$ and express π^* as a capped version of $\hat{\pi}^*$, i.e.,

$$\pi^*(t) = Cap(\hat{\pi}^*(t), \alpha, \beta) := \begin{cases} \alpha, & \hat{\pi}^*(t) < \alpha \\ \hat{\pi}^*(t), & \alpha \le \hat{\pi}^*(t) \le \beta \\ \beta, & \beta < \hat{\pi}^*(t). \end{cases}$$

In a true constrained context, i.e., $K \neq \mathbb{R}$, we may either determine an approximation of B by using a suitable numerical ODE solver (such as an Euler method) to solve the ODE (5.9) or directly derive an explicit expression for B by individually solving each of the three Riccati ODEs in (5.9) and merging the solutions at the transition points between the zones Z_{-} , Z_{0} and Z_{+} . To ensure that such solutions exist and do not explode before time T, we need to make the following assumption on \mathcal{M}_{H} and the constraints $K = [\alpha, \beta]$.

Assumption 5.2.7.

(i) Existence of Solution:

$$\max \left\{ \begin{array}{l} \frac{b}{1-b}\eta\left(\frac{\kappa\rho}{\sigma}+\frac{\eta}{2}\right),\\ b\alpha\left(\eta-\frac{1}{2}\alpha+\frac{\kappa\rho}{\sigma}+\frac{1}{2}\alpha b(1-\rho^2)\right),\\ b\beta\left(\eta-\frac{1}{2}\beta+\frac{\kappa\rho}{\sigma}+\frac{1}{2}\beta b(1-\rho^2)\right) \end{array} \right\} < \frac{\kappa^2}{2\sigma^2}$$

(ii) No Blow-Up:

The coefficients of each of the three Riccati ODEs satisfy $t_+(B_0) > T$ (cf. Lemma 2.2.4, (ii)) for each initial value

$$B_0 \in \left\{ \left(\frac{B_-}{\rho}\right) \mathbb{1}_{\{\rho \neq 0\}}, \left(\frac{B_+}{\rho}\right) \mathbb{1}_{\{\rho \neq 0\}}, 0 \right\}$$

Provided that Assumption 5.2.7 holds, the coefficients

$$r_3^- = \sqrt{(r_1^-)^2 + 2r_0^- r_2^-}, \quad r_3 = \sqrt{(r_1)^2 + 2r_0 r_2}, \quad r_3^+ = \sqrt{(r_1^+)^2 + 2r_0^+ r_2^+}$$
(5.10)

are well-defined and the solutions to each of the Riccati ODEs (5.9) do not blow up before time T when started at any of the transition points between the zones Z_{-} , Z_{0} and Z_{+} .³ For this reason, we define the following auxiliary functions:

- Let \hat{B}^+ , \hat{B} , and \hat{B}^- be the solution to Riccati ODE (2.22) with initial value 0 as well as coefficients r_0^+ , r_1^+ , r_2^+ , r_0 , r_1 , r_2 and r_0^- , r_1^- , r_2^- , respectively.
- If $\rho \neq 0$, let \hat{B}^+_+ , \hat{B}^-_- be the solution to Riccati ODE (2.22) with initial value $\frac{B_+}{\rho}$, $\frac{B_-}{\rho}$ and coefficients r_0^+ , r_1^+ , r_2^+ , r_0^- , r_1^- , r_2^- , respectively.
- If $\rho \neq 0$, let \hat{B}_+ , \hat{B}_- be the solution to Riccati ODE (2.22) with initial value $\frac{B_+}{\rho}$, $\frac{B_-}{\rho}$, respectively, and coefficients r_0 , r_1 , r_2 .

³Technically, one can formulate this assumption less restrictively by expressing 'No Blow-Up' in terms of the time spent in each of the zones Z_- , Z_0 and Z_+ . However, as this would significantly complicate the presentation without adding major additional insights, it is omitted here.

Moreover, if $\rho \neq 0$, we define the transition times⁴

$$\begin{aligned} \tau_1^+ &= \inf\left\{\tau \mid \hat{B}^+(\tau) = \frac{B^+}{\rho}\right\}, \quad \tau_2^+ = \inf\left\{\tau \mid \hat{B}_+(\tau) = \frac{B^-}{\rho}\right\}, \quad \tau_1 = \inf\left\{\tau \mid \hat{B}(\tau) \in \left\{\frac{B^-}{\rho}, \frac{B^+}{\rho}\right\}\right\}\\ \tau_1^- &= \inf\left\{\tau \mid \hat{B}^-(\tau) = \frac{B^-}{\rho}\right\} \quad \text{and} \quad \tau_2^- = \inf\left\{\tau \mid \hat{B}_-(\tau) = \frac{B^+}{\rho}\right\}.\end{aligned}$$

Note that each of the above functions and transition times, if finite, admit a closed-form expression, which can be obtained via Lemma 2.2.4 and Corollary 2.2.5. Having introduced these auxiliary functions and transition times, we can finally express a closed-form solution for B in terms of these processes via a piecewise construction. Possible trajectories of B are displayed in Figures 5.2.1, 5.2.1 and 5.2.1, depending on whether $0 \in Z_-$, $0 \in Z_0$ or $0 \in Z_+$. The monotonicity of B (cf. Remark 5.2.5) ensures that $\rho B(\tau)$ can only pass through each zone at most once.

Theorem 5.2.8. Let Assumption 5.2.7 hold. Then,

$$\left(\hat{B}^{-}(\tau) \mathbb{1}_{\{\tau \leq \tau_{1}^{-}\}} + \hat{B}_{-}(\tau - \tau_{1}^{-}) \mathbb{1}_{\{\tau_{1}^{-} < \tau \leq \tau_{1}^{-} + \tau_{2}^{-}\}} + \hat{B}_{+}^{+}(\tau - (\tau_{1}^{-} + \tau_{2}^{-})) \mathbb{1}_{\{\tau_{1}^{-} + \tau_{2}^{-} < \tau\}}, \quad if \ 0 \in \mathbb{Z}.$$

$$(\tau) = \begin{cases} \hat{B}(\tau) \mathbb{1}_{\{\tau \le \tau_1\}} + \hat{B}_{-}^{-}(\tau - \tau_1) \mathbb{1}_{\{\tau > \tau_1, \rho \hat{B}(\tau_1) = B_{-}\}} + \hat{B}_{+}^{+}(\tau - \tau_1) \mathbb{1}_{\{\tau > \tau_1, \rho \hat{B}(\tau_1) = B_{+}\}}, \quad if \ 0 \in Z_0 \end{cases}$$

$$\left(B^+(\tau) \mathbb{1}_{\{\tau \le \tau_1^+\}} + B_+(\tau - \tau_1^+) \mathbb{1}_{\{\tau_1^+ < \tau \le \tau_1^+ + \tau_2^+\}} + B^-(\tau - (\tau_1^+ + \tau_2^+)) \mathbb{1}_{\{\tau_1^+ + \tau_2^+ < \tau\}}, \quad \text{if } 0 \in \mathbb{Z}_+ \right)$$

satisfies ODE (5.5) for $0 \le \tau \le T$.⁵

B



Figure 5.1: Illustration of possible trajectories of $\rho B(\tau)$, if B is a solution to ODE (5.5) and $0 \in \mathbb{Z}_-$. The solid line represents an increasing trajectory that leaves \mathbb{Z}_- , whereas the dashed line represents a decreasing trajectory that stays in \mathbb{Z}_- .

⁴If $\rho = 0$ all of these transition times will be infinite.

⁵Using a similar separation with respect to the zones Z_- , Z_0 , and Z_+ and equation (2.24), it is also possible to determine a closed-form expression for A from Lemma 5.2.2.



Figure 5.2: Illustration of possible trajectories of $\rho B(\tau)$, if B is a solution to ODE (5.5) and $0 \in Z_0$. Depending on the sign of $\rho B'(0) = -\rho r_0$, $\rho B(\tau)$ is increasing or decreasing. In particular, $\rho B(\tau)$ can only transition to Z_+ or Z_- , but not both.



Figure 5.3: Illustration of possible trajectories of $\rho B(\tau)$, if B is a solution to ODE (5.5) and $0 \in Z_+$. The solid line represents a decreasing trajectory that leaves Z_+ , whereas the dashed line represents an increasing trajectory that stays in Z_+ .

5.2.2 Verification Theorem

Combining Remark 5.2.6 with Theorem 5.2.8 immediately yields a closed-form expression for the candidate optimal portfolio process π^* . It now just remains to prove a verification theorem which verifies that this candidate is indeed the optimal portfolio process corresponding to (**P**). This proof requires an additional assumption on the constraints $K = [\alpha, \beta]$, which ensures a certain boundedness of $\pi^*(t)$ for t close to maturity T as well as two auxiliary lemmas.

Assumption 5.2.9.

$$\max\left\{\frac{b\rho}{\kappa}\alpha, \ \frac{b\rho}{\kappa}\beta\right\} \le \frac{\kappa}{\sigma^2},\tag{5.11}$$

Lemma 5.2.10. Let Assumptions 5.2.7 and 5.2.9 hold and let B be given as in Theorem 5.2.8. Then, the following inequality holds for all $t \in [0, T]$

$$\frac{b\rho}{\sigma}\pi^*(t) + B(T-t) \le \frac{\kappa}{\sigma^2}.$$

Lemma 5.2.11. Let Assumptions 5.2.7 and 5.2.9 hold and let B be given as in Theorem 5.2.8. Then the following inequality holds for all $t \in [0, T]$

$$\begin{aligned} \frac{1}{2} \frac{b}{1-b} \eta^2 &- \frac{1}{2} \frac{b}{1-b} \left(\lambda^* \left(B(T-t) \right) + \sigma \rho B(T-t) \right)^2 - \frac{1}{2} b^2 \rho^2 \left(\pi^*(s) \right)^2 \\ &+ b \frac{\rho \kappa}{\sigma} \pi^*(t) + \frac{b}{1-b} \frac{\rho}{\sigma} \left[\left(\lambda^* \right)' \left(B(T-t) \right) + \sigma \rho \right] B'(T-t) \\ &< \frac{1}{2} \frac{\kappa^2}{\sigma^2} \left(\lambda^* \right)^2 \left(B(T-t) \right) + \sigma \rho \right] B'(T-t) \end{aligned}$$

Theorem 5.2.12 (Verification Theorem in \mathcal{M}_H).

Consider the financial market \mathcal{M}_H , let Assumptions 5.2.7 and 5.2.9 hold and let B be given as in Theorem 5.2.8. Then,

$$\pi^{*}(t) = \begin{cases} \alpha, & \rho B(T-t) < B_{-} \\ \frac{\eta + \sigma \rho B(T-t)}{1-b}, & B_{-} \le \rho B(T-t) \le B_{+} \\ \beta, & B_{+} < \rho B(T-t). \end{cases}$$
(5.12)

is optimal for (\mathbf{P}) .

5.2.3 Comparison to Unconstrained Portfolio

Unsurprisingly, we can immediately recover the solution to the unconstrained optimization problem, as discussed in [56], from Lemma 5.2.3 and Lemma 5.2.4.

Corollary 5.2.13. [Closed-form Unconstrained Optimal Portfolio as in [56]] Let $K = \mathbb{R}$ (i.e., $\alpha = -\infty$, $\beta = \infty$) and $B_u : [0,T] \to \mathbb{R}$ with $B_u(0) = 0$ satisfy

$$B'_{u}(\tau) = -r_{0} + r_{1}B_{u}(\tau) + \frac{1}{2}r_{2}B_{u}(\tau)^{2} \qquad \forall \tau \in [0, T].$$
(5.13)

Then, $\lambda^*(B) = 0 \ \forall B \in \mathbb{R}$ and the candidate optimal portfolio π^* is given by

$$\pi_u(t) := \pi^*(t) = \frac{1}{1-b} \left(\eta + \sigma \rho B_u(T-t) \right).$$

Remark 5.2.14. If the market parameters satisfy (cf. Assumption 5.2.7)

$$\frac{b}{1-b}\eta\left(\frac{\kappa\rho}{\sigma}+\frac{\eta}{2}\right) < \frac{\kappa^2}{2\sigma^2},\tag{5.14}$$

then

$$B_u(\tau) = \frac{2r_0(e^{r_3\tau} - 1)}{(r_1 - r_3)(e^{r_3\tau} - 1) - 2r_3}$$
(5.15)

and the optimality of π_u for the unconstrained portfolio optimization problem can be verified formally (see e.g. Theorem 5.3 in [56]).⁶

On an abstract level, when adding (allocation) constraints $K = [\alpha, \beta]$ to a portfolio optimization problem, the optimal constrained portfolio π^* for (**P**) will be given by a projection $\mathcal{P}_K : \Lambda \to \Lambda_K$ which maps the optimal unconstrained portfolio π_u onto Λ_K , i.e.,

$$\pi^* = \pi_u + (\pi^* - \pi_u) =: \mathcal{P}_K(\pi_u).$$

⁶Equation (5.14) corresponds to part (i) of Assumption 5.2.7. In the setting of [56], part (ii) of Assumption 5.2.7 is also implied by (5.14) and so does not have to be mentioned explicitly.

In a Black-Scholes financial market \mathcal{M}_{BS} with constant market coefficients (i.e., \mathcal{M}_H with $\sigma = \kappa = \rho = 0$.), the optimal unconstrained portfolio is a constant-mix strategy $\pi_u(t) = \pi_M := \frac{1}{1-b}\eta$, the so-called 'Merton portfolio'. Setting $\sigma = \rho = 0$ and $B \equiv 0$ in Remark 5.2.6, one can easily see that the projection $\mathcal{P}_K = \mathcal{P}_K^{BS}$ in the Black-Scholes market simply caps off π_M at the boundaries if $\pi_M \notin K = [\alpha, \beta]$, i.e.,

$$\mathcal{P}_{K}^{BS}(\pi_{M}) = \operatorname{Cap}\left(\pi_{M}, \alpha, \beta\right) = \begin{cases} \alpha, & \pi_{M} < \alpha \\ \pi_{M}, & \alpha \leq \pi_{M} \leq \beta \\ \beta, & \beta < \pi_{M}. \end{cases}$$

Given a solution B to (5.5) and considering Remark 5.2.6, it initially appears that the optimal constrained portfolio π^* in \mathcal{M}_H can be obtained from the same projection. However, if $K \neq \mathbb{R}$, then B_u as in Corollary 5.2.13 and B as in Theorem 5.2.8 are solutions to possibly different ODEs. In particular, this implies that the portfolios π_u and $\hat{\pi}^*$ may not be identical, in which case the projection \mathcal{P}_K^H for the Heston market does not necessarily coincide with the projection \mathcal{P}_K^{BS} for the Black-Scholes market either. In other words, in a financial market with Heston stochastic volatility we in general have

$$\pi^* = \mathcal{P}_K^H(\pi_u) = \operatorname{Cap}\left(\pi_u + \underbrace{(\hat{\pi}^* - \pi_u)}_{\neq 0}, \alpha, \beta\right) \neq \operatorname{Cap}\left(\pi_u, \alpha, \beta\right) = \mathcal{P}_K^{BS}(\pi_u)$$

In the following, we render this observation more precise by providing both conditions under which $\mathcal{P}_{K}^{H} = \mathcal{P}_{K}^{BS}$ and conditions under which $\mathcal{P}_{K}^{H} \neq \mathcal{P}_{K}^{BS}$. The former case is true, whenever either $\rho = 0$ or $\pi_{M} \in K$.

Lemma 5.2.15. Let π^* be as in Lemma 5.2.3, $\hat{\pi}^*$ be as in Remark 5.2.6 and π_u as in Corollary 5.2.13. If either

 $\rho = 0 \quad or \quad \pi_M \in K,$

then

$$\pi^* = \mathcal{P}_K^H(\pi_u) = Cap(\pi_u, \alpha, \beta) = \mathcal{P}_K^{BS}(\pi_u).$$

If $\rho = 0$, the stochasticity of the volatility is completely unhedgeable in \mathcal{M}_H . As a consequence, the optimal unconstrained portfolio processes coincide in \mathcal{M}_{BS} and in \mathcal{M}_H . Thus, the projections \mathcal{P}_K^{BS} and \mathcal{P}_K^H are identical if $\rho = 0$ too. In contrast, if $\rho \neq 0$, then the projections can only be different if the underlying ODE solutions B_u and B are different, specifically when π_u and $\hat{\pi}^*$ begin taking values inside K. This is the case if and only if π_u and $\hat{\pi}^*$ begin taking values inside K at different time points. This observation leads to an equivalent characterization of when the projections \mathcal{P}_K^{BS} and \mathcal{P}_K^H are different.

Lemma 5.2.16. Let π^* be as in Lemma 5.2.3, $\hat{\pi}^*$ be as in Remark 5.2.6 and π_u as in Corollary 5.2.13. The following statements are equivalent:

(i)

$$\pi^* = \mathcal{P}_K^H(\pi_u, \alpha, \beta) \neq \mathcal{P}_K^{BS}(\pi_u, \alpha, \beta) = Cap(\pi_u, \alpha, \beta)$$

(ii)

$$\pi_M \notin [\alpha, \beta] \quad and \quad \exists t \in (0, T) : \left| \left\{ \hat{\pi}^*(t), \pi_u(t) \right\} \cap (\alpha, \beta) \right| = 1$$

We can construct an extreme case which satisfies the requirements of Lemma 5.2.16 by choosing α such that $B(\tau)$ is constant and choosing the market parameters such that π_u changes sufficiently during the investment horizon to ensure that $\pi_u(t^*) \in (\alpha, \beta)$ for some $t^* \in [0, T]$. **Corollary 5.2.17.** Let π^* be as in Lemma 5.2.3, $\hat{\pi}^*$ be as in Remark 5.2.6 and π_u as in Corollary 5.2.13. Let $sign(x) \in \{-1, 0, 1\}$ denote the sign of $x \in \mathbb{R}$.

(*i*) If

$$0 < \pi_M = \frac{\alpha}{2} < \alpha$$
 and $\alpha < \pi_u(t^*) < \beta$ for some $t^* \in [0, T]$,

then $B(\tau) = 0$ for all $\tau \in [0,T]$, $\pi^*(t) = \alpha$ for all $t \in [0,T]$ and $\pi^* = \mathcal{P}_K^H(\pi_u, \alpha, \beta) \neq \mathcal{P}_K^{BS}(\pi_u, \alpha, \beta) = Cap(\pi_u, \alpha, \beta)$.

(ii) If $\pi_M > \beta > 0$, then

$$sign\left(\frac{\partial}{\partial t}\hat{\pi}^*(t)\right) = sign\left(\frac{\partial}{\partial t}\pi_u(t)\right) = -sign(\rho b) \quad \forall t \in [0,T].$$

Hence, if in addition b < 0 and $\rho < 0$, then $\mathcal{P}_K^H = \mathcal{P}_K^{BS}$.

Clearly, the requirements on α in Corollary 5.2.17, (i) are quite restrictive, but they still provide a valuable insight into when we can expect to see a large difference between the projections \mathcal{P}_{K}^{BS} and \mathcal{P}_{K}^{H} . Namely, if

- the optimal unconstrained portfolio π_u violates the constraint at maturity (i.e., $\pi_u(T) = \pi_M \notin K$) and there is sufficient change in $\pi_u(t)$ during the investment period such that $\pi_u(t^*) \in K$ for some $t^* \in K$.
- the derivatives of $B(\tau)$ and $B_u(\tau)$ are considerably different while $\pi_u \notin K$ (constant B being the extreme case).

As a matter of fact, we will later see in the numerical experiments in Section 5.4 that it is sufficient if $\alpha \approx 2\pi_M$ (i.e., $B(\tau)$ is nearly constant) to cause a considerable difference between the two projections.

As evidenced by the majority of empirical calibrations of Heston's stochastic volatility model to financial time series (see e.g. [31] for an overview), the parameter ρ is negative for most realistic applications. In the empirical study on risk preferences of mutual fund managers by [50], it is reported that the risk aversion parameter b has a median of b = -1.43 and a mean of b = -4.8. For more risk averse investors, such as insurance companies, reinsurance companies or pension funds, one can thus realistically assume negative values for b. Thus, for most realistic parameter configurations of \mathcal{M}_H with $\pi_M > \beta$, the projections \mathcal{P}_K^H and \mathcal{P}_K^{BS} coincide for investors with a high degree of risk aversion (i.e., for a low value of b).

5.3 Implications for Related Models

In this section, we consider a generalized version of the financial market \mathcal{M}_H with $d \in \mathbb{N}$ risky assets, m = d independent CIR processes as risk drivers and a generalized dependence of market price of risk and risky asset volatility on these risk drivers. Assume that we are given functions $\gamma: (0, \infty)^d \to \mathbb{R}^d$, $\Sigma: (0, \infty)^d \to \mathbb{R}^{d \times d}$, and parameters κ , θ , σ , $z_0 \in (0, \infty)^d$ such that their components satisfy $2\kappa_i\theta_i > \sigma_i^2$ for i = 1, ..., d. Then, we define the financial market $\mathcal{M}_H^{\gamma, \Sigma}$, as the stochastic factor model (4.3) with functions

$$\mu^{z}(t,z) = \kappa \odot (\theta - z), \ \Sigma^{z}(t,z) = \begin{pmatrix} \sqrt{z_{1}} & 0 \\ & \ddots & \\ 0 & \sqrt{z_{m}} \end{pmatrix}, \ \rho(t,z) = \rho,$$
$$r(t,z) = r, \ \mu(t,z) = r(t,z)\mathbb{1} + \Sigma(z)\gamma(z), \ \Sigma(t,z) = \Sigma(z).$$

Under these assumptions, the components of the stochastic factor $z = (z_1, ..., z_d)'$ are independent ent and satisfy the SDE

$$dz_i(t) = \kappa_i \left(\theta_i - z_i(t)\right) dt + \sigma_i \sqrt{z_i(t)} dW_i^z(t), \quad \text{for } i = 1, \dots, d.$$

Further, $\mathcal{M}_{H}^{\gamma,\Sigma}$ consists of one risk-free asset P_0 and d risky assets $P = (P_1, \ldots, P_d)'$ with dynamics

$$dP_0(t) = P_0(t)rdt$$

$$dP(t) = P(t) \odot \left[(r\mathbb{1} + \Sigma(z(t))\gamma(z(t))) dt + \Sigma(z(t))dW(t) \right].$$

Clearly, we can recover the financial market \mathcal{M}_H , as considered in Section 5.2, by assuming d = 1 and choosing $\gamma(z) = \eta \sqrt{z}$ and $\Sigma(z) = \sqrt{z}$ and the Black Scholes model \mathcal{M}_{BS} if both γ and Σ are constants. Similar, but slightly more general financial market models than $\mathcal{M}_H^{\gamma,\Sigma}$ have been considered in [33] or [61], for example. In $\mathcal{M}_H^{\gamma,\Sigma}$, the wealth process $V^{v_0,\pi}$ of an investor with initial wealth v_0 who trades continuously in time with \mathbb{R}^d -valued relative portfolio process $\pi \in \Lambda$, satisfies the SDE

$$dV^{v_0,\pi}(t) = V^{v_0,\pi}(t) \left[\left(r + \gamma(z(t))' \Sigma(z(t))' \pi(t) \right) dt + \pi(t)' \Sigma(z(t)) dW(t) \right]$$

and the allocation constrained portfolio optimization problem (**P**) in $\mathcal{M}_{H}^{\gamma,\Sigma}$ is then defined as

$$(\mathbf{P}) \begin{cases} \Phi(v_0) &= \sup_{\pi \in \Lambda'} \mathbb{E} \left[U(V^{v_0,\pi}(T)) \right] \\ \Lambda' &= \left\{ \pi \in \Lambda \mid \pi(t) \in K \ \mathcal{L}[0,T] \otimes Q - \text{a.e.} \right\}. \end{cases}$$

In the following two sections, we investigate the solvability of (**P**) for given choices of γ , Σ , and K. In Section 5.3.1, we consider the PCSV Model, as discussed in [30], and in Section 5.3.2, we consider inverse volatility constraints K, which impose stronger restrictions on an investor's portfolio during periods of high volatility.

5.3.1 PCSV Model

We recover the Principal Component Stochastic Volatility model ('PCSV model') \mathcal{M}^{PCSV} , as proposed in [37], from $\mathcal{M}_{H}^{\gamma,\Sigma}$ by considering an orthogonal matrix $A = (a_1, ..., a_d) \in \mathbb{R}^{d \times d}$, with columns $a_1, ..., a_d$, and defining market price of risk and volatility as

$$\gamma(z) = \operatorname{diag}(\sqrt{z})A'\eta = \Sigma(z)'\eta, \quad \Sigma(z) = A\operatorname{diag}(\sqrt{z}) \quad \forall z \in (0,\infty)^d,$$
(5.16)

where $\operatorname{diag}(x) \in \mathbb{R}^{d \times d}$ denotes the diagonal matrix with entries $x \in \mathbb{R}^d$, and $\eta \in \mathbb{R}^d$ is a constant.

If $A = I_d$, then \mathcal{M}^{PCSV} can be regarded as a canonical generalization of Heston's stochastic volatility model \mathcal{M}_H for a *d*-dimensional asset universe, where each risky asset's volatility is determined as the square root of one of the *d* independent CIR processes z_i . However, in its general form, the independent components of the *d*-dimensional CIR process *z* are not directly regarded as volatilities. Instead, the instantaneous covariance matrix of risky asset returns

$$\Sigma(z(t))\Sigma(z(t))' = A \operatorname{diag}(z(t))A'$$

is decomposed into its principal components, i.e., the columns a_i of the matrix A represent its eigenvectors, and the independent CIR processes z_i represent their (stochastic) eigenvalues. This approach not only enables the modelling of stochastic covariances of asset returns because of additional degrees of freedom in A, but also allows for an interpretation of z as hidden risk factors determining the volatility level in the financial market. Moreover, [30] demonstrated that several stylized facts are captured by the PCSV model, such as stochasticity of volatilities and correlation of risky asset returns, volatility and correlation leverage effect, volatility spillovers, and increasing correlation in periods of high market volatility.

Let Λ^{PCSV} be the set of admissible portfolios in \mathcal{M}^{PCSV} . Then, for any $\pi \in \Lambda^{PCSV}$, the wealth process $V^{v_0,\pi}$ satisfies the SDE

$$dV^{v_0,\pi}(t) = V^{v_0,\pi}(t) \left[\left(r + \eta' A \operatorname{diag}(z(t)) A' \pi(t) \right) dt + \pi(t)' A \operatorname{diag}\left(\sqrt{z(t)} \right) dW(t) \right],$$

for $t \in [0,T]$. The instantaneous variance of $V^{v_0,\pi}(t)$ can therefore be decomposed into a weighted sum of the risk factors z, since

$$\left\|\operatorname{diag}(\sqrt{z(t)})A'\pi(t)\right\| = \left\| \begin{pmatrix} a'_1\pi(t)\sqrt{z_1(t)} \\ \vdots \\ a'_d\pi(t)\sqrt{z_d(t)} \end{pmatrix} \right\|^2 = \sum_{i=1}^d \left(a'_i\pi(t)\right)^2 z_i(t).$$

In this sense, the portfolio weights determine a risk exposure $(a'_i \pi(t))^2$ to the risk factor z_i . Hence, it is very natural to impose risk limits on these exposures, i.e., for given upper bounds $\beta_1, ..., \beta_d > 0$ we require that

$$(a'_i \pi(t))^2 \leq \beta_i \quad \forall i = 1, ..., d \quad \Leftrightarrow \quad A' \pi(t) \in \underset{i=1}{\overset{d}{\underset{i=1}{\times}} [0, \sqrt{\beta_i}] \quad \Leftrightarrow \quad \pi(t) \in \underbrace{A \cdot \left(\underset{i=1}{\overset{d}{\underset{i=1}{\times}} [0, \sqrt{\beta_i}] \right)}_{=:K_{PCSV}}.$$

We can reuse the ideas and results from Section 5.2 to obtain the optimal portfolio to the portfolio optimization problem (**P**) in $\mathcal{M}_{H}^{\gamma,\Sigma} = \mathcal{M}_{PCSV}$ with constraints $K = K_{PCSV}$.

Theorem 5.3.1. Consider the portfolio optimization problem (**P**) in \mathcal{M}_{PCSV} with constraints $K = K_{PCSV}$, let the parameters b, $(\eta_A)_i := (A'\eta)_i$, κ_i , θ_i , σ_i , $\alpha_i = 0$, $\sqrt{\beta_i}$ satisfy Assumptions 5.2.7 and 5.2.9 and B_i be defined as in Theorem 5.2.8. Define the portfolio $\pi_A^*(t)$ via

$$(\pi_A^*(t))_i = Cap\left(\frac{1}{1-b}\left(\eta_i + \sigma\rho B_i(T-t)\right), 0, \sqrt{\beta_i}\right).$$

Then the portfolio $\pi^*(t) = A \cdot \pi^*_A(t)$ is optimal for (**P**).

The key argument in the proof of Theorem 5.3.1 lies in a change of control, which transforms the portfolio optimization problem (\mathbf{P}) into an equivalent portfolio optimization problem ($\mathbf{P}_{\mathbf{A}}$) in a financial market, which consists of d risky assets with independent Heston volatilities and interval constraints. Thanks to this familiar structure and the independence of the risky asset volatilities, we can extend the ideas from Section 5.2 to solve ($\mathbf{P}_{\mathbf{A}}$) and invert the change of control to obtain a solution to (\mathbf{P}).

5.3.2 Inverse Volatility Constraints

We now discuss a related problem, which we approach with a similar methodology to the one in Section 5.3.1. Consider again the one-dimensional setting with one risky asset, i.e., d = 1. In this section, we no longer assume that the convex constraints $K \subset \overline{\mathbb{R}}$ are static, but allow them to depend on the stochastic factor z. More specifically, we consider volatility-dependent constraints of the form

$$\pi(t) \in K(z(t)) \quad \mathcal{L}[0,T] \otimes Q - a.e.,$$

where $K: (0, \infty) \to \mathcal{B}(\mathbb{R})$ is a set-valued function, taking only closed-convex values in the Borel set $\mathcal{B}(\mathbb{R})$. The motivation for such constraints is quite clear: Depending on the current state of the financial market $\mathcal{M}_{H}^{\gamma,\Sigma}$, in particular the level of risky asset volatility $\Sigma(z(t))$ and the market price of risk $\gamma(z(t))$, investors may face different constraints on their portfolio, such as more relaxed bounds in periods of low volatility or stricter bounds in periods of high volatility. Further, in the spirit of mean-variance optimization, we can think of an investor seeking an optimal portfolio allocation subject to constraints on his instantaneous portfolio volatility

$$0 \le \pi(t)\Sigma(z(t)) \le \beta_z \quad \mathcal{L}[0,T] \otimes Q - a.e. \quad \Leftrightarrow \quad 0 \le \pi(t) \le \frac{\beta_z}{\Sigma(z(t))} \quad \mathcal{L}[0,T] \otimes Q - a.e.,$$

for a given volatility level $\beta_z > 0.7$ Keeping this motivation in mind, we thus define the portfolio optimization problem with volatility-dependent constraints ($\mathbf{P}^{\mathbf{z}}$) as

$$(\mathbf{P}^{\mathbf{z}}) = \begin{cases} \Phi^{z}(v_{0}) &= \sup_{\pi \in \Lambda_{K(\cdot)}} \mathbb{E}\left[U(V^{v_{0},\pi}(T))\right] \\ \Lambda' &= \left\{\pi \in \Lambda \mid \pi(t) \in K(z(t)) \ \mathcal{L}[0,T] \otimes Q - \text{a.e.}\right\}. \end{cases}$$

In its most general form, the portfolio optimization ($\mathbf{P}^{\mathbf{z}}$) is highly non-trivial, since closed-form solutions for its optimal portfolio process π_z^* can rarely be determined for general γ and Σ , even in the absence of (stochastic) allocation constraints. In particular, the portfolio optimization (\mathbf{P}) is included as a special case in the definition of ($\mathbf{P}^{\mathbf{z}}$). However, due to the results of [17] for the Black-Scholes model \mathcal{M}_{BS} with constant volatility, as well as the results from Section 5.2 for Heston's stochastic volatility model \mathcal{M}^H , we know of at least two different models in which solutions to (\mathbf{P}) (respectively ($\mathbf{P}^{\mathbf{z}}$)) with static constraints can be obtained in closed form. Using another change of control argument, we can therefore derive conditions on the market parameters γ , Σ and the constraints K, under which we can transform ($\mathbf{P}^{\mathbf{z}}$) into an equivalent, solvable portfolio optimization problem (\mathbf{P}) in either \mathcal{M}_{BS} or \mathcal{M}^H .

Theorem 5.3.2. Consider the financial market $\mathcal{M}_{H}^{\gamma,\Sigma}$ and the portfolio optimization problem $(\mathbf{P}^{\mathbf{z}})$ and constants $-\infty \leq \alpha < \beta \leq \infty$.

(i) If $\gamma(z) = \eta$ for some $\eta > 0$ and $K(z) = \frac{1}{\Sigma(z)} [\alpha_z, \beta_z]$, then the portfolio process

$$\pi_z^*(t) = \frac{1}{\Sigma(z(t))} Cap(\pi_M, \alpha, \beta)$$

is optimal for $(\mathbf{P}^{\mathbf{z}})$.

(ii) If $\gamma(z) = \eta \sqrt{z(t)}$ for some constant $\eta > 0$, $K(z) = \frac{\sqrt{z}}{\Sigma(z)} [\alpha, \beta]$, and Assumptions 5.2.7 and 5.2.9 are satisfied, then for π^* defined as in Theorem 5.2.12, the portfolio process

$$\pi_z^*(t) = \frac{\sqrt{z(t)}}{\Sigma(z(t))} \pi^*(t)$$

is optimal for $(\mathbf{P}^{\mathbf{z}})$.

The statements of Theorem 5.3.2 can be easily generalized to financial markets with d > 1 risky assets by an analogous change of control argument. Using the results for constant volatility markets from Example 15.2 in [17], one can prove a multi-dimensional analogue to statement (i) and using the results for the PCSV model from Section 5.3.1, one can prove a multi-dimensional analogue to statement (ii). For ease of presentation, we refrain from a detailed discussion of this generalization.

⁷Note that this is different from classic mean-variance optimization, where the variance of the terminal portfolio wealth $V^{v_0,\pi}(T)$ is constrained.

5.4 Numerical Studies

In this section, we illustrate the properties of the optimal portfolio π^* for (**P**) in Heston's stochastic volatility model \mathcal{M}_H , using a numerical example. In particular, we analyze the difference between π^* and two suboptimal 'naive' portfolio processes π , which either directly follow the optimal portfolio process in \mathcal{M}_{BS} (i.e., $\pi = \operatorname{Cap}(\pi_M, \alpha, \beta)$) or apply the projection \mathcal{P}_K^{BS} from \mathcal{M}_{BS} to the optimal unconstrained portfolio π_u in \mathcal{M}_H (i.e., $\pi = \operatorname{Cap}(\pi_u, \alpha, \beta)$). The suboptimality of these portfolios is quantified using the concept of wealth-equivalent loss.

Such an analysis is only meaningful if the differences between a financial market with stochastic (Heston) volatility \mathcal{M}_H and a financial market with constant volatility \mathcal{M}_{BS} are already reflected in the optimal unconstrained portfolios π_u for \mathcal{M}_H and π_M for \mathcal{M}_{BS} . Since allocation constraints further restrict the set of admissible portfolio allocations, any existing differences between π_u and π_M tend to be diminished further when adding allocation constraints. From an investor's perspective, the distinction between \mathcal{M}_{BS} and \mathcal{M}_H is only relevant if the volatility of risky asset log returns $\sqrt{z(t)}$ changes significantly and these changes are partially hedgeable through trading in the risky asset. This is the case if the volatility of the volatility (σ) is large, the mean reversion speed (κ) is small, and the correlation between risky asset and volatility diffusion (ρ) is close to either 1 or -1 (i.e., $|\rho|$ is large).

Based on these requirements, we choose the market parameters (see Table 5.1) for our numerical example such that the resulting market dynamics resemble a financial crisis. The only volatility

Parameter		Value	Explanation
End of Investment-Horizon	T	1	Limited duration of financial crises
Risk Aversion Parameter	b	-2.5	Within ranges estimated in Table 1, [50]
Initial Wealth	v_0	1	For convenience
Risk-Free Interest Rate	r	0	For convenience
Market Price of Risk Driver	η	3.0071	Table 2, [13]
Mean Reversion Speed	κ	3.15	Table 3 '%MSE', [69]
Volatility of Volatility	σ	0.76	Table 3 '%MSE', [69]
Correlation	ρ	-0.81	Table 3 '%MSE', [69]
Long-Term Mean	θ	0.35	Feller's Condition
Initial Variance	z_0	0.35	Chosen equal to θ

Table 5.1: Base parameters for the financial market \mathcal{M}_H .

parameters which influence the optimal portfolio allocation are σ , κ , and ρ . Our choices for these parameters in Table 5.1, follow the calibration results of [69], who calibrated Heston's stochastic volatility model using option prices on the Eurostoxx 50 during the 2008 financial crisis. The relatively short investment horizon of T = 1 year is chosen to reflect the limited duration of most financial crises.⁸

We quantify the sub-optimality of both naive portfolio processes in comparison to the optimal constrained portfolio process using the concept of wealth-equivalent loss ('WEL'). For an arbitrary portfolio process $\pi \in \Lambda_K$, we define the expected utility functional $J^{\pi} : [0, T] \times (0, \infty) \times (0, \infty) \to \mathbb{R}$ as

$$J^{\pi}(t, v, z) = \mathbb{E}\left[U\left(V^{v_0, \pi}(T)\right) \mid V^{v_0, \pi}(t) = v, \ z(t) = z\right].$$
(5.17)

When considering the optimal portfolio process π^* , the expected utility functional coincides with the value function of (**P**), i.e., $J^{\pi^*}(t, v, z) = \Phi(t, v, z)$ for all $(t, v, z) \in [0, T] \times (0, \infty) \times (0, \infty)$.

 $^{^{8}}$ [75] reported that the average length of an S&P500 bear market (defined as a period with drawdown in excess of 20%) was 289 days.

The WEL $L^{\pi} = L^{\pi}(t, z)$ of π is then defined as the solution to the equation⁹

$$\Phi(t, v(1 - L^{\pi}(t, z)), z) = J^{\pi}(t, v, z).$$

An investor following the optimal portfolio allocation π^* only needs $(1 - L^{\pi}(t, z))$ as much capital to achieve the same average utility as an investor following the sub-optimal strategy π . In this sense, $L^{\pi}(t, z)$ can be interpreted as a relative loss incurred for investing sub-optimally. If π is deterministic and J^{π} is the unique solution to the associated Feynman-Kac PDE, one can use an exponentially affine ansatz to characterise J^{π} in terms of the solutions to a system of ODEs. If the ODE solutions are given, then the WEL $L^{\pi}(0, z_0)$ is known in closed form. We provide a description of this approach in Lemma 5.4.1 and Corollary 5.4.2 below. In our studies, we approximated the corresponding ODE solutions by an Euler method.

Lemma 5.4.1. Consider a bounded deterministic relative portfolio process $\pi \in \Lambda_K$. Let A_{π} , B_{π} : $[0,T] \to \mathbb{R}$ with $A_{\pi}(0) = B_{\pi}(0) = 0$ be solutions to the system of ODEs

$$A'_{\pi}(T-t) = rb + \kappa \theta B_{\pi}(T-t)$$
(5.18)

$$B'_{\pi}(T-t) = -\left[\frac{1}{2}\pi(t)^{2}b(1-b) - b\eta\pi(t)\right] + \left[\sigma\rho b\pi(t) - \kappa\right]B_{\pi}(T-t) + \frac{1}{2}\sigma^{2}B_{\pi}(T-t)^{2}.$$
 (5.19)

If J^{π} (as defined in (5.17)) is the unique solution to the Feynman-Kac-PDE (omitting the argument (t, v, z) for readability)

$$0 = J_t^{\pi} + (r + \eta \pi(t)z) v J_v^{\pi} + \kappa (\theta - z) J_z^{\pi} + \sigma \rho \pi(t) z v J_{zv}^{\pi} + \frac{1}{2} v^2 \pi(t)^2 z J_{vv}^{\pi} + \frac{1}{2} \sigma^2 z J_{zz}^{\pi}, \quad (5.20)$$

with boundary condition $J^{\pi}(T, v, z) = U(v) = \frac{1}{b}v^{b}$, then

$$J^{\pi}(t, v, z) = \frac{1}{b}v^{b} \exp\left(A_{\pi}(T - t) + B_{\pi}(T - t)z\right).$$

Corollary 5.4.2. Let $\pi \in \Lambda_K$ be a deterministic portfolio process. Let A_{π} , B_{π} be as in Lemma 5.4.1 and A, B as in Lemma 5.2.2. Then, the wealth-equivalent loss L^{π} is given as

$$L^{\pi}(t,z) = 1 - \exp\left(\frac{1}{b}\left(A_{\pi}(T-t) - A(T-t) + \left[B_{\pi}(T-t) - B(T-t)\right]z\right)\right) \ \forall (t,z) \in [0,T] \times (0,\infty)$$

5.4.1 Optimal Constrained Merton Portfolio $\pi = \text{Cap}(\pi_M, \alpha, \beta)$

In this subsection, we compare π^* with the first naive portfolio process $\pi = \text{Cap}(\pi_M, \alpha, \beta)$. Although π is static, it is at least known that π is optimal for an allocation constrained portfolio optimization problem in the Black-Scholes market \mathcal{M}_{BS} , whereas no theoretical guarantees were available for the corresponding optimization problem in \mathcal{M}_H prior to this thesis. During this analysis, we consider the allocation constraint $K = [\alpha, \beta] = [0, 1]$, which corresponds to a no-borrowing constraint that prevents short-selling in the risk-free and risky asset.

⁹Since we exclusively work with power utility functions in this chapter, we may without loss of generality assume that the WEL is independent of wealth.



Figure 5.4: Portfolio weights $\pi(t)$ for $t \in [0, T]$, lower bound $\alpha = 0$, upper bound $\beta = 1$, and parameters as in Table 5.1.

Assuming the parameters in Table 5.1, the Merton portfolio satisfies the constraints, i.e., $\pi_M \in [\alpha, \beta]$ and thus $\operatorname{Cap}(\pi_M, \alpha, \beta) = \pi_M$. In contrast to this constant allocation in the interior of $[\alpha, \beta]$, π^* initially takes a constant value at the upper bound β and later decreases towards π_M at the end of the investment horizon. Therefore, other than $\operatorname{Cap}(\pi_M, \alpha, \beta)$, π^* is able to realise and benefit from a higher allocation to the risky asset. Note that $\pi_M \in [\alpha, \beta]$ implies $\pi^* = \operatorname{Cap}(\pi_u, \alpha, \beta)$, as shown in Lemma 5.2.15.

In the following, we quantify the impact of the suboptimal allocations $\pi = \text{Cap}(\pi_M, \alpha, \beta)$ by computing the annual WEL $L^{\pi}(0, z)$ at the beginning of the investment horizon and analyze its sensitivity with respect to the risk-aversion parameter b and the volatility drivers σ , κ and ρ . The ranges of the volatility parameter are chosen to be within the minimum and maximum parameter values obtained in individual calibrations in Table 5 of [69].



Figure 5.5: Dependence of the annual wealth-equivalent loss $L^{\pi}(0, z_0)$ on individual parameters for $\pi = \text{Cap}(\pi_M, \alpha, \beta) = \pi_M$, lower bound $\alpha = 0$, upper bound $\beta = 1$, and parameters as in Table 5.1. Figure 5.5(a) illustrates the dependence on $b \in [-10, 0.7] \setminus \{0\}$, whereas Figures 5.5(b), 5.5(c) and 5.5(d) display the dependence on $\sigma \in [0.2, 1.0]$, $\kappa \in [1.5, 5.0]$ and $\rho \in [-0.9, -0.4]$.

For small values of b, where the allocation constraint K = [0, 1] is largely satisfied by the unconstrained portfolios π_M and π_u , the WELs displayed in Figure 5.5(a) are increasing in b. However, as b increases past an inflection point of approximately b = -3, the allocation constraint K becomes active. From this point onwards, K forces π^* and $\operatorname{Cap}(\pi_M, \alpha, \beta)$ closer towards each other for increasing b and therefore leads to decreasing WELs. Ultimately, for $b \geq -2$, we have $\pi^*(t) = \operatorname{Cap}(\pi_M, \alpha, \beta) = \beta = 1$ for all $t \in [0, T]$ and thus the WEL is zero. Figures 5.5(b), 5.5(c) and 5.5(d) display WELs which are increasing in σ as well as decreasing in κ and ρ . This confirms the intuition voiced at the beginning of Section 5.4, in which we argued that \mathcal{M}_H is 'more different' to \mathcal{M}_{BS} under these circumstances. Within the chosen parameter ranges, we observe the largest annual WEL of 3.2% for small κ , whereas increasing σ and decreasing ρ leads to WELs of 3.0% and 2.5%, respectively. Changing any one of the volatility parameters σ , κ or ρ to less extreme levels, which are obtained during calibrations on long-term data sets¹⁰ leads to significant decreases in annual WELs. Even if only $\sigma < 0.5$, while κ and ρ remain at crisis level, the annual WEL still drops to values of 0.75% or lower. Note that Feller's condition is satisfied for all parameter values that were considered in our analysis.

5.4.2 Capped Optimal Unconstrained Heston Portfolio $\pi = \text{Cap}(\pi_u, \alpha, \beta)$

In this subsection, we compare π^* to the second naive portfolio process, the capped optimal unconstrained Heston portfolio Cap (π_u, α, β) . In particular, we aim to illustrate the phenomenon described in Section 5.2.3. Despite a theoretical guarantee that Cap (π_u, α, β) is indeed different from π^* for certain parameter settings, these differences appear to be mostly meaningful in terms of WEL for extreme market scenarios in combination with specific, large lower bounds α . According to Lemma 5.2.16 and Corollary 5.2.17, we know that Cap (π_u, α, β) and π^* are identical in the parameter setting of Table 5.1, unless $\pi_M < \alpha$. Further, Corollary 5.2.17 suggests that we should consider market parameters which ensure $\pi_u(t) > 2\pi_M$ for some $t \in [0, T]$. For these reasons, we adjust the previously considered parameter setting. In the following, we choose the most extreme volatility parameters from the sensitivity analysis in Figure 5.5, i.e., we set $\sigma = 1.0$, $\kappa = 1.5$ and $\rho = -0.9$, and increase the risk aversion coefficient to b = -15 to obtain realistic portfolio allocations.



¹⁰For an overview, consider e.g. Table 4 in [31]. Here, the authors consider values of $\kappa = 3.5$, $\sigma = 0.3$, $\rho = -0.4$ as an 'Average Case' for their reviewed literature.

Figure 5.6: Portfolio weights $\pi(t)$ for $t \in [0, T]$, upper bound $\beta = 1$, and parameters as in Table 5.1, except for b = 15, $\sigma = 1.0$, $\kappa = 1.5$ and $\rho = -0.9$. Figure 5.6(a) considers a lower bound $\alpha = 1.75\pi_M$, and Figure 5.6(b) considers a lower bound $\alpha = 2\pi_M$.

In Figure 5.6, we compare the portfolio weights of $\pi^*(t)$ and $\operatorname{Cap}(\pi_u(t), \alpha, \beta)$ for lower bounds $\alpha \in \{1.75\pi_M, 2\pi_M\}$ such that the corresponding ODE solution *B* is (nearly) constant, as described in Corollary 5.2.17. If $\alpha = 1.75\pi_M$ then π^* is initially larger than the lower bound α , then decreases until α is reached, while π^* is constant for $\alpha = 2\pi_M$. Additionally, we observe that

$$\pi^*(t) \leq \operatorname{Cap}(\pi_u(t), \alpha, \beta) \quad \forall t \in [0, T],$$

with equality only if $\pi_u(t) \leq \alpha$. In both cases illustrated in Figure 5.6, π^* lowers the portfolio allocation early throughout the investment horizon, thus accounting for the fact that the lower bound forces the portfolio allocation to be larger than the optimal unconstrained allocation later during the investment horizon.



Figure 5.7: Lower bounds $\alpha \in \{1.5\pi_M, 1.75\pi_M, 1.9\pi_M, 1.95\pi_M, 2\pi_M\}$, upper bound $\beta = 1, t \in [0, T]$, and parameters as in Table 5.1, except for b = 15, $\sigma = 1.0$, $\kappa = 1.5$ and $\rho = -0.9$. Figure 5.7(a) displays the portfolio weights $\pi^*(t)$, and Figure 5.7(b) displays the WEL $L^{\pi}(t, z_0)$ for $\pi = \text{Cap}(\pi_u, \alpha, \beta)$.

Figure 5.7(a) illustrates the behaviour of π^* for varying α . When increasing α , we observe that π^* decreases to the lower bound at earlier time points t. Despite π^* being constant for $\alpha = 1.95\pi_M$, the solution B to the ODE (5.9) for B is not stationary, but $\rho B(\tau)$ does not leave zone Z_- for $\tau \leq T$. Thus, we expect π^* to be constant for all $\alpha \in [1.95\pi_M, 2\pi_M]$ in our parameter setting. Figure 5.7(a) displays WELs $L^{\pi}(t, z_0)$ of $\pi = \operatorname{Cap}(\pi_u, \alpha, \beta)$, which are increasing in α at t = 0. However, this monotonicity does not hold throughout the entire investment horizon, as increasing the lower bound α implies that π^* and $\operatorname{Cap}(\pi_u, \alpha, \beta)$ coincide for longer parts of the investment horizon.

Clearly, Figures 5.6 and 5.7 suggest a strong link between the value of α and the difference between π^* and $\operatorname{Cap}(\pi_u, \alpha, \beta)$. Therefore, we quantify this difference not only using WELs, but additionally define the maximum absolute weight difference between π^* and a portfolio π as

$$\Delta_{\max}^{\pi} := \max_{t \in [0,T]} \left| \pi(t) - \pi^{*}(t) \right|.$$
(5.21)

The relationship between the lower bound α and the maximum absolute difference Δ_{\max}^{π} and the annual WEL $L^{\pi}(0, z_0)$ are analyzed in Figure 5.8.



Figure 5.8: Lower bounds $\alpha \in [\pi_M, 2\pi_M]$, upper bound $\beta = 1, \pi = \text{Cap}(\pi_u, \alpha, \beta)$, and parameters as in Table 5.1, except for b = 15, $\sigma = 1.0$, $\kappa = 1.5$ and $\rho = -0.9$. For $\pi = \text{Cap}(\pi_u, \alpha, \beta)$, Figure 5.8(a) displays the maximum absolute difference Δ^{π} to the portfolio weights $\pi^*(t)$, and Figure 5.8(b) displays the WEL $L^{\pi}(0, z_0)$ to π^* .

For $\pi = \operatorname{Cap}(\pi_u, \alpha, \beta)$, the maximum absolute difference Δ_{\max}^{π} is generally increasing with α except for larger lower bounds α . For large α , the optimal constrained portfolio π^* is constant throughout the investment horizon, as illustrated in Figure 5.7(a). Then, the monotonicity of π_u (see Remark 5.2.5) and $\pi^*(T) = \alpha = \operatorname{Cap}(\pi_u(T), \alpha, \beta)$ ensure that the maximum in (5.21) is attained at t = 0. Further, $\pi_u(0) > 2\pi_M \ge \alpha$ does not depend on α . Therefore, if $\pi^*(t) = \alpha$ for all $t \in [0, T]$, then the difference

$$\Delta_{\max}^{\operatorname{Cap}(\pi_u,\alpha,\beta)} = \max_{t \in [0,T]} \left| \operatorname{Cap}(\pi_u(t),\alpha,\beta) - \pi^*(t) \right| = \left| \pi_u(0) - \alpha \right| = \pi_u(0) - \alpha$$

decreases linearly in α , which causes a slight kink in Figure 5.8(a) for large lower bounds α . Irrespectively, the annual WEL for $\pi = \text{Cap}(\pi_u, \alpha, \beta)$ is increasing with α . However, note that for all but very large lower bounds (e.g. $\alpha \geq 1.75\pi_M$), the annual WEL is still negligible.

5.5 Conclusion

In this chapter, we considered a portfolio optimization problem with allocation constraints in Heston's stochastic volatility model. We derived an explicit expression for the optimal portfolio and analyzed its properties. Surprisingly, this portfolio can be different from the naive constrained portfolio which caps off the optimal unconstrained portfolio at the boundaries of the constraint. In light of this fact, we have shown that the addition of allocation constraints can have a fundamentally different impact on the optimal portfolio in markets with stochastic volatility as compared to a Black-Scholes market with constant volatility - even in financial markets with only one risky asset. Irrespective of these theoretical certainties, we observed in a numerical study that the annual wealth equivalent loss incurred due to trading according to this naive portfolio is relatively small for the majority of realistic scenarios. In this sense, the naive 'capped' portfolio is nearly optimal for most applications. However, in turbulent financial markets, such as the financial crisis of 2008, investors with a high degree of risk aversion and lower bound on their portfolio allocation can suffer high wealth equivalent losses. For such scenarios, investors should be mindful of the difference between the optimal constrained portfolio and the naive capped portfolio.

6 Conclusion

To conclude this thesis, we briefly recall for the main Chapters 3, 4, and 5 what we have done, what we have learned and what could be investigated in future research.

What we have done: In Chapter 3, we began by generalizing the classic martingale approach in such a way that we can characterize the optimal terminal wealth to a portfolio optimization problem with constraints on terminal wealth in complete financial markets. The generalization is applicable for (simultaneous) pointwise constraints on terminal wealth and expected value constraints. Afterwards, we integrated this generalized martingale approach into the auxiliary market framework of [17] to derive equivalent dual optimality conditions which can be used to characterize the optimal terminal wealth and optimal portfolio process of an investor who faces simultaneous constraints on his relative portfolio allocation and his terminal wealth. If the allocation constraint $K \subset \mathbb{R}^d$ is a convex cone and the financial market is a Black-Scholes market, then we were able to show that the 'optimal' auxiliary market can be determined as the solution λ^* to a deterministic, convex optimization problem and does not depend on the employed utility function or the wealth constraint. For additional deterministic pointwise bounds on terminal wealth and VaR- or ES-constraints, this allowed us to derive a representation for the investor's optimal terminal wealth which is explicit up to a constant $(y^*, \lambda^*) \in [0, \infty)^2 \times \mathbb{R}^d$.

In the subsequent Chapter 4, we restricted our analysis to (incomplete) stochastic factor models, allocation constraints and power utility functions. As a consequence, the standard martingale approach is no longer applicable and we had to develop a different duality approach. Using a duality result from constrained real optimization, we were able to rewrite the associated HJB PDE as a dual HJBI PDE. The min-max problem associated with the dual HJBI PDE is the dual minimization problem in Condition (\tilde{C}) from Section 3.4, which establishes a connection to the classic auxiliary market framework. By exploiting this connection, we were able to derive Condition (EAS), which under some additional regularity conditions guarantees that the value function is of exponentially affine form and yields a closed-form expression for the optimal portfolio allocation. We verified Condition (EAS) for several examples of financial markets and constraints.

Lastly, in Chapter 5, we specifically focussed on allocation constrained portfolio optimization in a Heston market with one risky asset. We verified Condition (EAS), solved the associated HJB(I) PDE and thereby derived a closed-form expression for the optimal constrained portfolio process for the Heston market. We concluded the chapter by analyzing the sub-optimality of two 'naive' portfolio processes in comparison to the optimal constrained portfolio process in the Heston market in a numerical study.

What we have learned: In a complete financial market, the addition of pointwise constraints $(B_1 \leq V^{v_0,\pi}(T) \leq B_2)$ and expected value constraints $(\mathbb{E}[g(V^{v_0,\pi}(T))] \leq 0)$ on terminal wealth is, under some conditions, equivalent to adjusting the investor's utility function from $U \in \mathcal{U}$ to $U(\cdot; B_1, B_2) - y_1^*g(\cdot)$, where the adjustment y_1^* can be determined via the dual optimization problem $(\mathbf{D}_0^{\mathbf{V}})$. In particular, we can obtain a closed-form characterization for the optimal terminal wealth (resp. portfolio process) under terminal wealth constraints, if the closed-form

6 Conclusion

characterization is known for the adjusted utility function. Similarly, [17] argued that, under some conditions, the introduction of convex allocation constraints is equivalent to adjusting the original financial market \mathcal{M} to \mathcal{M}_{λ^*} , where the adjustment λ^* can be determined via the dual minimization problem ($\mathbf{D}^{\mathbf{K}}$). What we have shown in Section 3.4 is that simultaneous constraints on terminal wealth and portfolio allocation can be regarded as a simultaneous adjustment to both utility function and financial market (y^*, λ^*), which can be computed via the joint dual minimization problem ($\mathbf{D}^{\mathbf{K},\mathbf{V}}$).

Even without constraints, closed-form solutions for the optimal portfolio process in stochastic factor models are mainly available for the class of exponentially affine (respectively exponentially quadratic) models and power utility functions. Due to the lack of flexibility in the choice of utility function, we thus had to neglect wealth constraints in the ensuing analysis and only focussed on allocation constraints. Within this context, Condition (EAS) allows us to characterize the optimal constrained portfolio process in a stochastic factor model if the market parameters adjusted by a dual optimizer (i.e. in the auxiliary market \mathcal{M}_{λ^*}) leave the model exponentially affine. Surprisingly, even if the original model is exponentially affine, the adjusted model may no longer be exponentially affine, as illustrated in Section 4.4.2.

Since Condition (EAS) is satisfied in a one-dimensional Heston market, we were able to make another surprising observation: while interval constraints $K = [\alpha, \beta]$ in a one-dimensional Black-Scholes market lead to an optimal portfolio process which 'caps' the optimal unconstrained portfolio process, this is no longer the case in a Heston market. This suggests that allocation constraints have a different impact on the investor's portfolio process in financial markets with stochastic volatility. During numerical experiments we observed that this difference is particularly relevant for investors who trade during turbulent financial markets, exhibit a high degree of risk aversion and are constrained by a high lower bound on their relative portfolio allocation.

What could be investigated in future research: Naturally, one could extend Chapter 3 to incomplete financial markets (m > 0), which can be completed through the addition of fictitious assets, such as Heston's stochastic volatility model (see e.g., [62], [9] and [30]). If a trading restriction on these fictitious assets is imposed, then the framework of Chapter 3 can in principle be applied. It then remains to see for which financial market models, utility functions, allocation constraints and wealth constraints we can still derive useful characterizations of the optimal terminal wealth akin to the results in Section 3.5.3. A convenient choice for the fictitious assets will likely be crucial in this aspect.

The pointwise duality approach presented in Chapter 4 on the level of the HJB PDE is in principle also applicable for portfolio optimization problems with simultaneous constraints on allocation and wealth, as long as these constraints are time-consistent (i.e., we need to be able to formulate an HJB PDE). Similarly, our approach may also lead to new insights for portfolio optimization problems with time-inconsistent allocation and wealth constraints when following the equilibrium approach of [7]. Whether or not such an application can lead to new closed-form solutions for optimal portfolio processes will depend crucially on the right combination of market model, utility function and constraints. This is required to ensure that one can derive a semi-closed-form solution to the associated HJB(I) PDEs, which is possible only in rare cases – even without the presence of allocation constraints.

Lastly, it would be interesting to analyze the impact of allocation constraints on the optimal allocation via a similar projection argument as in Section 5.2.3 for multi-dimensional market models. The form of the projection \mathcal{P}_{K}^{BS} in the Black-Scholes market \mathcal{M}_{BS} is already suggested by Remark 4.4.11. It would be compelling to see if one can derive similar conditions as in Lemma 5.2.16, under which the projection in \mathcal{M}_{CIR} and even more general markets coincide with \mathcal{P}_{K}^{BS} .

A.1 Proofs Chapter 2

Proof of Lemma 2.1.4. Noting that any vector space V with scalar product is a locally convex space, this statement is a special case of Chapter 1, Corollary 1.4 in [29]. \Box

Proof of Lemma 2.1.8. Let f satisfy (2.1). Further, let $(x, \alpha), (y, \beta) \in H_f$ and $\lambda \in (0, 1)$. Then, by the definition of H_f , we have $f(x) \ge \alpha$, $f(y) \ge \beta$. Moreover, we have

$$f(\lambda x + (1-\lambda)y) \stackrel{(2.1)}{\geq} \lambda f(x) + (1-\lambda)f(y) \stackrel{\text{Def. } H_f}{\geq} \lambda \alpha + (1-\lambda)\beta$$

Thus,

$$\lambda(x,\alpha) + (1-\lambda)(y,\beta) = (\lambda x + (1-\lambda)y, \lambda \alpha + (1-\lambda)\beta) \in H_f.$$

Therefore, H_f is convex.

Let H_f be convex, $x, y \in V$ and $\lambda \in (0, 1)$. We have to differentiate between two cases:

1. $f(x) > -\infty, f(y) > -\infty$: Then, f(x), f(y) are finite and $(x, f(x)), (y, f(y)) \in H_f$. Since H_f is convex, we have

$$(\lambda x + (1 - \lambda)y, \lambda f(x) + (1 - \lambda)f(y)) = \lambda(x, f(x)) + (1 - \lambda)(y, f(y)) \in H_f.$$

However, this implies (2.1) by the definition of H_f .

2. $f(x) = -\infty$ or $f(y) = -\infty$: Then we immediately have

$$f(\lambda x + (1 - \lambda)y) \ge -\infty = \lambda f(x) + (1 - \lambda)f(y).$$

L		
L		

Equivalence of (2.2) and (2.3).

(2.2) ' \Rightarrow ' (2.3): Let $\hat{x} \in V$. If $f(\hat{x}) = \infty$, then (2.3) holds trivially. If $f(\hat{x}) < \infty$, assume that $\limsup_{x \to \hat{x}} f(x) > f(\hat{x})$. Then there exists a sequence $(x_n)_{n \in \mathbb{N}}$ taking values in V with $\lim_{n \to \infty} x_n = \hat{x}$ and $\limsup_{n \to \infty} f(x_n) = \hat{f} > f(\hat{x})$. As $M_f(\alpha)$ is closed for arbitrary α , we know that for any $\epsilon > 0$ the complement

$$M_f(f(\hat{x}) + \epsilon)^c = \left\{ x \in V \mid f(x) < f(\hat{x}) + \epsilon \right\}$$

is open. In particular, $\hat{x} \in M_f(f(\hat{x}) + \epsilon)^c$ and thus there exists $N \in \mathbb{N}$ such that $x_n \in M_f(f(\hat{x}) + \epsilon)^c$ for all n > N. Setting $\epsilon = \frac{1}{2}(\hat{f} - f(\hat{x}))$ leads to a contradiction. Hence,

 $\limsup_{x \to \hat{x}} f(x) \le f(\hat{x}).$

 $(2.3) \Rightarrow (2.2)$: Let $\alpha \in \mathbb{R}$ be arbitrary. If $M_f(\alpha) = \emptyset$, then $M_f(\alpha)$ is closed. Otherwise, let $(x_n)_{n \in \mathbb{N}}$ be a convergent sequence taking values in $M_f(\alpha)$ with limit $\hat{x} \in V$. We need to show that $\hat{x} \in M_f(\alpha)$.

As $x_n \in M_f(\alpha) \ \forall n \in \mathbb{N}$, we have $f(x_n) \ge \alpha \ \forall n \in \mathbb{N}$. In combination with (2.3) we obtain

$$f(\hat{x}) \ge \limsup_{x \to \hat{x}} f(x) \ge \limsup_{n \to \infty} f(x_n) \ge \alpha$$

and therefore $\hat{x} \in M_f(\alpha)$ and $M_f(\alpha)$ is closed.

Proof of Lemma 2.1.11.

(i) ' \Rightarrow ': Let f be use and $M_f(\alpha)$ as in Definition 2.1.9. Then,

$$M_f(\alpha)^c \times (\alpha, \infty) = \left\{ x \mid f(x) < \alpha \right\} \times (\alpha, \infty)$$

is open in $V_{\mathbb{R}}$ for all $\alpha \in \mathbb{R}$. Hence,

$$H_f^c = \{ (x, \alpha) \mid f(x) < \alpha \}$$
$$= \bigcup_{\alpha \in \mathbb{R}} \{ x \mid f(x) < \alpha \} \times (\alpha, \infty)$$

is open in $V_{\mathbb{R}}$, i.e., H_f is closed in $V_{\mathbb{R}}$.

'⇐': Let H_f be closed in $V_{\mathbb{R}} = V \times \mathbb{R}$. For any $\alpha \in \mathbb{R}$ the set $V \times \{\alpha\}$ is closed in $V_{\mathbb{R}}$. Hence, $H_f \cap (V \times \{\alpha\}) = M_f(\alpha) \times \{\alpha\}$ is closed in $V_{\mathbb{R}}$. This implies that $M_f(\alpha)$ is closed in V. Hence, f is usc.

(ii) As $usc f \ge f$ is closed, we know by (i) that $H_{usc f}$ is closed and therefore,

$$H_f \subset H_{usc\ f} \Rightarrow cl\ H_f \subset cl\ H_{usc\ f} \stackrel{(i)}{\subset} H_{usc\ f}.$$

Further, $cl \ H_f$ is the hypograph H_g of a function $g: V \to \overline{\mathbb{R}}$

$$g(x) := \sup \left\{ \alpha \mid (x, \alpha) \in cl \ H_f \right\}.$$

Since $H_f \subset cl \ H_f = H_g$, we have $f \leq g$ and g is usc. In particular, this implies $usc \ f \leq g$ and therefore

$$H_{usc\ f} \subset H_g = cl\ H_f.$$

(iii) As $f \leq co f$ and $H_{co f}$ is convex, we must have $co H_f \subset H_{co f}$. Analogously to the proof of (ii), $co H_f$ defines the hypograph H_g of a concave function

$$g(x) := \sup \left\{ \alpha \mid (x, \alpha) \in co \ H_f \right\}.$$

As g is concave and $H_f \subset co \ H_f = H_g$, we have $f \leq co \ f \leq g$ and thus $H_{co \ f} \subset H_g = co \ H_f$.

(iv) For $y = (z, \gamma) \in V_{\mathbb{R}}, \beta \in \mathbb{R}$, let $H(y, \beta)$ denote a closed half space in $V_{\mathbb{R}}$. Then, we may use Lemma 2.1.4 and the previous derivations to realize

$$\bigcap_{\substack{H_f \subset H(y,\beta)\\(y,\beta) \in V_{\mathbb{R}} \times \mathbb{R}}} H(y,\beta) \stackrel{L.2.1.4}{=} cl \ co \ H_f \stackrel{(iv)}{=} cl \ H_{co \ f} \stackrel{(ii)}{=} H_{usc \ cl \ f}.$$

In particular, we have

$$usc \ cl \ f(x) = \sup\left\{\alpha \ \Big| \ (x,\alpha) \in \bigcap_{\substack{H_f \subset H(y,\beta) \\ (y,\beta) \in V_{\mathbb{R}} \times \mathbb{R}}} H(y,\beta)\right\}$$

Our aim is to now characterize $H(y,\beta)$ and $H_{usc\ cl\ f}$ in such a way that (iv) follows immediately. We make a distinction between the three cases $\gamma < 0$, $\gamma = 0$ and $\gamma > 0$, for $y = (z, \gamma)$.

For $\gamma < 0$, we have $(x, \alpha) \in H(y, \beta)$ if and only if

$$\langle x, z \rangle + \alpha \gamma \leq \beta \quad \stackrel{\gamma \leq 0}{\Leftrightarrow} \quad \alpha \geq \frac{1}{\gamma} \left(\beta - \langle x, z \rangle \right)$$

The function f is proper and therefore there exists $\hat{x} \in V$ such that $-\infty < f(\hat{x})$. Hence, $(\hat{x}, \alpha) \in H_f$ for all $\alpha \leq f(\hat{x})$. However, when choosing

$$\hat{\alpha} := \min\left(f(\hat{x}), \frac{1}{\gamma}\left(\beta - \langle \hat{x}, z \rangle\right)\right) - 1,$$

we then have $(\hat{x}, \hat{\alpha}) \notin H(y, \beta)$ and $(\hat{x}, \hat{\alpha}) \in H_f$. Hence, $H_f \not\subset H(y, \beta)$ and

$$\bigcap_{\substack{H_f \subset H((z,\gamma),\beta)\\(z,\gamma,\beta) \in V \times (-\infty,0) \times \mathbb{R}}} H(y,\beta) = V_{\mathbb{R}}.$$
(A.1)

For $\gamma = 0$, we have $(x, \alpha) \in H(y, \beta)$ if and only if

 $\langle x,z\rangle\leq\beta.$

Therefore,

$$\begin{split} H_f \subset H(y,\beta) \Leftrightarrow \forall (x,\alpha) \in H_f : (f(x) \ge \alpha \Rightarrow \langle x,z \rangle \le \beta) \\ \Leftrightarrow \forall (x,\alpha) \in H_f : (f(x) > -\infty \Rightarrow \langle x,z \rangle \le \beta) \\ \Leftrightarrow \ dom_f \subset H(z,\beta). \end{split}$$

As $(x, \alpha) \in H((z, 0), \beta)$ is true or false irrespective of the value of α , we thus obtain

$$\bigcap_{\substack{H_f \subset H((z,0),\beta) \\ (z,\beta) \in V \times \mathbb{R}}} H(y,\beta) = \left(\bigcap_{\substack{dom_f \subset H(z,\beta) \\ (z,\beta) \in V \times \mathbb{R}}} H(z,\beta)\right) \times \mathbb{R} \stackrel{L.2.1.4}{=} (cl \ co \ dom_f) \times \mathbb{R}.$$
(A.2)

For $\gamma > 0$, we have $(x, \alpha) \in H(y, \beta)$ if and only if

$$\langle x, z \rangle + \alpha \gamma \leq \beta \Leftrightarrow \underbrace{\frac{1}{\gamma} \left(\beta - \langle x, z \rangle \right)}_{=:h(x):=h(x,z,\gamma,\beta)} \geq \alpha.$$

$$\Rightarrow H(y,\beta) = \left\{ (x,\alpha) \mid h(x;z,\gamma,\beta) \geq \alpha \right\} = H_h.$$

If we consider h as a function of x for fixed z, γ, β , then h is affine in x. Moreover, for an appropriate choice of parameters z, γ, β , we can obtain an arbitrary affine function $h: V \to \mathbb{R}$. Note that we do not need the degree of freedom in γ , as we can still obtain any affine function by setting $\gamma = 1$, changing the sign of $\langle z, x \rangle$ and varying z, β freely. Thus, we obtain

$$\bigcap_{\substack{H_f \subset H((z,\gamma),\beta) \\ (z,\gamma,\beta) \in V \times (0,\infty) \times \mathbb{R}}} H(y,\beta) = \bigcap_{\substack{h: V \to \mathbb{R} \\ H_f \subset H_h}} H_h = \bigcap_{\substack{h: V \to \mathbb{R} \\ H_f \subset H_h}} H_h$$
(A.3)

In total, (A.1), (A.2) and (A.3) yield

$$\bigcap_{\substack{H_f \subset H(y,\beta) \\ (y,\beta) \in V_{\mathbb{R}} \times \mathbb{R}}} H(y,\beta) = \left(V_{\mathbb{R}}\right) \cap \left(\left(cl \ co \ dom_f\right) \times \mathbb{R}\right) \cap \left(\bigcap_{\substack{h: V \to \mathbb{R} \\ f \leq h}} H_h\right)$$
$$= \left(\left(cl \ co \ dom_f\right) \times \mathbb{R}\right) \cap \left(\bigcap_{\substack{h: V \to \mathbb{R} \\ f \leq h}} H_h\right)$$

(v) We make a distinction between $cl \ co \ f \equiv \infty$ and $cl \ co \ f = usc \ co \ f$.

If $cl \ co \ f \equiv \infty$, then $usc \ co \ f$ is not proper and there does not exist an affine function $h: V \to \mathbb{R}$ such that $f \leq h$. This implies that for any $(z, \beta) \in V_{\mathbb{R}}$, there exists an $\bar{x} = \bar{x}(z, \beta)$ such that

$$\begin{split} \beta + \langle \bar{x}, z \rangle &< f(\bar{x}) \Rightarrow \beta < f(\bar{x}) - \langle \bar{x}, z \rangle \leq \sup_{\hat{x} \in V} \left(f(\hat{x}) - \langle \hat{x}, z \rangle \right) \\ \Rightarrow \quad \infty &= \sup_{\beta \in \mathbb{R}} \beta \leq \sup_{\hat{x} \in V} \left(f(\hat{x}) - \langle \hat{x}, z \rangle \right). \end{split}$$

In particular, taking the infimum over $z \in V$ yields for any $x \in V$

$$\inf_{z \in V} \left(\underbrace{\sup_{\hat{x} \in V} \left(f(\hat{x}) - \langle z, \hat{x} \rangle \right)}_{=\infty} + \langle z, x \rangle \right) = \infty = cl \ co \ f(x).$$

If $cl \ co \ f = usc \ co \ f$, then $usc \ co \ f$ is proper and there exists an affine function $h: V \to \mathbb{R}$ such that $f \le h$. In particular, we have

$$\bigcap_{\substack{H_f \subset H(y,\beta) \\ (y,\beta) \in V_{\mathbb{R}} \times \mathbb{R}}} H(y,\beta) = \left((cl \ co \ dom_f) \times \mathbb{R} \right) \cap \left(\bigcap_{\substack{h: V \to \mathbb{R} \text{ affine} \\ f \leq h}} H_h \right)$$

$$= \bigcap_{\substack{h: V \to \mathbb{R} \text{ affine} \\ f \leq h}} H_h$$

$$= \bigcap_{\substack{(z,\beta) \in V_{\mathbb{R}} \\ f(x) \leq \beta + \langle z, x \rangle}} \{(x,\alpha) \mid \beta + \langle z, x \rangle \geq \alpha \}. \quad (A.4)$$

For any $z \in V$ we have,

$$f(\hat{x}) \le \beta + \langle z, \hat{x} \rangle \ \forall \hat{x} \in V \ \Leftrightarrow \sup_{\hat{x} \in V} \left(f(\hat{x}) - \langle z, \hat{x} \rangle \right) \le \beta$$

and therefore, by (A.4),

$$H_{usc\ co\ f} = \bigcap_{\substack{H_f \subset H(y,\beta)\\(y,\beta) \in V_{\mathbb{R}} \times \mathbb{R}}} H(y,\beta) = \bigcap_{z \in V} \left\{ (x,\alpha) \mid \sup_{x \in V} \left(f(\hat{x}) - \langle z, \hat{x} \rangle \right) + \langle z, x \rangle \ge \alpha \right\}$$

$$= \left\{ (x,\alpha) \mid \inf_{z \in V} \left(\sup_{x \in V} \left(f(\hat{x}) - \langle z, \hat{x} \rangle \right) + \langle z, x \rangle \right) \ge \alpha \right\}$$

$$\Rightarrow \ cl \ co \ f(x) = usc \ co \ f(x) = \inf_{z \in V} \left(\sup_{x \in V} \left(f(\hat{x}) - \langle z, \hat{x} \rangle \right) + \langle z, x \rangle \right).$$

Proof of Lemma 2.1.15.

(i) For any $y \in V$, we have

$$-f^*(y) = -\sup_{x \in V} \left(f(x) - \langle y, x \rangle \right) = \inf_{x \in V} \left(-f(x) + \langle y, v \rangle \right).$$

Thus,

$$H_{-f^*} = \left\{ (y, \alpha) \mid \inf_{x \in V} (-f(x) + \langle y, x \rangle) \ge \alpha \right\}$$
$$= \left\{ (y, \alpha) \mid \underbrace{-f(x) + \langle y, x \rangle}_{=:h(y;x)} \ge \alpha \quad \forall x \in V \right\}$$
$$= \left\{ (y, \alpha) \mid h(y; x) \ge \alpha \quad \forall x \in V \right\}$$
$$= \bigcup_{x \in V} H_{h(\cdot;x)}.$$

For fixed x, the function h(y;x) is affine in y and thus use concave in y. Hence, each hypograph $H_{h(\cdot;x)}$ is closed convex. Therefore, H_{-f^*} is closed convex, which is equivalent to $-f^*(y)$ being use concave by Definition 2.1.6 and Lemma 2.1.11, (i). The statements about $f^{**} = cl \ co \ f$ follow directly from Lemma 2.1.11 (iv).

(ii) Let $f = f^{**}$. By Lemma 2.1.11, we have $f^{**} = cl \ co \ f$, which is use and concave by construction. Hence, $f = f^{**} = cl \ co \ f$ is use and concave, too. Let f be use and concave. Then,

$$f \stackrel{D. 2.1.10}{=} co f$$
 and $f \stackrel{D. 2.1.10}{=} usc f \stackrel{f \text{ proper}}{=} cl co f \stackrel{L. 2.1.11}{=} \stackrel{(iv)}{=} f^{**}.$

Proof of Lemma 2.1.17.

(i) Let $y \in V$, $\alpha \ge 0$. Then,

$$\delta_K(\alpha y) = -\inf_{x \in K} (\langle x, \alpha y \rangle) \stackrel{\alpha \ge 0}{=} -\alpha \inf_{x \in K} (\langle x, y \rangle) = \alpha \delta_K(y).$$

(ii) Let $y_1, y_2 \in V$. Then,

$$\delta_K(y_1 + y_2) = -\inf_{x \in K} \left(\langle y_1 + y_2, x \rangle \right) = -\inf_{x \in K} \left(\langle y_1, x \rangle + \langle y_2, x \rangle \right)$$

$$\leq -\inf_{x \in K} \left(\langle y_1, x \rangle \right) - \inf_{x \in K} \left(\langle y_2, x \rangle \right) = \delta_K(y_1) + \delta_K(y_2).$$

(iii) The first equivalence follows directly from (2.4). By scaling any non-zero $y \in X_K$ to

$$\hat{y} = \frac{1}{\sqrt{\langle y, y \rangle}} y$$

such that $\langle \hat{y}, \hat{y} \rangle \leq 1$, we can use the positive homogeneity of δ_K to see that

$$0 \le \delta_K(y) + \langle x, y \rangle \ \forall y \in X_K \ \Leftrightarrow 0 \le \delta_K(\hat{y}) + \langle x, \hat{y} \rangle \ \forall \hat{y} \in X_K \text{ with } \langle \hat{y}, \hat{y} \rangle \le 1,$$

i.e., the second equivalence holds.

(iv) ' \Rightarrow ': Let $\delta_K(y) = 0 \ \forall y \in X_K$. Let $x \in K, \alpha \ge 0$. Since $x \in K$, we obtain from (iii):

$$0 \leq \underbrace{\delta_K(y)}_{=0} + \langle y, x \rangle = \langle y, x \rangle \quad \forall y \in X_K$$

$$\Rightarrow 0 \leq \langle y, \alpha x \rangle = \delta_K(y) + \langle y, \alpha x \rangle \quad \forall y \in X_K \Rightarrow \alpha x \in K \quad \text{by (iii)}.$$

Hence, K is a convex cone.

' \Leftarrow ': Let K be a convex cone. Then, $0 \in K$ and thus

$$\delta_K(y) = -\inf_{x \in K} \left(\langle x, y \rangle \right) \ge \langle 0, y \rangle = 0 \quad \forall y \in V.$$

Further, if $\delta_K(y) > 0$, then there exists $x^* \in K$ such that $\langle x^*, y \rangle < 0$. Since K is a convex cone, $\alpha x^* \in K$ for all $\alpha \ge 0$. Hence,

$$\delta_{K}(y) = -\inf_{x \in K} \left(\langle x, y \rangle \right) \ge -\inf_{\alpha \ge 0} \left(\langle \alpha x^{*}, y \rangle \right) = \sup_{\alpha \ge 0} \left(\underbrace{-\alpha}_{\le 0} \underbrace{\langle x^{*}, y \rangle}_{< 0} \right) = \infty$$

Thus, $\delta_K(y) > 0 \Rightarrow y \notin X_K \stackrel{\delta \ge 0}{\Rightarrow} \delta_K(y) = 0 \ \forall y \in X_K.$

Proof of Lemma 2.1.22.

- (i): For every fixed $x \in V$, $L(x, \lambda) \langle \bar{u}, \lambda \rangle$ is the concave conjugate of F(x, u) w.r.t. u. If $F(x, \cdot)$ is proper, then by Lemma 2.1.15, (i) this implies $-L(x, \cdot) + \langle \bar{u}, \lambda \rangle$ is use and concave in u and therefore $-L(x, \cdot)$ is use and concave in u, too.
- (ii): $\inf_{\lambda \in H} L(x, \lambda)$ is the bi-conjugate of $F(x, \cdot)$ evaluated at \bar{u} . Hence, we have by Lemma 2.1.11 (iv)

$$\inf_{\lambda \in H} L(x,\lambda) = \inf_{\lambda \in H} \left(\sup_{u \in H} \left(F(x,u) - \langle u, \lambda \rangle \right) + \langle \bar{u}, \lambda \rangle \right)$$
$$= cl \ co \ F(x,\bar{u})$$
$$= f(x).$$

(iii) By (ii), we now have

$$f(x) = \inf_{\lambda \in H} L(x, \lambda) \ \forall x \in V$$

$$\Rightarrow \quad \Phi_P = \sup_{x \in V} f(x) = \sup_{x \in V} \inf_{\lambda \in H} L(x, \lambda).$$

-	-	-	-	٦

Proof of Theorem 2.1.23. We have $\Phi(\bar{u}) = \sup_{x \in V} f(x) = \Phi_P$ by definition of Φ and Φ_P . Moreover, from Lemma 2.1.15 we obtain

$$\Psi_{D} = \inf_{\lambda \in H} \sup_{x \in V} L(x, \lambda)$$

$$= \inf_{\lambda \in H} \sup_{x \in V} \sup_{u \in H} \left(\left(F(x, u) - \langle u, \lambda \rangle \right) + \langle \bar{u}, \lambda \rangle \right)$$

$$= \inf_{\lambda \in H} \sup_{u \in H} \left(\left(\sup_{x \in V} F(x, u) - \langle u, \lambda \rangle \right) + \langle \bar{u}, \lambda \rangle \right)$$

$$= \inf_{\lambda \in H} \sup_{u \in H} \left(\left(\Phi(u) - \langle u, \lambda \rangle \right) + \langle \bar{u}, \lambda \rangle \right)$$

$$= \inf_{\lambda \in H} \left(\Phi^{*}(\lambda) + \langle \bar{u}, \lambda \rangle \right)$$

$$= \Phi^{**}(\bar{u})$$

$$= cl \ co \ \Phi(\bar{u}).$$

Finally, as noted in Remark 2.1.13, weak duality of the concave conjugate yields

$$\Phi_P = \Phi(\bar{u}) \le \Phi^{**}(\bar{u}) = cl \ co \ \Phi(\bar{u}) = \Psi_D.$$

Proof of Corollary 2.1.26. As (x^*, λ^*) is a saddle-point of L, we have

$$\sup_{x \in V} L(x, \lambda^*) = L(x^*, \lambda^*) \text{ and } \inf_{\lambda \in H} L(x^*, \lambda) = L(x^*, \lambda^*).$$

However, as we additionally have

$$L(x^*, \lambda^*) = \sup_{x \in V} L(x, \lambda^*) \ge \sup_{x \in V} \inf_{\lambda \in H} L(x, \lambda) \ge \inf_{\lambda \in H} L(x^*, \lambda) = L(x^*, \lambda^*)$$

and

$$L(x^*,\lambda^*) = \inf_{\lambda \in H} L(x^*,\lambda) \le \inf_{\lambda \in H} \sup_{x \in V} L(x,\lambda) \le \sup_{x \in V} L(x,\lambda^*) = L(x^*,\lambda^*),$$

we obtain

$$\Psi_D = \inf_{\lambda \in H} \sup_{x \in V} L(x, \lambda) = \sup_{x \in V} L(x, \lambda^*),$$

i.e., λ^* is optimal for (**D**). Moreover, if $F(x^*, \cdot)$ is use and concave in u, then we get

$$\sup_{x \in V} f(x) = \sup_{x \in V} \inf_{\lambda \in H} L(x, \lambda) = \inf_{\lambda \in H} L(x^*, \lambda) = f(x^*),$$

i.e., x^* is optimal for (**P**).

Proof of Theorem 2.1.27. The equivalence of (i) and (ii) is immediate from Theorem 2.1.23 and the implication (iv) \Rightarrow (iii) holds true according to Corollary 2.1.26. It remains to show that (iii) \Rightarrow (iv). Assume that (iii) holds. Then,

$$\Phi_P = \sup_{x \in V} f(x) \stackrel{x^* \text{ optimal for } (\mathbf{P})}{=} f(x^*) = F(x^*, \bar{u}) \stackrel{\text{L. 2.1.22}}{=} \inf_{\lambda \in H} L(x^*, \lambda)$$

$$\leq L(x^*, \lambda^*) \leq \sup_{x \in V} L(x, \lambda^*) \stackrel{\lambda^* \text{ optimal for } (\mathbf{D})}{=} \inf_{\lambda \in H} \sup_{x \in V} L(x, \lambda) = \Psi_D$$
$$= \Phi_P$$

This yields

$$\sup_{x \in V} L(x, \lambda^*) = L(x^*, \lambda^*) = \inf_{\lambda \in H} L(x^*, \lambda)$$

and therefore (x^*, λ^*) is a saddle-point.

Proof of Theorem 2.1.28. Follows immediately by considering -F in Theorem 18a) in [76]. \Box

Proof of Lemma 2.1.29. Recalling Example 2.1.20, we realize that L is the Lagrangian corresponding to the perturbed objective function

$$F(x,u) = f(x) + \mathcal{X}_K(x+u) = f(x) + \mathcal{X}_{K-u}(x)$$

We aim to apply Corollary 2.1.26 and thus need to verify that $F(x^*, \cdot)$ is proper, use and concave in u. However, as $f(x^*) \in \mathbb{R}$, it is sufficient to show that the mapping

$$u \to \mathcal{X}_K(x^* + u)$$

proper, use and concave in u. However, this is trivial because K is non-empty and closed convex and $u \to x^* + u$ is affine in u. Hence, by virtue of Corollary 2.1.26, x^* is optimal for (**P**).

Proof of Corollary 2.1.30. $\Rightarrow:$ Assume that (x^*, λ^*) is a saddle-point of L. Then, we have

$$L(x,\lambda^*) \le L(x^*,\lambda^*) \le L(x^*,\lambda) \qquad \forall (x,\lambda) \in \mathbb{R}^d \times \mathbb{R}^d.$$
(A.5)

In particular, this implies for all $x \in \mathbb{R}^d$:

$$f(x) + \langle x, \lambda^* \rangle + \delta_K(\lambda^*) = L(x, \lambda^*) \le L(x^*, \lambda^*) = f(x^*) + \langle x^*, \lambda^* \rangle + \delta_K(\lambda^*),$$

i.e., (i) is satisfied. Assume now that $\langle x^*, \lambda^* \rangle + \delta_K(\lambda^*) > 0$. But then we have

$$L(x^*, 0) = f(x^*) + \langle x^*, 0 \rangle + \delta_K(0) = f(x^*)$$

$$< f(x^*) + \langle x^*, \lambda^* \rangle + \delta_K(\lambda^*) = L(x^*, \lambda^*),$$

which is a contradiction to (A.5). Hence, we must have $\langle x^*, \lambda^* \rangle + \delta_K(\lambda^*) \leq 0$. Similarly, assuming $\langle x^*, \lambda^* \rangle + \delta_K(\lambda^*) < 0$ yields

$$L(x^*, 2\lambda^*) = f(x^*) + \langle x^*, 2\lambda^* \rangle + \delta_K(2\lambda^*)$$

= $f(x^*) + 2(\langle x^*, \lambda^* \rangle + \delta_K(\lambda^*))$
< $f(x^*) + \langle x^*, \lambda^* \rangle + \delta_K(\lambda^*) = L(x^*, \lambda^*)$

which is again a contradiction to (A.5). Hence, we must have $\langle x^*, \lambda^* \rangle + \delta_K(\lambda^*) \ge 0$. In total, this yields $\langle x^*, \lambda^* \rangle + \delta_K(\lambda^*) = 0$, i.e., (*ii*) is satisfied. This allows us to show that for any $\lambda \in \mathbb{R}^d$

$$\langle x^*, \lambda \rangle + \delta_K(\lambda) = -f(x^*) + f(x^*) + \langle x^*, \lambda \rangle + \delta_K(\lambda)$$

	_	_	_	

$A \ Proofs$

$$= -f(x^*) + L(x^*, \lambda)$$

$$\stackrel{(A.5)}{\geq} -f(x^*) + L(x^*, \lambda^*)$$

$$= \underbrace{-f(x^*) + f(x^*)}_{= 0} + \underbrace{\langle x^*, \lambda^* \rangle + \delta_K(\lambda^*)}_{= 0}$$

$$\stackrel{(ii)}{=} 0.$$

According to (2.4), this implies $x^* \in K$ and thus (iii) is satisfied. ' \Leftarrow ': Let (i), (ii), (iii) be satisfied for $(x^*, \lambda^*) \in V \times H$. From (i) we know that

$$L(x,\lambda^*) = f(x) + \langle x,\lambda^* \rangle + \delta_K(\lambda^*) \le f(x^*) + \langle x^*,\lambda^* \rangle + \delta_K(\lambda^*) \le L(x^*,\lambda^*) \quad \forall x \in V.$$

Moreover, using (ii) and (iii), we obtain

$$L(x^*, \lambda^*) = f(x^*) + \langle x^*, \lambda^* \rangle + \delta_K(\lambda^*)$$

$$\stackrel{(ii)}{=} f(x^*)$$

$$\stackrel{(iii)}{\leq} f(x^*) + \overbrace{\langle x^*, \lambda \rangle + \delta_K(\lambda)}^{\geq 0}$$

$$= L(x^*, \lambda) \quad \forall \lambda \in H.$$

Thus, (x^*, λ^*) is a saddle-point of L.

Proof of Theorem 2.1.31. Consider $0 < \epsilon < 1$, $\Delta \lambda \in X_K$ with $\|\Delta \lambda\| \leq 1$ and define $\lambda_{\epsilon} = \lambda^* + \epsilon \Delta \lambda$. Since λ^* is optimal for (**D**), we know that

$$0 \leq \frac{1}{\epsilon} \left(L(x^*(\lambda_{\epsilon}), \lambda_{\epsilon}) - \underbrace{L(x^*(\lambda^*), \lambda^*)}_{\geq L(x^*(\lambda_{\epsilon}), \lambda^*)} \right)$$

$$\leq \frac{1}{\epsilon} \left(L(x^*(\lambda_{\epsilon}), \lambda_{\epsilon}) - L(x^*(\lambda_{\epsilon}), \lambda^*) \right)$$

$$= \frac{1}{\epsilon} \left(f(x^*(\lambda_{\epsilon})) + (x^*(\lambda_{\epsilon}))'\lambda_{\epsilon} + \delta_K(\lambda_{\epsilon}) - f(x^*(\lambda_{\epsilon})) - (x^*(\lambda_{\epsilon}))'\lambda^* - \delta_K(\lambda^*) \right)$$

$$= \frac{1}{\epsilon} \left((x^*(\lambda_{\epsilon}))' \underbrace{(\lambda_{\epsilon} - \lambda^*)}_{=\epsilon\Delta\lambda} + \delta_K(\lambda_{\epsilon}) - \delta_K(\lambda^*) \right)$$

$$= (x^*(\lambda_{\epsilon}))'\Delta\lambda + \frac{1}{\epsilon} \left(\delta_K(\lambda_{\epsilon}) - \delta_K(\lambda^*) \right).$$

Taking the upper limit $\epsilon \downarrow 0$ and using the fact that $x^*(\lambda)'\Delta\lambda$ is use at $\lambda = \lambda^*$, yields

$$0 \leq \limsup_{\epsilon \downarrow 0} \frac{1}{\epsilon} \left(L(x^*(\lambda_{\epsilon}), \lambda_{\epsilon}) - L(x^*(\lambda^*), \lambda^*) \right)$$

$$\leq \limsup_{\epsilon \downarrow 0} \left((x^*(\lambda_{\epsilon}))' \Delta \lambda + \frac{1}{\epsilon} \left(\delta_K(\lambda_{\epsilon}) - \delta_K(\lambda^*) \right) \right)$$

$$\leq x^*(\lambda^*)' \Delta \lambda + \limsup_{\epsilon \downarrow 0} \frac{1}{\epsilon} \left(\delta_K(\lambda_{\epsilon}) - \delta_K(\lambda^*) \right).$$
(A.6)

	_	

Using the sub-additivity and positive homogeneity of δ_K (cf. Lemma 2.1.17), finally gives

$$0 \le x^* (\lambda^*)' \Delta \lambda + \limsup_{\epsilon \downarrow 0} \frac{1}{\epsilon} \left(\delta_K(\lambda^*) + \epsilon \delta_K(\Delta \lambda) - \delta_K(\lambda^*) \right)$$
$$= x^* (\lambda^*)' \Delta \lambda + \delta_K(\Delta \lambda).$$

Since $\Delta \lambda \in X_K$ was chosen arbitrarily, Lemma 2.1.17 implies that $x^*(\lambda^*) \in K$. On the other hand, repeating the same arguments with $\Delta \lambda = -\lambda^*$ up to (A.6) yields

$$0 \leq -x^* (\lambda^*)' \lambda^* + \limsup_{\epsilon \downarrow 0} \frac{1}{\epsilon} \left(\delta_K ((1-\epsilon)\lambda^*) - \delta_K (\lambda^*) \right)$$
$$= -x^* (\lambda^*)' \lambda^* + \limsup_{\epsilon \downarrow 0} \frac{1}{\epsilon} \left((1-\epsilon)\delta_K (\lambda^*) - \delta_K (\lambda^*) \right)$$
$$= -x^* (\lambda^*)' \lambda^* - \delta_K (\lambda^*)$$
$$\Leftrightarrow \quad 0 \geq x^* (\lambda^*)' \lambda^* + \delta_K (\lambda^*)$$
$$\lambda^* \in X_K \qquad 0 = x^* (\lambda^*)' \lambda^* + \delta_K (\lambda^*).$$

Hence, $(x^*(\lambda^*), \lambda^*)$ satisfies all three KKT conditions from Corollary 2.1.30. Therefore, $(x^*(\lambda^*), \lambda^*)$ is a saddle point of L, $x^*(\lambda^*)$ is optimal for (**P**) and $\Phi_P = \Psi_D$.

Proof of Corollary 2.1.32. For given $\lambda \in \mathbb{R}^d$, the Lagrangian L is continuously differentiable and concave in x. Hence, by the first order optimality condition and A' = A, the maximizer of the Lagrangian $L(\cdot, \lambda)$ is given as

$$x^*(\lambda) = \frac{1}{2}A^{-1}(b+\lambda).$$

Further,

$$L(x^{*}(\lambda),\lambda) = -\frac{1}{4}(b+\lambda)'\underbrace{(A^{-1})'}_{=A'}AA^{-1}(b+\lambda) + \frac{1}{2}(b+\lambda)'\underbrace{(A^{-1})'}_{=A^{-1}}(b+\lambda) + c + \delta_{K}(\lambda)$$
$$= \frac{1}{4}(b+\lambda)'A^{-1}(b+\lambda) + c + \delta_{K}(\lambda)$$

 A^{-1} is positive definite, since A is positive definite and δ_K is convex and positive homogeneous of order 1, by Lemma 2.1.17. Hence, $L(x^*(\lambda), \lambda)$ is strictly convex in λ . Moreover, $\delta_K(\lambda)$ grows at most linearly in $\|\lambda\|$ and thus $L(x^*(\lambda), \lambda) \to \infty$ as $\|\lambda\| \to \infty$. Therefore, there exists a unique minimizer $\lambda = \lambda^*$ of $L(x^*(\lambda), \lambda)$, i.e., a unique λ^* which is optimal for (**D**). As $x^*(\lambda)$ is continuous in $\lambda \in \mathbb{R}^d$, the requirements of Theorem 2.1.31 are satisfied. In particular, $(x^*(\lambda^*), \lambda^*)$ is a saddle-point of L and satisfies (2.14).

Proof of Lemma 2.1.33. Following the line of argument in Example 2.1.21, L as in (2.16) defines the Lagrangian corresponding to the perturbation function $F: L_Q^2 \times \mathbb{R}^n \to \overline{\mathbb{R}}$ defined by

$$F(D, u) = \mathbb{E}\left[U(D)\right] + \mathcal{X}_K\left(\mathbb{E}\left[g(D)\right] + u\right).$$

We aim to apply Corollary 2.1.26 and thus need to verify that $F(D^*, \cdot)$ is proper, use and concave in u. It is sufficient to show that the mapping

$$u \to \mathcal{X}_K \left(\mathbb{E}\left[g(D^*) \right] + u \right).$$
 (A.7)

is use and concave in u. However, as K is non-empty and closed convex, \mathcal{X}_K is use concave. Moreover, for every $D \in L^2_Q$ the real function $u \to \mathbb{E}[g(D)] + u$ is affine and thus continuous. Hence, (A.7) is use and concave in u and so F is use and concave in u, too. Thus, by virtue of Corollary 2.1.26, D^* is optimal for (**P**).

Proof of Lemma 2.1.35. We obtain

$$\begin{split} \Psi_D &= \inf_{y \in \mathbb{R}^n} \sup_{D \in L^2_Q} L(D, y) \\ &= \inf_{y \in \mathbb{R}^n} \left(\sup_{D \in L^2_Q} \left(\mathbb{E} \left[U(D) \right] + y' \mathbb{E} \left[g(D) \right] \right) + \delta_K(y) \right) \\ &= \inf_{y \in \mathbb{R}^n} \left(\sup_{D \in L^2_Q} \left(\mathbb{E} \left[\underbrace{U(D) + y'g(D)}_{\leq \sup_{x \in \mathbb{R}} U(x) + y'g(x)} \right] \right) + \delta_K(y) \right) \\ &\stackrel{(*)}{\leq} \inf_{y \in \mathbb{R}^n} \left(\mathbb{E} \left[\sup_{x \in \mathbb{R}} \left(U(x) + y'g(x) \right) \right] + \delta_K(y) \right) \\ &= \inf_{y \in \mathbb{R}^n} \left(\mathbb{E} \left[U^*_g(y) \right] + \delta_K(y) \right). \end{split}$$

If y^* attains the infimum over $y \in \mathbb{R}^n$ and $\mathcal{I}_g(y^*) \in L^2_Q$, then we have equality in (*).

Proof of Corollary 2.1.36. For any $D \in L^2_Q$ we have

$$\begin{split} L(D, y^{*}) &= \mathbb{E} \left[U(D) \right] + (y^{*})' \mathbb{E} \left[g(D) \right] + \delta_{K}(y^{*}) \\ &\leq \sup_{D \in L^{2}_{Q}} \mathbb{E} \left[U(D) + (y^{*})'g(D) \right] + \delta_{K}(y^{*}) \\ &\leq \mathbb{E} \left[\sup_{x \in \mathbb{R}} \left(U(x) + (y^{*})'g(x) \right) \right] + \delta_{K}(y^{*}) \\ &= \mathbb{E} \left[U(\mathcal{I}_{g}(y^{*})) + (y^{*})'g(\mathcal{I}_{g}(y^{*})) \right] + \delta_{K}(y^{*}) \\ &= L(\mathcal{I}_{g}(y^{*}), y^{*}). \end{split}$$

On the other hand, as (2.20) holds, we can apply Lemma 2.1.17 to obtain for any $y \in \mathbb{R}^n$

$$L(\mathcal{I}_g(y^*), y) = \mathbb{E} \left[U(\mathcal{I}_g(y^*)) \right] + \underbrace{y' \mathbb{E} \left[g(\mathcal{I}_g(y^*)) \right] + \delta_K(y)}_{\geq 0}$$

$$\geq \mathbb{E} \left[U(\mathcal{I}_g(y^*)) \right]$$

$$= \mathbb{E} \left[U(\mathcal{I}_g(y^*)) \right] + \underbrace{(y^*)' \mathbb{E} \left[g(\mathcal{I}_g(y^*)) \right] + \delta_K(y^*)}_{=0}$$

$$= L(\mathcal{I}_g(y^*), y^*).$$

Therefore, $(\mathcal{I}_g(y^*), y^*)$ is a saddle-point of L.
Proof of Theorem 2.1.37. Consider $0 < \epsilon < 1$, $\Delta y \in X_K$ with $||\Delta y|| \leq 1$ and define $y_{\epsilon} = y^* + \epsilon \Delta y$. Since y^* attains the infimum in (2.19), we know that

$$\mathbb{E}\left[U_g^*\left(y^*\right)\right] + \delta_K\left(y^*\right) \le \mathbb{E}\left[U_g^*\left(y_\epsilon\right)\right] + \delta_K\left(y_\epsilon\right).$$
(A.8)

Moreover, we also have the Q-a.s. inequality

$$\frac{1}{\epsilon} \left(U^* \left(y_{\epsilon} \right) + \delta_K \left(y_{\epsilon} \right) - U^* \left(y^* \right) - \delta_K \left(y^* \right) \right) \\
\leq \frac{1}{\epsilon} \left(U \left(\mathcal{I}_g(y_{\epsilon}) \right) + \left(y_{\epsilon} \right)' g(\mathcal{I}_g(y_{\epsilon})) - \sup_{\substack{x \in \mathbb{R} \\ \geq U(\mathcal{I}_g(y_{\epsilon})) + (y^*)' g(\mathcal{I}_g(y_{\epsilon}))}} \left(U(x) - \left(y^* \right)' g(x) \right) + \delta_K \left(y_{\epsilon} \right) - \delta_K \left(y^* \right) \right) \\
\leq \frac{1}{\epsilon} \left(\left(y_{\epsilon} - y^* \right)' g \left(\mathcal{I}_g(y_{\epsilon}) \right) + \delta_K \left(y_{\epsilon} \right) - \delta_K \left(y^* \right) \right) \\
= \left(\Delta y \right)' g \left(\mathcal{I}_g(y_{\epsilon}) \right) + \frac{1}{\epsilon} \left(\delta_K \left(y_{\epsilon} \right) - \delta_K \left(y^* \right) \right). \tag{A.9}$$

Thus, due to the usc assumption on $\mathbb{E}[g(\mathcal{I}_g(y))]' \Delta y$, we can combine equations (A.8), (A.9) and take the upper limit $\epsilon \downarrow 0$ to obtain

$$0 \leq \limsup_{\epsilon \downarrow 0} \frac{1}{\epsilon} \left(\mathbb{E} \left[U^* \left(y_\epsilon \right) \right] + \delta_K \left(y_\epsilon \right) - \left[U^* \left(y^* \right) \right] - \delta_K \left(y^* \right) \right)$$

$$\leq \limsup_{\epsilon \downarrow 0} \left((\Delta y)' \mathbb{E} \left[g \left(\mathcal{I}_g(y_\epsilon) \right) \right] \right) + \limsup_{\epsilon \downarrow 0} \frac{1}{\epsilon} \left(\delta_K \left(y_\epsilon \right) - \delta_K \left(y^* \right) \right)$$

$$\leq (\Delta y)' \mathbb{E} \left[g \left(\mathcal{I}_g(y^*) \right) \right] + \limsup_{\epsilon \downarrow 0} \left(\frac{1}{\epsilon} \left(\delta_K \left(y_\epsilon \right) - \delta_K \left(y^* \right) \right) \right)$$

As δ_K is sub-additive and positive-homogenous of order 1 (cf. Lemma 2.1.17), this implies

$$0 \leq (\Delta y)' \mathbb{E} \left[g \left(\mathcal{I}_g(y^*) \right) \right] + \limsup_{\epsilon \downarrow 0} \frac{1}{\epsilon} \left(\delta_K \left(y^* \right) + \epsilon \delta_K \left(\Delta y \right) - \delta_K \left(y^* \right) \right) \\ = (\Delta y)' \mathbb{E} \left[g \left(\mathcal{I}_g(y^*) \right) \right] + \delta_K \left(\Delta y \right).$$

This inequality holds for all $\Delta y \in X_K$ with $\Delta y \leq 1$ and therefore Lemma 2.1.17, (iii) implies that $\mathbb{E}\left[g\left(\mathcal{I}_g(y^*)\right)\right] \in K$. Following the same steps with $\Delta y = -y^*$ yields $\delta_K(y_\epsilon) = (1-\epsilon)\delta_K(y^*)$ and thus

$$0 \leq -(y^*)' \mathbb{E} \left[g \left(\mathcal{I}_g(y^*) \right) \right] + \limsup_{\epsilon \downarrow 0} \frac{1}{\epsilon} \left((1-\epsilon) \delta_K \left(y^* \right) - \delta_K \left(y^* \right) \right)$$
$$= -(y^*)' \mathbb{E} \left[g \left(\mathcal{I}_g(y^*) \right) \right] - \delta_K \left(y^* \right)$$
$$\Leftrightarrow \quad 0 \geq (y^*)' \mathbb{E} \left[g \left(I_g(y^*) \right) \right] + \delta_K \left(y^* \right),$$

which implies $0 = (y^*)' \mathbb{E} [g(\mathcal{I}_g(y^*))] + \delta_K(y^*)$. According to Corollary 2.1.36, $(\mathcal{I}_g(y^*), y^*)$ is a saddle-point of L and $\mathcal{I}_g(y^*)$ is optimal for (**P**).

Proof of Lemma 2.1.38. We begin by proving statements (i), (ii) and (iii) for the generalized conjugate U_q^* :

(i) By the definition of U_q^* , we have

$$U_g^*(y) = \sup_{x \in \mathbb{R}} \left(U(x) + y'g(x) \right) = \sup_{x \in \mathbb{R}} \left(U(x) + \sum_{i=1}^n y_i g_i(x) \right).$$

Hence, if $g_i(x) \leq 0$, then U_g^* is non-increasing in y_i .

(ii) Similarly, we have $y_i \ge 0$ and $g_i(x) \le \hat{g}_i(x)$ for all $x \in dom_U$. Hence,

$$U_g^*(y) = \sup_{x \in \mathbb{R}} \left(U(x) + \underbrace{y'g(x)}_{\leq y'\hat{g}(x)} \right) \leq \sup_{x \in \mathbb{R}} \left(U(x) + y'\hat{g}(x) \right) = U_{\hat{g}}^*(y).$$

(iii) Let $\alpha \in (0, 1), y_1, y_2 \in \mathbb{R}^n$. Then,

$$U_{g}^{*}(\alpha y_{1} + (1 - \alpha)y_{2}) = \sup_{x \in \mathbb{R}} \left(U(x) + (\alpha y_{1} + (1 - \alpha)y_{2})'g(x) \right)$$

$$= \sup_{x \in \mathbb{R}} \left(\alpha \left[U(x) + y_{1}'g(x) \right] + (1 - \alpha) \left[U(x) + y_{2}'g(x) \right] \right)$$

$$\leq \alpha \sup_{x \in \mathbb{R}} \left(U(x) + y_{1}'g(x) \right) + (1 - \alpha) \sup_{x \in \mathbb{R}} \left(U(x) + y_{2}'g(x) \right)$$

$$= \alpha U_{g}^{*}(y_{1}) + (1 - \alpha)U_{g}^{*}(y_{1}),$$

i.e., U_g^* is convex.

We continue by proving statements (i) and (ii) for the maximizing argument \mathcal{I}_g . For this purpose, consider an arbitrary $\hat{x} \geq 0$ such that $\mathcal{I}_g(y) \leq \hat{x}$. Then:

$$U_{g}^{*}(y) = U(\mathcal{I}_{g}(y)) + y'g(\mathcal{I}_{g}(y)) = U(\mathcal{I}_{g}(y)) + \sum_{j=1}^{n} y_{j}g_{j}(\mathcal{I}_{g}(y))$$
$$\geq U(\hat{x}) + \sum_{j=1}^{n} y_{j}g_{j}(\hat{x}) = U(\hat{x}) + y'g(\hat{x}).$$
(A.10)

(i) If now $\hat{y} = (y_1, \dots, y_{i-1}, \hat{y}_i, y_{i+1}, \dots, y_n)'$ and $\hat{y}_i \ge y_i$, then we can use the fact that $\mathcal{I}_g(y) \le \hat{x}$ and g_i is non-increasing to see that

$$U_{g}^{*}(\hat{y}) \geq U(\mathcal{I}_{g}(y)) + \hat{y}'g(\mathcal{I}_{g}(y)) = \underbrace{U(\mathcal{I}_{g}(y)) + y'g(\mathcal{I}_{g}(y))}_{\stackrel{(A.10)}{\geq} U(\hat{x}) + y'g(\hat{x})} + \underbrace{(\hat{y}_{i} - y_{i})}_{\geq 0} \underbrace{g_{i}(\mathcal{I}_{g}(y))}_{\geq g_{i}(\hat{x})}$$

$$\geq U(\hat{x}) + y'g(\hat{x}) + (\hat{y}_{i} - y_{i})g_{i}(\hat{x})$$

$$= U(\hat{x}) + \hat{y}'g(\hat{x}).$$
(A.11)

Thus, by the definition of the maximizing argument \mathcal{I}_g as an infimum, we obtain

$$\begin{aligned} \mathcal{I}_g(\hat{y}) &= \inf \left\{ x \in \mathbb{R} \mid U_g^*(\hat{y}) \le U(x) + \hat{y}'g(x) \right\} \\ &\stackrel{(A.11)}{=} \inf \left\{ x \le \mathcal{I}_g(y) \mid U_g^*(\hat{y}) \le U(x) + \hat{y}'g(x) \right\} \le \mathcal{I}_g(y), \end{aligned}$$

i.e., \mathcal{I}_g is non-increasing in y_i .

(ii) If $\hat{g}_i(x) \ge g_i(x)$ for all $x \in dom_U$, if both functions are non-increasing and if $y_i \ge 0$, then

$$U_{\hat{g}}^{*}(y) \geq U(\mathcal{I}_{g}(y)) + y'\hat{g}(\mathcal{I}_{g}(y)) = \underbrace{U(\mathcal{I}_{g}(y)) + y'g(\mathcal{I}_{g}(y))}_{\stackrel{(A.10)}{\geq} U(\hat{x}) + y'g(\hat{x})} + y_{i}\left(\underbrace{\hat{g}_{i}(x)}_{\geq \hat{g}_{i}(\hat{x})} - g_{i}(\mathcal{I}_{g}(y))\right)$$

$$\geq U(\hat{x}) + y'g(\hat{x}) + y_{i}\left(\widehat{g}_{i}(\hat{x}) - g_{i}(\mathcal{I}_{g}(y))\right)$$

$$= U(\hat{x}) + y'\hat{g}(\hat{x}) + y_{i}\underbrace{\left(g_{i}(\hat{x}) - g_{i}(\mathcal{I}_{g}(y))\right)}_{\geq 0}}$$

$$\geq U(\hat{x}) + y'\hat{g}(\hat{x}). \tag{A.12}$$

Again, by the definition of the maximizing argument $\mathcal{I}_{\hat{g}}$ as an infimum, we obtain

$$\mathcal{I}_{\hat{g}}(y) = \inf \left\{ x \in \mathbb{R} \mid U_{\hat{g}}^{*}(y) \leq U(x) + y'\hat{g}(x) \right\} \\
\stackrel{(A.12)}{=} \inf \left\{ x \leq \mathcal{I}_{g}(y) \mid U_{\hat{g}}^{*}(y) \leq U(x) + y'\hat{g}(x) \right\} \leq \mathcal{I}_{g}(y).$$

Proof of Lemma 2.2.1.

Proof of (i): Since f is convex on \mathbb{R} , f is continuous and thereby bounded on any compact subset of \mathbb{R} . In particular, there exists $m, M \in \mathbb{R}$ such that

$$m \le f(x) \le M$$
 for all x with $|x - x_0| \le \epsilon$.

Hence, the statement follows from Section A, Lemma 3.1.1 in [43]. Proof of (ii): Let L_1, L_2 denote the Lipschitz constant of f_1, f_2 respectively. For any real numbers $x, y \in \mathbb{R}$ recall the identity

$$\min(x,y) = \frac{1}{2} \left(x + y - |x - y| \right).$$
(A.13)

Therefore, for any $x, y \in I$ we have

$$\begin{split} |f(x) - f(y)| &= \left| \min\left(f_{1}(x), f_{2}(x)\right) - \min\left(f_{1}(y), f_{2}(y)\right) \right| \\ \stackrel{(A.13)}{=} \frac{1}{2} \left| f_{1}(x) + f_{2}(x) - |f_{1}(x) - f_{2}(x)| - f_{1}(y) - f_{2}(y) + |f_{1}(y) - f_{2}(y)| \right| \\ &\leq \frac{1}{2} \left(|f_{1}(x) - f_{1}(y)| + |f_{2}(x) - f_{2}(y)| + \left| |f_{1}(x) - f_{2}(x)| - |f_{1}(y) - f_{2}(y)| \right| \right) \\ &\leq \frac{1}{2} \left(|f_{1}(x) - f_{1}(y)| + |f_{2}(x) - f_{2}(y)| + |f_{1}(x) - f_{2}(x) - (f_{1}(y) - f_{2}(y))| \right) \\ &= \frac{1}{2} \left(|f_{1}(x) - f_{1}(y)| + |f_{2}(x) - f_{2}(y)| + \underbrace{|f_{1}(x) - f_{2}(x) - f_{1}(y) + f_{2}(y)|}_{\leq |f_{1}(x) - f_{1}(y)| + |f_{2}(x) - f_{2}(y)|} \right) \\ &\leq \underbrace{|f_{1}(x) - f_{1}(y)|}_{\leq L_{1}|x - y|} \underbrace{|f_{2}(x) - f_{2}(y)|}_{\leq L_{2}|x - y|} \\ &\leq (L_{1} + L_{2})|x - y|. \end{split}$$

Hence, f is Lipschitz continuous on I with Lipschitz constant $L = L_1 + L_2$.

Proof of 2.2.2. This statement follows directly from Chapter §6, 'VII. Existence and Uniqueness Theorem' in [83]. $\hfill \square$

Proof of Lemma 2.2.3. We first consider the case $f(B_0) \neq 0$. The following argument proceeds by contradiction. Assume that B is not a strictly monotone function of τ . Since f is locally Lipschitz continuous and therefore continuous, B is continuously differentiable and there must exist $\bar{\tau} \in I$ with $B(\bar{\tau}) = \bar{B}$ such that

$$B'(\bar{\tau}) = f(\bar{B}) = 0$$
 and $B'(\tau) \neq 0$ for all τ with $|\tau - \tau_0| < |\bar{\tau} - \tau_0|$.

In particular, B is also a solution to the ODE

$$B'(\tau) = f(B(\tau)), \quad B(\bar{\tau}) = \bar{B}.$$
 (A.14)

On the other hand, as $f(\bar{B}) = f(B(\bar{\tau})) = B'(\bar{\tau}) = 0$, the function $\hat{B}(\tau) := \bar{B}$ is also a solution to the ODE (A.14). However, for all τ with $|\tau - \tau_0| < |\bar{\tau} - \tau_0|$ we know that $f(B(\tau)) = B'(\tau) \neq 0$ and thus $B(\tau) \neq \hat{B}(\tau)$. This is a contradiction to Theorem 2.2.2. Therefore, B must be strictly monotone in τ .

If $f(B_0) = 0$, by an analogous argument, $B(\tau) = B_0$ is the unique solution to ODE (2.21) and therefore B is constant in τ .

Proof of Lemma 2.2.4. Statement (i) follows directly from Lemma 10.12 in [38] with $A = \frac{1}{2}r_2$, $B = r_1, C = r_0, u = B_0$ and

$$-\frac{2r_0 (e^{r_3\tau} - 1) - (r_3 (e^{r_3\tau} + 1) + r_1 (e^{r_3\tau} - 1)) B_0}{r_3 (e^{r_3\tau} + 1) - r_1 (e^{r_3\tau} - 1) - r_2 (e^{r_3\tau} - 1) B_0}$$

$$= -\frac{-2r_3 B_0 + (e^{r_3\tau} - 1) (2r_0 - (r_1 + r_3) B_0)}{2r_3 - (e^{r_3\tau} - 1) (r_1 + r_2 B_0 - r_3)}$$

$$= \frac{2r_2 r_3 B_0 - (e^{r_3\tau} - 1) (2r_0 r_2 - (r_1 + r_3) r_2 B_0)}{2r_2 r_3 - r_2 (e^{r_3\tau} - 1) (r_1 + r_2 B_0 - r_3)}$$

$$= \frac{2r_2 r_3 B_0 - (e^{r_3\tau} - 1) (r_3 - r_1) (r_3 + r_1) - (r_1 + r_3) r_2 B_0)}{2r_2 r_3 - r_2 (e^{r_3\tau} - 1) (r_1 + r_2 B_0 - r_3)}$$

$$= \frac{2r_2 r_3 B_0 + (e^{r_3\tau} - 1) (r_1 + r_3) (r_1 + r_2 B_0 - r_3)}{2r_2 r_3 - r_2 (e^{r_3\tau} - 1) (r_1 + r_3) (r_1 + r_2 B_0 - r_3)}.$$

The function B as in (2.23) is well-defined and continuously differentiable as long as the denominator in (2.23) is non-zero. In particular, taking derivatives verifies that B satisfies (2.22) as long as its denominator is non-zero. This is true for $\tau = 0$, when the denominator is equal to $r_2r_3 > 0$, and the denominator stays positive if $r_1 + r_2B_0 - r_3 \leq 0$, i.e., $t_+(B_0) = \infty$. However, if $r_1 + r_2B_0 - r_3 > 0$, then

$$0 \stackrel{!}{=} 2r_2r_3 - r_2 \left(e^{r_3t_+(B_0)} - 1 \right) (r_1 + r_2B_0 - r_3)$$

$$\Leftrightarrow 2r_3 = \left(e^{r_3t_+(B_0)} - 1 \right) (r_1 + r_2B_0 - r_3)$$

$$\Leftrightarrow t_+(B_0) = \frac{1}{r_3} \ln \left(\frac{r_1 + r_2B_0 + r_3}{r_1 + r_2B_0 - r_3} \right).$$

Proof of Corollary 2.2.5. If $\tau_{\hat{B}} < t_+(B_0)$, then

$$B(\tau_{\hat{B}}) = \frac{2r_2r_3B_0 + (e^{r_3\tau_{\hat{B}}} - 1)(r_1 + r_3)(r_1 + r_2B_0 - r_3)}{2r_2r_3 - r_2(e^{r_3\tau_{\hat{B}}} - 1)(r_1 + r_2B_0 - r_3)}$$

This in turn implies

$$\hat{B} \stackrel{!}{=} B(\tau_{\hat{B}}) = \frac{2r_2r_3B_0 + (e^{r_3\tau_{\hat{B}}} - 1)(r_1 + r_3)(r_1 + r_2B_0 - r_3)}{2r_2r_3 - r_2(e^{r_3\tau_{\hat{B}}} - 1)(r_1 + r_2B_0 - r_3)}$$

$$\Leftrightarrow \quad \hat{B}\left(2r_2r_3 - r_2\left(e^{r_3\tau_{\hat{B}}} - 1\right)(r_1 + r_2B_0 - r_3)\right)$$

$$= 2r_2r_3B_0 + r_2\left(e^{r_3\tau_{\hat{B}}} - 1\right)(r_1 + r_3)(r_1 + r_2B_0 - r_3)$$

$$\Leftrightarrow \quad 2r_2r_3\left(\hat{B} - B_0\right) = (e^{r_3\tau_{\hat{B}}} - 1)(r_1 + r_2B_0 - r_3)\left(r_1 + \hat{B}r_2 + r_3\right)$$

$$\Leftrightarrow \quad (e^{r_3 \tau_{\hat{B}}} - 1) = \frac{2r_2 r_3 \left(\hat{B} - B_0\right)}{\left(r_1 + r_2 B_0 - r_3\right) \left(r_1 + \hat{B} r_2 + r_3\right)} \Leftrightarrow \quad \tau_{\hat{B}} = \frac{1}{r_3} \ln \left(\frac{2r_2 r_3 \left(\hat{B} - B_0\right)}{\left(r_1 + r_2 B_0 - r_3\right) \left(r_1 + \hat{B} r_2 + r_3\right)} + 1\right) = \frac{1}{r_3} \ln \left(\frac{2r_2 r_3 \left(\hat{B} - B_0\right) + \left(r_1 + r_2 B_0 - r_3\right) \left(r_1 + \hat{B} r_2 + r_3\right)}{\left(r_1 + r_2 B_0 - r_3\right) \left(r_1 + \hat{B} r_2 + r_3\right)}\right)$$

Proof of Theorem 2.3.2. This is a special case of Theorem 3.41 in [51].

A.2 Proofs Chapter 3

Proof of Lemma 3.3.1. The proof follows along the lines of Proposition 7.3 in [17]. Consider the non-negative Q-martingale M defined as

$$M(t) = \mathbb{E}\left[D\tilde{Z}(T) \mid \mathcal{F}_t\right].$$

By a variation of the martingale representation theorem (see Theorem 4.15 in [48]), there exists an \mathbb{R}^d -valued process ψ with $\int_0^T \|\psi(s)\|^2 ds < \infty$ Q-a.s, such that M can be expressed as a stochastic integral

$$M(t) = v_0 + \int_0^t \psi(t)' dW(t).$$

Due to the non-negativity and martingale property of M, M(t) = 0 for one $t \in (0, T)$ implies $M(s) = 0 \ \forall s \in [t, T]$. We define the processes

$$X(t) = \frac{M(t)}{\tilde{Z}(t)}$$

and $\pi(t) := \left(\Sigma(t)'\right)^{-1} \left[\frac{\psi(t)}{M(t)} + \gamma(t)\right] \mathbb{1}_{\{M(t)>0\}}$

Applications of Itô's lemma and Itô's product rule yield

$$d\left(\frac{1}{\tilde{Z}(t)}\right) = -\frac{1}{(\tilde{Z}(t))^2}d\tilde{Z}(t) + \frac{1}{(\tilde{Z}(t))^3}d\langle\tilde{Z}\rangle_t = \frac{1}{\tilde{Z}(t)}\left(\left[r(t) + \|\gamma(t)\|^2dt\right] + \gamma(t)'dW(t)\right)$$

and

$$\begin{split} dX(t) &= \frac{1}{\tilde{Z}(t)} dM(t) + M(t) d\left(\frac{1}{\tilde{Z}(t)}\right) dt + \left\langle M, \frac{1}{\tilde{Z}} \right\rangle_t \\ &= \frac{1}{\tilde{Z}(t)} \left(\psi(t)' dW(t) + M(t) \left[r(t) + \|\gamma(t)\|^2 \right] dt + M(t)\gamma(t)' dW(t) + \gamma(t)'\psi(t) dt \right) \\ &= \frac{1}{\tilde{Z}(t)} \left(\left[M(t)r(t) + \gamma(t)' \underbrace{\left(M(t)\gamma(t) + \psi(t) \right)}_{=M(t)\Sigma(t)'\pi(t)} \right] dt + \underbrace{\left(M(t)\gamma(t) + \psi(t) \right)}_{=M(t)\Sigma(t)'\pi(t)} dW(t) \right) \\ &= X(t) \left(\left[r(t) + (\mu(t) - r(t)\mathbb{1})'\pi(t) \right] dt + \pi(t)'\Sigma(t) dW(t) \right). \end{split}$$

Hence, $V^{v_0,\pi}$ and X satisfy the same SDE. Further, as $X(0) = M(0) = v_0$, the wealth process $V^{v_0,\pi}$ and X coincide.

Proof of Lemma 3.3.2.

As $C(v_0, B_1, B_2, g) \subset C(v_0, g)$ and $U(D) = U(D; B_1, B_2)$ for all $D \in C(v_0, B_1, B_2, g)$, we clearly have $\Phi_0(v_0) \leq \tilde{\Phi}_0(v_0)$.

To show the opposite inequality, we take any $D \in C(v_0, g)$ with $Q(D > B_2) > 0$, and show that there exists a $D' \in C(v_0, B_1, B_2, g)$ with $\mathbb{E}[U(D')] = \mathbb{E}[U(D'; B_1, B_2)] > \mathbb{E}[U(D; B_1, B_2)]$. For this purpose, let $D \in C(v_0, g)$ with $Q(D > B_2) > 0$.

• If $Q(D < B_1) > 0$, then $\mathbb{E}[U(D; B_1, B_2)] = -\infty$ and by Assumption 3.2.1, we can choose $D' = v_0 P_0(T) \in C(v_0, B_1, B_2, g)$ to obtain

$$E[U(D; B_1, B_2)] = -\infty < \mathbb{E}[U(v_0 P_0(T))] = \mathbb{E}[U(D')] = \mathbb{E}[U(D'; B_1, B_2)].$$

• If $Q(D < B_1) = 0$, then we define for $\alpha \in [0, 1]$:

$$D_{\alpha} = B_2 \mathbb{1}_{\{D > B_2\}} + D\mathbb{1}_{\{D \le B_2\}} + \alpha \left(B_2 - D\right) \mathbb{1}_{\{D \le B_2\}}$$

By construction, we have $B_1 \leq D < D_{\alpha} \leq B_2$ for all $\alpha \in (0, 1]$. In particular, as U is strictly increasing on its domain and g is non-increasing, this implies for any $\alpha \in (0, 1]$:

$$\mathbb{E}\left[U(D)\right] < \mathbb{E}\left[U(D_{\alpha})\right]$$

and

$$\mathbb{E}\left[g(D_{\alpha})\right] \leq \mathbb{E}\left[g(D)\right] \stackrel{D \in C(v_0,g)}{\leq} 0.$$

Hence, we only need to find a specific $\hat{\alpha} \in (0, 1]$ such that $v_0 = \mathbb{E}[\tilde{Z}(T)D_{\hat{\alpha}}]$ and choose $D' = D_{\hat{\alpha}}$. Following Lemma 3.3.1 and standing assumption (3.3), the risk-free portfolio $\pi \equiv 0$ satisfies

$$\mathbb{E}\left[\tilde{Z}(T)B_1\right] < \underbrace{\mathbb{E}\left[\tilde{Z}(T)V^{v_0,0}(T)\right]}_{=v_0} < \mathbb{E}\left[\tilde{Z}(T)B_2\right].$$
(A.15)

We define the quantities

$$\bar{v} = \mathbb{E}\left[\tilde{Z}(T)D_{0}\right] < v_{0}$$
$$\hat{v} = \mathbb{E}\left[\tilde{Z}(T)\left(D - B_{2}\right)\mathbb{1}_{\{D > B_{2}\}}\right] = v_{0} - \bar{v} > 0,$$
$$\tilde{v} = \mathbb{E}\left[\tilde{Z}(T)\left(B_{2} - D\right)\mathbb{1}_{\{D \le B_{2}\}}\right] > 0.$$

Due to $D \in C(v_0, g)$ and (A.15), we obtain

$$\hat{v} - \tilde{v} = \mathbb{E}\left[\tilde{Z}(T)\left(D - B_2\right)\right] = v_0 - \mathbb{E}\left[\tilde{Z}(T)B_2\right] < 0.$$

Hence, choosing $0 < \hat{\alpha} = \frac{\hat{v}}{\tilde{v}} < 1$ yields

$$\mathbb{E}[\tilde{Z}(T)D_{\hat{\alpha}}] = \mathbb{E}\left[\tilde{Z}(T)\left(B_{2}\mathbb{1}_{\{D>B_{2}\}} + D\mathbb{1}_{\{D\leq B_{2}\}} + \hat{\alpha}\left(B_{2} - D\right)\mathbb{1}_{\{D\leq B_{2}\}}\right)\right]$$
$$= \mathbb{E}\left[\tilde{Z}(T)D_{0}\right] + \hat{\alpha}\mathbb{E}\left[\tilde{Z}(T)\left(B_{2} - D\right)\mathbb{1}_{\{D\leq B_{2}\}}\right]$$
$$= \bar{v} + \frac{\hat{v}}{\tilde{v}}\tilde{v} = v_{0}.$$

In particular, $D_{\hat{\alpha}} \in C(v_0, B_1, B_2, g)$ and $\mathbb{E}[U(D)] < \mathbb{E}[U(D_{\hat{\alpha}})]$.

In total, this implies

$$\begin{split} \tilde{\Phi}_0(v_0) &= \sup_{D \in C(v_0,g)} \mathbb{E}[U(D;B_1,B_2)] \\ &= \sup_{D \in C(v_0,B_1,B_2,g)} \mathbb{E}[U(D;B_1,B_2)] \\ &= \sup_{D \in C(v_0,B_1,B_2,g)} \mathbb{E}[U(D)] \\ &= \Phi_0(v_0). \end{split}$$

Proof of Theorem 3.3.3. Following Lemma 2.1.35, we may write the dual optimization problem associated with $(\tilde{\mathbf{P}}_0)$ as

$$(\mathbf{D}_{\mathbf{0}})\left\{\Psi_{D_0}=\inf_{y\in\mathbb{R}^2}\left(\mathbb{E}\left[U_{g_0}^*(y;B_1,B_2)\right]+\delta_{K_V}(y)\right).\right.$$

According to Theorem 2.1.37, if $y^* = (y_0^*, y_1^*) \in \mathbb{R}^2$ is optimal for (\mathbf{D}_0) , $\mathcal{I}_{g_0}(y^*; B_1, B_2) \in L^2_Q$ and

$$y \to \mathbb{E}\left[g_0(\mathcal{I}_{g_0}(y; B_1, B_2))\right]' \Delta y \tag{A.16}$$

is use at $y = y^*$ for all $\Delta y \in X_{K_V}$, then $\mathcal{I}_{g_0}(y^*; B_1, B_2)$ is (admissible and) optimal for $(\tilde{\mathbf{P}}_0)$ and

 $(y^*)' \mathbb{E} \left[g_0(\mathcal{I}_{g_0}(y^*; B_1, B_2)) \right] + \delta_{K_V}(y^*) = 0.$

Thus, it remains to show that $\delta_{K_V}(y) = y_0 v_0$ for all $y = (y_0, y_1) \in [0, \infty)^2$ and that the minimization in (\mathbf{D}_0) can be restricted to $[0, \infty)^2$.

For $y = (y_0, y_1)' \in \mathbb{R}^2$ the support function δ_{K_V} can be computed explicitly as

$$\delta_{K_V}(y) = -\inf_{x \in \{-v_0\} \times [0,\infty)} \left(y'x \right) = y_0 v_0 - \inf_{x_1 \in [0,\infty)} \left(y_1 x_1 \right) = \begin{cases} y_0 v_0 & y_1 \ge 0\\ \infty & y_1 < 0. \end{cases}$$
(A.17)

In particular, $X_K = \mathbb{R} \times [0, \infty)$ and $\delta_K(y) = y_0 v_0$ for all $y \in X_K$. Moreover, U is strictly increasing on its domain, $Q(\tilde{Z}(T) > 0) = 1$, g is non-increasing and $V^{v_0,0}(T) < B_2$ (Assumption 3.2.1). Thus, we have for any $y = (y_0, y_1)' \in (-\infty, 0) \times [0, \infty)$

$$U_{g_0}^*(y; B_1, B_2) + y_0 v_0 = \sup_{B_1 \le x \le B_2} \left(\underbrace{U(x) - y_0 \tilde{Z}(T) x - y_1 g(x)}_{\text{strictly increasing in } x} \right) - y_0 v_0$$

= $U(B_2) - y_0 \left(\tilde{Z}(T) B_2 - v_0 \right) - y_1 g(B_2)$
> $U(B_2) - y_0 \left(\tilde{Z}(T) V^{v_0,0}(T) - v_0 \right) - y_1 g(V^{v_0,0}(T))$

Further, by Assumption, $\mathbb{E}[g(B_2)] \leq \mathbb{E}[g(V^{v_0,0}(T))] \leq 0$. Hence,

$$\mathbb{E}[U_{g_0}^*(y; B_1, B_2)] - y_0 v_0 > \mathbb{E}[U(B_2)] - y_0 \left(\mathbb{E}[\tilde{Z}(T)V^{v_0,0}(T)] - v_0\right) - y_1\mathbb{E}[g(V^{v_0,0}(T))] \\ \stackrel{A.3.2.1}{\geq} \mathbb{E}[U(B_2)] - y_0 \left(\mathbb{E}[\tilde{Z}(T)V^{v_0,0}(T)] - v_0\right) \\ \stackrel{L.3.3.1}{=} \mathbb{E}[U(B_2)] \\ = \mathbb{E}[U_{g_0}^*(0; B_1, B_2)] + 0 \cdot v_0.$$
(A.18)

Thus, we may restrict the minimization in (\mathbf{D}_0) to

$$\Psi_{\hat{D}_{0}} = \inf_{\substack{y \in \mathbb{R}^{2} \\ = \\ y = (y_{0}, y_{1})' \in \mathbb{R} \times [0, \infty)}} \left(\mathbb{E} \left[U_{g_{0}}^{*}(y; B_{1}, B_{2}) \right] + y_{0} v_{0} \right)$$

$$\stackrel{(A.18)}{=} \inf_{\substack{y = (y_{0}, y_{1})' \in [0, \infty)^{2} \\ y = (y_{0}, y_{1})' \in [0, \infty)^{2}}} \left(\mathbb{E} \left[U_{g_{0}}^{*}(y; B_{1}, B_{2}) \right] + y_{0} v_{0} \right).$$

Proof of Lemma 3.4.1. The poof is completely analogous to the proof of Lemma 3.3.1.

Proof of Lemma 3.4.2.

- (i) This statement follows from the analogous steps as in the proof of Lemma 3.3.2.
- (ii) By assumption, we know that $\frac{v_0}{v_\lambda(B_2)} > 1$, $D^* = \frac{v_0}{v_\lambda(B_2)}B_2 > B_2$,

$$\mathbb{E}[D^* \tilde{Z}_{\lambda}(T)] = \frac{v_0}{v_{\lambda}(B_2)} \mathbb{E}[B_2 \tilde{Z}_{\lambda}(T)] = v_0.$$

and due to the monotonicity of g, (3.3) and (3.4)

$$\mathbb{E}[g(D^*)] \le \mathbb{E}[g(B_2)] \le \mathbb{E}[g(v_0 P_0)] \le 0.$$

Hence, $D^* \in C_{\lambda}(v_0, g)$ and

$$\tilde{\Phi}_{\lambda}(v_0) = \sup_{D \in C_{\lambda}(v_0,g)} \mathbb{E}\left[\underbrace{U(D; B_1, B_2)}_{\leq U(B_2)}\right] \leq \mathbb{E}[U(B_2)] \leq \mathbb{E}[U(D^*)] \leq \tilde{\Phi}_{\lambda}(v_0),$$

i.e., D^* is optimal for $(\tilde{\mathbf{P}}_{\lambda})$.

Proof of Lemma 3.4.3. The argument goes along the lines of [17], Proposition 8.3. As π_{λ^*} satisfies the slackness condition

$$\delta_K(\lambda^*(t)) + \pi_{\lambda^*}(t)'\lambda^*(t) = 0 \quad \mathcal{L}[0,T) \otimes Q - a.e.,$$

we have $V_{\lambda^*}^{v_0, \pi_{\lambda^*}}(0) = V^{v_0, \pi_{\lambda^*}}(0) = v_0$ and

$$dV_{\lambda^{*}}^{v_{0},\pi_{\lambda^{*}}}(t) = V_{\lambda^{*}}^{v_{0},\pi_{\lambda^{*}}}(t) \left([r(t) + (\mu(t) - r(t)\mathbb{1})'\pi_{\lambda^{*}}(t)] + \pi_{\lambda^{*}}(t)'\Sigma(t)dW(t) \right) + \underbrace{V_{\lambda^{*}}^{v_{0},\pi_{\lambda^{*}}}(t) [\delta_{K}(\lambda^{*}(t)) + \pi_{\lambda^{*}}(t)'\lambda^{*}(t)]}_{=0} dt = V_{\lambda^{*}}^{v_{0},\pi_{\lambda^{*}}}(t) \left([r(t) + (\mu(t) - r(t)\mathbb{1})'\pi_{\lambda^{*}}(t)] + \pi_{\lambda^{*}}(t)'\Sigma(t)dW(t) \right).$$

Therefore, $V_{\lambda^*}^{v_0,\pi_{\lambda^*}}$ and $V^{v_0,\pi_{\lambda^*}}$ are solutions to the same SDE and $V_{\lambda^*}^{v_0,\pi_{\lambda^*}}(T) = V^{v_0,\pi_{\lambda^*}}(T)$ holds *Q*-a.s.. Further,

- $\pi_{\lambda^*} \in K \mathcal{L}[0,T] \otimes Q a.e.$
- $V^{v_0,\pi_{\lambda^*}}(T) = V^{v_0,\pi_{\lambda^*}}_{\lambda^*}(T) \ge B_1$ Q-a.s., as $V^{v_0,\pi_{\lambda^*}}_{\lambda^*}(T)$ is optimal for $(\tilde{\mathbf{P}}_{\lambda^*})$.
- $V^{v_0,\pi_{\lambda^*}}(T) = V^{v_0,\pi_{\lambda^*}}_{\lambda^*}(T) \leq B_2 Q$ -a.s., as $V^{v_0,\pi_{\lambda^*}}_{\lambda^*}(T)$ is optimal for $(\tilde{\mathbf{P}}_{\lambda^*})$ and $v_0 \leq v_{\lambda^*}(B_2)$ (Lemma 3.4.2).
- $\mathbb{E}\left[g(V^{v_0,\pi_{\lambda^*}}(T))\right] = \mathbb{E}\left[g(V^{v_0,\pi_{\lambda^*}}_{\lambda^*}(T))\right] \le 0$, as $V^{v_0,\pi_{\lambda^*}}_{\lambda^*}(T)$ is admissible for $(\tilde{\mathbf{P}}_{\lambda^*})$.
- $\mathbb{E}\left[U(V^{v_0,\pi_{\lambda^*}}(T))\right] = \mathbb{E}\left[U(V^{v_0,\pi_{\lambda^*}}(T);B_1,B_2)\right] = \mathbb{E}\left[U(V^{v_0,\pi_{\lambda^*}}_{\lambda^*}(T);B_1,B_2)\right] = \tilde{\Phi}_{\lambda^*}(v_0).$

Thus, $\pi_{\lambda^*} \in \Lambda(v_0, K, B_1, B_2, g)$ and

$$\mathbb{E}\left[U(V^{v_0,\pi_{\lambda^*}}(T))\right] \le \Phi(v_0) \le \tilde{\Phi}_{\lambda^*}(v_0) = \mathbb{E}\left[U(V^{v_0,\pi_{\lambda^*}}(T))\right]$$

i.e., π_{λ^*} is optimal for (**P**).

Proof of Theorem 3.4.4. The poof is completely analogous to the proof of Lemma 3.4.4. \Box

Lemma A.2.1. Let $\lambda \in \mathcal{D}', y^* = (y_0^*, y_1^*)' \in (0, \infty) \times [0, \infty)$. Then,

(i)

 $Q\left(y_0 \to \mathcal{I}_{g_\lambda}(y_0, y_1^*; B_1, B_2) \text{ is continuous at } y_0 = y_0^*\right) = 1.$

(ii) There exists $y_0^+ < y_0^*$ such that

$$\mathcal{I}_{g_{\lambda}}(y_{0}^{*}, y_{1}^{*}; B_{1}, B_{2}) \leq \mathcal{I}_{g_{\lambda}}(y_{0}^{+}, y_{1}^{*}; B_{1}, B_{2}) \text{ and } \mathbb{E}\left[\mathcal{I}_{g_{\lambda}}(y_{0}^{+}, y_{1}^{*}; B_{1}, B_{2})\tilde{Z}_{\lambda}(T)\right] < \infty.$$

Proof of Lemma A.2.1.

(i) Consider the function

$$\hat{g}(x) := \begin{pmatrix} -x \\ -g(x) \end{pmatrix}.$$

Then, $\mathcal{I}_{\hat{g}}(y_0, y^*; B_1, B_2)$ is non-increasing in y_0 by Lemma 2.1.38, (i). Hence, according to the Darboux-Froda Theorem (cf. Theorem 8-2 in [1]), the function

$$y_0 \to \mathcal{I}_{\hat{g}}(y_0, y_1^*; B_1, B_2)$$
 (A.19)

has at most countably infinite points of discontinuity on $(0, \infty)$. Further,

$$\mathcal{I}_{g_{\lambda}}(y_0, y_1^*; B_1, B_2) = \mathcal{I}_{\hat{g}}(y_0 Z_{\lambda}(T), y_1^*; B_1, B_2) \quad Q\text{-a.s}$$

and the random variable $\tilde{Z}_{\lambda}(T)$ has a continuous distribution on $(0, \infty)$. As the points of discontinuity of (A.19) are a null set with respect to the Lebesgue measure $\mathcal{L}(0, \infty)$, this implies

$$Q\left(y_0 \to \mathcal{I}_{g_\lambda}(y_0, y_1^*; B_1, B_2) \text{ is continuous at } y_0 = y_0^*\right) = 1.$$

(ii) Since $\lambda \in \mathcal{D}'$, we know that

$$y \to \mathbb{E}\left[g_{\lambda}(\mathcal{I}_{g_{\lambda}}(y; B_1, B_2))\right]' \Delta y$$

is use at $y = y^* = (y_0^*, y_1^*)'$ for all $\Delta y \in \mathbb{R} \times [0, \infty)$. Choosing $\Delta y = (-1, 0)'$ implies

$$\limsup_{y \to y^*} \mathbb{E} \left[\mathcal{I}_{g_{\lambda}}(y; B_1, B_2) \tilde{Z}_{\lambda}(T) \right] = \limsup_{y \to y^*} \mathbb{E} \left[g_{\lambda} \left(\mathcal{I}_{g_{\lambda}}(y; B_1, B_2) \right) \right]' \Delta y$$
$$\stackrel{usc}{\leq} \mathbb{E} \left[g_{\lambda} \left(\mathcal{I}_{g_{\lambda}}(y^*; B_1, B_2) \right) \right]' \Delta y = \mathbb{E} \left[\mathcal{I}_{g_{\lambda}}(y^*; B_1, B_2) \tilde{Z}_{\lambda}(T) \right] \stackrel{\lambda \in \mathcal{D}'}{=} v_0 \quad (A.20)$$

Hence, there must exist a $y_0^+ < y_0^*$, as described in the statement of this lemma, since otherwise

$$\limsup_{y \to y^*} \mathbb{E} \left[\mathcal{I}_{g_{\lambda}}(y; B_1, B_2) \tilde{Z}_{\lambda}(T) \right] \ge \limsup_{y_0 \uparrow y_0^*} \mathbb{E} \left[\mathcal{I}_{g_{\lambda}}(y_0, y_1^*; B_1, B_2) \tilde{Z}_{\lambda}(T) \right] = \infty,$$

which contradicts (A.20).

г		п.
L		н

Proof of Theorem 3.4.5.

 $(\tilde{B}) \Rightarrow (\tilde{A})$: This is the statement of Lemma 3.4.3.

 $(\tilde{B}) \Rightarrow (\tilde{C})$: By Lemma 3.4.3 and (3.6), we have for any $\lambda \in \mathcal{D}$:

$$\tilde{\Phi}_{\lambda^*}(v_0) \stackrel{L.3.4.3}{=} \Phi(v_0) \stackrel{(3.6)}{\leq} \tilde{\Phi}_{\lambda}(v_0).$$

 $(\tilde{B}) \Rightarrow (\tilde{E}):$

Due to condition (\tilde{B}) , the additional drift from the portfolio $\pi_{\lambda^*} \in \Lambda(v_0, K, B_1, B_2, g)$ in \mathcal{M}_{λ^*} is zero $\mathcal{L}[0, T] \otimes Q$ -a.e. and therefore, $V_{\lambda^*}^{v_0, \pi_{\lambda^*}}(t) = V^{v_0, \pi_{\lambda^*}}(t) \mathcal{L}[0, T] \otimes Q$ -a.s. However, for any $\lambda \in \mathcal{D}$, we can use Itô's product rule to see

$$\begin{split} d(Z_{\lambda}(t)V^{v_{0},\pi_{\lambda^{*}}}(t)) &= Z_{\lambda}(t)dV^{v_{0},\pi_{\lambda^{*}}}(t) + V^{v_{0},\pi_{\lambda^{*}}}(t)dZ_{\lambda}(t) + d\langle Z_{\lambda}, V^{v_{0},\pi_{\lambda^{*}}}\rangle_{t} \\ &= -\tilde{Z}_{\lambda}(t)V^{v_{0},\pi_{\lambda^{*}}}(t) \cdot \left(\left(r(t) + \delta_{K}(\lambda(t)) \right) dt + \gamma_{\lambda}(t) dW(t) \right) \\ &+ \tilde{Z}_{\lambda}(t)V^{v_{0},\pi_{\lambda^{*}}}(t) \cdot \left(r(t) + \left[\mu(t) - r(t)\mathbb{1} \right]'\pi_{\lambda^{*}}(t) \right) dt \\ &+ \tilde{Z}_{\lambda}(t)V^{v_{0},\pi_{\lambda^{*}}}(t)\pi_{\lambda^{*}}(t)'\Sigma(t)dW(t) \\ &- \tilde{Z}_{\lambda}(t)V^{v_{0},\pi_{\lambda^{*}}}(t) \underbrace{\gamma_{\lambda}(t)'\Sigma(t)'\pi_{\lambda^{*}}(t)}_{= \left[\mu(t) - r(t)\mathbb{1} + \lambda(t) \right]'\pi_{\lambda^{*}}(t)} dt \\ &= -\tilde{Z}_{\lambda}(t)V^{v_{0},\pi_{\lambda^{*}}}(t) \left(\delta_{K}(\lambda(t)) + \lambda(t)'\pi_{\lambda^{*}}(t) \right) dt \\ &+ \tilde{Z}_{\lambda}(t)V^{v_{0},\pi_{\lambda^{*}}}(t) \left(\pi_{\lambda^{*}}(t)'\Sigma(t) - \gamma_{\lambda}(t)' \right) dW(t) \end{split}$$

Integrating this SDE then shows that

$$\tilde{Z}_{\lambda}(t)V^{v_0,\pi_{\lambda^*}}(t) + \int_0^t \tilde{Z}_{\lambda}(s)V^{v_0,\pi_{\lambda^*}}(s)\underbrace{\left(\delta_K(\lambda(s)) + \lambda(s)'\pi_{\lambda^*}(s)\right)}_{\ge 0 \text{ by L. 2.1.17, (iii)}} ds$$
(A.21)

$$= v_0 + \int_0^t \tilde{Z}_{\lambda}(s) V^{v_0, \pi_{\lambda^*}}(s) \big(\pi_{\lambda^*}(s)' \Sigma(s) - \gamma_{\lambda}(s) \big)' dW(s).$$
(A.22)

As (A.21) is non-negative, so is (A.22). Thus, both (A.21) and (A.22) are non-negative local martingales and therefore supermartingales. Hence, we obtain from the supermartingale property

$$\mathbb{E}[\mathcal{I}_{g_{\lambda^*}}(y^*; B_1, B_2)\tilde{Z}_{\lambda}(T)] = \mathbb{E}[V_{\lambda^*}^{v_0, \pi_{\lambda^*}}(T)\tilde{Z}_{\lambda}(T)] = \mathbb{E}[V^{v_0, \pi_{\lambda^*}}(T)\tilde{Z}_{\lambda}(T)]$$

$$\leq \mathbb{E}[V^{v_0, \pi_{\lambda^*}}(T)\tilde{Z}_{\lambda}(T) + \int_0^T \underbrace{\tilde{Z}_{\lambda}(t)V^{v_0, \pi_{\lambda^*}}(t)}_{\geq 0} \underbrace{\delta_K(\lambda(t)) + \lambda(t)'\pi(t)}_{\geq 0 \text{ by L. 2.1.17, (iii)}} dt] \leq v_0.$$

 $(\tilde{E}) \Rightarrow (\tilde{B})$:

Follows from exactly the same arguments as Theorem 9.1 in [17] with the choice

$$B = V_{\lambda^*}^{v_0, \pi_{\lambda^*}}(T) = \mathcal{I}_{g_{\lambda^*}}(y^*; B_1, B_2).$$

 $(\tilde{C}) \Rightarrow (\tilde{D})$: Let λ^* satisfy Condition (\tilde{C}) , let $\lambda \in \mathcal{D}$ and $D \in C_{\lambda}(v_0, g)$ be an admissible terminal wealth for $(\tilde{\mathbf{P}}_{\lambda})$. Then,

$$\mathbb{E}\left[D\tilde{Z}_{\lambda}(T)\right] = v_0 \text{ and } \mathbb{E}[g(D; B_2)] \le 0.$$

Thus, we obtain for any $y = (y_0, y_1)' \in [0, \infty)$

$$\mathbb{E}[U(D; B_{1}, B_{2})] = \mathbb{E}\left[\underbrace{U(D; B_{1}, B_{2}) - y_{0}\tilde{Z}_{\lambda}(T)D - y_{1}g(D; B_{2})}_{\leq U_{g_{\lambda}}^{*}(y; B_{1}, B_{2})}\right] + y_{0}v_{0} + \underbrace{y_{0}\left(\mathbb{E}\left[\tilde{Z}_{\lambda}(T)D\right] - v_{0}\right)}_{=0} + \underbrace{\mathbb{E}\left[y_{1}g(D; B_{2})\right]}_{\leq 0}$$

$$\leq \mathbb{E}[U_{g_{\lambda}}^{*}(y; B_{1}, B_{2})] + y_{0}v_{0}. \tag{A.23}$$

Noting that the right-hand side is independent of D, we can take the supremum over all terminal wealths $D \in C_{\lambda}(v_0, g(\cdot, B_2))$ admissible for $(\tilde{\mathbf{P}}_{\lambda})$ and receive

$$\tilde{\Phi}_{\lambda}(v_0) \leq \mathbb{E}[U_{q_{\lambda}}^*(y; B_1, B_2)] + y_0 v_0.$$

Due to Theorem 3.4.4, the choice of $\lambda = \lambda^*$, $y = y^*$ and $D = \mathcal{I}_{g_{\lambda^*}}(y^*; B_1, B_2)$ achieves equality in (A.23). Combining this fact with Condition (\tilde{C}) finally gives

$$\mathbb{E}[U_{g_{\lambda}}^{*}(y^{*}; B_{1}, B_{2})] + y_{0}^{*}v_{0} \geq \tilde{\Phi}_{\lambda}(v_{0}) \geq \tilde{\Phi}_{\lambda^{*}}(v_{0}) = \mathbb{E}[U_{g_{\lambda^{*}}}^{*}(y^{*}; B_{1}, B_{2})] + y_{0}^{*}v_{0}$$

$$\Rightarrow \mathbb{E}[U_{g_{\lambda}}^{*}(y^{*}; B_{1}, B_{2})] \geq \mathbb{E}[U_{g_{\lambda^{*}}}^{*}(y^{*}; B_{1}, B_{2})].$$

 $(\tilde{D}) \Rightarrow (\tilde{B}):$

Let $\lambda^* \in \mathcal{D}'$ satisfy Condition (\tilde{D}) . Let $\pi := \pi_{\lambda^*}$ be the portfolio process that attains $\mathcal{I}_{g_{\lambda^*}}(y^*; B_1, B_2)$ in \mathcal{M}_{λ^*} . We now need to show two things:

$$\pi(t) \in K \quad \mathcal{L}[0,T] \otimes Q - a.e., \tag{A.24}$$

$$[\pi(t)'\lambda^*(t) + \delta_K(\lambda^*(t))] = 0 \quad \mathcal{L}[0,T] \otimes Q - a.e..$$
(A.25)

Before doing this, we need to introduce a variety of different notations. Let $\nu \in \mathcal{D} \cup \{-\lambda^*\}$ be arbitrary but fixed. Let $\epsilon \in (0, 1), t \in [0, T]$ and $n \in \mathbb{N}$. We define:

$$\hat{\delta}^{(\nu)}(\lambda^*(t)) := \begin{cases} -\delta_K(\lambda^*(t)), & \nu = -\lambda^*\\ \delta_K(\nu(t)), & \text{else} \end{cases}$$

$$L_t := L_t^{(\nu)} := \int_0^t \hat{\delta}^{(\nu)}(\lambda^*(s))ds$$

$$W_{\lambda^*}(t) := W(t) + \int_0^t \gamma_{\lambda^*}(s)ds$$

$$N_t := N_t^{(\nu)} := \int_0^t \left(\Sigma(s)^{-1}\nu(s)\right)' dW_{\lambda^*}(s)$$

$$\tau_n := T \wedge \inf \left\{ t \in [0,T] \mid |L_t| \ge n, \text{ or } |N_t| \ge n, \text{ or } or \ \int_0^t ||\Sigma(s)^{-1}\nu(s)||^2 ds \ge n, or \ \int_0^t ||\gamma(t) + \Sigma(t)^{-1}\lambda^*(s)||^2 ds \ge n, or \ \int_0^t ||\gamma(t) + \Sigma(t)^{-1}\lambda^*(s)||^2 ds \ge n, or \ \int_0^t \left(\frac{V_{\lambda^*}^{\nu_0,\pi}(s)}{P_0^{\lambda^*}(s)}\right)^2 ||\Sigma(s)^{-1}\nu(s) + (L_s + N_s)\Sigma(s)'\pi(s)||^2 ds \ge n \right\}$$

•
$$\lambda_{\epsilon,n}(t) := \lambda^*(t) + \epsilon \nu(t) \mathbb{1}_{\{t \le \tau_n\}} \Rightarrow \lambda_{\epsilon,n} \in \mathcal{D}$$

The ensuing proof is split into three parts and is structured as follows. In Part 1 and Part 2 we prove (A.24) and (A.25) by showing the analogue of (9.14) and (9.15) from [17]:

$$0 \stackrel{(\tilde{D})}{\leq} \limsup_{\epsilon \downarrow 0} \frac{1}{\epsilon} \Big(\mathbb{E} \Big[U_{g_{\lambda_{\epsilon,n}}}^{*}(y^{*}; B_{1}, B_{2}) - U_{g_{\lambda^{*}}}^{*}(y^{*}; B_{1}, B_{2}) \Big] \Big)$$

$$\stackrel{\text{Part 1}}{\leq} y_{0}^{*} \mathbb{E} \Big[\mathcal{I}_{g_{\lambda^{*}}}(y^{*}; B_{1}, B_{2}) \tilde{Z}_{\lambda^{*}}(T) (L_{\tau_{n}} + N_{\tau_{n}}) \Big]$$

$$= y_{0}^{*} \mathbb{E} \Big[V_{\lambda^{*}}^{v_{0,\pi}}(T) \tilde{Z}_{\lambda^{*}}(T) (L_{\tau_{n}} + N_{\tau_{n}}) \Big]$$

$$\stackrel{\text{Part 2}}{\leq} y_{0}^{*} \mathbb{E} \Big[Z_{\lambda^{*}}(\tau_{n}) \int_{0}^{\tau_{n}} \underbrace{\frac{V_{\lambda^{*}}^{v_{0,\pi}}(t)}{P_{0}^{0}(t)}}_{=:\tilde{V}_{\lambda^{*}}^{v_{0,\pi}}(t)} [\pi(t)'\nu(t) + \hat{\delta}^{(\nu)}(\lambda^{*}(t))] dt \Big]$$

$$(A.26)$$

Part 1 closely follows the argument of the proof of Theorem 10.1 in [17]. In the end, the main objective of Part 1 is showing that we can apply Fatou's Lemma to interchange limsup and expectation by using properties of the conveniently stopped processes. Part 2 closely follows the argument in Step 5 of the proof of Theorem 9.1 in [17]. In Part 3, we choose arbitrary $\nu \in \mathcal{D}$ to show (A.24) and $\nu := -\lambda^*$ to show (A.25). This corresponds to Step 3 of the proof of Theorem 9.1 in [17].

We will elaborate further on the third inequality in (A.26), the last equality in (A.26) and the concluding argument mentioned previously in chronological order. This splits the proof into three parts.

Part 1: We start off by proving a useful set of inequalities that will form an integral part of this proof. First of all, in either case $\nu \in \mathcal{D}$ and $\nu = -\lambda^*$ we have for every $t \in [0, \tau_n]$:

$$\delta_K(\lambda_{\epsilon,n}(t)) - \delta_K(\lambda^*(t)) = \delta_K(\lambda^*(t) + \epsilon\nu(t)) - \delta_K(\lambda^*(t)) \le \epsilon \hat{\delta}^{(\nu)}(\lambda^*(t))$$
(A.27)

Case $\nu \in \mathcal{D}$:

$$\delta_K(\lambda^*(t) + \epsilon\nu(t)) - \delta_K(\lambda^*(t)) \le \delta_K(\lambda^*(t)) + \epsilon\delta_K(\nu(t)) - \delta_K(\lambda^*(t)) = \epsilon\hat{\delta}^{(\nu)}(\lambda^*(t)),$$

by the positive homogeneity and subadditivity of δ_K .

Case $\nu = -\lambda^*$:

$$\delta_K(\lambda^*(t) + \epsilon\nu(t)) - \delta_K(\lambda^*(t)) = (1 - \epsilon)\delta_K(\lambda^*(t)) - \delta_K(\lambda^*(t)) = -\epsilon\delta_K(\lambda^*(t)) = \epsilon\hat{\delta}^{(\nu)}(\lambda^*(t))$$

Furthermore, note that on $\{t \leq \tau_n\}$

$$\gamma_{\lambda_{\epsilon,n}}(t) - \gamma_{\lambda^*}(t) = \epsilon \Sigma(t)^{-1} \nu(t)$$

$$\Rightarrow \|\gamma_{\lambda_{\epsilon,n}}(t)\|^2 - \|\gamma_{\lambda^*}(t)\|^2 = 2\epsilon \big(\Sigma(t)^{-1} \nu(t)\big)' \gamma_{\lambda^*}(t) + \epsilon^2 \|\Sigma(t)^{-1} \nu(t)\|^2$$
(A.28)

and $\gamma_{\lambda_{\epsilon,n}}(t) = \gamma_{\lambda^*}(t)$ on $\{t > \tau_n\}$.

From these findings we can derive the following inequalities:

$$\frac{\tilde{Z}_{\lambda_{\epsilon,n}}(T)}{\tilde{Z}_{\lambda^*}(T)} \ge \underbrace{\exp\left(-\epsilon(L_{\tau_n} + N_{\tau_n}) - \frac{\epsilon^2}{2} \int_0^{\tau_n} \|\Sigma(t)^{-1}\nu(t)\|^2 dt\right)}_{=:\alpha_n^{(\nu)}(\epsilon)} \ge e^{-3\epsilon n} \ge e^{-3n} \qquad (A.29)$$

and indeed, upon more detailed inspection of (A.29) we find that

$$\begin{split} \frac{\tilde{Z}_{\lambda_{\epsilon,n}}(T)}{\tilde{Z}_{\lambda^*}(T)} &= \exp\left(-\int_0^{\tau_n} \underbrace{\delta_K(\lambda^*(t) + \epsilon\nu(t)) - \delta_K(\lambda^*(t))}_{\leq \epsilon \delta^{(\nu)}(\lambda^*(t)) \text{ by } (A.27)} dt \\ &- \frac{1}{2} \int_0^{\tau_n} \|\gamma_{\lambda_{\epsilon,n}}(t)\|^2 - \|\gamma_{\lambda^*}(t)\|^2 dt - \int_0^{\tau_n} (\gamma_{\lambda_{\epsilon,n}}(t) - \gamma_{\lambda^*}(t))' dW(t)\right) \\ \stackrel{(A.28)}{\geq} \exp\left(-\epsilon \int_0^{\tau_n} \delta^{(\nu)}(\lambda^*(t)) dt - \frac{\epsilon^2}{2} \int_0^{\tau_n} \|\Sigma(t)^{-1}\nu(t)\|^2 dt \\ &- \epsilon \int_0^{\tau_n} (\Sigma(t)^{-1}\nu(t))' \gamma_{\lambda^*}(t) dt - \epsilon \int_0^{\tau_n} (\Sigma(t)^{-1}\nu(t))' dW(t)\right) \\ &= \exp\left(-\epsilon \underbrace{\int_0^{\tau_n} \delta^{(\nu)}(\lambda^*(t)) dt}_{=L_{\tau_n}} - \frac{\epsilon^2}{2} \int_0^{\tau_n} \|\Sigma(t)^{-1}\nu(t)\|^2 dt \\ &- \epsilon \underbrace{\int_0^{\tau_n} (\Sigma(t)^{-1}\nu(t))' dW_{\lambda^*}(t)}_{=N_{\tau_n}}\right) \\ &= \exp\left(-\epsilon(L_{\tau_n} + N_{\tau_n}) - \frac{\epsilon^2}{2} \int_0^{\tau_n} \|\Sigma(t)^{-1}\nu(t)\|^2 dt\right) \\ &\geq \exp\left(-3\epsilon n\right) \ge \exp\left(-3n\right), \end{split}$$

where the second-to-last inequality follows from the choice of the stopping time τ_n . Note that the expression $\alpha_n^{(\nu)}(\cdot)$ is differentiable in ϵ with

$$\frac{\partial}{\partial \epsilon} \alpha_n^{(\nu)}(\epsilon) \Big|_{\epsilon=0} = \left(\alpha_n^{(\nu)} \right)'(0) = \lim_{\epsilon \to 0} \frac{\alpha_n^{(\nu)}(\epsilon) - 1}{\epsilon} = -(L_{\tau_n} + N_{\tau_n}).$$
(A.30)

We go on to prove the second inequality in (A.26) by validating the following inequality:

$$\limsup_{\epsilon \downarrow 0} \frac{1}{\epsilon} \left(\mathbb{E} \left[U_{g_{\lambda_{\epsilon,n}}}^*(y^*; B_1, B_2) - U_{g_{\lambda^*}}^*(y^*; B_1, B_2) \right] \right)$$

$$\leq y_0^* \mathbb{E} \left[\mathcal{I}_{g_{\lambda^*}}(y^*; B_1, B_2) \tilde{Z}_{\lambda^*}(T) (L_{\tau_n} + N_{\tau_n}) \right]$$
(A.31)

We first derive a pointwise majorant with finite expectation to justify the application of Fatou's lemma:

$$\frac{1}{\epsilon} \left(U_{g_{\lambda_{\epsilon,n}}}^{*}(y^{*}; B_{1}, B_{2}) - U_{g_{\lambda^{*}}}^{*}(y^{*}; B_{1}, B_{2}) \right) \\
= \frac{1}{\epsilon} \left(U(\mathcal{I}_{g_{\lambda_{\epsilon,n}}}(y^{*}; B_{1}, B_{2})) - y_{0}^{*}\tilde{Z}_{\lambda_{\epsilon,n}}(T)\mathcal{I}_{g_{\lambda_{\epsilon,n}}}(y^{*}; B_{1}, B_{2}) - y_{1}^{*}g\left(\mathcal{I}_{g_{\lambda_{\epsilon,n}}}(y^{*}; B_{1}, B_{2})\right) \right) \\
- \frac{1}{\epsilon} \underbrace{\sup_{B_{1} \leq x \leq B_{2}} \left(U(x) - y_{0}^{*}\tilde{Z}_{\lambda^{*}}(T)x - y_{1}^{*}g(x) \right)}_{\geq U(x) - y_{0}^{*}\tilde{Z}_{\lambda^{*}}(T)x - y_{1}^{*}g(x)} \underbrace{\int_{Z_{\lambda^{*}}(x)}^{E_{\lambda^{*}}(x)} \int_{Z_{\lambda^{*}}(x)}^{E_{\lambda^{*}}(x)} \int_{Z_{\lambda^{*}}(x)$$

Again, recalling (A.29) yields for any $0 \le B_1 \le x \le B_2$

$$g_{\lambda_{\epsilon,n}}(x) = \begin{pmatrix} -\tilde{Z}_{\lambda_{\epsilon,n}}(T)x\\ -g(x) \end{pmatrix} \le \begin{pmatrix} -e^{-3\epsilon n}\tilde{Z}_{\lambda^*}(T)x\\ -g(x) \end{pmatrix} := \hat{g}(x)$$
(A.33)

and thus by Lemma 2.1.38, (ii), $\mathcal{I}_{g_{\lambda_{\epsilon,n}}}(y^*; B_1, B_2) \leq \mathcal{I}_{\hat{g}}(y^*; B_1, B_2)$. However, we also have

$$U_{\hat{g}}^{*}(y^{*}; B_{1}, B_{2}) = \sup_{B_{1} \le x \le B_{2}} \left(U(x) - y_{0}^{*} e^{-3\epsilon n} \tilde{Z}_{\lambda^{*}}(T) x - y_{1}^{*} g(x) \right)$$
$$= U_{g_{\lambda^{*}}}^{*} \left(y_{0}^{*} e^{-3\epsilon n}, y_{1}^{*}; B_{1}, B_{2} \right)$$

and in particular $\mathcal{I}_{\hat{g}}(y^*; B_1, B_2) = \mathcal{I}_{g_{\lambda^*}}(y_0^* e^{-3\epsilon n}, y_1^*; B_1, B_2)$. Combining these observations with (A.32), we have for any fixed $0 < \epsilon^+$ and for all $0 < \epsilon \le \epsilon^+$

$$\frac{1}{\epsilon} \left(U_{g_{\lambda_{\epsilon,n}}}^{*}(y^{*}; B_{1}, B_{2}) - U_{g_{\lambda^{*}}}^{*}(y^{*}; B_{1}, B_{2}) \right) \\
\stackrel{(A.33)}{\leq} y_{0}^{*} \mathcal{I}_{\hat{g}}(y^{*}; B_{1}, B_{2}) \tilde{Z}_{\lambda^{*}}(T) \frac{1}{\epsilon} \left(1 - \alpha_{n}^{(\nu)}(\epsilon) \right) \\
= y_{0}^{*} \mathcal{I}_{g_{\lambda^{*}}} \left(y_{0}^{*} e^{-3\epsilon n}, y_{1}^{*}; B_{1}, B_{2} \right) \tilde{Z}_{\lambda^{*}}(T) \frac{1}{\epsilon} \left(1 - \alpha_{n}^{(\nu)}(\epsilon) \right) \\
\stackrel{L. 2.1.38, \ (i)}{\leq} y_{0}^{*} \mathcal{I}_{g_{\lambda^{*}}} \left(y_{0}^{*} e^{-3\epsilon^{+}n}, y_{1}^{*}; B_{1}, B_{2} \right) \tilde{Z}_{\lambda^{*}}(T) \frac{1}{\epsilon} \left(1 - \alpha_{n}^{(\nu)}(\epsilon) \right) \\
\leq y_{0}^{*} \mathcal{I}_{g_{\lambda^{*}}} \left(y_{0}^{*} e^{-3\epsilon^{+}n}, y_{1}^{*}; B_{1}, B_{2} \right) \tilde{Z}_{\lambda^{*}}(T) \underbrace{\sup_{0 < \epsilon < 1} \left(\frac{1}{\epsilon} \left(1 - e^{-3\epsilon n} \right) \right)}_{<\infty} \quad (A.35)$$

Since $\lambda^* \in \mathcal{D}'$, the auxiliary Lemma A.2.1, (ii) implies that we can choose ϵ^+ small enough such that

$$\mathbb{E}\left[y_0^*\mathcal{I}_{g_{\lambda^*}}\left(y_0^*e^{-3\epsilon^+n}, y_1^*; B_1, B_2\right)\tilde{Z}_{\lambda^*}(T)\right] < \infty$$

Then (A.35) yields an integrable majorant for all $0 < \epsilon \leq \epsilon^+$ and we take expectations and

apply Fatou's lemma to obtain

$$\begin{split} & \limsup_{\epsilon \downarrow 0} \frac{1}{\epsilon} \Big(\mathbb{E} \left[U_{g_{\lambda\epsilon,n}}^* \left(y^*; B_1, B_2 \right) - U_{g_{\lambda^*}}^* \left(y^*; B_1, B_2 \right) \right] \Big) \\ \stackrel{(A.34)}{\leq} & \limsup_{\epsilon \downarrow 0} \mathbb{E} \left[y_0^* \mathcal{I}_{g_{\lambda^*}} \left(y_0^* e^{-3\epsilon n}, y_1^*; B_1, B_2 \right) \tilde{\mathcal{Z}}_{\lambda^*}(T) \frac{1}{\epsilon} \left(1 - \alpha_n^{(\nu)}(\epsilon) \right) \right] \\ \stackrel{\text{Fatou}}{\leq} & y_0^* \mathbb{E} \left[\tilde{\mathcal{Z}}_{\lambda^*}(T) \limsup_{\epsilon \downarrow 0} \left(\underbrace{\mathcal{I}_{g_{\lambda^*}} \left(y_0^* e^{-3\epsilon n}, y_1^*; B_1, B_2 \right)}_{\substack{\epsilon \to 0 \\ e \to 0 \\ \mathcal{I}_{g_{\lambda^*}} \left(y^*; B_1, B_2 \right) \text{ by } L.A.2.1, (i)} \cdot \underbrace{\frac{1}{\epsilon} \left(1 - \alpha_n^{(\nu)}(\epsilon) \right)}_{\substack{\epsilon \to 0 \\ \mathcal{L}_{\tau_n} + N_{\tau_n} \text{ by } (A.30)} \right) \right] \\ \stackrel{(A.30)}{=} & y_0^* \mathbb{E} \left[\tilde{\mathcal{Z}}_{\lambda^*}(T) \mathcal{I}_{g_{\lambda^*}} \left(y^*; B_1, B_2 \right) \left(L_{\tau_n} + N_{\tau_n} \right) \right], \end{split}$$

This proves (A.31) and hence, the second inequality in (A.26).

Part 2: We continue by proving the last inequality in (A.26). Let $V_{\lambda^*}^{v_0,\pi}$ be the wealth process of π in \mathcal{M}_{λ^*} and set $\tilde{V}_{\lambda^*}^{v_0,\pi}(t) := P_0^{\lambda^*}(t)^{-1} \cdot V_{\lambda^*}^{v_0,\pi}(t)$. Then, through an application of Itô's product rule and a change of the diffusion from dW(t) to $dW_{\lambda^*}(t) = dW(t) + \gamma_{\lambda^*}(t)dt$ we see:

$$d(\tilde{V}_{\lambda^{*}}^{v_{0},\pi}(t)) = d(P_{0}^{\lambda^{*}}(t)^{-1} \cdot V_{\lambda^{*}}^{v_{0},\pi}(t)) = P_{0}^{\lambda^{*}}(t)^{-1} dV_{\lambda^{*}}^{v_{0},\pi}(t) + V_{\lambda^{*}}^{v_{0},\pi}(t) d(P_{0}^{\lambda^{*}}(t)^{-1})$$

$$\stackrel{(3.5)}{=} \tilde{V}_{\lambda^{*}}^{v_{0},\pi}(t) \Big([r(t) + (\mu(t) - r(t)\mathbb{1})'\pi(t)] dt + \pi(t)'\Sigma(t)dW(t) \Big)$$

$$+ \tilde{V}_{\lambda^{*}}^{v_{0},\pi}(t) \left[\delta_{K}(\lambda^{*}(t)) + \pi(t)'\lambda^{*}(t) \right] dt - \tilde{V}_{\lambda^{*}}^{v_{0},\pi}(t) [r(t) + \delta_{K}(\lambda^{*}(t))] dt$$

$$= \tilde{V}_{\lambda^{*}}^{v_{0},\pi}(t)\pi(t)'\Sigma(t)dW_{\lambda^{*}}(t)$$

Furthermore,

$$d(\tilde{V}_{\lambda^{*}}^{v_{0},\pi}(t) \cdot (L_{t} + N_{t}))$$

$$= \tilde{V}_{\lambda^{*}}^{v_{0},\pi}(t) (dL_{t} + dN_{t}) + (L_{t} + N_{t}) d\tilde{V}_{\lambda^{*}}^{v_{0},\pi}(t) + d\langle \tilde{V}_{\lambda^{*}}^{v_{0},\pi}, N \rangle_{t}$$

$$= \tilde{V}_{\lambda^{*}}^{v_{0},\pi}(t) (dL_{t} + (\Sigma(t)^{-1}\nu(t))' dW_{\lambda^{*}}(t)) + \tilde{V}_{\lambda^{*}}^{v_{0},\pi}(t) (L_{t} + N_{t})\pi(t)'\Sigma(t) dW_{\lambda^{*}}(t) + \tilde{V}_{\lambda^{*}}^{v_{0},\pi}(t)\pi(t)'\nu(t) dt$$

$$= \tilde{V}_{\lambda^{*}}^{v_{0},\pi}(t) \left(\left[(\Sigma(t)^{-1}\nu(t))' + (L_{t} + N_{t})\pi(t)'\Sigma(t) \right] dW_{\lambda^{*}}(t) + \pi(t)'\nu(t) dt + dL_{t} \right)$$
(A.36)

Define the probability measure $\tilde{Q}_{\lambda^*}^{(n)}$ via its Radon-Nikoydm derivative:

$$\left. \frac{d\tilde{Q}_{\lambda^*}^{(n)}}{dQ} \right|_{\mathcal{F}_T} := Z_{\lambda^*}(\tau_n)$$

Then, according to Girsanov's theorem, the process

$$W_{\lambda^*}^{(n)}(t) := W(t) + \int_0^t \gamma_{\lambda^*}(s) \mathbb{1}_{\{s \le \tau_n\}} ds$$

is a Wiener process with respect to $\tilde{Q}_{\lambda^*}^{(n)}$. From now on, let $\mathbb{E}_{\tilde{Q}_{\lambda^*}^{(n)}}[X]$ denote the expectation of a random variable X w.r.t $\tilde{Q}_{\lambda^*}^{(n)}$. Then, by the choice of τ_n ,

$$\mathbb{E}_{\tilde{Q}_{\lambda^{*}}^{(n)}} \left[\int_{0}^{\tau_{n}} \tilde{V}_{\lambda^{*}}^{v_{0},\pi}(t) \left[(\Sigma(t)^{-1}\nu(t))' + (L_{t}+N_{t})\pi(t)'\Sigma(t) \right] dW_{\lambda^{*}}(t) \right] \\
= \mathbb{E}_{\tilde{Q}_{\lambda^{*}}^{(n)}} \left[\int_{0}^{T} \tilde{V}_{\lambda^{*}}^{v_{0},\pi}(t) \left[(\Sigma(t)^{-1}\nu(t))' + (L_{t}+N_{t})\pi(t)'\Sigma(t) \right] \mathbb{1}_{\{t \leq \tau_{n}\}} dW_{\lambda^{*}}(t) \right] \\
= \mathbb{E}_{\tilde{Q}_{\lambda^{*}}^{(n)}} \left[\int_{0}^{T} \tilde{V}_{\lambda^{*}}^{v_{0},\pi}(t) \left[(\Sigma(t)^{-1}\nu(t))' + (L_{t}+N_{t})\pi(t)'\Sigma(t) \right] \mathbb{1}_{\{t \leq \tau_{n}\}} dW_{\lambda^{*}}^{(n)}(t) \right] \\
= 0.$$
(A.37)

Furthermore, integrating over (A.36) and taking expectations with respect to $\tilde{Q}_{\lambda^*}^{(n)}$ results in

$$\mathbb{E}[\tilde{Z}_{\lambda^{*}}(\tau_{n})V_{\lambda^{*}}^{v_{0},\pi}(\tau_{n})(L_{\tau_{n}}+N_{\tau_{n}})] = \mathbb{E}[Z_{\lambda^{*}}(\tau_{n})\tilde{V}_{\lambda^{*}}^{v_{0},\pi}(\tau_{n})(L_{\tau_{n}}+N_{\tau_{n}})] \\
= \mathbb{E}_{\tilde{Q}_{\lambda^{*}}^{(n)}}[\tilde{V}_{\lambda^{*}}^{v_{0},\pi}(\tau_{n})(L_{\tau_{n}}+N_{\tau_{n}})] \\
\stackrel{(A.36)}{=} \mathbb{E}_{\tilde{Q}_{\lambda^{*}}^{(n)}}\left[\int_{0}^{\tau_{n}}\tilde{V}_{\lambda^{*}}^{v_{0},\pi}(t)[\pi(t)'\nu(t)dt+dL_{t}]\right] \\
+\mathbb{E}_{\tilde{Q}_{\lambda^{*}}^{(n)}}\left[\int_{0}^{\tau_{n}}\tilde{V}_{\lambda^{*}}^{v_{0},\pi}(t)[(\Sigma(t)^{-1}\nu(t))'+(L_{t}+N_{t})\pi(t)'\Sigma(t)]dW_{\lambda^{*}}(t)\right] \\
\stackrel{(A.37)}{=} \mathbb{E}_{\tilde{Q}_{\lambda^{*}}^{(n)}}\left[\int_{0}^{\tau_{n}}\tilde{V}_{\lambda^{*}}^{v_{0},\pi}(t)[\pi(t)'\nu(t)dt+dL_{t}]\right] \\
= \mathbb{E}\left[Z_{\lambda^{*}}(\tau_{n})\int_{0}^{\tau_{n}}\tilde{V}_{\lambda^{*}}^{v_{0},\pi}(t)[\pi(t)'\nu(t)dt+dL_{t}]\right].$$
(A.38)

Furthermore, since $(\tilde{Z}_{\lambda^*}(t) \cdot V_{\lambda^*}^{v_0,\pi}(t))_{0 \le t \le T}$ is a *Q*-supermartingale, the tower property of conditional expectations ensures

$$\mathbb{E}\left[V_{\lambda^*}^{v_0,\pi}(T)\tilde{Z}_{\lambda^*}(T)(L_{\tau_n}+N_{\tau_n})\right] = \mathbb{E}\left[\mathbb{E}\left[V_{\lambda^*}^{v_0,\pi}(T)\tilde{Z}_{\lambda^*}(T)(L_{\tau_n}+N_{\tau_n})|\mathcal{F}_{\tau_n}\right]\right]$$
$$= \mathbb{E}\left[(L_{\tau_n}+N_{\tau_n})\mathbb{E}\left[V_{\lambda^*}^{v_0,\pi}(T)\tilde{Z}_{\lambda^*}(T)|\mathcal{F}_{\tau_n}\right]\right]$$
$$\leq E\left[V_{\lambda^*}^{v_0,\pi}(\tau_n)\tilde{Z}_{\lambda^*}(\tau_n)(L_{\tau_n}+N_{\tau_n})\right].$$

This, in combination with (A.38), proves the last inequality in (A.26).

Part 3: We can finally wrap up this proof. First, let $\nu \in \mathcal{D}$. Then, by the inequalities in (A.26)

$$0 \leq \mathbb{E}\left[Z_{\lambda^*}(\tau_n) \int_0^{\tau_n} \tilde{V}_{\lambda^*}^{v_0,\pi}(t) [\pi(t)'\nu(t) + \delta_K(\nu(t))] dt\right]$$

= $\mathbb{E}\left[Z_{\lambda^*}(\tau_n) \int_0^T \tilde{V}_{\lambda^*}^{v_0,\pi}(t) [\pi(t)'\nu(t) + \delta_K(\nu(t))] \mathbb{1}_{\{t \leq \tau_n\}} dt\right] \quad \forall n \in \mathbb{N}.$

Assume now that the set

$$A := \left\{ [\pi(t)'\nu(t) + \delta_K(\nu(t))] \mathbb{1}_{\{t \le \tau_n\}} < 0 \right\} \subset \Omega \times [0,T]$$

has positive product measure. Setting $\tilde{\nu} := \nu \mathbb{1}_A \in \mathcal{D}$, we see that

$$\mathbb{E}\left[Z_{\lambda^*}(\tau_n)\int_0^{\tau_n}\tilde{V}_{\lambda^*}^{v_0,\pi}(t)[\pi(t)'\tilde{\nu}(t)+\delta_K(\tilde{\nu}(t))]dt\right]$$
$$=\mathbb{E}\left[\underbrace{Z_{\lambda^*}(\tau_n)}_{>0}\int_0^{\tau_n}\underbrace{\tilde{V}_{\lambda^*}^{v_0,\pi}(t)}_{>0}\underbrace{[\pi(t)'\nu(t)+\delta_K(\nu(t))]\mathbb{1}_A}_{<0}dt\right]<0.$$

But as $\tilde{\nu} \in \mathcal{D}$, this contradicts (A.26). Hence,

$$[\pi(t)'\nu(t) + \delta_K(\nu(t))]\mathbb{1}_{\{t \le \tau_n\}} \ge 0 \quad \mathcal{L}[0,T] \otimes Q - a.e.$$
$$\downarrow n \to \infty$$
$$\pi(t)'\nu(t) + \delta_K(\nu(t)) \ge 0 \quad \mathcal{L}[0,T] \otimes Q - a.e.$$

as $\tau_n \to T \ Q - a.s.$ for $n \to \infty$. Since the choice of $\nu \in \mathcal{D}$ was arbitrary, this allows us to conclude (A.24) by virtue of Lemma 2.1.17, (iii). In particular, the choice of $\nu = \lambda^*$ yields

$$\pi(t)'\lambda^*(t) + \delta_K(\lambda^*(t)) \ge 0 \quad \mathcal{L}[0,T] \otimes Q - a.e..$$
(A.39)

On the other hand, by setting $\nu := -\lambda^*$, we obtain from (A.26)

$$0 \leq -\mathbb{E}\Big[Z_{\lambda^*}(\tau_n) \int_0^{\tau_n} \tilde{V}_{\lambda^*}^{v_0,\pi}(t) \underbrace{[\pi(t)'\lambda^*(t) + \delta_K(\lambda^*(t))]}_{\geq 0 \text{ by } (A.39)} dt\Big]$$

and this implies (A.25).

Proof of Theorem 3.4.6. This is a direct consequence of the remarks above the statement of the corollary. $\hfill \Box$

Proof of Lemma 3.5.2. Due to G being convex and strictly decreasing, $\operatorname{RRA}(t, z; y) \geq 0$. As $0 \in K$, we have $\delta_K(x) = -\inf_{v \in K} (v'x) \stackrel{v=0}{\geq} 0$. For given (t, z; y), we aim to find a minimizer $x = \lambda^*(t, z; y) \in \mathbb{R}^d$ of the function

$$f(x) := \operatorname{RRA}(t, z; y) \|\gamma + \Sigma^{-1}x\|^2 + \delta_K(x).$$

If RRA(t, z; y) = 0, then $x = \lambda^*(t, z; y) = 0 \in X_K$ minimizes f globally. Otherwise, f is strictly convex in x and $f(x) \to \infty$ as $||x|| \to \infty$. Hence, f has bounded level sets and is continuous on X_K (cf. Assumption 3.5.1). Thus, f attains its global minimum at some $x = \lambda^*(t, z; y) \in$ X_K . Hence, there exists a global minimizer $\lambda^*(t, z; y) \in X_K$ and it remains to show that this minimizer is uniformly bounded in (t, z; y).

Since Σ^{-1} is non-singular, the matrix $C = (\Sigma^{-1})' \Sigma^{-1}$ is symmetric and positive definite. By the symmetry of C, the spectral theorem implies that C has d orthonormal eigenvectors and dreal eigenvalues. Furthermore, as C is strictly positive definite, all of its eigenvalues are strictly positive. Thus, if $c_{-} > 0$ is the square root of the smallest eigenvalue of C, we obtain

$$\|\Sigma^{-1}x\|^2 = x'Cx \ge (c_{-})^2 \|x\|^2 \text{ for all } x \in \mathbb{R}^d.$$

Further, for a given minimizer $\lambda^*(t, z; y) \in X_K$, define

$$\nu(t, z; y) := \lambda^*(t, z; y) \cdot \mathbb{1}_{\{\|\lambda^*(t, z; y)\| \le \frac{2}{\alpha} \|\gamma\|\}}$$

Then, $\nu \in X_K$ and ν coincides with λ^* whenever $\|\lambda^*(t,z;y)\| \leq \frac{2}{c_-} \|\gamma\|$. Otherwise let $\|\lambda^*(t,z;y)\| > \frac{2}{c_-} \|\gamma\|$. Then, $\nu(t,z;y) = 0$ and an application of the reverse Cauchy-Schwarz inequality

$$\|x_1 + x_2\|^2 = \|x_1\|^2 + 2x_1'x_2 + \|x_2\|^2 \ge \|x_1\|^2 - 2\|x_1\|\|x_2\| + \|x_2\|^2 \quad \forall x_1, x_2 \in \mathbb{R}^d, \quad (A.40)$$
 yield

$$\begin{split} \frac{1}{2} \mathrm{RRA}(t,z;y) \|\gamma + \Sigma^{-1} \lambda^{*}(t,z;y)\|^{2} + \delta_{K}(\lambda^{*}(t,z;y)) \\ & \stackrel{(A.40)}{\geq} \quad \frac{1}{2} \mathrm{RRA}(t,z;y) \left(\|\gamma\|^{2} - 2\|\gamma\| \|\Sigma^{-1} \lambda^{*}(t,z;y)\| + \|\Sigma^{-1} \lambda^{*}(t,z;y)\|^{2} \right) + \delta_{K}(\lambda^{*}(t,z;y)) \\ & = \quad \frac{1}{2} \mathrm{RRA}(t,z;y) \left(\|\gamma\|^{2} + \|\Sigma^{-1} \lambda^{*}(t,z;y)\| (-2\|\gamma\| + \underbrace{\|\Sigma^{-1} \lambda^{*}(t,z;y)\|}_{\geq c_{-} \|\lambda^{*}(t,z;y)\|} \right) + \underbrace{\delta_{K}(\lambda^{*}(t,z;y))}_{\geq 0} \\ & \geq \quad \frac{1}{2} \mathrm{RRA}(t,z;y) \|\gamma\|^{2} \\ & \stackrel{\nu(t,z;y)=0}{=} \quad \frac{1}{2} \mathrm{RRA}(t,z;y) \|\gamma + \Sigma^{-1} \nu(t,z;y)\|^{2} + \delta_{K}(\nu(t,z;y)). \end{split}$$

Hence, $\nu(t, z; y)$ is also a minimizer of (3.11) and $\|\nu(t, z; y)\| \leq \frac{2}{c_{-}} \|\gamma\|$, for all (t, z; y), which concludes the proof.

Proof of Theorem 3.5.3. The uniform boundedness of $\lambda^*(t, z; y) \in X_K$ and the continuity of δ_K on X_K guarantee that $\|\lambda^*(t)\|$ and $|\delta_K(\lambda^*(t))|$ are Q-a.s. bounded on [0, T]. Due to measurable selection theorems (for example Corollary 3.48 in [77]) we may w.l.o.g. assume the mapping $(t, z) \to \lambda^*(t, z; y)$ to be Borel-measurable. Hence, the corresponding stochastic process λ^* is progressively measurable and thus an element of \mathcal{D} .

For convenience, we define the characteristic operator $\mathcal{H}^{\lambda}G$ of G with respect to $\lambda \in \mathcal{D}$ as

$$\mathcal{H}^{\lambda}G(t,z;y) = G_t(t,z;y) - \left(r + \delta_K(\lambda(t))\right) z G_z(t,z;y) + \frac{1}{2} \|\gamma + \Sigma^{-1}\lambda(t)\|^2 z^2 G_{zz}(t,z;y).$$

Note $\mathcal{H}^{\lambda}G(t,z;y) \geq 0$ for any $\lambda \in \mathcal{D}$ and $\mathcal{H}^{\lambda}G(t,z;y) = 0$ if $\lambda = \lambda^*$.

Let now $\lambda \in \mathcal{D}$ with $\|\lambda(t) - \lambda^*(t)\| \leq 1 \mathcal{L}[0, T] \otimes \mathbb{Q}$ -a.s., be arbitrary but fixed. Due to Lemma 3.5.2, we can assume that C > 0 from the polynomial growth condition was chosen large enough such that

 $\max\left(\|\gamma_{\lambda}(t)\|^{2}, \delta_{K}(\lambda^{*}(t)), \delta_{K}(\lambda(t))\right) \leq C \quad \mathcal{L}[0, T] \otimes \mathbb{Q}\text{-a.s.}.$

Finally, for any p > 0, we define the stopping times

$$\bar{\tau}_p = \inf\left\{ S \in [t,T] \; \Big| \int_t^S \left(G_z(s, \tilde{Z}_\lambda(s); y) \tilde{Z}_\lambda(s) \| \gamma_\lambda(s) \| \right)^2 ds \ge p \right\}$$
$$\tau_p = \min\left(\bar{\tau}_p, T\right)$$

This choice of τ_p ensures that

$$\mathbb{E}\left[\underbrace{\int_{t}^{\tau_{p}} \left(G_{z}(s,\tilde{Z}_{\lambda}(s);y)\tilde{Z}_{\lambda}(s)\|\gamma_{\lambda}(s)\|\right)^{2} ds}_{\leq p} \middle| \mathcal{F}_{t}\right] \leq p$$

$$\Rightarrow \mathbb{E}\left[\int_{t}^{\tau_{p}} G_{z}(s,\tilde{Z}_{\lambda}(s);y)\tilde{Z}_{\lambda}(s)\gamma_{\lambda}(s)' dW(s) \middle| \mathcal{F}_{t}\right] = 0.$$

Therefore,

$$\mathbb{E}[G(\tau_{p}, Z_{\lambda}(\tau_{p}); y) | \mathcal{F}_{t}] \\
\stackrel{\text{Itô}}{=} G(t, \tilde{Z}_{\lambda}(t); y) + \mathbb{E}\left[\int_{t}^{\tau_{p}} \underbrace{\mathcal{H}^{\lambda}G(s, \tilde{Z}_{\lambda}(s); y)}_{\geq 0} ds \middle| \mathcal{F}_{t}\right] \\
+ \mathbb{E}\left[\int_{t}^{\tau_{p}} G_{z}(s, \tilde{Z}_{\lambda}(s); y) \tilde{Z}_{\lambda}(s) \gamma_{\lambda}(s)' dW(s) \middle| \mathcal{F}_{t}\right] \\
\stackrel{=0}{=} 0 \\
\geq G(t, \tilde{Z}_{\lambda}(t); y) \qquad (A.41)$$

Clearly, $\tau_p \to T$, for $p \to \infty$ and hence $G(\tau_p, \tilde{Z}_\lambda(\tau_p); y) \to G(T, \tilde{Z}_\lambda(T); y) = U^*_{g_\lambda}(y; B_1, B_2)$ for $p \to \infty$.

Inequality (A.41) holds with equality for $\lambda = \lambda^*$. Next, we show that we can interchange limit and expectation. This will be achieved using Doob's martingale inequality and the dominated convergence theorem. The polynomial growth condition gives us a dominating random variable for $G(\tau_p, \tilde{Z}_\lambda(\tau_p); y)$:

$$|G(\tau_p, \tilde{Z}_{\lambda}(\tau_p); y)| \le C \left| (\tilde{Z}_{\lambda}(\tau_p))^{-\alpha} + (\tilde{Z}_{\lambda}(\tau_p))^{\alpha} \right|$$

$$\leq C \left(\sup_{t \leq s \leq T} (\tilde{Z}_{\lambda}(s))^{-\alpha} + \sup_{t \leq s \leq T} (\tilde{Z}_{\lambda}(s))^{\alpha} \right).$$

We now show that the right-hand side has finite expectation, due to the uniform boundedness of λ , γ_{λ} and $\delta_{K}(\lambda)$:

$$\begin{split} \tilde{Z}_{\lambda}(s)^{-\alpha} &= \tilde{Z}_{\lambda}(t)^{-\alpha} \exp\left(\alpha \int_{t}^{s} r + \delta_{K}(\lambda(u)) + \frac{1}{2} \|\gamma_{\lambda}(u)\|^{2} du + \alpha \int_{t}^{s} \gamma_{\lambda}(u)' dW(u)\right) \\ &= \tilde{Z}_{\lambda}(t)^{-\alpha} \exp\left(\alpha \int_{t}^{s} r + \underbrace{\delta_{K}(\lambda(u))}_{\leq C} + \frac{1}{2}(1+\alpha) \underbrace{\|\gamma_{\lambda}(u)\|^{2}}_{\leq C} du\right) \\ & \cdot \underbrace{\exp\left(-\frac{\alpha^{2}}{2} \int_{t}^{s} \|\gamma_{\lambda}(u)\|^{2} du + \alpha \int_{t}^{s} \gamma_{\lambda}(u)' dW(u)\right)}_{=: M_{-}(s)} \\ &\leq \tilde{Z}_{\lambda}(t)^{-\alpha} \exp\left(\left[\alpha(r+C) + \frac{1}{2}(\alpha+\alpha^{2})C\right](s-t)\right) M_{-}(s) \end{split}$$

and M_{-} is a martingale by Novikov's condition (again due to the uniform boundedness of λ). Therefore, through an application of the general inequality $x \leq 1 + x^2$ for all $x \in \mathbb{R}$ in (i) and an application of Doob's martingale inequality in (ii) we obtain

$$\mathbb{E}\left[\sup_{t\leq s\leq T} (\tilde{Z}_{\lambda}(s))^{-\alpha} \left| \mathcal{F}_{t} \right]\right]$$

$$\leq \tilde{Z}_{\lambda}(t)^{-\alpha} \exp\left(\left[\alpha|r+C| + \frac{1}{2}(\alpha+\alpha^{2})C\right](T-t)\right) \mathbb{E}\left[\sup_{t\leq s\leq T} M_{-}(s) \left| \mathcal{F}_{t} \right]\right]$$

$$\stackrel{(i)}{\leq} \tilde{Z}_{\lambda}(t)^{-\alpha} \exp\left(\left[\alpha|r+C| + \frac{1}{2}(\alpha+\alpha^{2})C\right](T-t)\right) \mathbb{E}\left[1 + \sup_{t\leq s\leq T} \left(M_{-}(s)\right)^{2} \left| \mathcal{F}_{t} \right]\right]$$

$$\stackrel{(ii)}{\leq} \tilde{Z}_{\lambda}(t)^{-\alpha} \exp\left(\left[\alpha|r+C| + \frac{1}{2}(\alpha+\alpha^{2})C\right](T-t)\right) \left(1 + 4\underbrace{\mathbb{E}\left[M_{-}(T)^{2} \left| \mathcal{F}_{t}\right]\right]}_{<\infty}\right) < \infty,$$

where the finiteness of the last expectation follows again using Novikov's condition and the uniform boundedness of λ . Similarly,

$$\begin{split} \tilde{Z}_{\lambda}(t,s)^{\alpha} &= \tilde{Z}_{\lambda}(t)^{\alpha} \exp\left(-\alpha \int_{t}^{s} r + \underbrace{\delta_{K}(\lambda(u))}_{\geq 0} + \underbrace{\frac{1}{2} \|\gamma_{\lambda}(u)\|^{2}}_{\geq 0} du - \alpha \int_{t}^{s} \gamma_{\lambda}(u)' dW(u)\right) \\ &\leq \tilde{Z}_{\lambda}(t)^{\alpha} \exp\left(-\alpha \int_{t}^{s} r - \frac{1}{2} \alpha \underbrace{\|\gamma_{\lambda}(u)\|^{2}}_{\leq C} du\right) \\ & \cdot \underbrace{\exp\left(-\frac{\alpha^{2}}{2} \int_{t}^{s} \|\gamma_{\lambda}(u)\|^{2} du - \alpha \int_{t}^{s} \gamma_{\lambda}(u)' dW(u)\right)}_{=: M_{+}(s)} \\ &\leq \tilde{Z}_{\lambda}(t)^{\alpha} \exp\left(\left[-\alpha r + \frac{1}{2}\alpha^{2}C\right](s-t)\right) M_{+}(s) \end{split}$$

and M_+ is a martingale by Novikov's condition (again due to the uniform boundedness of λ). Therefore, through another application of the inequality $x \leq 1 + x^2$ for all $x \in \mathbb{R}$ in (iii) and

an application of Doob's martingale inequality in (iv) we obtain

$$\mathbb{E}\left[\sup_{t\leq s\leq T} (\tilde{Z}_{\lambda}(s))^{\alpha} \middle| \mathcal{F}_{t}\right]$$

$$\leq \tilde{Z}_{\lambda}(t)^{\alpha} \exp\left(\left[\alpha|r| + \frac{1}{2}\alpha^{2}C\right](T-t)\right) \mathbb{E}\left[\sup_{t\leq s\leq T} M_{+}(s) \middle| \mathcal{F}_{t}\right]$$

$$\stackrel{(iii)}{\leq} \tilde{Z}_{\lambda}(t)^{\alpha} \exp\left(\left[\alpha|r| + \frac{1}{2}\alpha^{2}C\right](T-t)\right) \mathbb{E}\left[1 + \sup_{t\leq s\leq T} (M_{+}(s))^{2} \middle| \mathcal{F}_{t}\right]$$

$$\stackrel{(iv)}{\leq} \tilde{Z}_{\lambda}(t)^{\alpha} \exp\left(\left[\alpha|r| + \frac{1}{2}\alpha^{2}C\right](T-t)\right) \left(1 + 4\underbrace{\mathbb{E}\left[M_{+}(T)^{2} \middle| \mathcal{F}_{t}\right]}_{<\infty}\right) < \infty.$$

The finiteness of the last expectation can again easily be verified using Novikov's condition and the uniform boundedness of λ . This allows us to apply the dominated convergence theorem

$$\begin{aligned} G(t, \tilde{Z}_{\lambda}(t); y) &= \mathbb{E} \left[G(t, \tilde{Z}_{\lambda}(t); y) \big| \mathcal{F}_{t} \right] \\ &\stackrel{(A.41)}{\leq} \lim_{p \to \infty} \mathbb{E} \left[G(\tau_{p}, \tilde{Z}_{\lambda}(\tau_{p}); y) \big| \mathcal{F}_{t} \right] \\ &\stackrel{\text{dominated}}{=} \mathbb{E} \left[\lim_{p \to \infty} G(\tau_{p}, \tilde{Z}_{\lambda}(\tau_{p}); y) \big| \mathcal{F}_{t} \right] \\ &= \mathbb{E} \left[G(T, \tilde{Z}_{\lambda}(T); y) \big| \mathcal{F}_{t} \right] \\ &= \mathbb{E} \left[U_{g_{\lambda}}^{*}(y; B_{1}, B_{2}) \big| \mathcal{F}_{t} \right], \end{aligned}$$

with equality if $\lambda = \lambda^*$. Conditioning on $\tilde{Z}_{\lambda}(t) = z$ concludes the proof.

Proof of Lemma 3.5.4. Follows immediately from Chapter 5, Theorem 6.1 in [22].

Proof of Lemma 3.5.5. Note that for any given constant $\beta > 0$, we have

$$\beta u^2 \ge |u| \quad \forall u \in \mathbb{R} \text{ with } |u| \ge \frac{1}{\beta}$$

In particular, we then have

$$\beta u^2 + \frac{1}{\beta} \ge |u| \quad \forall u \in \mathbb{R}.$$
 (A.42)

Furthermore, for $u \in \mathbb{R}$, define $f(u) := U_{\hat{g}}^*\left(y; B_1, B_2, \exp\left(\frac{\|\gamma_{\lambda^*}\|}{\sqrt{2}}u\right)\right)$. Then f satisfies the prerequisites of Lemma 3.5.4, since for any $u \in \mathbb{R}$ and any $\beta > 0$

$$|f(u)| = \left| U_{\hat{g}}^{*}\left(y; B_{1}, B_{2}, \exp\left(\frac{\|\gamma_{\lambda^{*}}\|}{\sqrt{2}}u\right)\right) \right|$$

$$\stackrel{(3.17)}{\leq} C\left(e^{-\alpha \frac{\|\gamma_{\lambda^{*}}\|}{\sqrt{2}}u} + e^{\alpha \frac{\|\gamma_{\lambda^{*}}\|}{\sqrt{2}}u}\right)$$

$$\leq 2Ce^{\alpha \frac{\|\gamma_{\lambda^*}\|}{\sqrt{2}}|u|}$$

$$\leq 2Ce^{\alpha \frac{\|\gamma_{\lambda^*}\|}{\sqrt{2}}(\beta u^2 + \frac{1}{\beta})}.$$

Choosing β small enough such that $\alpha \frac{\|\gamma_{\lambda^*}\|}{\sqrt{2}}\beta < \frac{1}{4T}$ guarantees that F, as in (3.16), defines a continuously differentiable function in $\tau \in (0,T]$ and twice continuously differentiable function in $u \in \mathbb{R}$, which satisfies the heat equation $F_{\tau}(\tau, u) = F_{uu}(\tau, u)$ for all $(\tau, u) \in (0, \infty) \times \mathbb{R}$ with initial condition F(0, u) = f(u). We substitute

$$t := T - \tau$$
 and $u := \frac{\sqrt{2}}{\|\gamma_{\lambda^*}\|} \left(\log(z) - \left(r + \frac{1}{2} \|\gamma_{\lambda^*}\|^2 \right) (T - t) \right)$

and define a function $G: [0,T] \times (0,\infty) \to \mathbb{R}$ as

$$\begin{aligned} G(t,z;y) &:= F(\tau,u) = F\left(T - t, \frac{\sqrt{2}}{\|\gamma_{\lambda^*}\|} \left(\log(z) - (r + \frac{1}{2}\|\gamma_{\lambda^*}\|^2)(T - t)\right)\right) \\ &= \frac{1}{\sqrt{4\pi(T - t)}} \int_{\mathbb{R}} e^{-\frac{x^2}{4(T - t)}} f\left(\frac{\sqrt{2}}{\|\gamma_{\lambda^*}\|} \left(\log(z) - (r + \frac{1}{2}\|\gamma_{\lambda^*}\|^2)(T - t)\right) - x\right) dx \\ &= \frac{1}{\sqrt{4\pi(T - t)}} \int_{\mathbb{R}} e^{-\frac{x^2}{4(T - t)}} U_{\hat{g}}^*\left(y; B_1, B_2, ze^{-(r + \frac{1}{2}\|\gamma_{\lambda^*}\|^2)(T - t) - \frac{\|\gamma_{\lambda^*}\|}{\sqrt{2}}x}\right) dx. \end{aligned}$$

The function

$$z \to U_{\hat{g}}^*(y; B_1, B_2, z) = \sup_{B_1 \le x \le B_2} \left(U(x) - y_0 z x - y_1 g(x) \right)$$

is convex and strictly decreasing in z for any $y = (y_0, y_1) \in [0, \infty)^2$. Therefore, by the linearity of the integral, G is convex and strictly decreasing in z. Furthermore, using

$$\frac{\partial \tau}{\partial t} = -1, \quad \frac{\partial u}{\partial t} = \frac{\sqrt{2}}{\|\gamma_{\lambda^*}\|} r + \frac{\|\gamma_{\lambda^*}\|}{\sqrt{2}}, \quad \frac{\partial u}{\partial z} = \frac{\sqrt{2}}{\|\gamma_{\lambda^*}\|} \frac{1}{z} \quad \text{and} \quad \frac{\partial^2 u}{\partial^2 z} = -\frac{\sqrt{2}}{\|\gamma_{\lambda^*}\|} \frac{1}{z^2}.$$

we have

$$\begin{aligned} G_t(t,z;y) &= \frac{\partial}{\partial t} G(t,z;y) = \frac{\partial \tau}{\partial t} F_\tau(\tau,u) + \frac{\partial u}{\partial t} F_u(\tau,u) \\ &= -F_\tau(\tau,u) + \left(\frac{\sqrt{2}}{\|\gamma_{\lambda^*}\|}r + \frac{\|\gamma_{\lambda^*}\|}{\sqrt{2}}\right) F_u(\tau,u), \\ -rzG_z(t,z;y) &= -rz\frac{\partial}{\partial z} G(t,z;y) = -rz\frac{\partial u}{\partial z} F_u(\tau,u) = -r\frac{\sqrt{2}}{\|\gamma_{\lambda^*}\|} F_u(\tau,u) \\ \frac{z^2}{2} \|\gamma_{\lambda^*}\|^2 G_{zz}(t,z;y) &= \frac{z^2}{2} \|\gamma_{\lambda^*}\|^2 \frac{\partial}{\partial z} G_z(t,z;y) = \frac{z^2}{\sqrt{2}} \|\gamma_{\lambda^*}\| \frac{\partial}{\partial z} \left(\frac{1}{z} F_u(\tau,u)\right) \\ &= -\frac{\|\gamma_{\lambda^*}\|}{\sqrt{2}} F_u(\tau,u) + z\frac{\|\gamma_{\lambda^*}\|}{\sqrt{2}} \frac{\partial u}{\partial z} F_{uu}(\tau,u) \\ &= -\frac{\|\gamma_{\lambda^*}\|}{\sqrt{2}} F_u(\tau,u) + F_{uu}(\tau,u) \end{aligned}$$

and therefore

$$G_t(t,z;y) - rzG_z(t,z;y) + \frac{z^2}{2} \|\gamma_{\lambda^*}\|^2 G_{zz}(t,z;y) = -F_\tau(\tau,u) + F_{uu}(\tau,u) = 0,$$

for all $(t, z) \in [0, T) \times (0, \infty)$. The substitution of $\tau = 0 \Leftrightarrow t = T$ and the initial condition for F gives the terminal condition for G:

$$G(T, z; y) \stackrel{t=T \Leftrightarrow \tau=0}{=} F\left(0, \frac{\sqrt{2}}{\|\gamma_{\lambda^*}\|} \log(z)\right)$$
$$= U_{\hat{g}}^*\left(y; B_1, B_2, \exp\left(\frac{\|\gamma_{\lambda^*}\|}{\sqrt{2}} \frac{\sqrt{2}}{\|\gamma_{\lambda^*}\|} \log(z)\right)\right)$$
$$= U_{\hat{g}}^*(y; B_1, B_2, z).$$

In total, G satisfies the PDE

$$0 = G_t(t,z;y) - rzG_z(t,z;y) + \frac{z^2}{2} \|\gamma_{\lambda^*}\|^2 G_{zz}(t,z;y)$$

$$\stackrel{\text{Def},\lambda^*}{=} G_t(t,z;y) - rzG_z(t,z;y) + \frac{z^2}{2} \inf_{x \in X_K} \left(\|\gamma + \Sigma^{-1}x\|^2 \right) G_{zz}(t,z;y)$$

$$\stackrel{G_{zz} \ge 0}{=} G_t(t,z;y) - rzG_z(t,z;y) + \frac{z^2}{2} \inf_{x \in X_K} \left(\|\gamma + \Sigma^{-1}x\|^2 G_{zz}(t,z;y) \right)$$

with terminal condition $G(T, z; y) = U_{\hat{g}}^*(y; B_1, B_2)$. Furthermore, we can bound the absolute value of G, since for t = T the polynomial growth condition on $U_{\hat{g}}^*(y; B_1, B_2, z)$ holds and for t < T

$$\begin{split} |G(t,z;y)| &\leq \frac{1}{\sqrt{4\pi(T-t)}} \int_{\mathbb{R}} e^{-\frac{x^2}{4(T-t)}} \left| U_{\hat{g}}^* \left(y; B_1, B_2, z e^{-(r+\frac{1}{2} \|\gamma_{\lambda^*}\|^2)(T-t) - \frac{\|\gamma_{\lambda^*}\|}{\sqrt{2}} x} \right) \right| dx \\ \stackrel{(3.17)}{\leq} \frac{C}{\sqrt{4\pi(T-t)}} \int_{\mathbb{R}} e^{-\frac{x^2}{4(T-t)}} \left(z e^{-(r+\frac{1}{2} \|\gamma_{\lambda^*}\|^2)(T-t) - \frac{\|\gamma_{\lambda^*}\|}{\sqrt{2}} x} \right)^{-\alpha} dx \\ &+ \frac{C}{\sqrt{4\pi(T-t)}} \int_{\mathbb{R}} e^{-\frac{x^2}{4(T-t)}} \left(z e^{-(r+\frac{1}{2} \|\gamma_{\lambda^*}\|^2)(T-t) - \frac{\|\gamma_{\lambda^*}\|}{\sqrt{2}} x} \right)^{\alpha} dx \\ &= \frac{C}{\sqrt{4\pi(T-t)}} e^{\alpha(r+\frac{1}{2} \|\gamma_{\lambda^*}\|^2)(T-t)} \left(\int_{\mathbb{R}} e^{-\frac{x^2}{4(T-t)} - \frac{\alpha\|\gamma_{\lambda^*}\|}{\sqrt{2}} x} dx} \right) z^{-\alpha} \\ &+ \frac{C}{\sqrt{4\pi(T-t)}} e^{-\alpha(r+\frac{1}{2} \|\gamma_{\lambda^*}\|^2)(T-t)} \left(\int_{\mathbb{R}} e^{-\frac{x^2}{4(T-t)} - \frac{\alpha\|\gamma_{\lambda^*}\|}{\sqrt{2}} x} dx} \right) z^{\alpha} \\ \stackrel{(i)}{\stackrel{(i)}{=} C e^{\alpha(r+\frac{1}{2} \|\gamma_{\lambda^*}\|^2)(T-t)} \underbrace{\left(\frac{1}{\sqrt{2\pi}} \int_{\mathbb{R}} e^{-\frac{u^2}{2} + \alpha} \|\gamma_{\lambda^*}\| \sqrt{T-tu} du \right)}_{I_2} z^{\alpha} \\ &+ C e^{-\alpha(r+\frac{1}{2} \|\gamma_{\lambda^*}\|^2)(T-t)} \underbrace{\left(\frac{1}{\sqrt{2\pi}} \int_{\mathbb{R}} e^{-\frac{u^2}{2} - \alpha} \|\gamma_{\lambda^*}\| \sqrt{T-tu} du \right)}_{I_2} z^{\alpha} \\ \stackrel{(ii)}{\stackrel{(ii)}{=} C e^{\alpha|r+\frac{1}{2} \|\gamma_{\lambda^*}\|^2|T+\frac{1}{2}\alpha^2\|\gamma_{\lambda^*}\|^2T}}_{=: \tilde{C}} \left(z^{-\alpha} + z^{\alpha} \right) \end{split}$$

where we substituted $u = \frac{x}{\sqrt{2(T-t)}}$ in (i) and used that $I_1 = I_2 = \exp(\frac{1}{2}\alpha^2 \|\gamma_{\lambda^*}\|^2 (T-t))$ in (ii), since the integrals I_1 and I_2 are the moment generating function of a standard normal random variable evaluated at $\pm \alpha \|\gamma_{\lambda^*}\| \sqrt{T-t}$.

Proof of Lemma 3.5.6. We begin by deriving the concave conjugate of a power utility function on $(0, \infty)$. Thus, for b < 1, $b \neq 0$ and y > 0, we aim to compute

$$\sup_{x\geq 0} \left(\frac{1}{b}x^b - yx\right).$$

The corresponding first order optimality condition for the optimizer x^* yields

$$0 \stackrel{!}{=} (x^*)^{b-1} - y \iff x^* = y^{\frac{1}{b-1}}$$

As $\frac{1}{b}x^b - yx$ is strictly concave in b, the first order optimality condition is necessary and sufficient. Hence,

$$\sup_{x \ge 0} \left(\frac{1}{b}x^b - yx\right) = \frac{1}{b}y^{\frac{b}{b-1}} - yy^{\frac{1}{b-1}} = \frac{1-b}{b}y^{\frac{b}{b-1}}.$$
(A.43)

We continue by showing that (i) and (ii) imply that the growth condition (3.17) is satisfied.

(i) If
$$\max(U(x), -g(x)) \le C_+ \left(\frac{1}{b_+}x^{b_+} + 1\right)$$
, then we have for any $(y_0, y_1)' \in [0, \infty)^2$

$$\begin{aligned} U_{\hat{g}}^{*}(y; B_{1}, B_{2}, z) &= \sup_{B_{1} \leq x \leq B_{2}} \left(U(x) - y_{0}xz - y_{1}g(x) \right) \\ &\leq \sup_{B_{1} \leq x \leq B_{2}} \left(C_{+}(1 + y_{1}) \left(\frac{1}{b_{+}}x^{b_{+}} + 1 \right) - y_{0}xz \right) \\ &\leq \sup_{x \geq 0} \left(C_{+}(1 + y_{1}) \left(\frac{1}{b_{+}}x^{b_{+}} + 1 \right) - y_{0}xz \right) \\ &= C_{+}(1 + y_{1}) \left(1 + \sup_{x \geq 0} \left(\frac{1}{b_{+}}x^{b_{+}} - \frac{y_{0}z}{(1 + y_{1})C_{+}}x \right) \right) \\ & \stackrel{(A.43)}{=} C_{+}(1 + y_{1}) \left(1 + \frac{1 - b_{+}}{b_{+}} \left(\frac{y_{0}z}{(1 + y_{1})C_{+}} \right)^{\frac{b_{+}}{b_{+} - 1}} \right) \\ &= C_{+}(1 + y_{1}) \left(1 + \frac{1 - b_{+}}{b_{+}} \left(\frac{y_{0}}{(1 + y_{1})C_{+}} \right)^{\frac{b_{+}}{b_{+} - 1}} z^{\frac{b_{+}}{b_{+} - 1}} \right). \end{aligned}$$

Using analogous arguments, we can show that $C_{-}\left(\frac{1}{b_{-}}x^{b_{-}}-1\right) \leq \min\left(U(x),-g(x)\right)$ implies

$$C_{-}(1+y_{1})\left(-1+\frac{1-b_{-}}{b_{-}}\left(\frac{y_{0}}{(1+y_{1})C_{-}}\right)^{\frac{b_{-}}{b_{-}-1}}z^{\frac{b_{-}}{b_{-}-1}}\right) \leq U_{\hat{g}}^{*}(y;B_{1},B_{2},z).$$

Thus, the growth condition (3.17) is satisfied with $\alpha = \max\left(\left|\frac{b_{-}}{b_{-}-1}\right|, \left|\frac{b_{+}}{b_{+}-1}\right|\right)$ and a suitably large constant C > 0.

(ii) If $C = |U(B_2)| + y_0(B_1 + B_2) + y_1|g(B_2)| < \infty$, then condition (3.17) is satisfied with $\alpha = 1$ because for all z > 0

$$U_{\hat{g}}^{*}(y; B_{1}, B_{2}, z) = \sup_{B_{1} \le x \le B_{2}} \left(U(x) - y_{0}zx - y_{1}g(x) \right) \stackrel{x=B_{2}}{\ge} U(B_{2}) - y_{0}zB_{2} - y_{1}g(B_{2})$$

and $U_{\hat{g}}^{*}(y; B_{1}, B_{2}, z) = \sup_{B_{1} \le x \le B_{2}} \left(\underbrace{U(x)}_{\le U(B_{2})} - y_{0}zx \underbrace{-y_{1}g(x)}_{\le y_{1}g(B_{2})} \right)$

$$\leq U(B_2) - y_1 g(B_2) + \sup_{B_1 \leq x \leq B_2} \left(-y_0 z x \right) = U(B_2) - y_0 z B_1 - y_1 g(B_2)$$

$$\Rightarrow |U_{\hat{g}}^*(y; B_1, B_2, z)| \leq |U(B_2)| + y_0 (B_1 + B_2) z + y_1 |g(B_2)|$$

$$\leq (|U(B_2)| + y_1 |g(B_2)|) \frac{1}{z} + (|U(B_2)| + y_0 (B_1 + B_2) + y_1 |g(B_2)|) z$$

$$\leq C(z^{-\alpha} + z^{\alpha}).$$

Proof of Corollary 3.5.7. All prerequisites of Lemma 3.5.5 are satisfied and thus G (as defined in Lemma 3.5.5) satisfies the HJB equation (3.10), is convex, strictly decreasing in z and satisfies a polynomial growth condition. According to Theorem 3.5.3, as the minimizing $\lambda^*(t, z; y) = \lambda^*$ is independent of $y \in [0, \infty)^2$, this implies for all $y \in [0, \infty)^2$:

$$\begin{aligned} G(0,z;y) &= \mathbb{E}[U_{g_{\lambda^*}}^*(y;B_1,B_2)] \\ &\leq \mathbb{E}[U_{g_{\lambda}}^*(y;B_1,B_2)], \quad \forall \lambda \in \mathcal{D} \text{ with } \|\lambda^* - \lambda(t)\|^2 \leq 1 \quad \mathcal{L}[0,T] \otimes Q\text{-a.s..} \end{aligned}$$

In particular, this holds for the choice of $y = y^*$, where y^* is optimal for $(\mathbf{D}_{\lambda^*}^{\mathbf{V}})$. Finally, as $\lambda^* \in \mathcal{D}'$, Condition (\tilde{D}) is satisfied locally and the statement of this Corollary now follows due to the equivalence of Condition (\tilde{B}) and Condition (\tilde{D}) by virtue of Corollary 3.4.6 and Theorem 3.4.5.

Proof of Example 3.5.10. Following Corollary 3.5.7 and Remark 3.5.8, we still need to

- verify that $U_{\hat{a}}^*$ satisfies a growth condition (3.17) in z,
- show that $\mathcal{I}_{g_{\lambda^*}}(y^*; B_1, B_2) = \operatorname{Cap}(\mathcal{I}(y_0^*), B_1, B_2),$
- verify that $\mathcal{I}_{g_{\lambda^*}}(y^*; B_1, B_2) \in L^2_Q$,
- verify that $\mathbb{E}\left[g_{\lambda^*}(\mathcal{I}_{g_{\lambda^*}}(y; B_1, B_2))\right]' \Delta y$ is use at $y = y^*$ for all $\Delta \in \mathbb{R} \times [0, \infty)$.

As U is a power utility function and $g \equiv 0$, condition (i) of Lemma 3.5.6 is satisfied. Furthermore, $U(x) - y_0 \tilde{Z}_{\lambda^*}(T)$ is strictly concave in $x \ge 0$ and is maximized by $\mathcal{I}(y_0)$ over $x \in \mathbb{R}$. Hence, its maximum over $x \in [B_1, B_2]$ is attained by

$$\mathcal{I}_{g_{\lambda^*}}(y; B_1, B_2) = B_1 + (\mathcal{I}(y_0) - B_1)^+ - (\mathcal{I}(y_0) - B_2)^+ = \operatorname{Cap}(\mathcal{I}(y_0), B_1, B_2).$$

By Assumption 3.5.1, the market coefficients are deterministic and so is λ^* . Hence, $\tilde{Z}_{\lambda^*}(T)$ is log-normally distributed and all of its moments are finite. In particular, we have for all $y_0 \in (0, \infty)$

$$\mathbb{E}\left[\mathcal{I}_{g_{\lambda^{*}}}(y; B_{1}, B_{2})^{2}\right] \leq 2B_{1}^{2} + 2\mathbb{E}\left[\mathcal{I}(y_{0})\right] = 2B_{1}^{2} + 2\mathbb{E}\left[\left(y_{0}\tilde{Z}_{\lambda^{*}}(T)\right)^{\frac{2}{b-1}}\right] < \infty$$

Similarly,

$$\mathbb{E}\left[\mathcal{I}_{g_{\lambda^*}}(y; B_1, B_2)\tilde{Z}_{\lambda^*}(T)\right] \leq B_1 \mathbb{E}\left[\tilde{Z}_{\lambda^*}(T)\right] + \mathbb{E}\left[\mathcal{I}(y_0)\tilde{Z}_{\lambda^*}(T)\right]$$
$$\leq B_1 \mathbb{E}\left[\tilde{Z}_{\lambda^*}(T)\right] + y_0^{\frac{1}{b-1}} \mathbb{E}\left[\tilde{Z}_{\lambda^*}(T)^{\frac{b}{b-1}}\right] < \infty.$$
(A.44)

Lastly, $\mathcal{I}_{g_{\lambda^*}}(y; B_1, B_2)$ is a concatenation of continuous functions and thereby continuous in $y_0 \in (0, \infty)$. Therefore, the monotonicity of $\mathcal{I}_{g_{\lambda^*}}(y; B_1, B_2)$ in y_0 , equation (A.44) and the dominated convergence theorem imply the continuity of

$$y \to \mathbb{E}\left[g_{\lambda^*}\left(\mathcal{I}_{g_{\lambda^*}}(y; B_1, B_2)\tilde{Z}_{\lambda^*}(T)\right)\right] = \begin{pmatrix}\mathbb{E}\left[\mathcal{I}_{g_{\lambda^*}}(y; B_1, B_2)\tilde{Z}_{\lambda^*}(T)\right]\\0\end{pmatrix}$$

in $y \in [0, \infty)^2$. This is also true for $y = (y_0^*, 0)'$ and thus $\lambda^* \in \mathcal{D}'$. The optimality of $\mathcal{I}_{g_{\lambda^*}}(y^*; B_1, B_2)$ for (**P**) follows from Corollary 3.5.7.

Proof of Example 3.5.11. Following Corollary 3.5.7 and Remark 3.5.8, we still need to

- verify that $U_{\hat{a}}^*$ satisfies growth condition (3.17) in z,
- show that $\mathcal{I}_{g_{\lambda^*}}(y^*)$ is given as in the statement of the example,
- verify that $\mathcal{I}_{g_{\lambda^*}}(y^*) \in L^2_Q$,
- verify that $\mathbb{E}\left[g_{\lambda^*}(\mathcal{I}_{g_{\lambda^*}}(y))\right]' \Delta y$ is use at $y = y^*$ for all $\Delta \in \mathbb{R} \times [0, \infty)$.

As U is a power utility function and $-\epsilon \leq g(x) \leq 1 - \epsilon$ for all $x \geq 0$, $U_{\hat{g}}$ satisfies the growth condition (3.17) by Lemma 3.5.6, (i). Further, we have for any $y = (y_0, y_1)' \in [0, \infty)^2$:

$$\begin{split} U_{g_{\lambda^{*}}}^{*}(y) &= \sup_{x \ge 0} \left(U(x) - y_{0} \tilde{Z}_{\lambda^{*}}(T) x - y_{1} g(x) \right) \\ &= \sup_{x \ge 0} \left(U(x) - y_{0} \tilde{Z}_{\lambda^{*}}(T) x - y_{1} \mathbb{1}_{\{x < B_{VaR}\}} \right) + y_{1} \epsilon \\ &= \max \left(\sup_{B_{VaR} \le x} \underbrace{\left(U(x) - y_{0} \tilde{Z}_{\lambda^{*}}(T) x \right)}_{=:f_{1}(x)}, \sup_{0 \le x \le B_{VaR}} \underbrace{\left(U(x) - y_{0} \tilde{Z}_{\lambda^{*}}(T) x - y_{1} \right)}_{=:f_{2}(x)} \right) + y_{1} \epsilon \\ &= \max \left(\sup_{B_{VaR} \le x} f_{1}(x), \sup_{0 \le x \le B_{VaR}} f_{2}(x) \right) + y_{1} \epsilon \end{split}$$

Since $f_1(B_{VaR}) \ge f_2(B_2)$, the last inequality holds despite the discontinuity of $U(x) - y_0 \tilde{Z}_{\lambda^*}(T)x - y_1 \mathbb{1}_{\{x < B_{VaR}\}}$ at $x = B_{VaR}$. The functions f_1 and f_2 are both strictly concave in x and thus their maximizer over bounded intervals is obtained by capping their global maximizer at the boundaries. Hence, $x = \max(\mathcal{I}(y_0), B_{VaR})$ maximizes $f_1(x)$ over $x \in [B_{VaR}, \infty)$ and $x = \min(\mathcal{I}(y_0), B_{VaR})$ maximizes f_2 over $x \in [0, B_{VaR}]$. It remains to determine which of these two local maximizers yields the global maximizer $\mathcal{I}_{g_{\lambda^*}}(y)$. We make a distinction between two cases:

• $B_{VaR} < \mathcal{I}(y_0)$: This implies

$$\max(\mathcal{I}(y_0), B_{VaR}) = \mathcal{I}(y_0), \qquad \min\left(\mathcal{I}(y_0), B_{VaR}\right) = B_{VaR}.$$

Hence,

$$\sup_{0 \le x \le B_{VaR}} f_2(x) = f_2(B_{VaR}) \le f_1(B_{VaR}) \le \sup_{B_{VaR} \le x} f_1(x) = f_1(\mathcal{I}(y_0))$$

and therefore $\mathcal{I}_{g_{\lambda^*}}(y) = \mathcal{I}(y_0).$

• $\mathcal{I}(y_0) \leq B_{VaR}$: This implies

$$\max(\mathcal{I}(y_0), B_{VaR}) = B_{VaR}, \qquad \min\left(\mathcal{I}(y_0), B_{VaR}\right) = \mathcal{I}(y_0).$$

We then have

$$\begin{split} \sup_{0 \le x \le B_{VaR}} f_2(x) \ge \sup_{B_{VaR} \le x} f_1(x) & \Leftrightarrow \quad f_2(\mathcal{I}(y_0)) \ge f_1(B_{VaR}) \\ \Leftrightarrow \quad U(\mathcal{I}(y_0)) - \underbrace{y_0 \tilde{Z}_{\lambda^*}(T)}_{=\mathcal{I}(y_0)^{b-1}} \mathcal{I}(y_0) - y_1 \ge U(B_{VaR}) - y_0 \tilde{Z}_{\lambda^*}(T) B_{VaR} \\ \Leftrightarrow \quad \frac{1-b}{b} \mathcal{I}(y_0)^b - y_1 \ge \frac{1}{b} B_{VaR}^b - y_0 \tilde{Z}_{\lambda^*}(T) B_{VaR} \\ \Leftrightarrow \quad \frac{1-b}{b} \mathcal{I}(y_0)^b + y_0 \tilde{Z}_{\lambda^*}(T) B_{VaR} \ge \frac{1}{b} B_{VaR}^b + y_1. \end{split}$$

Thus, if additionally

$$\circ \ \frac{1-b}{b}\mathcal{I}(y_0)^b + y_0\tilde{Z}_{\lambda^*}(T)B_{VaR} \ge \frac{1}{b}B_{VaR}^b + y_1, \text{ then } \mathcal{I}_{g_{\lambda^*}}(y) = \mathcal{I}(y_0).$$

$$\circ \ \frac{1-b}{b}\mathcal{I}(y_0)^b + y_0\tilde{Z}_{\lambda^*}(T)B_{VaR} < \frac{1}{b}B_{VaR}^b + y_1, \text{ then } \mathcal{I}_{g_{\lambda^*}}(y) = B_{VaR}.$$

In total, the maximizing argument $\mathcal{I}_{g_{\lambda^*}}(y^*)$, where y^* is optimal for $(\mathbf{D}_{\lambda^*}^{\mathbf{V}})$, is as specified in the statement of this Example. By Assumption 3.5.1, the market coefficients are deterministic and so is λ^* . Hence, $\tilde{Z}_{\lambda^*}(T)$ is log-normally distributed and all of its moments are finite. In particular, we have for arbitrary $y = (y_0, y_1)' \in (0, \infty) \times [0, \infty)$

$$\mathbb{E}\left[\mathcal{I}_{g_{\lambda^*}}(y)^2\right] \leq \mathbb{E}\left[\left(B_{VaR} + \mathcal{I}(y_0)\right)^2\right]$$
$$\leq 2B_{VaR}^2 + 2\mathbb{E}\left[\left(\mathcal{I}(y_0)\right)^2\right]$$
$$= 2B_{VaR}^2 + 2y_0^{\frac{2}{b-1}}\mathbb{E}\left[\tilde{Z}_{\lambda^*}(T)^{\frac{2}{b-1}}\right] < \infty,$$

i.e., $\mathcal{I}_{g_{\lambda^*}}(y) \in L^2_Q$. Similarly, we have for arbitrary $y = (y_0, y_1)' \in (0, \infty) \times [0, \infty)$

$$\mathbb{E}\left[\mathcal{I}_{g_{\lambda^*}}(y)\tilde{Z}_{\lambda^*}(T)\right] \le B_{VaR}\mathbb{E}[\tilde{Z}_{\lambda^*}(T)] + y_0^{\frac{1}{b-1}}\mathbb{E}\left[\tilde{Z}_{\lambda^*}(T)^{\frac{b}{b-1}}\right] < \infty.$$

For a fixed value of $\tilde{Z}_{\lambda^*}(T)$, the functions $y \to \mathcal{I}_{g_{\lambda^*}}(y)\tilde{Z}_{\lambda^*}(T)$ and $y \to g(\mathcal{I}_{g_{\lambda^*}}(y))$ are continuous at $y = y^*$ if

$$\frac{1-b}{b} \Big(y_0^* \tilde{Z}_{\lambda^*}(T) \Big)^{\frac{b}{b-1}} + y_0^* \tilde{Z}_{\lambda^*}(T) B_{VaR} \neq \frac{1}{b} B_{VaR}^b + y_1^*.$$
(A.45)

The left side of (A.45) is strictly convex in $\tilde{Z}_{\lambda^*}(T)$, while the right side is a deterministic constant. Thus, there are at most 2 values of $\tilde{Z}_{\lambda^*}(T) \in (0,\infty)$ such that we have equality in (A.45). However, $\tilde{Z}_{\lambda^*}(T)$ has a continuous distribution on $(0,\infty)$ and thus

$$Q\left(g_{\lambda^*}\left(\mathcal{I}_{g_{\lambda^*}}(y)\right) \text{ is continuous at } y = y^*\right)$$
$$\geq Q\left(\frac{1-b}{b}\left(y_0^*\tilde{Z}_{\lambda^*}(T)\right)^{\frac{b}{b-1}} + y_0^*\tilde{Z}_{\lambda^*}(T)B_{VaR} \neq \frac{1}{b}B_{VaR}^b + y_1^*\right) = 1,$$

i.e., $y \to g_{\lambda^*}(\mathcal{I}_{g_{\lambda^*}}(y))$ is continuous at $y = y^*$ with probability 1. Further, for any sequence of $(y_n)_{n \in \mathbb{N}}$ taking values in $(0, \infty) \times [0, \infty)$ and converging to $y^* \in (0, \infty) \times [0, \infty)$, we know that

$$\hat{y}_0 := \inf_{n \in \mathbb{N}} \left((y_n)_0 \right) > 0$$

$$\begin{aligned} \left| \mathcal{I}_{g_{\lambda^*}}(y_n) \tilde{Z}_{\lambda^*}(T) \right| &\leq \left(B_{VaR} + \mathcal{I}\left((y_n)_0 \right) \right) \tilde{Z}_{\lambda^*}(T) \leq \left(B_{VaR} + \mathcal{I}\left(\hat{y}_0 \right) \right) \tilde{Z}_{\lambda^*}(T) \\ \left| g\left(\mathcal{I}_{g_{\lambda^*}}(y_n) \right) \right| &\leq 1. \end{aligned}$$

Thus, the dominated convergence theorem implies

$$\lim_{n \to \infty} \mathbb{E} \left[g_{\lambda^*}(\mathcal{I}_{g_{\lambda^*}}(y_n)) \right] = \lim_{n \to \infty} \begin{pmatrix} \mathbb{E} \left[-\mathcal{I}_{g_{\lambda^*}}(y_n) \tilde{Z}_{\lambda^*}(T) \right] \\ \mathbb{E} \left[-g \left(\mathcal{I}_{g_{\lambda^*}}(y_n) \right) \right] \end{pmatrix} \\ = \begin{pmatrix} \mathbb{E} \left[-\mathcal{I}_{g_{\lambda^*}}(y) \tilde{Z}_{\lambda^*}(T) \right] \\ \mathbb{E} \left[-g \left(\mathcal{I}_{g_{\lambda^*}}(y) \right) \right] \end{pmatrix} = \mathbb{E} \left[g_{\lambda^*}(\mathcal{I}_{g_{\lambda^*}}(y)) \right] \end{pmatrix}$$

Hence, $y \to \mathbb{E}\left[g_{\lambda^*}(\mathcal{I}_{g_{\lambda^*}}(y))\right]$ is continuous at $y = y^*$ and therefore $\mathbb{E}\left[g_{\lambda^*}(\mathcal{I}_{g_{\lambda^*}}(y))\right]' \Delta y$ is usc at $y = y^*$ for all $\Delta \in \mathbb{R} \times [0, \infty)$ and thus $\lambda^* \in \mathcal{D}'$. In total, the optimality of $\mathcal{I}_{g_{\lambda^*}}(y^*)$ for (**P**) now follows from Corollary 3.5.7.

Proof of Example 3.5.12. Following the same arguments as in the proof of Example 3.5.12, we have for any $y = (y_0, y_1) \in [0, \infty)^2$

$$\begin{aligned} U_{g_{\lambda^*}}^*(y; B_1, B_2) &= \sup_{B_1 \le x \le B_2} \left(U(x) - y_0 \tilde{Z}_{\lambda^*}(T) x - y_1 g(x) \right) \\ &= \sup_{B_1 \le x \le B_2} \left(U(x) - y_0 \tilde{Z}_{\lambda^*}(T) x - y_1 \mathbb{1}_{\{x < B_{VaR}\}} \right) + y_1 \epsilon \\ &= \max \left(\sup_{B_{VaR} \le x \le B_2} \underbrace{\left(U(x) - y_0 \tilde{Z}_{\lambda^*}(T) x \right)}_{=f_1(x)}, \sup_{B_1 \le x \le B_{VaR}} \underbrace{\left(U(x) - y_0 \tilde{Z}_{\lambda^*}(T) x - y_1 \right)}_{=f_2(x)} \right) + y_1 \epsilon \\ &= \max \left(\sup_{B_{VaR} \le x \le B_2} f_1(x), \sup_{B_1 \le x \le B_{VaR}} f_2(x) \right) + y_1 \epsilon. \end{aligned}$$

Since f_1 and f_2 are strictly concave in $x \in (0, \infty)$, their constrained maximizers over intervals can be obtained by capping their global maximizers at the boundaries of the interval, i.e.,

$$\sup_{\substack{B_{VaR} \le x \le B_2}} f_1(x) = f_1\left(\operatorname{Cap}(\mathcal{I}(y_0), B_{VaR}, B_2)\right)$$
$$\sup_{\substack{B_1 \le x \le B_{VaR}}} f_2(x) = f_2\left(\operatorname{Cap}(\mathcal{I}(y_0), B_1, B_{VaR})\right).$$

We make a distinction between three cases:

• $B_2 < \mathcal{I}_{g_{\lambda^*}}(y)$: As $B_{VaR} < B_2$, this implies $\mathcal{I}_{g_{\lambda^*}}(y) = \mathcal{I}(y_0)$ and

$$\operatorname{Cap}(\mathcal{I}(y_0), B_{VaR}, B_2) = B_2, \qquad \operatorname{Cap}(\mathcal{I}(y_0), B_1, B_{VaR}) = B_{VaR}.$$

Hence,

$$\sup_{B_1 \le x \le B_{VaR}} f_2(x) = f_2(B_{VaR}) \le f_1(B_{VaR}) \le \sup_{B_{VaR} \le x \le B_2} f_1(x) = f_1(B_2)$$

and therefore $\mathcal{I}_{g_{\lambda^*}}(y; B_1, B_2) = B_2$.

• $B_1 \leq \mathcal{I}_{g_{\lambda^*}}(y) \leq B_2$: This implies that

$$U(\mathcal{I}_{g_{\lambda^*}}(y)) - y_0 \tilde{Z}_{\lambda^*}(T) \mathcal{I}_{g_{\lambda^*}}(y) - y_1 g(\mathcal{I}_{g_{\lambda^*}}(y)) = U_{g_{\lambda^*}}^*(y)$$

$$= \sup_{0 \le x} \left(U(x) - y_0 \tilde{Z}_{\lambda^*}(T) x - y_1 g(x) \right)$$

$$\geq \sup_{B_1 \le x \le B_2} \left(U(x) - y_0 \tilde{Z}_{\lambda^*}(T) x - y_1 g(x) \right)$$

$$\geq U(\mathcal{I}_{g_{\lambda^*}}(y)) - y_0 \tilde{Z}_{\lambda^*}(T) \mathcal{I}_{g_{\lambda^*}}(y) - y_1 g(\mathcal{I}_{g_{\lambda^*}}(y))$$

and therefore $\mathcal{I}_{g_{\lambda^*}}(y; B_1, B_2) = \mathcal{I}_{g_{\lambda^*}}(y).$

•
$$\mathcal{I}_{g_{\lambda^*}}(y) < B_1$$
: As $B_1 < B_{VaR}$, this implies $\mathcal{I}_{g_{\lambda^*}}(y) = \mathcal{I}(y_0)$ and

$$\operatorname{Cap}(\mathcal{I}(y_0), B_{VaR}, B_2) = B_{VaR}, \qquad \operatorname{Cap}(\mathcal{I}(y_0), B_1, B_{VaR}) = B_1$$

We then have

$$\sup_{B_1 \le x \le B_{VaR}} f_2(x) \ge \sup_{B_{VaR} \le x \le B_2} f_1(x) \quad \Leftrightarrow \quad f_2(B_1) \ge f_1(B_{VaR})$$
$$\Leftrightarrow \quad U(B_1) - y_0 \tilde{Z}_{\lambda^*}(T) B_1 - y_1 \ge U(B_{VaR}) - y_0 \tilde{Z}_{\lambda^*}(T) B_{VaR}$$
$$\Leftrightarrow \quad \tilde{Z}_{\lambda^*}(T) \ge \frac{\frac{1}{b} (B_{VaR}^b - B_1^b) + y_1}{y_0(B_{VaR} - B_1)}.$$

Thus, if additionally

$$\circ \quad \tilde{Z}_{\lambda^*}(T) < \frac{\frac{1}{b}(B_{VaR}^b - B_1^b) + y_1}{y_0(B_{VaR} - B_1)}, \text{ then } \mathcal{I}_{g_{\lambda^*}}(y; B_1, B_2) = B_{VaR}$$

$$\circ \quad \tilde{Z}_{\lambda^*}(T) \ge \frac{\frac{1}{b}(B_{VaR}^b - B_1^b) + y_1}{y_0(B_{VaR} - B_1)}, \text{ then } \mathcal{I}_{g_{\lambda^*}}(y; B_1, B_2) = B_1.$$

In total, the maximizing argument $\mathcal{I}_{g_{\lambda^*}}(y^*)$ for the optimal y^* is as specified in the statement of this Example. All integrability conditions for $\mathcal{I}_{g_{\lambda^*}}(y^*; B_1, B_2)$ are satisfied by analogous arguments as in Example 3.5.11. For a fixed value of $\tilde{Z}_{\lambda^*}(T)$, the functions $y \to \mathcal{I}_{g_{\lambda^*}}(y; B_1, B_2)\tilde{Z}_{\lambda^*}(T)$ and $y \to g(\mathcal{I}_{g_{\lambda^*}}(y; B_1, B_2))$ are continuous at $y = y^*$ if

$$\frac{1-b}{b} \left(y_0^* \tilde{Z}_{\lambda^*}(T) \right)^{\frac{b}{b-1}} + y_0^* \tilde{Z}_{\lambda^*}(T) B_{VaR} \neq \frac{1}{b} B_{VaR}^b + y_1^*.$$
(A.46)

and

$$\tilde{Z}_{\lambda^*}(T) \neq \frac{\frac{1}{b}(B_{VaR}^b - B_1^b) + y_1^*}{y_0^*(B_{VaR} - B_1)}.$$
(A.47)

However, as argued in the proof of Example 3.5.11, there are at most 2 values of $\tilde{Z}_{\lambda^*}(T)$ such that equality holds in (A.46) and there is at most one value of $\tilde{Z}_{\lambda^*}(T)$ such that equality holds in (A.47). Since $\tilde{Z}_{\lambda^*}(T)$ has a continuous distribution on $(0, \infty)$ this again implies

 $Q\left(g_{\lambda^*}\left(\mathcal{I}_{g_{\lambda^*}}(y)\right) \text{ is continuous at } y = y^*\right) \ge Q\left((A.46) \text{ and } (A.47 \text{ hold.}) = 1.$

The rest of this example can be proved using analogous steps as in the proof of Example 3.5.13. $\hfill \Box$

Proof of Example 3.5.13. Following Corollary 3.5.7 and Remark 3.5.8, we still need to

- verify that $U_{\hat{q}}^*$ satisfies growth condition (3.17) in z,
- show that $\mathcal{I}_{g_{\lambda^*}}(y^*)$ is given as in the statement of the example,
- verify that $\mathcal{I}_{g_{\lambda^*}}(y^*) \in L^2_Q$,

• verify that $\mathbb{E}\left[g_{\lambda^*}(\mathcal{I}_{g_{\lambda^*}}(y))\right]' \Delta y$ is use at $y = y^*$ for all $\Delta \in \mathbb{R} \times [0, \infty)$.

As U is a power utility and $-\epsilon \leq g(x) \leq B_{ES} - \epsilon$ for all $x \geq 0$, $U^{\hat{g}}$ satisfies the growth condition (3.17) by Lemma 3.5.6, (i). Further, we have for any $(y_0, y_1)' \in [0, \infty)^2$:

$$\begin{split} U_{g_{\lambda^*}}^*(y_0, y_1) &= \sup_{x \ge 0} \left(U(x) - y_0 \tilde{Z}_{\lambda^*}(T) x - y_1 g(x) \right) \\ &= \sup_{x \ge 0} \left(U(x) - y_0 \tilde{Z}_{\lambda^*}(T) x - y_1 \left(B_{ES} - x \right) \mathbb{1}_{\{x < B_{ES}\}} \right) + y_1 \epsilon \\ &= \max \left(\sup_{B_{ES} \le x} \underbrace{\left(U(x) - y_0 \tilde{Z}_{\lambda^*}(T) x \right)}_{=:f_1(x)}, \sup_{0 \le x \le B_{ES}} \underbrace{\left(U(x) - (y_0 z - y_1) x - y_1 B_{ES} \right)}_{=:f_2(x)} \right) + y_1 \epsilon \\ &= \max \left(\sup_{B_{ES} \le x} f_1(x), \sup_{0 \le x \le B_{ES}} f_2(x) \right) + y_1 \epsilon \end{split}$$

 f_1 and f_2 are both strictly concave in x and thus their maximizer over bounded intervals is obtained by capping their global maximizer at the boundaries. Hence, $x = \max(\mathcal{I}(y_0), B_{ES})$ maximizes $f_1(x)$ over $x \in [B_{ES}, \infty)$ and $x = \min\left(\mathcal{I}\left(y_0 - \frac{y_1}{Z_{\lambda^*}(T)}\right), B_{ES}\right)$ maximizes f_2 over $x \in [0, B_{ES}]$. It remains to determine which of these two local maximizers yields the global maximizer $\mathcal{I}_{g_{\lambda^*}}(y)$. We make a distinction between three cases

• $B_{ES} < \mathcal{I}(y_0)$: As $\mathcal{I}(y_0)$ is non-increasing in y_0 , this implies $\mathcal{I}\left(y_0 - \frac{y_1}{\tilde{Z}_{\lambda^*}(T)}\right) \geq \mathcal{I}(y_0) > B_{ES}$ and thus

$$\max(\mathcal{I}(y_0), B_{ES}) = \mathcal{I}(y_0), \qquad \min\left(\mathcal{I}\left(y_0 - \frac{y_1}{\tilde{Z}_{\lambda^*}(T)}\right), B_{ES}\right) = B_{ES}.$$

Hence,

$$\sup_{0 \le x \le B_2} f_2(x) = f_2(B_{ES})$$

$$= U(B_{ES}) - \left(y_0 \tilde{Z}_{\lambda^*}(T) - y_1\right) B_{ES} - y_1 B_{ES}$$

$$= \underbrace{U(B_{ES}) - y_0 \tilde{Z}_{\lambda^*}(T) B_{ES}}_{=f_1(B_{ES})}$$

$$< \sup_{B_{ES} \le x} f_1(x)$$

$$= f_1(\mathcal{I}(y_0))$$

and therefore $\mathcal{I}_{g_{\lambda^*}}(y) = \mathcal{I}(y_0).$

• $\mathcal{I}(y_0) \leq B_{ES} \leq \mathcal{I}\left(y_0 - \frac{y_1}{\tilde{Z}_{\lambda^*}(T)}\right)$: This implies

$$\max(\mathcal{I}(y_0), B_{ES}) = B_{ES}, \qquad \min\left(\mathcal{I}\left(y_0 - \frac{y_1}{\tilde{Z}_{\lambda^*}(T)}\right), B_{ES}\right) = B_{ES}$$

and therefore $\mathcal{I}_{g_{\lambda^*}} = B_{ES}$.

• $\mathcal{I}\left(y_0 - \frac{y_1}{\tilde{Z}_{\lambda^*}(T)}\right) < B_{ES}$: As $\mathcal{I}(y_0)$ is non-increasing in y_0 , this implies $\mathcal{I}(y_0) < \mathcal{I}\left(y_0 - \frac{y_1}{\tilde{Z}_{\lambda^*}(T)}\right) < B_{ES}$ and thus and thus

$$\max(\mathcal{I}(y_0), B_{ES}) = B_{ES}, \quad \min\left(\mathcal{I}\left(y_0 - \frac{y_1}{\tilde{Z}_{\lambda^*}(T)}\right), B_{ES}\right) = \mathcal{I}\left(y_0 - \frac{y_1}{\tilde{Z}_{\lambda^*}(T)}\right).$$

Hence,

$$\sup_{B_{ES} \le x} f_1(x) = f_1(B_{ES})$$

$$= U(B_2) - y_0 \tilde{Z}_{\lambda^*}(T) B_{ES}$$

$$= \underbrace{U(B_2) - \left(y_0 \tilde{Z}_{\lambda^*}(T) - y_1\right) B_{ES} - y_1 B_{ES}}_{=f_2(B_{ES})}$$

$$\leq \sup_{0 \le x \le B_{ES}} f_2(x)$$

$$= f_2 \left(\mathcal{I} \left(y_0 - \frac{y_1}{\tilde{Z}_{\lambda^*}(T)} \right) \right)$$

and therefore $\mathcal{I}_{g_{\lambda^*}} = \mathcal{I}\left(y_0 - \frac{y_1}{\tilde{Z}_{\lambda^*}(T)}\right)$.

In total, the maximizing argument $\mathcal{I}_{g_{\lambda^*}}(y^*)$, where y^* is optimal for $(\mathbf{D}_{\lambda^*}^{\mathbf{V}})$, is as specified in the statement of this Example. By Assumption 3.5.1, the market coefficients are deterministic and so is λ^* . Hence, $\tilde{Z}_{\lambda^*}(T)$ is log-normally distributed and all of its moments are finite. In particular, we have for arbitrary $y = (y_0, y_1)' \in (0, \infty) \times [0, \infty)$

$$\mathbb{E}\left[\mathcal{I}_{g_{\lambda^*}}(y)^2\right] \leq \mathbb{E}\left[\left(B_{ES} + \mathcal{I}(y_0)\right)^2\right]$$
$$\leq 2B_{ES}^2 + 2\mathbb{E}\left[\left(\mathcal{I}(y_0)\right)^2\right]$$
$$= 2B_{ES}^2 + 2y_0^{\frac{2}{b-1}}\mathbb{E}\left[\tilde{Z}_{\lambda^*}(T)^{\frac{2}{b-1}}\right] < \infty,$$

i.e., $\mathcal{I}_{g_{\lambda^*}}(y) \in L^2_Q$. Similarly, we have for arbitrary $y = (y_0, y_1)' \in (0, \infty) \times [0, \infty)$

$$\mathbb{E}\left[\mathcal{I}_{g_{\lambda^*}}(y)\tilde{Z}_{\lambda^*}(T)\right] \le B_{ES}\mathbb{E}[\tilde{Z}_{\lambda^*}(T)] + y_0^{\frac{1}{b-1}}\mathbb{E}\left[\tilde{Z}_{\lambda^*}(T)^{\frac{b}{b-1}}\right] < \infty.$$

For a fixed value of $\tilde{Z}_{\lambda^*}(T)$, the functions $y \to \mathcal{I}_{g_{\lambda^*}}(y)\tilde{Z}_{\lambda^*}(T)$ and $y \to g(\mathcal{I}_{g_{\lambda^*}}(y))$ are continuous in $y \in (0,\infty) \times [0,\infty)$. Further, for any sequence of $(y_n)_{n \in \mathbb{N}}$ taking values in $(0,\infty) \times [0,\infty)$ and converging to the optimal $y^* \in (0,\infty) \times [0,\infty)$, we know that

$$\begin{split} \hat{y}_0 &:= \inf_{n \in \mathbb{N}} \left((y_n)_0 \right) > 0 \\ \left| \mathcal{I}_{g_{\lambda^*}}(y_n) \tilde{Z}_{\lambda^*}(T) \right| &\leq \left(B_{ES} + \mathcal{I}\left((y_n)_0 \right) \right) \tilde{Z}_{\lambda^*}(T) \leq \left(B_{ES} + \mathcal{I}\left(\hat{y}_0 \right) \right) \tilde{Z}_{\lambda^*}(T) \\ \left| g \left(\mathcal{I}_{g_{\lambda^*}}(y_n) \right) \right| &\leq B_{ES}. \end{split}$$

Thus, the dominated convergence theorem implies

$$\lim_{n \to \infty} \mathbb{E} \left[g_{\lambda^*}(\mathcal{I}_{g_{\lambda^*}}(y_n)) \right] = \lim_{n \to \infty} \begin{pmatrix} \mathbb{E} \left[-\mathcal{I}_{g_{\lambda^*}}(y_n) \tilde{Z}_{\lambda^*}(T) \right] \\ \mathbb{E} \left[-g \left(\mathcal{I}_{g_{\lambda^*}}(y_n) \right) \right] \end{pmatrix} \\ = \begin{pmatrix} \mathbb{E} \left[-\mathcal{I}_{g_{\lambda^*}}(y) \tilde{Z}_{\lambda^*}(T) \right] \\ \mathbb{E} \left[-g \left(\mathcal{I}_{g_{\lambda^*}}(y) \right) \right] \end{pmatrix} = \mathbb{E} \left[g_{\lambda^*}(\mathcal{I}_{g_{\lambda^*}}(y)) \right].$$

Hence, $\mathbb{E}\left[g_{\lambda^*}(\mathcal{I}_{g_{\lambda^*}}(y))\right]$ is continuous in $y = y^*$ and therefore $\mathbb{E}\left[g_{\lambda^*}(\mathcal{I}_{g_{\lambda^*}}(y)\right]' \Delta y$ is use in y for all $\Delta \in \mathbb{R} \times [0, \infty)$ and thus, $\lambda^* \in \mathcal{D}'$. In total, the optimality of $\mathcal{I}_{g_{\lambda^*}}(y^*)$ for (**P**) follows from Corollary 3.5.7.

Proof of Example 3.5.14. We may proceed analogously to Example 3.5.14. Let $y \in [0, \infty)^2$. As g is convex, the maximization objective

$$U(x) - y_0 \tilde{Z}_{\lambda^*}(T) x - y_1 g(x)$$

remains strictly concave in $x \ge 0$ and is maximized globally by $x = \mathcal{I}_{g_{\lambda^*}}(y)$ over $x \in \mathbb{R}$. Hence, its maximum over $[B_1, B_2]$ is attained by

$$\mathcal{I}_{q_{\lambda^*}}(y; B_1, B_2) = \operatorname{Cap}(\mathcal{I}(y), B_1, B_2).$$

Since 'capping' is a continuous operation which only further bounds its argument, the rest of this example can be proved using analogous steps as in the proof of Example 3.5.13. \Box

A.3 Proofs Chapter 4

Proof of Lemma 4.3.1. From Corollary 2.1.32 it is known that

$$0 = G_t + vrG_v + (\mu^z)' (\nabla_z G) + \frac{1}{2} \operatorname{Trace} \left[\Sigma^z (\Sigma^z)' \nabla_z^2 G \right] + v \sup_{\pi \in K} \left\{ (\mu - r\mathbb{1})' \pi G_v + (\Sigma^z \rho \Sigma' \pi)' \nabla_z (G_v) + \frac{1}{2} v \|\Sigma' \pi\|^2 G_{vv} \right\} = G_t + vrG_v + (\mu^z)' (\nabla_z G) + \frac{1}{2} \operatorname{Trace} \left[\Sigma^z (\Sigma^z)' \nabla_z^2 G \right] + v \inf_{\lambda \in \mathbb{R}^d} \sup_{\pi \in \mathbb{R}^d} \left\{ \delta_K(\lambda) + \lambda' \pi + (\mu - r\mathbb{1})' \pi G_v + (\Sigma^z \rho \Sigma' \pi)' \nabla_z (G_v) + \frac{1}{2} v \|\Sigma' \pi\|^2 G_{vv} \right\} = G_t + vrG_v + (\mu^z)' (\nabla_z G) + \frac{1}{2} \operatorname{Trace} \left[\Sigma^z (\Sigma^z)' \nabla_z^2 G \right] + v \inf_{\lambda \in \mathbb{R}^d} \sup_{\pi \in \mathbb{R}^d} \left\{ \delta_K(\underbrace{G_v}_{>0} \cdot \lambda) + \underbrace{G_v}_{>0} \cdot \lambda' \pi + (\mu - r\mathbb{1})' \pi G_v + (\Sigma^z \rho \Sigma' \pi)' \nabla_z (G_v) + \frac{1}{2} v \|\Sigma' \pi\|^2 G_{vv} \right\}$$

By assumption, v and $G_v(t, v, z)$ are positive¹. Using that δ_K is a support function and thus positive homogenous of order 1, yields

$$0 = G_t + vrG_v + (\mu^z)' (\nabla_z G) + \frac{1}{2} \operatorname{Trace} \left[\Sigma^z (\Sigma^z)' \nabla_z^2 G \right] + v \inf_{\lambda \in \mathbb{R}^d} \sup_{\pi \in \mathbb{R}^d} \left\{ \left[\delta_K(\lambda) + (\mu + \lambda - r\mathbb{1})'\pi \right] G_v + \left(\Sigma^z \rho \Sigma' \pi \right)' \nabla_z (G_v) + \frac{1}{2} v \|\Sigma' \pi\|^2 G_{vv} \right\}.$$
(A.48)

Further, for every fixed $\lambda \in \mathbb{R}^d$, the first-order optimality condition for the maximization over $\pi \in \mathbb{R}^d$ yields the candidate optimizer π_{λ} through

$$0 \stackrel{!}{=} [\mu + \lambda - r\mathbb{1}] G_v + (\Sigma^z \rho \Sigma')' \nabla_z (G_v) + v \Sigma \Sigma' \pi_\lambda G_{vv}$$

$$\Leftrightarrow \quad \pi_\lambda = -\frac{1}{v G_{vv}} (\Sigma \Sigma')^{-1} \left([\mu + \lambda - r\mathbb{1}] G_v + (\Sigma^z \rho \Sigma')' \nabla_z (G_v) \right).$$

Note that

$$\left[\mu + \lambda - r\mathbb{1}\right]' \pi_{\lambda} G_{v} + \left(\Sigma^{z} \rho \Sigma' \pi_{\lambda}\right)' \nabla_{z} \left(G_{v}\right) = \pi_{\lambda}' \left[\left(\mu + \lambda - r\mathbb{1}\right) G_{v} + \left(\Sigma^{z} \rho \Sigma'\right)' \nabla_{z} \left(G_{v}\right)\right]$$

¹Note that this assumption is valid due to the exponential structure of the wealth process $V^{v_0,\pi}$ and the strict monotonicity of the utility function U.

$$= \pi'_{\lambda} \left[-v G_{vv} \Sigma \Sigma' \pi_{\lambda} \right]$$
$$= -v \| \Sigma' \pi_{\lambda} \|^2 G_{vv}.$$

Since we are maximizing a quadratic function with respect to π , the first-order optimality condition is both necessary and sufficient. Hence, plugging π_{λ} into (A.48) finally yields

$$0 = G_t + vrG_v + (\mu^z)' (\nabla_z G) + \frac{1}{2} \operatorname{Trace} \left[\Sigma^z (\Sigma^z)' \nabla_z^2 G \right] + v \inf_{\lambda \in \mathbb{R}^d} \left\{ \delta_K(\lambda) - \frac{1}{2} v \|\Sigma' \pi_\lambda\|^2 G_{vv} \right\}$$
$$= G_t + vrG_v + (\mu^z)' (\nabla_z G) + \frac{1}{2} \operatorname{Trace} \left[\Sigma^z (\Sigma^z)' \nabla_z^2 G \right]$$
$$+ v \inf_{\lambda \in \mathbb{R}^d} \left\{ \delta_K(\lambda) G_v - \frac{1}{2} \frac{1}{vG_{vv}} \|\Sigma^{-1} [\mu + \lambda - r\mathbb{1}] G_v + (\Sigma^z \rho)' \nabla_z (G_v) \|^2 \right\}.$$

Proof of Lemma 4.3.3. For any $\pi \in \Lambda$ ($\underline{\pi} \in \underline{\Lambda}$) and $\lambda \in \mathcal{D}$ ($\underline{\lambda} \in \underline{\mathcal{D}}$) define the operator

$$\mathcal{H}^{\pi}_{\lambda}[G](t,v,z) = G_t + vrG_v + (\mu^z)' (\nabla_z G) + \frac{1}{2} \operatorname{Trace} \left[\Sigma^z (\Sigma^z)' \nabla_z^2 G \right] + v \Big\{ \left[\delta_K(\lambda) + (\mu + \lambda - r\mathbb{1})'\pi \right] G_v + \left(\Sigma^z \rho \Sigma' \pi \right)' \nabla_z (G_v) + \frac{1}{2} v \|\Sigma' \pi\|^2 G_{vv} \Big\},$$

Let $(\underline{\lambda}^*, \pi) \in \underline{\mathcal{D}} \times \Lambda$ satisfy Condition (UI_{λ}) . Then, by Itô's lemma and the boundedness of the integrands, due to the definition of $\tau_{n,t}^{\underline{\lambda}^*}$, we have

$$\begin{split} \mathbb{E} \left[G\left(\tau_{n,t}^{\lambda^{*}}, V_{\underline{\lambda}^{*}}^{v_{0}, \pi}(\tau_{n,t}^{\lambda^{*}}), z(\tau_{n,t}^{\lambda^{*}})\right) \mid \mathcal{F}_{t} \right] \\ &= G\left(t, V_{\underline{\lambda}^{*}}^{v_{0}, \pi}(t), z(t)\right) + \mathbb{E} \left[\int_{t}^{\tau_{n,t}^{\lambda^{*}}} \underbrace{\mathcal{H}_{\underline{\lambda}^{*}}^{\pi}[G]\left(s, V_{\underline{\lambda}^{*}}^{v_{0}, \pi}(s), z(s)\right)}_{\leq \mathcal{H}_{\underline{\lambda}^{*}}^{\pi^{*}}[G]\left(s, V_{\underline{\lambda}^{*}}^{v_{0}, \pi}(s), z(s)\right)} ds \mid \mathcal{F}_{t} \right] \\ &+ \underbrace{\mathbb{E} \left[\int_{t}^{\tau_{n,t}^{\lambda^{*}}} V_{\underline{\lambda}^{*}}^{v_{0}, \pi}(s) G_{v}\left(s, V_{\underline{\lambda}^{*}}^{v_{0}, \pi}(s), z(s)\right) \cdot \pi(s)' \Sigma(s, z(s)) dW(s) \mid \mathcal{F}_{t} \right]}_{=0} \right] \\ &+ \underbrace{\mathbb{E} \left[\int_{t}^{\tau_{n,t}^{\lambda^{*}}} \nabla_{z} \left(G\right) \left(s, V_{\underline{\lambda}^{*}}^{v_{0}, \pi}(s), z(s)\right)' \Sigma^{z}(s, z(s)) dW^{z}(s) \mid \mathcal{F}_{t} \right]}_{=0} \right] \\ &\leq G(t, V_{\underline{\lambda}^{*}}^{v_{0}, \pi}(t), z(t)) \end{split}$$

Taking conditional expectations and the limit $n \to \infty$ on both sides as well as recalling Remark 4.3.2 yields

$$\begin{split} \mathbb{E}\Big[U(V_{\underline{\lambda}^*}^{v_0,\pi}(T)) \mid \mathcal{F}_t\Big] &= \mathbb{E}\Big[G(T,V_{\underline{\lambda}^*}^{v_0,\pi}(T),z(T)) \mid \mathcal{F}_t\Big] \\ &= \mathbb{E}\Big[\lim_{n \to \infty} G(\tau_{n,t}^{\underline{\lambda}^*},V_{\underline{\lambda}^*}^{v_0,\pi}(\tau_{n,t}^{\underline{\lambda}^*}),z(\tau_{n,t}^{\underline{\lambda}^*})) \mid \mathcal{F}_t\Big] \\ &\stackrel{(\mathrm{UI}_{\lambda})}{=} \lim_{n \to \infty} \mathbb{E}\Big[G(\tau_{n,t}^{\underline{\lambda}^*},V_{\underline{\lambda}^*}^{v_0,\pi}(\tau_{n,t}^{\underline{\lambda}^*}),z(\tau_{n,t}^{\underline{\lambda}^*})) \mid \mathcal{F}_t\Big] \\ &\leq \lim_{n \to \infty} G\big(t,V_{\underline{\lambda}^*}^{v_0,\pi}(t),z(t)\big) \\ &= G\big(t,V_{\underline{\lambda}^*}^{v_0,\pi}(t),z(t)\big). \end{split}$$

Conditioning on $V_{\underline{\lambda}^*}^{v_0,\pi}(t) = v, \, z(t) = z$ leads to (4.10).

Letting $(\lambda, \pi^*) \in \overline{\mathcal{D}} \times \Lambda$ satisfy Condition (UI_{λ}) and following the analogous steps as before, we can prove equation (4.11).

As all previous inequalities become equalities if we instead consider $(\underline{\lambda}^*, \underline{\pi}^*) \in \mathcal{D} \times \underline{\Lambda}$ satisfying Condition (UI_{λ}) , equation (4.12) follows immediately. \Box

Proof of Theorem 4.3.5. We first determine the derivatives of G = G(t, v, z) in terms of A = A(T - t) and B = B(T - t) as

$$G_t = -\left(A_\tau + B'_\tau z\right)G, \quad G_v = \frac{b}{v}G, \quad G_{vv} = \frac{b(b-1)}{v^2}G$$

$$\nabla_z G = GB, \quad \nabla_z^2 G = GBB', \quad \nabla_z \left(G_v\right) = \frac{b}{v}GB.$$
(A.49)

Plugging the derivatives into the dual HJBI PDE (4.7), factoring $\frac{bG}{2v(1-b)} > 0$ out of the minimization, dividing by $G \neq 0$ and plugging in the optimizer $\hat{\lambda}^* = \hat{\lambda}^*(t, z, B)$ yields

$$\begin{split} (*) &:= G_t + vrG_v + (\mu^z)' (\nabla_z G) + \frac{1}{2} \text{Trace} \left[\Sigma^z (\Sigma^z)' \nabla_z^2 G \right] \\ &+ v \inf_{\lambda \in \mathbb{R}^d} \left\{ \delta_K(\lambda) G_v - \frac{1}{2} \frac{1}{vG_{vv}} \| \Sigma^{-1} [\mu + \lambda - r\mathbf{1}] G_v + (\Sigma^z \rho)' \nabla_z (G_v) \|^2 \right\} \\ &= - (A_\tau - B'_\tau z) G + brG + (\mu^z)' BG + \frac{1}{2} \text{Trace} \left[\Sigma^z (\Sigma^z)' BB' \right] G \\ &+ v \inf_{\lambda \in \mathbb{R}^d} \left\{ \delta_K(\lambda) \frac{b}{v} G - \frac{1}{2} \frac{v}{b(b-1)G} \right\| \Sigma^{-1} [\mu + \lambda - r\mathbf{1}] \frac{b}{v} G + (\Sigma^z \rho)' \frac{b}{v} BG \right\|^2 \right\} \\ &= - (A_\tau - B'_\tau z) G + brG + (\mu^z)' BG + \frac{1}{2} \text{Trace} \left[\Sigma^z (\Sigma^z)' BB' \right] G \\ &+ \frac{1}{2} \frac{bG}{1 - b} \inf_{\lambda \in \mathbb{R}^d} \left\{ 2(1 - b) \delta_K(\lambda) + \| \Sigma^{-1} [\mu + \lambda - r\mathbf{1}] + (\Sigma^z \rho)' B \|^2 \right\} \\ &= -A_\tau - B'_\tau z + br + (\mu^z)' B + \frac{1}{2} \text{Trace} \left[\Sigma^z (\Sigma^z)' BB' \right] \\ &+ \frac{1}{2} \frac{b}{1 - b} \inf_{\lambda \in \mathbb{R}^d} \left\{ 2(1 - b) \delta_K(\lambda) + \| \Sigma^{-1} [\mu + \lambda - r\mathbf{1}] + (\Sigma^z \rho)' B \|^2 \right\} \\ &= -A_\tau - B'_\tau z + br + (\mu^z)' B + \frac{1}{2} \text{Trace} \left[\Sigma^z (\Sigma^z)' BB' \right] \\ &+ \frac{1}{2} \frac{b}{1 - b} \inf_{\lambda \in \mathbb{R}^d} \left\{ 2(1 - b) \delta_K(\lambda) + 2\lambda' (\Sigma\Sigma')^{-1} (\mu - r\mathbf{1} + (\Sigma^z \rho\Sigma')' B) + \| \Sigma^{-1} \lambda \|^2 \right\} \\ &+ \frac{1}{2} \frac{b}{1 - b} \left\| \Sigma^{-1} [\mu - r\mathbf{1}] + (\Sigma^z \rho)' B \right\|^2 \\ &= -A_\tau - B'_\tau z + br + (\mu^z)' B + \frac{1}{2} \text{Trace} \left[\Sigma^z (\Sigma^z)' BB' \right] \\ &+ \frac{1}{2} \frac{b}{1 - b} \left\| \Sigma^{-1} [\mu - r\mathbf{1}] + (\Sigma^z \rho)' B \right\|^2 \\ &= -A_\tau - B'_\tau z + br + (\mu^z)' B + \frac{1}{2} \text{Trace} \left[\Sigma^z (\Sigma^z)' BB' \right] \\ &+ \frac{1}{2} \frac{b}{1 - b} \left\| \Sigma^{-1} [\mu - r\mathbf{1}] + (\Sigma^z \rho)' B \right\|^2 \\ &= -A_\tau - B'_\tau z + br + (\mu^z)' B + \frac{1}{2} \text{Trace} \left[\Sigma^z (\Sigma^z)' BB' \right] \\ &+ \frac{1}{2} \frac{b}{1 - b} \left\| \Sigma^{-1} [\mu - r\mathbf{1}] + (\Sigma^z \rho)' B \right\|^2 \\ &= -A_\tau - B'_\tau z + b \left(r + \delta_K (\hat{\lambda}^*) \right) + (\mu^z)' B + \frac{1}{2} \text{Trace} \left[\Sigma^z (\Sigma^z)' BB' \right] \\ &+ \frac{1}{2} \frac{b}{1 - b} \left\| (\| \Sigma^{-1} (\mu + \hat{\lambda}^* - r\mathbf{1}) \|^2 + 2B' \Sigma^z \rho \Sigma^{-1} (\mu + \hat{\lambda}^* - r\mathbf{1}) + B' \Sigma^z \rho (\Sigma^z \rho)' B \right)$$

Condition (EAS) allows us to replace the market coefficients by affine functions in z. Thus, we obtain

$$\begin{aligned} (*) &= -A_{\tau} - B_{\tau}'z + b\left(p_{0} + p_{1}'z\right) + (k_{0} + k_{1}z)'B + \frac{1}{2}\operatorname{Trace}\left[\left(h_{0} + h_{1}[z]\right)BB'\right] \\ &+ \frac{1}{2}\frac{b}{1-b}\left(q_{0} + q_{1}'z + 2B'\left(g_{0} + g_{1}z\right) + B'\left(l_{0} + h_{0} + l_{1}[z] + h_{1}[z]\right)B\right) \\ &= -A_{\tau} + bp_{0} + k_{0}'B + \frac{1}{2}\operatorname{Trace}\left[h_{0}BB'\right] + \frac{1}{2}\frac{b}{1-b}\left(q_{0} + 2g_{0}'B + B'\left(l_{0} + h_{0}\right)B\right) \\ &- B_{\tau}'z + bp_{1}'z + B'k_{1}z + \frac{1}{2}\operatorname{Trace}\left[h_{1}[z]BB'\right] + \frac{1}{2}\frac{b}{1-b}\left(q_{1}'z + 2B'g_{1}z + B'\left(l_{1}[z] + h_{1}[z]\right)B\right) \end{aligned}$$

Making use of the commutative property of the trace of a matrix as well as the matrix representation of $h_1[\cdot]$ and $l_1[\cdot]$, we have

$$\operatorname{Trace} \left[h_0 BB'\right] = \operatorname{Trace} \left[\underbrace{B'h_0 B}_{\in \mathbb{R}}\right] = B'h_0 B$$
$$\operatorname{Trace} \left[h_1[z]BB'\right] = \operatorname{Trace} \left[\underbrace{B'h_1[z]B}_{\in \mathbb{R}}\right] = B'h_1[z]B = \left(B'h_1[\cdot]B\right)' z$$
$$B'\left(l_1[z] + h_1[z]\right) B = \left(B'\left(l_1[\cdot] + h_1[\cdot]\right)B\right)' z.$$

Thus, as A and B are solutions to the ODEs (4.15) and (4.16)

$$\begin{aligned} (*) &= -A_{\tau} + bp_0 + k'_0 B + \frac{1}{2} B' h_0 B + \frac{1}{2} \frac{b}{1-b} \Big(q_0 + 2g'_0 B + B' \left(l_0 + h_0 \right) B \Big) \\ &+ \left(-B_{\tau} + bp_1 + k'_1 B + \frac{1}{2} B' h_1 [\cdot] B + \frac{1}{2} \frac{b}{1-b} \Big(q_1 + 2g'_1 B + B' \left(l_1 [\cdot] + h_1 [\cdot] \right) B \Big) \Big)' z \\ &= 0. \end{aligned}$$

Hence, G is a solution to the dual HJBI PDE (4.7) and thereby, according to Lemma 4.3.1, also a solution to the primal HJB PDE (4.5). \Box

Proof of Theorem 4.3.7. We first derive the explicit expression (4.17) for $\underline{\pi}^*$ in terms of $\underline{\lambda}^*$ and B. Following the arguments in the proof of Lemma 4.3.1, we realize that the candidate optimal portfolio $\underline{\pi}^*$ is given as the maximizing argument π_{λ^*} , i.e.,

$$\underline{\pi}^* = \pi_{\lambda^*} = -\frac{1}{vG_{vv}} \left(\Sigma\Sigma'\right)^{-1} \left[G_v(\mu + \underline{\lambda}^* - r\mathbb{1}) + \Sigma\rho'(\Sigma^z)'\nabla_z(G_v)\right].$$
(A.50)

On the other hand, since Condition (EAS) is satisfied, the exponentially affine structure of G from (4.13) implies that the derivatives of G are given by (A.49). Plugging these derivatives into (A.50) yields the candidate optimal portfolio

$$\overline{\pi}^* = -\left(\frac{b(b-1)}{v}G\right)^{-1} \left(\Sigma\Sigma'\right)^{-1} \left[\frac{b}{v}G(\mu+\underline{\lambda}^*-r\mathbb{1}) + \Sigma\rho'\left(\Sigma^z\right)'\frac{b}{v}GB\right] \\
= \frac{1}{1-b} \left(\Sigma\Sigma'\right)^{-1} \left[\mu+\underline{\lambda}^*-r\mathbb{1} + \left(\Sigma^z\rho\Sigma'\right)'B\right].$$

We continue by proving (4.18). As G is a solution to the (dual) HJB PDE (4.7) and $\underline{\pi}^*$, $\underline{\lambda}^*$ attain the optimum in (4.7), we know for every $t \in [0, T]$

$$dG(t, V^{v_0, \underline{\pi}^*}(t), z(t)) = V^{v_0, \underline{\pi}^*}(t)G_v\left(t, V^{v_0, \underline{\pi}^*}(t), z(t)\right)\left(\underline{\pi}^*(t, V^{v_0, \underline{\pi}^*}(t), z(t))\right)' \Sigma dW(t)$$

+
$$\left(\nabla_{z}(G)(t, V^{v_{0}, \underline{\pi}^{*}}(t), z(t))\right)' \Sigma^{z} dW^{z}(t).$$
 (A.51)

In particular, noting that $V^{v_0, \underline{\pi}^*} = V_0^{v_0, \underline{\pi}^*}$ and considering the stopping times $\tau_{n,t}^0$ for $n \in \mathbb{N}$ from Condition (UI₀) yields

$$\begin{split} \mathbb{E}\Big[U(V^{v_0,\pi^*}(T)) \mid \mathcal{F}_t\Big] &= \mathbb{E}\Big[G\left(T, V^{v_0,\pi^*}(T), z(T)\right) \mid \mathcal{F}_t\Big] \\ &= \mathbb{E}\Big[\lim_{n \to \infty} G\left(\tau_{n,t}^0, V^{v_0,\pi^*}(\tau_{n,t}^0), z(\tau_{n,t}^0)\right) \mid \mathcal{F}_t\Big] \\ \stackrel{(\mathrm{UI}_0)}{=} \lim_{n \to \infty} \mathbb{E}\Big[G\left(\tau_{n,t}^0, V^{v_0,\pi^*}(\tau_{n,t}^0), z(\tau_{n,t}^0)\right) \mid \mathcal{F}_t\Big] \\ &= \lim_{n \to \infty} \left(G\left(t, V^{v_0,\pi^*}(t), z(t)\right) \\ &+ \mathbb{E}\Big[\int_t^{\tau_{n,t}^0} V^{v_0,\pi^*}(u)G_v\left(u, V^{v_0,\pi^*}(u), z(u)\right)\left(\pi^*(u, V^{v_0,\pi^*}(u), z(u))\right)'\Sigma(u, z(u))dW(u) \mid \mathcal{F}_t\Big] \\ &= 0, \text{ by choice of } \tau_{n,t}^0 \\ &+ \mathbb{E}\Big[\int_t^{\tau_{n,t}^0} \left(\nabla_z\left(G\right)\left(u, V^{v_0,\pi^*}(u), z(u)\right)\right)'\Sigma^z(u, z(u))dW^z(u) \mid \mathcal{F}_t\Big] \Big) \\ &= 0, \text{ by choice of } \tau_{n,t}^0 \\ &= G\left(t, V^{v_0,\pi^*}(t), z(t)\right). \end{split}$$

Conditioning on $V^{v_0,\pi^*}(t) = v, \ z(t) = z$ leads to equation (4.18).

It remains to show that $\underline{\pi}^*$ dominates all other $\pi \in \Lambda_K$, i.e., we need to verify inequality (4.19). The proof idea is similar to that of Theorem 4.3 in [13], but adapted for our constrained setting. Let $t \in [0, T]$ and $\pi \in \Lambda_K(t)$ be arbitrary but fixed. Define the process $L = (L(u))_{u \in [t,T]}$ as

$$L(u) = b \frac{V^{v_0, \pi}(u)}{V^{v_0, \underline{\pi}^*}(u)} G(u, V^{v_0, \underline{\pi}^*}(u), z(u)).$$

We proceed by first deriving the SDE of L and then showing that L is a supermartingale. By Itô's product rule we have with $\pi^*(u) = \pi^*(u, V^{v_0, \pi^*}(u), z(u))$

$$\begin{split} d\left(\frac{V^{v_{0},\pi}(u)}{V^{v_{0},\pi^{*}}(u)}\right) &= \frac{1}{V^{v_{0},\pi^{*}}(u)} dV^{v_{0},\pi}(u) + V^{v_{0},\pi}(u) d\left(\frac{1}{V^{v_{0},\pi^{*}}(u)}\right) + d\langle V^{v_{0},\pi}(u), \frac{1}{V^{v_{0},\pi^{*}}(u)}\rangle_{u} \\ &= \left(\frac{V^{v_{0},\pi}(u)}{V^{v_{0},\pi^{*}}(u)}\right) \left(\left[r + (\mu - r\mathbf{1})'\pi(u)\right] du + \pi(u)'\Sigma dW(u) \\ &- \left[r + (\mu - r\mathbf{1})'\pi^{*}(u) - \|\Sigma'\pi^{*}(u)\|^{2}\right] du - \pi^{*}(u)'\Sigma dW(u) - \pi^{*}(u)'\Sigma\Sigma'\pi(u) du\right) \\ &= \left(\frac{V^{v_{0},\pi}(u)}{V^{v_{0},\pi^{*}}(u)}\right) \left(\left[(\mu - r\mathbf{1})'(\pi(u) - \pi^{*}(u)) \underbrace{-\pi^{*}(u)'\Sigma\Sigma'\pi(u) + \|\Sigma'\pi^{*}(u)\|^{2}}_{=-\pi^{*}(u)'\Sigma\Sigma'(\pi(u) - \pi^{*}(u))} + (\pi(u) - \pi^{*}(u))'\Sigma dW(u)\right) \\ &+ (\pi(u) - \pi^{*}(u))'\Sigma dW(u) \right) \\ &= \left(\frac{V^{v_{0},\pi}(u)}{V^{v_{0},\pi^{*}}(u)}\right) (\pi(u) - \pi^{*}(u))' \left(\left[\mu - r\mathbf{1} - \Sigma\Sigma'\pi^{*}(u)\right] du + \Sigma dW(u)\right). \end{split}$$

Moreover, due to (A.49) and (A.51),

$$dG(u, V^{v_0, \pi^*}(u), z(u)) \stackrel{(A.51)}{=} G(u, V^{v_0, \pi^*}(u), z(u)) \Big(b\pi^*(u)' \Sigma dW(u) + B(T-u)' \Sigma^z dW^z(u) \Big)$$

and

$$d \left\langle \frac{V^{v_0,\pi}}{V^{v_0,\pi^*}}, G(\cdot, V^{v_0,\pi^*}, z) \right\rangle_u \\ = \frac{V^{v_0,\pi}(u)}{V^{v_0,\pi^*}(u)} G(u, V^{v_0,\pi^*}(u), z(u)) (\pi(u) - \pi^*(u))' \Sigma \left(b \Sigma' \pi^*(u) + (\Sigma^z \rho)' B(T-u) \right) du.$$

In total, this yields with $\lambda^*(u) = \lambda^*(u, V^{v_0, \underline{\pi}^*}(u), z(u))$

$$\begin{split} dL(u) &= bd\left(\frac{V^{v_0,\pi}(u)}{V^{v_0,\pi^*}(u)}G(u,V^{v_0,\pi^*}(u),z(u))\right) \\ &= b\left[\frac{V^{v_0,\pi}(u)}{V^{v_0,\pi^*}(u)}dG(u,V^{v_0,\pi^*}(u),z(u))G(u,V^{v_0,\pi^*}(u),z(u))d\left(\frac{V^{v_0,\pi}(u)}{V^{v_0,\pi^*}(u)}\right)\right] \\ &\quad + bd\left\langle\frac{V^{v_0,\pi}}{V^{v_0,\pi^*}},G(\cdot,V^{v_0,\pi^*},z)\right\rangle_u \\ &= L(u)\Big[(\pi(u)-\pi^*(u))'\left[\mu-r\mathbf{1}-\Sigma\Sigma'\pi^*(u)\right]du+\Sigma dW(u)+b\pi^*(u)'\Sigma dW(u) \\ &\quad + B(T-u)'\Sigma^z dW^z(u)+(\pi(u)-\pi^*(u))'\Sigma\left(b\Sigma'\pi^*(u)+(\Sigma^z\rho)'B(T-u)\right)du\Big] \\ &= L(u)(\pi(u)-\pi^*(u))'\left[\mu-r\mathbf{1}+(\Sigma^z\rho\Sigma')'B(T-u)-(1-b)\Sigma\Sigma'\pi^*(u)\right]du \\ &\quad + L(u)\Big((\pi(u)-(1-b)\pi^*(u))'\Sigma dW(u)+B(T-u)'\Sigma^z dW^z(u)\Big). \\ &= L(u)\Big((\pi^*(u)-\pi(u))'\lambda^*(u)du+(\pi(u)-(1-b)\pi^*(u))'\Sigma dW(u)+B(T-u)'\Sigma^z dW^z(u)\Big). \end{split}$$

However, Corollary 2.1.32 (resp. Theorem 2.1.31) implies

$$\delta_K(\underline{\lambda}^*(u,v,z)) + \underline{\pi}^*(u,v,z)'\underline{\lambda}^*(u,v,z) = 0 \quad \forall (u,v,z) \in [0,T] \times (0,\infty) \times \mathbb{R}^m$$

$$\Rightarrow \delta_K(\underline{\lambda}^*(u)) + \underline{\lambda}^*(u)'\underline{\pi}^*(u) = 0 \quad \mathcal{L}[t,T] \otimes Q - \text{a.e.}.$$

Moreover, as $\pi \in \Lambda_K(t)$, $\pi(u) \in K$ holds $\mathcal{L}[t,T] \otimes Q$ – a.e. and thus

$$\delta_K(\lambda^*(u)) + \lambda^*(u)'\pi(u) = -\inf_{\substack{x \in K \\ \leq \lambda^*(u)'\pi(u)}} (\lambda^*(u)'x) + \lambda^*(u)'\pi(u) \ge 0 \quad \mathcal{L}[t,T] \otimes Q - \text{a.e.}.$$

We finally obtain

$$dL(u) = L(u) \left(\underbrace{-\left(\delta_K(\lambda^*(u)) + \lambda^*(u)'\pi(u)\right)}_{=(*)} du + (\pi(u) - (1-b)\pi^*(u))'\Sigma dW(u) + B(T-u)'\Sigma^z dW^z(u) \right)$$

and for any $s \in [t,T]$, L(s) can therefore be expressed as

$$L(s) = L(t) \exp\left(-\int_t^s \delta_K(\lambda^*(u)) + \lambda^*(u)'\pi(u)du\right) M(s),$$

for a supermartingale $M=(M(u))_{u\in[t,T]}$ which satisfies the SDE

$$dM(u) = M(u) \left((\pi(u) - (1-b)\pi^*(u))' \Sigma dW(u) + B(T-u)' \Sigma^z dW^z(u) \right), \quad M(t) = 1.$$
Hence, as $(*) \leq 0$, L is a supermartingale, too.

To conclude the proof, recall that U is concave and therefore $U(y) \leq U(x) + U'(x)(y-x)$ for all $x, y \in (0, \infty)$. This leads to

$$\begin{split} \mathbb{E} \left[U(V^{v_{0},\pi}(T)) \mid \mathcal{F}_{t} \right] &\leq \mathbb{E} \left[U(V^{v_{0},\underline{\pi}^{*}}(T)) \mid \mathcal{F}_{t} \right] + \mathbb{E} \left[U'(V^{v_{0},\underline{\pi}^{*}}(T)) \left(V^{v_{0},\pi}(T) - V^{v_{0},\underline{\pi}^{*}}(T) \right) \mid \mathcal{F}_{t} \right] \\ &= \mathbb{E} \left[U(V^{v_{0},\underline{\pi}^{*}}(T)) \mid \mathcal{F}_{t} \right] + \underbrace{\mathbb{E} \left[L(T) \mid \mathcal{F}_{t} \right]}_{\leq L(t)} - \underbrace{\mathbb{E} \left[bU(V^{v_{0},\underline{\pi}^{*}}(T)) \mid \mathcal{F}_{t} \right]}_{=bG(t,V^{v_{0},\underline{\pi}^{*}}(t),z(t))} \\ &\leq \mathbb{E} \left[U(V^{v_{0},\underline{\pi}^{*}}(T)) \mid \mathcal{F}_{t} \right] + L(t) - bG(t,V^{v_{0},\underline{\pi}^{*}}(t),z(t)) \\ &= \mathbb{E} \left[U(V^{v_{0},\underline{\pi}^{*}}(T)) \mid \mathcal{F}_{t} \right] + b \left(G(t,V^{v_{0},\pi}(t),z(t)) \frac{V^{v_{0},\pi}(t)}{V^{v_{0},\underline{\pi}^{*}}(t)} - G(t,V^{v_{0},\underline{\pi}^{*}}(t),z(t)) \right). \end{split}$$

Finally, conditioning on $V^{v_0,\pi}(t) = V^{v_0,\pi^*}(t) = v, \ z(t) = z$ yields (4.19).

Proof of Lemma 4.4.2. In \mathcal{M}_{BS} , the optimizer $\hat{\lambda}^*$ from (4.14) is given as

$$\hat{\lambda}^*(t,z,B) = \underset{\lambda \in \mathbb{R}^d}{\operatorname{argmin}} \left\{ 2(1-b)\delta_K(\lambda) + \left\| \underbrace{\Sigma(t,z)^{-1}}_{\Sigma(t)^{-1}} \left(\underbrace{\mu(t,z) - r(t,z)\mathbb{1}}_{=\mu(t) - r(t)\mathbb{1}} + \lambda \right) + \underbrace{(\Sigma^z(t,z)\rho(t,z))'}_{=0} B \right\|^2 \right\}$$
$$= \underset{\lambda \in \mathbb{R}^d}{\operatorname{argmin}} \left\{ 2(1-b)\delta_K(\lambda) + \left\| \Sigma(t)^{-1} \left(\mu(t) - r(t)\mathbb{1} + \lambda \right) \right\|^2 \right\}$$
$$= \lambda^*(t),$$

i.e., $\hat{\lambda}^*(t, v, z) = \lambda^*(t)$ is a deterministic function independent of z and B. Moreover, by defining

$$p_{0}(t, B) = r(t) + \delta_{K} (\lambda^{*}(t))$$

$$q_{0}(t, B) = \left\| \Sigma(t)^{-1} (\mu(t) - r(t)\mathbb{1} + \lambda^{*}(t)) \right\|^{2}$$

and setting the remaining coefficients k_0 , k_1 , h_0 , h_1 , l_0 , l_1 , p_1 , q_1 , g_0 and g_1 to zero, Condition (EAS) is satisfied. The corresponding ODEs (4.15) and (4.16) simplify to

$$A_{\tau}(\tau) = b \left(r(T-\tau) + \delta_K \left(\lambda^*(T-\tau) \right) \right) + \frac{1}{2} \frac{b}{1-b} \left\| \Sigma(t)^{-1} \left(\mu(T-\tau) - r(T-\tau) \mathbb{1} + \lambda^*(T-\tau) \right) \right\|^2$$

= $br(T-\tau) + \frac{1}{2} \frac{b}{1-b} \inf_{\lambda \in \mathbb{R}^d} \left\{ 2(1-b)\delta_K(\lambda) + \left\| \Sigma(T-\tau)^{-1} \left(\mu(T-\tau) - r(T-\tau) \mathbb{1} + \lambda \right) \right\|^2 \right\}$
 $B_{\tau}(\tau) = 0.$

Hence, $B \equiv 0$ and A can be obtained through simple integration. In particular, the candidate optimal portfolio (4.17) is given through

$$\underline{\pi}^{*}(t,v,z) = \frac{1}{1-b} \left(\Sigma(t)\Sigma(t)' \right)^{-1} \left(\mu(t) - r(t)\mathbb{1} + \lambda^{*}(t) \right).$$

Proof of Corollary 4.4.3. We verify Condition (UI_{λ}) by showing the L^q boundedness of

$$G\left(\tau_{n,t}^{0}, V_{0}^{v_{0}, \underline{\pi}^{*}}(\tau_{n,t}^{0}), z(\tau_{n,t}^{0})\right)$$

in $n \in \mathbb{N}$ for arbitrary q > 1.

The market coefficients $\Sigma(t)$, $\mu(t)$ and r(t) are continuous and therefore uniformly bounded in $t \in [0, T]$. In particular, this has the consequence that $\lambda^*(t)$ is uniformly bounded in $t \in [0, T]$, since the quadratic term in (4.14) dominates for large $\|\lambda\|$. Hence, A(T - t) and the candidate optimal portfolio $\pi^*(t) := \pi^*(t, v, z)$ are uniformly bounded in $t \in [0, T]$, too. For arbitrary q > 1, we can thus find a constant $C_q > 0$ such that for all $t \in [0, T]$

$$\begin{aligned} \left| G(t, V^{v_0, \overline{\pi}^*}(t), z(t)) \right|^{q} \stackrel{\text{T. } \underline{4}.3.5}{=} \left| \frac{1}{b} \exp\left(b \ln\left(V^{v_0, \overline{\pi}^*}(t) \right) + A(T-t) + B(T-t)'z(t) \right) \right|^{q} \\ &= \frac{1}{|b|} \exp\left(bq \int_{0}^{t} r(s) + (\mu(s) - r(s)\mathbb{1})' \pi^*(s) - \frac{1}{2} \|\Sigma(s)'\pi^*(s)\|^2 ds + bq \int_{0}^{t} \pi^*(s)'\Sigma(s) dW(s) + qA(T-t) \right) \\ &= \frac{1}{|b|} \exp\left(bq \int_{0}^{t} r(s) + (\mu(s) - r(s)\mathbb{1})' \pi^*(s) - \frac{1 - bq}{2} \|\Sigma(s)'\pi^*(s)\|^2 ds + qA(T-t) \right) \\ &\quad - \frac{1}{2} \int_{0}^{t} b^2 q^2 \|\Sigma(s)'\pi^*(s)\|^2 ds + bq \int_{0}^{t} \pi^*(s)'\Sigma(s) dW(s) \right) \\ &\leq C_q \exp\left(-\frac{1}{2} \int_{0}^{t} b^2 q^2 \|\Sigma(s)'\pi^*(s)\|^2 ds + bq \int_{0}^{t} \pi^*(s)'\Sigma(s) dW(s) \right) \\ &= M_t \end{aligned}$$
(A.52)

The process $M = (M_t)_{t \in [0,T]}$ is a non-negative local martingale and thus a supermartingale. Doob's optional sampling theorem ('O.S.') implies

$$\sup_{n\in\mathbb{N}}\mathbb{E}\left[\left|G\left(\tau_{n,t}^{0},V_{0}^{v_{0},\underline{\pi}^{*}}(\tau_{n,t}^{0}),z(\tau_{n,t}^{0})\right)\right|^{q}\right] \stackrel{(A.52)}{\leq} C_{q}\sup_{n\in\mathbb{N}}\mathbb{E}\left[M_{\tau_{n,t}^{0}}\right] \stackrel{O.S.}{\leq} C_{q}M_{0}=C_{q}<\infty.$$

Hence,

$$\left(G\left(\tau_{n,t}^0,V_0^{v_0,\underline{\pi}^*}(\tau_{n,t}^0),z(\tau_{n,t}^0)\right)\right)_{n\in\mathbb{N}}$$

is bounded in L^q for any q > 1 and $t \in [0, T]$ and is thus uniformly integrable for any $t \in [0, T]$ (see Theorem 4.5.9 in [8] with $G(t) = t^q$). Hence, Condition (UI_{λ}) is satisfied and $\underline{\pi}^*$ is optimal for (**P**) by virtue of Theorem 4.3.7.

Proof of Lemma 4.4.6. We may rewrite any $\lambda \in \mathbb{R}^d$ as

$$\lambda = \begin{pmatrix} \lambda_1 \\ \vdots \\ \lambda_m \end{pmatrix}, \quad \text{with} \quad \lambda_i \in \mathbb{R}^{d_i}, \quad i = 1, ..., m$$

Hence, we obtain for every $\lambda \in \mathbb{R}^d$

$$\delta_K(\lambda) = -\inf_{x \in K} \left(x'\lambda \right) = -\inf_{\substack{x_i \in K_i \\ i=1,\dots,m}} \left(\sum_{i=1}^m x'_i \lambda_i \right) = -\sum_{i=1}^m \inf_{x_i \in K_i} \left(x'_i \lambda_i \right) = \sum_{i=1}^m \delta_{K_i}(\lambda_i).$$

Note that we may restrict the minimization (4.14) to $z \in (0, \infty)^m$ because Feller's condition (4.21) ensures that the *m*-dimensional CIR process has positive components $\mathcal{L}[0, T] \otimes Q$ -a.e.. Hence, for any $(t, z, B) \in [0, T] \times (0, \infty)^m \times \mathbb{R}^m$ we have

$$\begin{bmatrix} \left(\Sigma(t,z)\Sigma(t,z)'\right)^{-1}\lambda \end{bmatrix}' [\mu(t,z) - r(t,z)\mathbb{1}] = \begin{pmatrix} \frac{1}{z_1} \left(\Sigma_1 \Sigma_1'\right)^{-1} \lambda_1 \\ \vdots \\ \frac{1}{z_m} \left(\Sigma_m \Sigma_m'\right)^{-1} \lambda_m \end{pmatrix}' \begin{pmatrix} \eta_1 z_1 \\ \vdots \\ \eta_m z_m \end{pmatrix}$$
$$= \sum_{i=1}^m \lambda_i \left(\Sigma_i \Sigma_i'\right)^{-1} \eta_i,$$

$$\begin{split} \left[\Sigma(t,z)^{-1}\lambda \right]' \rho' \Sigma^{z}(t,z) B &= \begin{pmatrix} \frac{1}{\sqrt{z_{1}}} \Sigma_{1}^{-1}\lambda_{1} \\ \vdots \\ \frac{1}{\sqrt{z_{m}}} \Sigma_{m}^{-1}\lambda_{m} \end{pmatrix}' \begin{pmatrix} \rho_{1} & 0 \\ \ddots \\ 0 & \rho_{m} \end{pmatrix} \begin{pmatrix} \sigma_{1}\sqrt{z_{1}}B_{1} \\ \vdots \\ \sigma_{m}\sqrt{z_{m}}B_{m} \end{pmatrix} \\ &= \begin{pmatrix} \frac{1}{\sqrt{z_{1}}} \Sigma_{1}^{-1}\lambda_{1} \\ \vdots \\ \frac{1}{\sqrt{z_{m}}} \Sigma_{m}^{-1}\lambda_{m} \end{pmatrix}' \begin{pmatrix} \sigma_{1}\sqrt{z_{1}}B_{1}\rho_{1} \\ \vdots \\ \sigma_{m}\sqrt{z_{1}}B_{m}\rho_{m} \end{pmatrix} \\ &= \sum_{i=1}^{m} \sigma_{i}B_{i} \left(\Sigma_{i}^{-1}\lambda_{i}\right)' \rho_{i}, \end{split}$$

and

$$\lambda' \left(\Sigma(t, z) \Sigma(t, z)' \right)^{-1} \lambda = \sum_{i=1}^{m} \left\| \Sigma_i^{-1} \lambda_i \right\|^2 z_i.$$

Hence, the minimizer of (4.14) can be rewritten as

$$\hat{\lambda}^{*}(t,z,B) = \underset{\lambda \in \mathbb{R}^{d}}{\operatorname{argmin}} \left\{ 2(1-b)\delta_{K}(\lambda) + 2\lambda' \left(\Sigma(t,z)\Sigma(t,z)'\right)^{-1} \left[\mu(t,z) - r(t,z)\mathbb{1}\right] \\
+ 2\lambda' \left(\Sigma(t,z)\Sigma(t,z)'\right)^{-1} \left(\Sigma^{z}(t,z)\rho(t,z)\Sigma(t,z)'\right)'B + \left\|\Sigma(t,z)^{-1}\lambda\right\|^{2} \right\} \\
= \underset{\lambda=(\lambda_{1},\dots,\lambda_{m})'}{\operatorname{argmin}} \left\{ \sum_{i=1}^{m} 2(1-b)\delta_{K_{i}}(\lambda_{i}) + 2\left(\Sigma_{i}^{-1}\lambda_{i}\right)' \left(\Sigma_{i}^{-1}\eta_{i} + \sigma_{i}B_{i}\rho_{i}\right) + \left\|\Sigma_{i}^{-1}\lambda_{i}\right\|^{2} z_{i} \right\} \\
= \underset{\lambda=(\lambda_{1},\dots,\lambda_{m})'}{\operatorname{argmin}} \left\{ \sum_{i=1}^{m} z_{i} \left[2(1-b)\delta_{K_{i}}\left(\frac{\lambda_{i}}{z_{i}}\right) + 2\left(\Sigma_{i}^{-1}\frac{\lambda_{i}}{z_{i}}\right)' \left(\Sigma_{i}^{-1}\eta_{i} + \sigma_{i}B_{i}\rho_{i}\right) + \left\|\Sigma_{i}^{-1}\frac{\lambda_{i}}{z_{i}}\right\|^{2} \right] \right\} \\
= \underset{\lambda=(\lambda_{1},\dots,\lambda_{m})'}{\operatorname{argmin}} \left\{ \sum_{i=1}^{m} z_{i} \left[2(1-b)\delta_{K_{i}}\left(\frac{\lambda_{i}}{z_{i}}\right) + \left\|\Sigma_{i}^{-1}\left(\eta_{i} + \frac{\lambda_{i}}{z_{i}}\right) + \sigma_{i}B_{i}\rho_{i} \right\|^{2} \right] \right\} \quad (A.53)$$

Using the change of control $\hat{\lambda}_i = \frac{\lambda_i}{z_i}$, we see that $\hat{\lambda}^*(t, z, B) = \lambda^*(t, z, B)$ from the statement of the Lemma. Letting e_i denote the *i*-th unit vector in \mathbb{R}^m , we can express the market coefficients in Condition (EAS) as

$$\mu^{z}(t,z) = \kappa \odot (\theta - z) = \underbrace{\kappa \odot \theta}_{=:k_{0}(t)} + \underbrace{\begin{pmatrix} -\kappa_{1} & 0 \\ & \ddots \\ 0 & -\kappa_{m} \end{pmatrix}}_{=:k_{1}(t)} z$$

 $A \ Proofs$

$$\begin{split} \Sigma^{z}(t,z)\Sigma^{z}(t,z) &= \begin{pmatrix} \sigma_{1}^{2}z_{1} & 0 \\ 0 & \sigma_{m}^{2}z_{m} \end{pmatrix} = \underbrace{\begin{pmatrix} z'(\sigma_{1}^{2}e_{1}) & 0 \\ 0 & z'(\sigma_{m}^{2}e_{m}) \end{pmatrix}}_{=:h_{1}(t)[z]} \\ \Sigma^{z}(t,z)\rho(t,z) (\Sigma^{z}(t,z)\rho(t,z))' - \Sigma^{z}(t,z)\Sigma^{z}(t,z)' \\ &= \begin{pmatrix} \sigma_{1}\sqrt{z_{1}}\rho_{1}' & 0 \\ 0 & \sigma_{m}\sqrt{z_{m}}\rho_{m} \end{pmatrix} \begin{pmatrix} \sigma_{1}\sqrt{z_{1}}\rho_{1} & 0 \\ 0 & \sigma_{m}\sqrt{z_{m}}\rho_{m} \end{pmatrix} \begin{pmatrix} \sigma_{1}\sqrt{z_{1}}\rho_{1} & 0 \\ 0 & \sigma_{m}\sqrt{z_{m}}\rho_{m} \end{pmatrix} \\ &= \begin{pmatrix} \sigma_{1}^{2}z_{1} (\|\rho_{1}\|^{2}-1) & 0 \\ 0 & \sigma_{m}^{2}z_{m} (\|\rho_{m}\|^{2}-1) \end{pmatrix} \\ &= \begin{pmatrix} z'(\sigma_{1}^{2}(\|\rho_{1}\|^{2}-1)e_{1}) & 0 \\ 0 & \sigma_{m}^{2}z_{m} (\|\rho_{m}\|^{2}-1)e_{m} \end{pmatrix} \\ &= \underbrace{\begin{pmatrix} z'(\sigma_{1}^{2}(\|\rho_{1}\|^{2}-1)e_{1}) & 0 \\ 0 & \sigma_{m}^{2}z_{m} (\|\rho_{m}\|^{2}-1)e_{m} \end{pmatrix} \\ &= \underbrace{\begin{pmatrix} z'(\sigma_{1}^{2}(\|\rho_{1}\|^{2}-1)e_{1}) & 0 \\ 0 & \sigma_{m}^{2}z_{m} (\|\rho_{m}\|^{2}-1)e_{m} \end{pmatrix}}_{=:l_{1}(t)[z]} \\ r(t,z) + \delta_{K}(\hat{\lambda}^{*}(t,z,B)) = \underbrace{\sum_{i=1}^{r} \frac{\delta_{K_{i}}(\lambda^{*}_{i}(B_{i}))}{i=:(\rho_{1}(t,B))_{i}} z_{i} \\ &= \underbrace{\begin{pmatrix} z'(\sigma_{1}^{2}(\|\rho_{1}\|^{2}-1)e_{1}) & 0 \\ 0 & \sigma_{m}^{2}z_{m} \end{pmatrix}}_{=:q_{1}(t,B)} \begin{pmatrix} (\mu_{1},z) + \hat{\lambda}^{*}(t,z,B) - r(t,z)\mathbf{1} \end{pmatrix} \\ = \sum \begin{bmatrix} \sigma_{1}\sqrt{z_{1}} & 0 \\ 0 & \sigma_{m}\sqrt{z_{m}} \end{pmatrix} \begin{pmatrix} \rho_{1}' & 0 \\ 0 & \rho_{m}' \end{pmatrix} \begin{pmatrix} (\Sigma_{1}\sqrt{z_{1}})^{-1} & 0 \\ 0 & (\Sigma_{m}\sqrt{z_{m}})^{-1} \end{pmatrix} \begin{pmatrix} (\eta_{1} + \lambda^{*}_{i}(B_{1}))z_{1} \\ \vdots \\ (\eta_{m} + \lambda^{*}_{m}(B_{m}))z_{m} \end{pmatrix} \\ = \underbrace{\begin{pmatrix} \sigma_{1}\rho_{1}'\Sigma_{1}^{-1}(\eta_{1} + \lambda^{*}_{1}(B_{1})) & 0 \\ 0 & 0 \end{pmatrix}_{=:q_{1}(t,B)} z_{i} \end{pmatrix} \\ z_{i} \\ \\ = \underbrace{\begin{pmatrix} \sigma_{1}\rho_{1}'\Sigma_{1}^{-1}(\eta_{1} + \lambda^{*}_{1}(B_{1})) & 0 \\ 0 & 0 \end{pmatrix}_{=:q_{1}(t,B)} z_{i} \end{pmatrix}}_{=:q_{1}(t,B)} z_{i} \\ \\ \end{bmatrix}$$

By setting the remaining coefficients h_0 , l_0 , q_0 , and g_0 as zero, Condition (EAS) is satisfied. Moreover, the ODEs (4.15) and (4.16) simplify to

$$\begin{split} A_{\tau}(\tau) &= br + (\kappa \odot \theta)' B(\tau) \\ (B_{\tau}(\tau))_{i} &= b \left(p_{1}(T - \tau, B(\tau)) \right)_{i} + \left(k_{1}(T - \tau)B(\tau) \right)_{i} + \frac{1}{2} \left(B(\tau)'h_{1}[\cdot]B(\tau) \right)_{i} \\ &+ \frac{1}{2} \frac{b}{1 - b} \Big[q_{1}(T - \tau, B(\tau)) + 2g_{1}(T - \tau, B(\tau))B(\tau) + B(\tau)' \left(l_{1}[\cdot] + h_{1}[\cdot] \right) B(\tau) \Big]_{i} \\ &= b \delta_{K_{i}} \left(\lambda_{i}^{*}(B_{i}(\tau)) \right) - \kappa_{i}B_{i}(\tau) + \frac{1}{2} \sigma_{i}^{2} \left(B_{i}(\tau) \right)^{2} \\ &+ \frac{1}{2} \frac{b}{1 - b} \left[\left\| \Sigma_{i}^{-1} \left(\eta_{i} + \lambda_{i}^{*}(B_{i}(\tau)) \right) \right\|^{2} + 2\sigma_{i}B_{i}(\tau)\rho_{i}'\Sigma_{i}^{-1} \left(\eta_{i} + \lambda_{i}^{*}(B_{i}(\tau)) \right) + \sigma_{i}^{2} \|\rho_{i}\|^{2}B_{i}(\tau)^{2} \right] \\ &= -\kappa_{i}B_{i}(\tau) + \frac{1}{2} \sigma_{i}^{2} \left(B_{i}(\tau) \right)^{2} \\ &+ \frac{1}{2} \frac{b}{1 - b} \left[2(1 - b)\delta_{K_{i}} \left(\lambda_{i}^{*}(B_{i}(\tau)) \right) + \left\| \Sigma_{i}^{-1} \left(\eta_{i} + \lambda_{i}^{*}(B_{i}(\tau)) \right) + \sigma_{i}B_{i}(\tau)\rho_{i} \right\|^{2} \right] \end{split}$$

$$= -\kappa_{i}B_{i}(\tau) + \frac{1}{2}\sigma_{i}^{2}(B_{i}(\tau))^{2} + \frac{1}{2}\frac{b}{1-b}\inf_{\lambda_{i}\in\mathbb{R}^{d_{i}}}\left\{2(1-b)\delta_{K_{i}}(\lambda_{i}) + \left\|\Sigma_{i}^{-1}(\eta_{i}+\lambda_{i}) + \sigma_{i}B_{i}(\tau)\rho_{i}\right\|^{2}\right\}.$$

Hence, according to Theorem 4.3.5,

$$G(t, v, z) = \frac{1}{b}v^b \exp(A(T-t) + B(T-t)'z)$$

is a solution to the dual HJBI PDE (4.7) and the corresponding candidate optimal portfolio (4.17) is given as in the statement of the lemma. Furthermore,

$$\begin{split} & \left(\Sigma(t,z)\Sigma(t,z)' \right)^{-1} \left(\Sigma^{z}(t,z)\rho(t,z)\Sigma(t,z)' \right)' B(T-t) \\ &= \left(\Sigma(t,z)' \right)^{-1} \rho(t,z)'\Sigma^{z}(t,z)' B(T-t) \\ &= \begin{pmatrix} \left(\Sigma_{1}'\sqrt{z_{1}} \right)^{-1} & 0 \\ & \ddots \\ 0 & \left(\Sigma_{m}'\sqrt{z_{m}} \right)^{-1} \end{pmatrix} \begin{pmatrix} \rho_{1} & 0 \\ & \ddots \\ 0 & \rho_{m} \end{pmatrix} \begin{pmatrix} \sigma_{1}\sqrt{z_{1}}B_{1}(T-t) \\ \vdots \\ \sigma_{m}\sqrt{z_{m}}B_{m}(T-t) \end{pmatrix} \\ &= \begin{pmatrix} \sigma_{1}B_{1}(T-t) \left(\Sigma_{1}^{-1} \right)' \rho_{1} \\ \vdots \\ \sigma_{m}B_{m}(T-t) \left(\Sigma_{m}^{-1} \right)' \rho_{m} \end{pmatrix} \end{split}$$

and therefore

$$\begin{split} \bar{\pi}^*(t,v,z) &= \frac{1}{1-b} \left(\Sigma(t,z)\Sigma(t,z)' \right)^{-1} \left[\mu(t,z) + \underline{\lambda}^*(t,v,z) - r(t,z) \mathbb{1} \right] \\ &+ \frac{1}{1-b} \left(\Sigma(t,z)\Sigma(t,z)' \right)^{-1} \left(\Sigma^z(t,z)\rho(t,z)\Sigma(t,z)' \right)' B(T-t) \\ &= \frac{1}{1-b} \begin{pmatrix} (\Sigma_1 \Sigma_1' z_1)^{-1} & 0 \\ \ddots & \\ 0 & (\Sigma_m \Sigma_m' z_m)^{-1} \end{pmatrix} \begin{pmatrix} (\eta_1 + \lambda_1^*(B_1(T-t))) z_1 \\ \vdots \\ (\eta_m + \lambda_m^*(B_m(T-t))) z_m \end{pmatrix} \\ &+ \frac{1}{1-b} \begin{pmatrix} \sigma_1 B_1(T-t) \left(\Sigma_1^{-1} \right)' \rho_1 \\ \vdots \\ \sigma_m B_m(T-t) \left(\Sigma_m^{-1} \right)' \rho_m \end{pmatrix} \\ &= \frac{1}{1-b} \begin{pmatrix} (\Sigma_1 \Sigma_1')^{-1} \left(\eta_1 + \lambda_1^*(B_1(T-t)) + \sigma_1 B_1(T-t)\Sigma_1 \rho_1 \right) \\ \vdots \\ (\Sigma_m \Sigma_m')^{-1} \left(\eta_m + \lambda_m^*(B_m(T-t)) + \sigma_m B_m(T-t)\Sigma_m \rho_m \right) \end{pmatrix}. \end{split}$$

Proof of Lemma 4.4.8. In \mathcal{M}_{OU} , the minimizer $\hat{\lambda}^*$ from (4.14) is given as

$$\hat{\lambda}^*(t,z,B) = \operatorname*{argmin}_{\lambda \in \mathbb{R}^d} \left\{ 2(1-b)\delta_K(\lambda) + \left\| \underbrace{\Sigma(t,z)^{-1} \big(\mu(t,z) - r(t,z)\mathbb{1} + \lambda \big)}_{\eta + \big(b(t;\hat{T})'\sigma \big)^{-1}\lambda} + \underbrace{(\Sigma^z(t,z)\rho(t,z))'}_{=\sigma'} B \right\|^2 \right\}$$

 $A \ Proofs$

$$= \underset{\lambda \in \mathbb{R}^d}{\operatorname{argmin}} \left\{ 2(1-b)\delta_K(\lambda) + \left\| \eta + \sigma'B + \left(b(t;\hat{T})'\sigma \right)^{-1}\lambda \right\|^2 \right\}$$
$$= \lambda^*(t,B),$$

i.e., $\hat{\lambda}^*(t, v, z) = \lambda^*(t, B)$ is a deterministic function independent of z, but dependent on B. Considering the market coefficients in Condition (EAS), we get

$$\begin{split} \mu^{z}(t,z) &= \kappa \odot \left(\theta - z\right) = \underbrace{\kappa \odot \theta}_{=:k_{0}(t)} + \underbrace{\begin{pmatrix} -\kappa_{1} & 0 \\ \vdots \\ 0 & -\kappa_{m} \end{pmatrix}}_{=:k_{1}(t)} z \\ \Sigma^{z}(t,z)\Sigma^{z}(t,z) &= \underbrace{\sigma\sigma'}_{=:h_{0}(t)} \\ \Sigma^{z}(t,z)\underbrace{\rho(t,z)}_{=I_{m}} \left(\sum^{z}(t,z)\underbrace{\rho(t,z)}_{=I_{m}} \right)' - \Sigma^{z}(t,z)\Sigma^{z}(t,z)' = 0 \\ r(t,z) + \delta_{K}(\hat{\lambda}^{*}(t,z,B)) &= \underbrace{w_{0} + \delta_{K}(\lambda^{*}(t,B))}_{=:p_{0}(t,B)} + \underbrace{w_{1}'}_{=:p_{1}(t,B)'} z \\ \left\| \Sigma^{-1}(t,z) \left(\mu(t,z) + \hat{\lambda}^{*}(t,z,B) - r(t,z)\mathbf{1} \right) \right\|^{2} = \underbrace{\|\eta + \left(b(t;\hat{T})'\sigma\right)^{-1}\lambda^{*}(t,B)\|^{2}}_{=:q_{0}(t,B)} \\ \underbrace{\Sigma^{z}(t,z)}_{=\sigma} \underbrace{\rho(t,z)}_{=I_{m}} \underbrace{\Sigma^{-1}(t,z) \left(\mu(t,z) + \hat{\lambda}^{*}(t,z,B) - r(t,z)\mathbf{1} \right)}_{=\eta + \left(b(t;\hat{T})'\sigma\right)^{-1}\lambda^{*}(t,B)} = \underbrace{\sigma\eta + \left(b(t;\hat{T})'\right)^{-1}}_{=:g_{0}(t,B)} . \end{split}$$

By setting the remaining coefficients h_1 , l_0 , l_1 , q_1 and g_1 as zero, Condition (EAS) is satisfied. Moreover, the ODEs (4.15) and (4.16) simplify to

$$\begin{split} A_{\tau}(\tau) &= b \left(w_{0} + \delta_{K} (\lambda^{*}(T - \tau, B(\tau))) \right) + (\kappa \odot \theta)' B(\tau) + \frac{1}{2} \| \sigma' B(\tau) \|^{2} \\ &+ \frac{1}{2} \frac{b}{1 - b} \left(\left\| \eta + \left(b(T - \tau; \hat{T})' \sigma \right)^{-1} \lambda^{*}(T - \tau, B(\tau)) \right\|^{2} \right) \\ &+ 2 \left(\sigma \eta + \left(b(t; \hat{T})' \right)^{-1} \lambda^{*}(t, B(\tau)) \right)' B(\tau) + \left\| \sigma' B(\tau) \right\|^{2} \right) \\ &= b w_{0} + (\kappa \odot \theta)' B(\tau) + \frac{1}{2} \| \sigma' B(\tau) \|^{2} \\ &+ \frac{1}{2} \frac{b}{1 - b} \left(2(1 - b) \delta_{K} (\lambda^{*}(T - \tau, B(\tau))) \right) \\ &+ \left\| \eta + \sigma' B(\tau) + \left(b(T - \tau; \hat{T})' \right)^{-1} \lambda^{*}(T - \tau, B(\tau)) \right\|^{2} \right) \\ &= b w_{0} + (\kappa \odot \theta)' B(\tau) + \frac{1}{2} \| \sigma' B(\tau) \|^{2} \\ &+ \frac{1}{2} \frac{b}{1 - b} \inf_{\lambda \in \mathbb{R}^{d}} \left(2(1 - b) \delta_{K} (\lambda) + \left\| \eta + \sigma' B(\tau) + \left(b(T - \tau; \hat{T})' \right)^{-1} \lambda \right\|^{2} \right) \end{split}$$

and

$$B_{\tau}(\tau) = bw_1 - \kappa \odot B(\tau).$$

Hence, according to Theorem 4.3.5,

$$G(t, v, z) = \frac{1}{b}v^b \exp(A(T-t) + B(T-t)'z)$$

is a solution to the dual HJBI PDE (4.7). The corresponding candidate optimal portfolio (4.17) is given through

$$\underline{\pi}^{*}(t,v,z) = \frac{1}{1-b} \left(\Sigma(t,z)\Sigma(t,z)' \right)^{-1} \left[\mu(t,z) + \underline{\lambda}^{*}(t,v,z) - r(t,z)\mathbb{1} + \left(\Sigma^{z}(t,z)\rho(t,z)\Sigma(t,z)' \right)' B(T-t) \right] \\ = \frac{1}{1-b} \left(\sigma' b(t;\hat{T}) \right)^{-1} \left(\eta + \left(b(t;\hat{T})'\sigma \right)^{-1} \lambda^{*}(t,B(T-t)) + \sigma' B(T-t) \right).$$

Proof of Corollary 4.4.10. We again verify Condition (UI_{λ}) by showing the L^q boundedness of $G\left(\tau_{n,t}^0, V_0^{v_0,\pi^*}(\tau_{n,t}^0), z(\tau_{n,t}^0)\right)$ in $n \in \mathbb{N}$ for arbitrary q > 1. As per Remark 4.4.9, there exists a closed-form expression for B which is continuously differ-

As per Remark 4.4.9, there exists a closed-form expression for B which is continuously differentiable. Moreover, the matrix $b(t; \hat{T})$ is continuously differentiable in t and non-singular for all $t \in [0, T]$ and therefore uniformly bounded in $t \in [0, T]$. Following the same arguments as in the proof of Corollary 4.4.3, this has the consequence that the minimizer $\lambda^*(t, B(T-t))$, A(T-t)and the candidate optimal portfolio $\pi^*(t) := \underline{\pi}^*(t, v, z)$ are uniformly bounded in $t \in [0, T]$.

For arbitrary q > 1, we can thus find a constant $C_q > 0$ such that for all $t \in [0, T]$

$$\begin{aligned} \left| G(t, V^{v_0, \pi^*}(t), z(t)) \right|^q &= \frac{1}{|b|} \exp\left(bq \int_0^t w_0 + w_1' z(s) + \eta' \sigma' b(s; \hat{T}) \pi^*(s) - \frac{1}{2} \|\sigma' b(s; \hat{T}) \pi^*(s)\|^2 ds \\ &+ bq \int_0^t \pi^*(s)' b(s; \hat{T})' \sigma dW(s) + qA(T-t) + qB(T-t)' z(t) \right) \\ &\leq C_q \exp\left(\underbrace{bq \int_0^t w_1' z(s) ds + bq \int_0^t \pi^*(s)' b(s; \hat{T})' \sigma dW(s) + qB(T-t)' z(t)}_{=:X_t} \right) \\ &= C_q \exp\left(X_t \right). \end{aligned}$$

Since, B is continuously differentiable, we can use Itô's product rule to rewrite

$$B(T-t)'z(t) = B(T)'z_0 + \int_0^t B(T-s)'dz(s) + \int_0^t z(s)'d(B(T-s)) + \underbrace{\langle z, B(T-\cdot) \rangle_t}_{=0}$$

= $B(T)'z_0 + \int_0^t B(T-s)'\kappa \odot [\theta - z(s)] - z(s)'B_\tau(T-s)ds + \int_0^t B(T-s)'\sigma dW^z(s)$
(A.55)

Due to $\rho(t, z) = I_m$, we know that $W^z(t) = W(t)$ holds $\mathcal{L}[0, T] \otimes Q$ -a.e.. Hence, using (A.55) and disregarding terms of finite variation, the quadratic variation of X can be computed as

$$\begin{split} \langle X \rangle_t &= \left\langle bq \int_0^t w_1' z(s) ds + bq \int_0^\cdot \pi^*(s)' b(s; \hat{T})' \sigma \underbrace{dW(s)}_{=dW^z(s)} + qB(T-\cdot)' z(\cdot) \right\rangle_t \\ &= \left\langle bq \int_0^\cdot \pi^*(s)' b(s; \hat{T})' \sigma + \frac{1}{b} B(T-s)' \sigma dW^z(s) \right\rangle_t \\ &= b^2 q^2 \int_0^t \left\| \sigma' \left(b(s; \hat{T}) \pi^*(s) + \frac{1}{b} B(T-s) \right) \right\|^2 ds. \end{split}$$

Since all involved functions are bounded and deterministic, $\langle X \rangle_t \leq \langle X \rangle_T < \infty$ yields a deterministic upper bound on $\langle X \rangle_t$ for all $t \in [0, T]$. Therefore, we can continue equation (A.54) to obtain for all $t \in [0, T]$

$$\left| G(t, V^{v_0, \underline{\pi}^*}(t), z(t)) \right|^q \stackrel{(A.54)}{\leq} C_q \exp(X_t) \leq \underbrace{C_q \exp\left(\frac{1}{2}\langle X \rangle_T\right)}_{=:\tilde{C}_q} \underbrace{\exp\left(X_t - \frac{1}{2}\langle X \rangle_t\right)}_{=:M_t} = \tilde{C}_q M_t$$
(A.56)

The process $M = (M_t)_{t \in [0,T]}$ is a non-negative local martingale and thus a supermartingale. Doob's optional sampling theorem ('O.S.') implies

$$\sup_{n\in\mathbb{N}}\mathbb{E}\left[\left|G\left(\tau_{n,t}^{0}, V_{0}^{v_{0},\underline{\pi}^{*}}(\tau_{n,t}^{0}), z(\tau_{n,t}^{0})\right)\right|^{q}\right] \stackrel{(A.56)}{\leq} \tilde{C}_{q} \sup_{n\in\mathbb{N}}\mathbb{E}\left[M_{\tau_{n,t}^{0}}\right] \stackrel{O.S.}{\leq} \tilde{C}_{q}M_{0} = \tilde{C}_{q} < \infty.$$

Hence,

$$\left(G\left(\tau_{n,t}^{0}, V_{0}^{v_{0}, \underline{\pi}^{*}}(\tau_{n,t}^{0}), z(\tau_{n,t}^{0})\right)\right)_{n \in \mathbb{N}}$$

is bounded in L^q for any q > 1 and $t \in [0, T]$ and is thus uniformly integrable for any $t \in [0, T]$ (see Theorem 4.5.9 in [8] with $G(t) = t^q$). Hence, Condition (UI_{λ}) is satisfied and π^* is optimal for (**P**) by virtue of Theorem 4.3.7.

A.4 Proofs Chapter 5

Proof of Lemma 5.2.2. Follows immediately from Lemma 4.4.6 with d = m = 1 and $\Sigma_1 = 1$. \Box

Proof of Lemma 5.2.3. Noting that

$$G_v(t,v,z) = rac{b}{v}G(t,v,z)$$
 and $G_z(t,v,z) = B(T-t)G(t,v,z),$

this result follows immediately from Theorem 4.3.7 and Lemma 4.4.6 with d = m = 1 and $\Sigma_1 = 1$.

Proof of Lemma 5.2.4. The main objective of this proof is to determine the minimizing argument $\lambda^*(B)$ for the minimization problem

$$\inf_{\lambda \in \mathbb{R}} \underbrace{\left(2(1-b)\delta_K(\lambda) + (\eta + \lambda + \sigma\rho B)^2 \right)}_{=:D(\lambda,B)} = \inf_{\lambda \in \mathbb{R}} D(\lambda,B)$$

for any fixed value $B \in \mathbb{R}$. The remaining statement follows immediately after substituting the obtained minimizer $\lambda^*(B) = \lambda^*(B(\tau))$ in (5.5).

Given any constraints $K = [\alpha, \beta]$ with $-\infty \le \alpha \le \beta \le \infty$ (see (5.2)), we have for any $\lambda \in \mathbb{R}$

$$\delta_K(\lambda) = -\inf_{\alpha \le x \le \beta} (x\lambda) = -\alpha \lambda \mathbb{1}_{\{\lambda > 0\}} - \beta \lambda \mathbb{1}_{\{\lambda < 0\}}$$

Thus, taking the derivative of $D(\cdot, B)$ on $(-\infty, 0)$ and $(0, \infty)$ yields

$$\frac{\partial}{\partial\lambda}D(\lambda,B) = \begin{cases} -2(1-b)\alpha + 2(\eta+\lambda+\sigma\rho B), & \lambda > 0\\ -2(1-b)\beta + 2(\eta+\lambda+\sigma\rho B), & \lambda < 0. \end{cases}$$

Then, by the first-order optimality condition and the fact that $D(\cdot, B)$ is quadratic on $[0, \infty)$,

$$\lambda^{-}(B) = ((1-b)\alpha - (\eta + \sigma\rho B))\mathbb{1}_{\{(1-b)\alpha - (\eta + \sigma\rho B) > 0\}} = ((1-b)\alpha - (\eta + \sigma\rho B))\mathbb{1}_{\{\rho B < B_{-}\}}$$

minimises $D(\cdot, B)$ on $[0, \infty)$, and by the same argument

$$\lambda^{+}(B) = \left((1-b)\beta - (\eta + \sigma\rho B) \right) \mathbb{1}_{\{(1-b)\beta - (\eta + \sigma\rho B) < 0\}} = \left((1-b)\beta - (\eta + \sigma\rho B) \right) \mathbb{1}_{\{\rho B > B_+\}}$$

minimises $D(\cdot, B)$ on $(-\infty, 0]$. Therefore, noting that by construction

$$\max\left(D(\lambda^{-}(B), B), \ D(\lambda^{+}(B), B)\right) = \max\left(\underbrace{\min_{\lambda \in (-\infty, 0]} D(\lambda, B)}_{\leq D(0, B)}, \underbrace{\min_{\lambda \in [0, \infty)} D(\lambda, B)}_{\leq D(0, B)}\right)$$
$$\leq D(0, B) = (\eta + \sigma\rho B)^{2}$$
(A.57)

and $B_{-} \leq B_{+}$, we finally derive

$$\begin{split} &\inf_{\lambda \in \mathbb{R}} D(\lambda, B) \\ &= \min \left(D(\lambda^{-}(B), B), \ D(\lambda^{+}(B), B) \right) \\ &= \min \left(\left[2(1-b)(-\alpha)((1-b)\alpha - \eta - \sigma\rho B) + ((1-b)\alpha)^{2} \right] \mathbb{1}_{\{\rho B < B_{-}\}} + \left[\eta + \sigma\rho B \right]^{2} \mathbb{1}_{\{B_{-} \le \rho B\}}, \\ & \left[2(1-b)(-\beta)((1-b)\beta - \eta - \sigma\rho B) + ((1-b)\beta)^{2} \right] \mathbb{1}_{\{B_{+} < \rho B\}} + \left[\eta + \sigma\rho B \right]^{2} \mathbb{1}_{\{\rho B \le B_{+}\}} \right) \\ &\stackrel{(A.57)}{=} \left[2(1-b)(-\alpha)((1-b)\alpha - \eta - \sigma\rho B) + ((1-b)\alpha)^{2} \right] \mathbb{1}_{\{B_{-} \le \rho B\}} + \left[\eta + \sigma\rho B \right]^{2} \mathbb{1}_{\{B_{-} \le \rho B \le B_{+}\}} \\ &+ \left[2(1-b)(-\beta)((1-b)\beta - \eta - \sigma\rho B) + ((1-b)\beta)^{2} \right] \mathbb{1}_{\{B_{+} < \rho B\}} \\ &= D(\lambda^{-}(B), B) \mathbb{1}_{\{B_{-} \le \rho B\}} + D(0, B) \mathbb{1}_{\{B_{-} \le \rho B \le B_{+}\}} + D(\lambda^{+}(B), B) \mathbb{1}_{\{B_{+} < \rho B\}}. \end{split}$$

Thus, the minimizer $\lambda^*(B)$ is given as

$$\lambda^{*}(B) = \lambda^{-}(B)\mathbb{1}_{\{\rho B < B_{-}\}} + \lambda^{+}(B)\mathbb{1}_{\{B_{+} < \rho B\}}$$

= $((1-b)\alpha - (\eta + \sigma\rho B))\mathbb{1}_{\{\rho B < B_{-}\}} + ((1-b)\beta - (\eta + \sigma\rho B))\lambda^{+}(B)\mathbb{1}_{\{B_{+} < \rho B\}}.$

Substituting $\lambda^*(B(\tau))$ in (5.5) and factoring $B(\tau)$ and $(B(\tau))^2$ concludes the proof.

Proof of Theorem 5.2.8. B is specifically constructed in such a way that it is continuous and satisfies the corresponding Riccati ODE of (5.9), whenever $\rho B(\tau)$ is within each of the zones Z_{-} , Z_{0} , or Z_{+} . Moreover, as the right hand side of the ODE (5.5) is continuous, so are the derivatives of the constructed B. Thus, B is a solution to (5.9) and thereby (5.5).

Proof of Lemma 5.2.10. By rewriting π^* in terms of λ^* and B, we see that

$$\frac{b\rho}{\sigma}\pi^*(t) + B(T-t) = \begin{cases} \frac{b\rho}{\sigma}\alpha + B(T-t), & \text{if } \rho B(T-t) < B_-\\ \frac{b}{1-b}\frac{\rho}{\sigma}\eta + \left(1 + \frac{b}{1-b}\rho^2\right)B(T-t), & \text{if } B_- \le \rho B(T-t) \le B_+\\ \frac{b\rho}{\sigma}\beta + B(T-t), & \text{if } B_+ < \rho B(T-t) \end{cases}$$

is non-decreasing in B(T-t). Hence, as B is itself monotonous (see Remark 5.2.5), we have

$$\sup_{t\in[0,T]} \left(\frac{b\rho}{\sigma}\pi^*(t) + B(T-t)\right) = \max\left(\frac{b\rho}{\sigma}\pi^*(T) + B(0), \frac{b\rho}{\sigma}\pi^*(0) + B(T)\right).$$
(A.58)

Hence, when first assuming that the maximum in (A.58) is attained at t = T, we obtain

$$\frac{b\rho}{\sigma} \underbrace{\pi^*(T)}_{\in [\alpha,\beta]} + \underbrace{B(0)}_{=0} \le \max\left(\frac{b\rho}{\sigma}\alpha, \frac{b\rho}{\sigma}\beta\right) \stackrel{(5.11)}{\le} \frac{\kappa}{\sigma^2}.$$

On the other hand, if we assume that the maximum in (A.58) is attained at t = 0, we need to compare the different terminal values B(T) based on the coefficients of the corresponding Riccati ODE at T. Due to the monotonicity of B and the assumption that the maximum in (A.58) is attained at t = 0, it suffices if we can find a bound for $T \to \infty$.

• Case $\rho B(T) \in \mathbb{Z}_+$:

Then, $\pi^*(0) = \beta$ and B(T) is given as the solution (2.23) to the Riccati ODE (2.22) with coefficients r_0^+ , r_1^+ , and r_2^+ . In particular, we have

$$\begin{split} \frac{b\rho}{\sigma} \underbrace{\frac{\pi^*(0)}{\sigma}}_{=\beta} + B(T) &\leq \frac{b\rho}{\sigma}\beta + \lim_{T \to \infty} B(T) \\ &= \frac{b\rho}{\sigma}\beta + \lim_{T \to \infty} \left(\frac{2r_2^+ r_3^+ B_0 + \left(e^{r_3^+ T} - 1\right)\left(r_1^+ + r_3^+\right)\left(r_1^+ + r_2^+ B_0 - r_3^+\right)}{2r_2^+ r_3^+ - r_2^+ \left(e^{r_3^+ T} - 1\right)\left(r_1^+ + r_2^+ B_0 - r_3^+\right)} \right) \\ &= \frac{b\rho}{\sigma}\beta - \frac{r_1^+ + r_3^+}{r_2^+} = \frac{b\rho}{\sigma}\beta - \frac{b\sigma\rho\beta - \kappa}{\sigma^2} - \frac{r_3^+}{\sigma^2} = \frac{\kappa}{\sigma^2} - \underbrace{\frac{r_3^+}{\sigma^2}}_{>0} \leq \frac{\kappa}{\sigma^2}. \end{split}$$

• Case $\rho B(T) \in \mathbb{Z}_-$:

Analogous arguments as in the case $\rho B(T) \in \mathbb{Z}_+$ yield that

$$\frac{b\rho}{\sigma}\pi^*(0) + B(T) \le \frac{\kappa}{\sigma^2}$$

• Case $\rho B(T) \in Z_0$: Then, $\pi^*(0) = \frac{1}{1-b} (\eta + \sigma \rho B(T))$ and B(T) is given as the solution (2.23) to Riccati ODE (2.22) with coefficients r_0 , r_1 , and r_2 . In particular, we have

$$\frac{b\rho}{\sigma}\pi^{*}(0) + B(T) = \frac{\rho}{\sigma}\frac{b}{1-b}\left(\eta + \sigma\rho B(T)\right) + B(T) = \frac{\rho}{\sigma}\frac{b}{1-b}\eta + \left(1 + \frac{b}{1-b}\rho^{2}\right)B(T)$$

 $A \ Proofs$

$$\leq \frac{\rho}{\sigma} \frac{b}{1-b} \eta + \left(1 + \frac{b}{1-b} \rho^2\right) \underbrace{\lim_{T \to \infty} B(T)}_{= -\frac{r_1 + r_3}{r_2}} = \frac{\rho}{\sigma} \frac{b}{1-b} \eta - \left(1 + \frac{b}{1-b} \rho^2\right) \frac{r_1 + r_3}{r_2}$$

$$= \frac{\rho}{\sigma} \frac{b}{1-b} \eta - \left(1 + \frac{b}{1-b} \rho^2\right) \frac{r_1 + r_3}{\sigma^2 \left(1 + \frac{b}{1-b} \rho^2\right)} = \frac{\rho}{\sigma} \frac{b}{1-b} \eta - \frac{r_1 + r_3}{\sigma^2}$$

$$= \frac{\rho}{\sigma} \frac{b}{1-b} \eta - \frac{1}{\sigma^2} \left(\frac{b}{1-b} \eta \sigma \rho - \kappa\right) - \frac{r_3}{\sigma^2} = \frac{\kappa}{\sigma^2} - \underbrace{\frac{r_3}{\sigma^2}}_{\geq 0} \leq \frac{\kappa}{\sigma^2}.$$

Proof of Lemma 5.2.11. Note that we may express ODE (5.5) in terms of λ^* and π^* as

$$B'(\tau) = -\kappa B(\tau) + \frac{1}{2}\sigma^2 B(\tau)^2 + \frac{1}{2}\frac{b}{1-b}\left(2(1-b)\delta_K\left(\lambda^*\left(B(\tau)\right)\right) + (1-b)^2\left(\pi^*(T-\tau)\right)^2\right).$$

We distinguish between three cases, depending on whether the allocation constraint is active or not.

• Case
$$\pi^*(t) \in (\alpha, \beta)$$
:
Then, $\lambda^* (B(T-t)) = (\lambda^*)' (B(T-t)) = \delta_K (\lambda^* (B(T-t))) = 0$. Hence,

$$\frac{1}{2} \frac{b}{1-b} \eta^2 - \frac{1}{2} \frac{b}{1-b} (\lambda^* (B(T-t)) + \sigma \rho B(T-t))^2 - \frac{1}{2} b^2 \rho^2 (\pi^*(t))^2 + b \frac{\rho \kappa}{\sigma} \pi^*(t) + \frac{b}{1-b} \frac{\rho}{\sigma} [(\lambda^*)' (B(T-t)) + \sigma \rho] B'(T-t)$$

$$= \frac{1}{2} \frac{b}{1-b} \eta^2 - \frac{1}{2} \frac{b}{1-b} (\sigma \rho B(T-t))^2 - \frac{1}{2} b^2 \rho^2 (\pi^*(t))^2 + b \frac{\rho \kappa}{\sigma} \pi^*(t) + \frac{b}{1-b} \rho^2 \left[-\kappa B(T-t) + \frac{1}{2} \sigma^2 (B(T-t))^2 + \frac{1}{2} b(1-b) (\pi^*(t))^2 \right]$$

$$= \frac{1}{2} \frac{b}{1-b} \eta^2 - \frac{1}{2} \frac{b}{1-b} (\sigma \rho B(T-t))^2 + \frac{b}{1-b} \frac{\rho \kappa}{\sigma} (\eta + \sigma \rho B(T-t)) + \rho^2 \frac{b}{1-b} \left[-\kappa B(T-t) + \frac{1}{2} \sigma^2 (B(T-t))^2 \right]$$

$$= \frac{b}{1-b} \eta \left(\frac{\eta}{2} + \frac{\rho \kappa}{\sigma} \right)$$

• Case
$$\pi^*(t) = \alpha$$
:
Then, $\lambda^*(B(T-t)) = (1-b)\alpha - \eta - \sigma\rho B(T-t)$ and

$$\underbrace{\left[\left(\lambda^*\right)'\left(B(T-t)\right)+\sigma\rho\right]}_{=0}B'(T-t)=0.$$

Hence,

$$\frac{1}{2}\frac{b}{1-b}\eta^2 - \frac{1}{2}\frac{b}{1-b}\overbrace{(\lambda^*(B(T-t)) + \sigma\rho B(T-t))^2}^{=((1-b)\alpha - \eta)^2} - \frac{1}{2}b^2\rho^2\underbrace{(\pi^*(t))^2}_{=\alpha^2}$$

$$\begin{split} &+ b\frac{\rho\kappa}{\sigma}\underbrace{\pi^{*}(t)}_{=\alpha} + \frac{b}{1-b}\frac{\rho}{\sigma}\underbrace{\left[(\lambda^{*})'\left(B(T-t)\right) + \sigma\rho\right]B'(T-t)}_{=0} \\ &= \frac{1}{2}\underbrace{\frac{b}{1-b}}{\eta^{2}} - \frac{1}{2}\frac{b}{1-b}\left((1-b)\alpha - \eta\right)^{2} - \frac{1}{2}b^{2}\rho^{2}\alpha^{2} + b\frac{\rho\kappa}{\sigma}\alpha \\ &= \frac{1}{2}\frac{b}{1-b}\eta^{2} - \frac{1}{2}\frac{b}{1-b}\left((1-b)^{2}\alpha^{2} - 2(1-b)\alpha\eta + \eta^{2}\right) - \frac{1}{2}b^{2}\rho^{2}\alpha^{2} + b\frac{\rho\kappa}{\sigma}\alpha \\ &= -\frac{1}{2}\frac{b}{1-b}\left((1-b)^{2}\alpha^{2} - 2(1-b)\alpha\eta\right) - \frac{1}{2}b^{2}\rho^{2}\alpha^{2} + b\frac{\rho\kappa}{\sigma}\alpha \\ &= -\frac{1}{2}b(1-b)\alpha^{2} + b\alpha\eta - \frac{1}{2}b^{2}\rho^{2}\alpha^{2} + b\frac{\rho\kappa}{\sigma}\alpha \\ &= b\alpha\left[\eta - \frac{1}{2}\alpha + \frac{\rho\kappa}{\sigma} + \frac{1}{2}\alpha b(1-\rho^{2})\right]. \end{split}$$

• Case $\pi^*(t) = \beta$:

Following the same arguments as in the case $\pi^*(t) = \alpha$ yields

$$\begin{split} &\frac{1}{2} \frac{b}{1-b} \eta^2 - \frac{1}{2} \frac{b}{1-b} \left(\lambda^* \left(B(T-t) \right) + \sigma \rho B(T-t) \right)^2 - \frac{1}{2} b^2 \rho^2 \left(\pi^*(t) \right)^2 \\ &+ b \frac{\rho \kappa}{\sigma} \pi^*(t) + \frac{b}{1-b} \frac{\rho}{\sigma} \left[\left(\lambda^* \right)' \left(B(T-t) \right) + \sigma \rho \right] B'(T-t) \\ &= b \beta \left[\eta - \frac{1}{2} \beta + \frac{\rho \kappa}{\sigma} + \frac{1}{2} \beta b(1-\rho^2) \right]. \end{split}$$

Combining all three cases with Assumption 5.2.7 finally yields the claim.

Proof of Theorem 5.2.12. Let B be as in Theorem 5.2.8,

$$A(\tau) = rb\tau + \kappa\theta \int_0^\tau B(s)ds.$$

and define $G : [0,T] \times (0,\infty)^2 \to \mathbb{R}$ as $G(t,v,z) = \frac{1}{b}v^b \exp(A(T-t) + B(T-t)z)$. Then, A and B are solutions to the ODEs (5.4) and (5.5) and thus continuous and bounded on [0,T]. Moreover, according to Lemma 5.2.2, G is a solution to the HJB PDE (5.3). Hence, as per Lemma 5.2.3, it only remains to verify that the sequence

$$\left(G(\tau_{n,t}^{0}, V^{v_{0}, \pi^{*}}(\tau_{n,t}^{0}), z(\tau_{n,t}^{0}))\right)_{n \in \mathbb{N}}$$

is uniformly integrable for all $t \in [0, T]$. Let \mathcal{T} denote the set of all \mathbb{F} stopping times taking values in [0, T]. We verify the stronger statement that

$$\left(G(\tau, V^{v_0, \pi^*}(\tau), z(\tau))\right)_{\tau \in \mathcal{T}}$$

is bounded in L^q for some q > 1 (see e.g. Theorem 4.6.2 in [27]). The argument is an adaptation of the proof of Theorem 5.3 in [56]. Consider arbitrary $t \in [0, T]$ and $q = 1 + \epsilon$ with $\epsilon > 0$ and define

$$D(t) = bq\sqrt{1-\rho^2} \int_0^t \pi^*(s)\sqrt{z(s)}d\hat{W}(s).$$

Then, by defining the deterministic, continuous function

$$det_1(t) = q \left(-\ln(|b|) + b \ln(v_0) + brt + A(T-t) \right),$$

we obtain

$$\begin{split} q \ln \left(\left| G\left(t, V^{v_0, \pi^*}(t), z(t) \right) \right| \right) \\ &= q \left(-\ln(|b|) + b \ln \left(V^{v_0, \pi^*}(t) \right) + A(T-t) + B(T-t)z(t) \right) \\ &= q \left(-\ln(|b|) + b \ln(v_0) + b \int_0^t \left(r + \eta \pi^*(s)z(s) - \frac{1}{2} \left(\pi^*(s) \right)^2 z(s) \right) ds \\ &+ b \int_0^t \pi^*(s) \sqrt{z(s)} \frac{dW(s)}{dW(s)} + A(T-t) + B(T-t)z(t) \right) \\ &= \rho dW^z(s) + \sqrt{1-\rho^2} d\hat{W}(s) \\ &= det_1(t) + D(t) - \frac{1}{2} \langle D \rangle_t + \frac{1}{2} q^2 b^2 (1-\rho^2) \int_0^t \left(\pi^*(s) \right)^2 z(s) ds + q B(T-t)z(t) \\ &+ q b \rho \int_0^t \pi^*(s) \sqrt{z(s)} dW^z(s) + q b \int_0^t \left(\pi^*(s) \eta - \frac{1}{2} \left(\pi^*(s) \right)^2 \right) z(s) ds \\ &= det_1(t) + D(t) - \frac{1}{2} \langle D \rangle_t + q B(T-t)z(t) + q b \rho \int_0^t \pi^*(s) \sqrt{z(s)} dW^z(s) \\ &+ q \int_0^t z(s) \left\{ b \left(\pi^*(s) \eta - \frac{1}{2} (\pi^*(s))^2 \right) + \frac{1}{2} q b^2 (1-\rho^2) \left(\pi^*(s) \right)^2 \right\} ds \\ q^{q=1+\epsilon} det_1(t) + D(t) - \frac{1}{2} \langle D \rangle_t + q B(T-t)z(t) + q b \rho \int_0^t \pi^*(s) \sqrt{z(s)} dW^z(s) \\ &+ q \int_0^t z(s) \left\{ \frac{1}{2} \frac{b}{1-b} \eta^2 - \frac{1}{2} b(1-b) \left(\pi^*(s) - \frac{\eta}{1-b} \right)^2 \\ &- \frac{1}{2} b^2 \rho^2 (\pi^*(s))^2 + \frac{1}{2} \epsilon b^2 (1-\rho^2) (\pi^*(s))^2 \right\} ds \\ \binom{(5.7)}{=} det_1(t) + D(t) - \frac{1}{2} \langle D \rangle_t + q B(T-t)z(t) + q b \rho \int_0^t \pi^*(s) \sqrt{z(s)} dW^z(s) \\ &+ q \int_0^t z(s) \left\{ \frac{1}{2} \frac{b}{1-b} \eta^2 - \frac{1}{2} b(1-b) \left(\pi^*(B(T-s)) + \sigma \rho B(T-s) \right)^2 \\ &- \frac{1}{2} b^2 \rho^2 (\pi^*(s))^2 + \frac{1}{2} \epsilon b^2 (1-\rho^2) (\pi^*(s))^2 \right\} ds \end{aligned}$$
(A.59)

B is continuously differentiable and monotonous, since it is the solution to an autonomous ODE (see Remark 5.2.5). Hence, π^* is monotonous and differentiable and has finite variation. Therefore, Itô's product rule yields

$$d(\pi^{*}(t)z(t)) = \pi^{*}(t)dz(t) + z(t)d\pi^{*}(t)$$

= $\pi^{*}(t)dz(t) - \frac{z(t)}{1-b} \left[(\lambda^{*})' (B(T-t)) + \sigma \rho \right] B'(T-t)dt$
= $\pi^{*}(t)\kappa (\theta - z(t)) dt + \pi^{*}(t)\sigma \sqrt{z(t)} dW^{z}(t)$
 $- \frac{z(t)}{1-b} \left[(\lambda^{*})' (B(T-t)) + \sigma \rho \right] B'(T-t)dt.$

Hence,

$$\int_{0}^{t} \pi^{*}(s)\sqrt{z(s)}dW^{z}(s) = \frac{1}{\sigma} \left(\pi^{*}(t)z(t) - \pi^{*}(0)z(0)\right) - \frac{\kappa}{\sigma} \int_{0}^{t} \pi^{*}(s) \left(\theta - z(s)\right) ds + \frac{1}{\sigma} \frac{1}{1-b} \int_{0}^{t} \left[\left(\lambda^{*}\right)' \left(B(T-s)\right) + \sigma\rho\right] B'(T-s)z(s) ds$$
(A.61)

Substituting (A.61) in (A.60), while defining the deterministic and continuous function

$$det_2(t) = det_1(t) - \frac{qb\rho}{\sigma} \left[\pi^*(0)z(0) + \kappa \int_0^t \pi^*(s)\theta ds \right],$$

yields

$$q \ln \left(\left| G\left(t, V^{v_0, \pi^*}(t), z(t)\right) \right| \right)$$

= $det_2(t) + D(t) - \frac{1}{2} \langle D \rangle_t + qz(t) \left[\frac{b\rho}{\sigma} \pi^*(t) + B(T-t) \right]$
+ $q \int_0^t z(s) \left\{ \frac{1}{2} \frac{b}{1-b} \eta^2 - \frac{1}{2} \frac{b}{1-b} \left(\lambda^* \left(B(T-s) \right) + \sigma \rho B(T-s) \right)^2$
 $- \frac{1}{2} b^2 \rho^2 \left(\pi^*(s) \right)^2 + \frac{1}{2} \epsilon b^2 (1-\rho^2) \left(\pi^*(s) \right)^2$
 $+ b \frac{\rho \kappa}{\sigma} \pi^*(s) + \frac{b}{1-b} \frac{\rho}{\sigma} \left[(\lambda^*)' \left(B(T-s) \right) + \sigma \rho \right] B'(T-s) \right\} ds$ (A.62)

Combining Lemma 5.2.10 with (A.62) yields

$$\begin{split} q \ln \left(\left| G\left(t, V^{v_0, \pi^*}(t), z(t)\right) \right| \right) \\ &\leq det_2(t) + D(t) - \frac{1}{2} \langle D \rangle_t + q \frac{\kappa}{\sigma^2} z(t) \\ &+ q \int_0^t z(s) \left\{ \frac{1}{2} \frac{b}{1-b} \eta^2 - \frac{1}{2} \frac{b}{1-b} \left(\lambda^* \left(B(T-s) \right) + \sigma \rho B(T-s) \right)^2 \\ &- \frac{1}{2} b^2 \rho^2 \left(\pi^*(s) \right)^2 + \frac{1}{2} c b^2 (1-\rho^2) \left(\pi^*(s) \right)^2 \\ &+ b \frac{\rho \kappa}{\sigma} \pi^*(s) + \frac{b}{1-b} \frac{\rho}{\sigma} \left[\left(\lambda^* \right)' \left(B(T-s) \right) + \sigma \rho \right] B'(T-s) \right\} ds \\ &= det_2(t) + D(t) - \frac{1}{2} \langle D \rangle_t + q \frac{\kappa}{\sigma^2} \left(z(0) + \int_0^t \kappa(\theta - z(s)) ds + \sigma \int_0^t \sqrt{z}(s) dW^z(s) \right) \\ &+ q \int_0^t z(s) \left\{ \frac{1}{2} \frac{b}{1-b} \eta^2 - \frac{1}{2} \frac{b}{1-b} \left(\lambda^* \left(B(T-s) \right) + \sigma \rho B(T-s) \right)^2 \\ &- \frac{1}{2} b^2 \rho^2 \left(\pi^*(s) \right)^2 + \frac{1}{2} c b^2 (1-\rho^2) \left(\pi^*(s) \right)^2 \\ &+ b \frac{\rho \kappa}{\sigma} \pi^*(s) + \frac{b}{1-b\sigma} \frac{\rho}{\sigma} \left[\left(\lambda^* \right)' \left(B(T-s) \right) + \sigma \rho \right] B'(T-s) \right\} ds \\ &= \underbrace{det_2(t) + q \frac{\kappa}{\sigma^2} \left(z(0) + \int_0^t \kappa \theta ds \right) \\ &+ q \int_0^t z(s) \left\{ - \frac{\kappa^2}{\sigma^2} + \frac{1}{2} \frac{b}{1-b} \eta^2 - \frac{1}{2} \frac{b}{1-b} \left(\lambda^* \left(B(T-s) \right) + \sigma \rho B(T-s) \right)^2 \\ &- \frac{1}{2} b^2 \rho^2 \left(\pi^*(s) \right)^2 + \frac{1}{2} c b^2 (1-\rho^2) \left(\pi^*(s) \right)^2 \\ &+ b \frac{\rho \kappa}{\sigma} \pi^*(s) + \frac{b}{1-b\sigma} \frac{\rho}{\sigma} \left[\left(\lambda^* \right)' \left(B(T-s) \right) + \sigma \rho B(T-s) \right)^2 \\ &- \frac{1}{2} b^2 \rho^2 \left(\pi^*(s) \right)^2 + \frac{1}{2} c b^2 (1-\rho^2) \left(\pi^*(s) \right)^2 \\ &+ b \frac{\rho \kappa}{\sigma} \pi^*(s) + \frac{b}{1-b\sigma} \frac{\rho}{\sigma} \left[\left(\lambda^* \right)' \left(B(T-s) \right) + \sigma \rho B(T-s) \right)^2 \right] ds \\ &= det_3(t) + D(t) - \frac{1}{2} \langle D \rangle_t + M(t) - \frac{1}{2} \langle M \rangle_t \end{aligned}$$

$$+ q \int_{0}^{t} z(s) \Biggl\{ -\frac{\kappa^{2}}{\sigma^{2}} + \frac{1}{2} \underbrace{q}_{=1+\epsilon} \frac{\kappa^{2}}{\sigma^{2}} + \frac{1}{2} \frac{b}{1-b} \eta^{2} - \frac{1}{2} \frac{b}{1-b} \left(\lambda^{*} \left(B(T-s)\right) + \sigma \rho B(T-s)\right)^{2} \\ - \frac{1}{2} b^{2} \rho^{2} \left(\pi^{*}(s)\right)^{2} + \frac{1}{2} \epsilon b^{2} (1-\rho^{2}) \left(\pi^{*}(s)\right)^{2} \\ + b \frac{\rho \kappa}{\sigma} \pi^{*}(s) + \frac{b}{1-b} \frac{\rho}{\sigma} \left[(\lambda^{*})' \left(B(T-s)\right) + \sigma \rho \right] B'(T-s) \Biggr\} ds$$

$$= det_{3}(t) + D(t) - \frac{1}{2} \langle D \rangle_{t} + M(t) - \frac{1}{2} \langle M \rangle_{t}$$

$$+ q \int_{0}^{t} z(s) \Biggl\{ -\frac{\kappa^{2}}{2\sigma^{2}} + \frac{1}{2} \frac{b}{1-b} \eta^{2} - \frac{1}{2} \frac{b}{1-b} \left(\lambda^{*} \left(B(T-s)\right) + \sigma \rho B(T-s)\right)^{2} \\ - \frac{1}{2} b^{2} \rho^{2} \left(\pi^{*}(s)\right)^{2} + \frac{1}{2} \epsilon b^{2} (1-\rho^{2}) \left(\pi^{*}(s)\right)^{2} + \frac{1}{2} \epsilon \frac{\kappa^{2}}{\sigma^{2}} \\ + b \frac{\rho \kappa}{\sigma} \pi^{*}(s) + \frac{b}{1-b} \frac{\rho}{\sigma} \left[(\lambda^{*})' \left(B(T-s)\right) + \sigma \rho \right] B'(T-s) \Biggr\} ds$$

$$= det_{3}(t) + D(t) - \frac{1}{2} \langle D \rangle_{t} + M(t) - \frac{1}{2} \langle M \rangle_{t} + q \int_{0}^{t} z(s) \Biggl\{ (*) \Biggr\} ds$$

$$(A.64)$$

According to Lemma 5.2.11, the expression (*) in (A.64) is negative for all $s \in [0, T]$ if $\epsilon = 0$. Hence, due to the continuity of (*) in ϵ and the boundedness of all deterministic functions in (*) w.r.t. t, there exists $\epsilon > 0$ such that (*) < 0 for all $s \in [0, T]$. For such a choice of ϵ we obtain

$$\begin{split} q\ln\left(\left|G\left(t,V^{v_0,\pi^*}(t),z(t)\right)\right|\right) &\leq det_3(t) + D(t) - \frac{1}{2}\langle D\rangle_t + M(t) - \frac{1}{2}\langle M\rangle_t \\ &\leq \sup_{s\in[0,T]} \left(det_3(s)\right) + D(t) - \frac{1}{2}\langle D\rangle_t + M(t) - \frac{1}{2}\langle M\rangle_t \end{split}$$

$$\Leftrightarrow \left| G\left(t, V^{v_0, \pi^*}(t), z(t)\right) \right|^q \le \exp\left(\sup_{s \in [0, T]} \left(det_3(s)\right)\right) \cdot \underbrace{\exp\left(D(t) - \frac{1}{2} \langle D \rangle_t\right) \cdot \exp\left(M(t) - \frac{1}{2} \langle M \rangle_t\right)}_{=:(**)}.$$

Since D and M are local martingales with independent diffusions, (**) is a supermartingale. Hence, Doob's optional sampling theorem finally yields

$$\begin{split} \sup_{\tau \in \mathcal{T}} \mathbb{E} \left[\left| G(\tau, V^{v_0, \pi^*}(\tau), z(\tau)) \right|^q \right] \\ &\leq \exp \left(\sup_{s \in [0,T]} \left(\det_3(s) \right) \right) \sup_{\tau \in \mathcal{T}} \mathbb{E} \left[\exp \left(D(t) - \frac{1}{2} \langle D \rangle_t \right) \cdot \exp \left(M(t) - \frac{1}{2} \langle M \rangle_t \right) \right] \\ &\leq \exp \left(\sup_{s \in [0,T]} \left(\det_3(s) \right) \right) \mathbb{E} \left[\exp \left(D(0) - \frac{1}{2} \langle D \rangle_0 \right) \cdot \exp \left(M(0) - \frac{1}{2} \langle M \rangle_0 \right) \right] \\ &= \exp \left(\sup_{s \in [0,T]} \left(\det_3(s) \right) \right) < \infty. \end{split}$$

Thus, $(G(\tau, V^{v_0, \pi^*}(\tau), z(\tau)))_{\tau \in \mathcal{T}}$ is bounded in L^q for some q > 1 and therefore uniformly integrable. Hence, π^* is optimal for (**P**) as per Lemma 5.2.3.

Proof of Corollary 5.2.13. Noting that $\delta_K(\lambda) = \infty$ for all $x \neq 0$ if $K = \mathbb{R}$, the minimum in (5.6) is attained by $\lambda^*(B) = 0$ for all $B \in \mathbb{R}$ and the ODE (5.5) simplifies to

$$B'_{u}(\tau) = -r_{0} + r_{1}B_{u}(\tau) + \frac{1}{2}r_{2}(B_{u}(\tau))^{2}$$

with coefficients

$$r_0 = -\frac{b}{2(1-b)}\eta^2$$
, $r_1 = \frac{b}{1-b}\eta\sigma\rho - \kappa$, $r_2 = \sigma^2\left(1 + \frac{b}{1-b}\rho^2\right)$.

The expression for π_u can be directly read off from Lemma 5.2.3.

Proof of Lemma 5.2.15. Let $\rho = 0$. Then, according to Remark 5.2.6 and Corollary 5.2.13, both $\hat{\pi}^*(t) = \pi_M$ and $\pi_u(t) = \pi_M$ are constant in time $t \in [0, T]$. Thus,

$$\pi^*(t) = \operatorname{Cap}(\hat{\pi}(t), \alpha, \beta) = \operatorname{Cap}(\pi_M, \alpha, \beta) = \operatorname{Cap}(\pi_u(t), \alpha, \beta) \quad \forall t \in [0, T].$$

Consider now the case $\rho \neq 0$ and $\pi_M \in K = [\alpha, \beta]$. Then, $\alpha(1-b) \leq \eta \leq \beta(1-b)$ and thus $B_- \leq 0 \leq B_+$. Following Lemma 5.2.4 and Remark 5.2.6, $0 \in Z_0$ and the Riccati ODE (5.5) has coefficients r_0 , r_1 , and r_2 at $\tau = 0$. Hence, $B(\tau) = B_u(\tau)$ for all $\tau \in [0,T]$ such that $\rho B(\tau) \in Z_0$. Moreover, following Remark 5.2.5, $B_u(\tau)$ and $B(\tau)$ (respectively $\pi_u(t)$ and $\hat{\pi}^*(t)$) are solutions to autonomous ODEs and thus monotone functions in τ (respectively t). Noting that $\pi_u(t) \in K = [\alpha, \beta] \Leftrightarrow \rho B_u(T-t) \in Z_0$, we know if $\pi_u(t) \in K$, then

$$\pi_u(t) = \frac{1}{1-b} \left(\eta + \sigma \rho B_u(T-t) \right) = \frac{1}{1-b} \left(\eta + \sigma \rho B(T-t) \right) = \hat{\pi}(t).$$

Due to the monotonicity of π_u and $\hat{\pi}^*$, we have $\pi_u(t) \notin (\alpha, \beta)$ if and only if $\hat{\pi}(t) \notin (\alpha, \beta)$. Thus, we have in total

$$\pi^*(t) = \operatorname{Cap}\left(\hat{\pi}^*(t), \alpha, \beta\right) = \operatorname{Cap}\left(\pi_u(t), \alpha, \beta\right) \quad \forall t \in [0, T].$$

In particular, in both cases $\rho = 0$ and $\pi_M \in K$ we have $\mathcal{P}_K^H = \mathcal{P}_K^{BS}$.

Proof of Lemma 5.2.16. The implication (ii) \Rightarrow (i) is trivial, because if (ii) holds, then $\hat{\pi}^*(t) \neq \pi_u(t)$ and either $\hat{\pi}^*(t) \in (\alpha, \beta)$ or $\pi_u(t) \in (\alpha, \beta)$. Hence,

$$\pi^{*}(t) = \operatorname{Cap}\left(\hat{\pi}^{*}(t), \alpha, \beta\right) \neq \operatorname{Cap}\left(\pi_{u}(t), \alpha, \beta\right),$$

and therefore $\mathcal{P}_{K}^{H}(\pi_{u}, \alpha, \beta) \neq \mathcal{P}_{K}^{BS}(\pi_{u}, \alpha, \beta)$.

Assume now that (i) holds. According to Lemma 5.2.15, this implies $\rho \neq 0$ and $\pi_M \notin K = [\alpha, \beta]$. Moreover, as *B* and *B_u* are continuously differentiable,

$$\hat{\pi}^{*}(t) = \frac{1}{1-b} \left(\eta + \sigma \rho B(T-t) \right)$$
 and $\pi_{u}(t) = \frac{1}{1-b} \left(\eta + \sigma \rho B_{u}(T-t) \right)$

are continuously differentiable functions with $\hat{\pi}^*(T) = \pi_u(T) = \pi_M \notin K$. Thus, (i) can only be satisfied if there exists 0 < t < T such that

$$\{\hat{\pi}^*(t), \pi_u(t)\} \cap K \neq \emptyset, \tag{A.65}$$

as otherwise $\pi^*(t) = \operatorname{Cap}(\hat{\pi}^*(t), \alpha, \beta) = \operatorname{Cap}(\pi_M, \alpha, \beta) = \operatorname{Cap}(\pi_u(t), \alpha, \beta)$ for all $t \in [0, T]$. Thus, we define the latest such time point as

$$\hat{t} = \sup\left\{ 0 \le t \le T \mid \{\hat{\pi}^*(t), \pi_u(t)\} \cap K \neq \emptyset \right\}.$$

Since there exists a positive t > 0 satisfying (A.65), we know that $\hat{t} > 0$. Moreover, as $\pi_M \notin K$ and K is closed, we also know that $\hat{t} < T$. If $\hat{\pi}^*(\hat{t}) \neq \pi_u(\hat{t})$, then only one of these portfolio processes takes a value in K and (ii) holds for \hat{t} .

It thus remains to verify that $\hat{\pi}^*(\hat{t}) \neq \pi_u(\hat{t})$ to conclude the proof. This will be proven by contradiction. Assume that $\hat{\pi}^*(\hat{t}) = \pi_u(\hat{t}) \in K$ which is equivalent to $\rho B(T - \hat{t}) = \rho B_u(T - \hat{t}) \in Z_0$. Further,

$$T - \hat{t} = \inf \left\{ 0 \le \tau \le T \mid \{\rho B(\tau), \rho B_u(\tau)\} \cap Z_0 \neq \emptyset \right\}.$$

Moreover, as $\pi_M \notin K \stackrel{\rho \neq 0}{\Leftrightarrow} \rho B(0) = \rho B_u(0) = 0 \notin Z_0$, B and B_u are non-constant and thus strictly monotone functions. However, this implies that at $\tau = T - \hat{t}$, $\rho B(\tau)$ and $\rho B_u(\tau)$ have the same value at the boundary of Z_0 and satisfy the same ODE while taking values in Z_0 . Hence, $B(\tau) = B_u(\tau)$ for all $\tau \in [0, T]$ such that $B(\tau) \in Z_0$. Noting that

$$\rho B(\tau) \in Z_0 \Leftrightarrow \hat{\pi}^*(\tau) \in K \text{ and } \rho B_u(\tau) \in Z_0 \Leftrightarrow \pi_u(\tau) \in K,$$

finally yields for all $t \in [0, T]$:

$$\pi^*(t) = \operatorname{Cap}\left(\hat{\pi}^*(t), \alpha, \beta\right) = \operatorname{Cap}\left(\frac{1}{1-b}(\eta + \sigma \underbrace{\rho B(T-t)}_{=\rho B_u(T-t) \text{ while } \hat{\pi}^*(t) \in K}\right)$$
$$= \operatorname{Cap}\left(\frac{1}{1-b}(\eta + \sigma \rho B_u(T-t)), \alpha, \beta\right) = \operatorname{Cap}\left(\pi_u(t), \alpha, \beta\right),$$

which is a contradiction to (i). Thus, $\hat{\pi}^*(\hat{t}) \neq \pi_u(\hat{t})$.

Proof of Corollary 5.2.17.

Proof of (i): Let $\pi_M < \alpha$ and $0 < \alpha = 2\pi_M < \alpha \frac{2}{1-b}\eta$. Then, $\eta > 0$, $\pi_M > 0$ and $\eta = (1-b)\pi_M < (1-b)\alpha \Leftrightarrow B_- > 0$. Following Lemma 5.2.4 and Remark 5.2.6, $0 \in Z_-$ and the Riccati ODE (5.5) has coefficients r_0^- , r_1^- , and r_2^- at $\tau = 0$. However, due to the initial condition B(0) = 0, this implies

$$B'(0) = -r_0^- + r_1^- B(0) + \frac{1}{2}r_2^- B(0)^2 = -r_0^- = -\frac{1}{2}b\alpha\left((1-b)\alpha - 2\eta\right) = 0,$$

i.e., according to Remark 5.2.5, B is constant with $B(\tau) = 0$ for all $\tau \in [0, T]$. In particular, for all $t \in [0, T]$ we have $\hat{\pi}^*(t) = \pi_M$ and

$$\pi^*(t) = \operatorname{Cap}(\hat{\pi}^*(t), \alpha, \beta) = \operatorname{Cap}(\pi_M, \alpha, \beta) = \alpha.$$

On the other hand, the optimal unconstrained portfolio π_u does not violate the constraint at time t^* , i.e.,

$$\operatorname{Cap}(\pi_u(t^*), \alpha, \beta) = \pi_u(t^*) \in (\alpha, \beta),$$

and thus $\mathcal{P}_{K}^{H} \neq \mathcal{P}_{K}^{BS}$. Proof of (ii): According to Corollary 5.2.13,

$$\frac{\partial}{\partial t} \left(\pi_u(t) \right) = \frac{1}{1-b} \frac{\partial}{\partial t} \left(\eta + \sigma \rho B_u(T-t) \right)$$

Since B_u is the solution to the autonomous ODE (5.13), B_u is a monotone function (cf. Remark 5.2.5). Therefore,

$$\operatorname{sign}\left(\frac{\partial}{\partial t}\left(\pi_{u}(t)\right)\right) = \operatorname{sign}\left(\frac{-\sigma\rho}{1-b}B'_{u}(T-t)\right) = \operatorname{sign}\left(\frac{-\sigma\rho}{1-b}B'_{u}(0)\right) \stackrel{(5.13)}{=} \operatorname{sign}\left(\frac{\sigma\rho}{1-b}r_{0}\right)$$

r.		

$$\stackrel{(5.9)}{=} \operatorname{sign}\left(\frac{\sigma\rho}{1-b}\left(-\frac{b}{2(1-b)}\eta^2\right)\right)$$

As $\frac{\sigma \eta^2}{2(1-b)^2}$ is positive, this implies sign $\left(\frac{\partial}{\partial t}\left(\pi_u(t)\right)\right) = -\text{sign}\left(b\rho\right)$. As $\beta > 0$ and $\pi_M > \beta \Leftrightarrow 0 = \rho B(0) > B_+$, we follow the same line of argument for $\hat{\pi}^*$ to obtain

$$\operatorname{sign}\left(\frac{\partial}{\partial t}\left(\hat{\pi}^{*}(t)\right)\right) = \operatorname{sign}\left(\frac{-\sigma\rho}{1-b}B'(0)\right) = \operatorname{sign}\left(\frac{\sigma\rho}{1-b}r_{0}^{+}\right)$$
$$= \operatorname{sign}\left(\frac{\sigma\rho}{1-b}\frac{1}{2}b\beta\left([1-b]\beta - 2\eta\right)\right) = \operatorname{sign}\left(\frac{\sigma\rho b\beta}{2}\left(\beta - 2\pi_{M}\right)\right)$$

Disregarding the positive factor $\frac{\sigma\beta}{2}$ finally yields

$$\operatorname{sign}\left(\frac{\partial}{\partial t}\left(\hat{\pi}^{*}(t)\right)\right) = \operatorname{sign}\left(\rho b \underbrace{\left(\beta - 2\pi_{M}\right)}_{<0}\right) = -\operatorname{sign}\left(b\rho\right) = \operatorname{sign}\left(\frac{\partial}{\partial t}\left(\pi_{u}(t)\right)\right).$$

If now $\rho < 0$ and b < 0, then both portfolio allocation $\pi^*(t) = \operatorname{Cap}(\hat{\pi}^*(t), \alpha, \beta)$ and $\operatorname{Cap}(\pi_u(t), \alpha, \beta)$ are non-increasing in time. In particular, as both allocations are equal to $\beta < \pi_M$ at t = T, they must be equal (to β) throughout the entire investment horizon. Hence, the projections \mathcal{P}_K^H and \mathcal{P}_K^{BS} coincide under these assumptions.

Proof of Theorem 5.3.1. We transform the portfolio optimization problem (**P**) into an equivalent optimization problem by applying the change of control $\pi_A(t) := A'\pi(t)$ for any given $\pi \in \Lambda^{PCSV}$. Expressing the SDE of the wealth process $V^{v_0,\pi}$ in terms of π_A yields

$$dV^{v_0,\pi}(t) = V^{v_0,\pi}(t)V^{v_0,\pi}(t) \left[\left(r + \underbrace{\eta'A}_{=:\eta'_A} \operatorname{diag}(z(t)) \underbrace{A'\pi(t)}_{=\pi_A(t)} \right) dt + \underbrace{\pi(t)'A}_{=\pi_A(t)'} \operatorname{diag}(\sqrt{z(t)}) dW(t) \right]$$

= $V^{v_0,\pi}(t) \left[\left(r + \eta'_A \operatorname{diag}(z(t))\pi_A(t) \right) dt + \pi_A(t)' \operatorname{diag}(\sqrt{z(t)}) dW(t) \right]$ (A.66)

and $\pi(t) \in K_{PCSV} \Leftrightarrow \pi_A(t) \in \times_{i=1}^d [0, \sqrt{\beta_i}]$. In particular, we may equivalently rewrite the portfolio optimization problem in terms of π_A as

$$(\mathbf{P}_{\mathbf{A}}) \begin{cases} \Phi(v_0) = \sup_{\pi \in \Lambda_A} \mathbb{E}\left[U(V^{v_0,\pi}(T))\right] \\ \Lambda_A = \left\{\pi_A \in \Lambda_{PCSV} \mid \pi_A(t) \in \times_{i=1}^d [0,\sqrt{\beta_i}] \mathcal{L}[0,T] \otimes Q - \text{a.e.} \right\} \end{cases}$$

We proceed by solving $(\mathbf{P}_{\mathbf{A}})$ and then inverting the change of control to obtain a solution for the original optimization problem (\mathbf{P}) . As $K_A = \times_{i=1}^d [0, \sqrt{\beta_i}]$ is a d-dimensional interval, $(\mathbf{P}_{\mathbf{A}})$ fits into the setting of the financial market \mathcal{M}_{CIR} in Definition 4.4.5 with m = d and $\Sigma_i = 1$ for i = 1, ..., d. In particular, according to Lemma 4.4.6, if $A : [0, T] \to \mathbb{R}, B : [0, T] \to \mathbb{R}^d$ with $A(0) = B_i(0) = 0$ for i = 1, ..., d satisfy

$$A'(\tau) = br + \sum_{i=1}^{d} \kappa_{i}\theta_{i}$$

$$B'_{i}(\tau) = -\kappa_{i}B_{i}(\tau) + \frac{1}{2}\sigma_{i} (B_{i}(\tau))^{2}$$

$$+ \frac{1}{2}\frac{b}{1-b} \inf_{\lambda \in \mathbb{R}} \left\{ 2(1-b)\delta_{[0,\sqrt{\beta_{i}}]}(\lambda) + ((\eta_{A})_{i} + \lambda_{i} + \sigma_{i}\rho_{i}B_{i}(\tau))^{2} \right\}, \quad (A.67)$$

then the function $G(t, v, z) := \frac{1}{b}v^b \exp(A(T-t) + B(T-t)'z)$ is a solution to the HJB PDE associated with ($\mathbf{P}_{\mathbf{A}}$). However, for each i = 1, ..., d, equation (A.67) contains the same ODE and optimization that were considered in Lemma 5.2.4 and Section 5.2. Since Assumption 5.2.7 is satisfied, the solution B_i to (A.67) is given by Theorem 5.2.8. Finally, the candidate optimal portfolio π_A^* for (\mathbf{P}) is given by

$$(\pi_A^*)_i(t) = \operatorname{Cap}\left(\frac{1}{b}\left((\eta_A)_i + \sigma_i\rho_i B_i(T-t)\right), 0, \sqrt{\beta_i}\right).$$

We still need to formally verify the optimality of π_A^* for (\mathbf{P}_A) . For this purpose, we define the sequence of stopping times $\tau_{n,t}$ as $\tau_{n,t} = \min(T, \hat{\tau}_{n,t})$, with

$$\hat{\tau}_{n,t} = \inf \left\{ t \le u \le T \mid \int_t^u \left\| b\left(\sqrt{z(s)} \odot \pi_A(s)\right) G(s, V^{v_0, \pi^*}(s), z(s)) \right\|^2 ds \ge n, \\ \int_t^u \left\| \left(\sigma \odot \sqrt{z(s)} \odot B(T-s) \right) G(s, V^{v_0, \pi^*}(s), z(s)) \right\|^2 ds \ge n \right\}.$$

According to Theorem 3.12 in [33], the optimality of π_A^* is verified if

$$\left(G\left(\tau_{n,t}, V^{v_0, \pi^*}(\tau_{n,t}), z(\tau_{n,t})\right)\right)_{n \in \mathbb{N}}$$

is uniformly integrable for every $t \in [0, T]$. However, as Assumptions 5.2.7 and 5.2.9 are satisfied for every i = 1, ..., d, following the same steps as in the proof of Theorem 5.2.12, we can show that there exists a constant q > 1 and a bounded, deterministic function det(t) such that the local martingales

$$D_i(t) = bq\sqrt{1 - \rho_i^2} \int_0^t (\pi_A)_i^*(s)\sqrt{z_i(s)}d\hat{W}_i(s)$$
$$M_i(t) = \frac{q\kappa_i}{\sigma_i} \int_0^t \sqrt{z_i(s)}dW_i^z(s)$$

can be used to bound $|G(t, V^{v_0,\pi^*}(t), z(t))|^q$ for every $t \in [0, T]$ through

$$\left| G\left(t, V^{v_0, \pi^*}(t), z(t)\right) \right|^q \le \exp\left(\sup_{s \in [0, T]} det(s)\right) \underbrace{\exp\left(\sum_{i=1}^d D_i(t) - \frac{1}{2} \langle D_i \rangle_t + M_i(t) - \frac{1}{2} \langle M_i \rangle_t\right)}_{(*)}.$$

The diffusions of the local martingales D_i , M_i are independent and therefore (*) is a supermartingale. Since the above bound holds pointwise for every $t \in [0, T]$, Doob's optional sampling theorem yields the L^q -boundedness (and thereby uniform integrability) of

$$\left(G(\tau_{n,t}^{0}, V^{v_{0}, \pi^{*}}(\tau_{n,t}^{0}), z(\tau_{n,t}^{0}))\right)_{n \in \mathbb{N}}$$

Hence, π_A^* is optimal for (\mathbf{P}_A) and thus $\pi^*(t) = A\pi_A^*(t)$ is optimal for (\mathbf{P}) in \mathcal{M}_{PCSV} .

Proof of Theorem 5.3.2. The proof follows similar arguments as the proof of Corollary 5.4 in [56]. For given $\pi \in \Lambda^{\gamma, \Sigma}$, let $\hat{\pi}_{BS}$ and $\hat{\pi}_H$ be defined through

$$\hat{\pi}_{BS}(t) := \Sigma(z(t))\pi(t) \quad \text{or} \quad \hat{\pi}_H(t) := \frac{\Sigma(z(t))}{\sqrt{z(t)}}\pi(t).$$
 (A.68)

Proof of (i): By construction, $\pi \in \Lambda_{K(\cdot)}$ if and only if $\hat{\pi}_{BS}(t) \in \hat{K} := [\alpha, \beta] \mathcal{L}[0, T] \otimes Q$ -a.e., i.e., the constraint on $\hat{\pi}_{BS}$ is constant. Under the present assumptions, the wealth process satisfies

$$dV^{v_0,\pi}(t) = V^{v_0,\pi}(t) \left([r + \eta \Sigma(z(t))\pi(t)] dt + \pi(t)\Sigma(z(t)) dW(t) \right)$$

= $V^{v_0,\pi}(t) \left([r + \eta \hat{\pi}_{BS}(t)] dt + \hat{\pi}_{BS}(t) dW(t) \right).$

Hence, maximizing $\mathbb{E}\left[U\left(V^{v_0,\pi}(T)\right)\right]$ over $\hat{\pi}_{BS}$ subject to the constraint $\hat{\pi}_{BS}(t) \in \hat{K} = [\alpha, \beta]$ $\mathcal{L}[0,T] \otimes Q$ -a.e. is equivalent to the constrained portfolio optimization problem in a Black-Scholes market \mathcal{M}_{BS} with market price of risk η and volatility 1. Therefore, as discussed in Section 5.2.3, the constant-mix strategy

$$\hat{\pi}_{BS}^* = \operatorname{Cap}(\pi_M, \alpha, \beta) \Leftrightarrow \pi^*(t) = \Sigma(z(t))^{-1} \operatorname{Cap}(\pi_M, \alpha, \beta)$$

maximises the expected utility over all admissible $\hat{\pi}_{BS}$. Inverting the change of control through multiplication by $\Sigma(z(t))^{-1}$ yields the claim.

Proof of (ii): By construction, $\pi \in \Lambda_{K(\cdot)}$ if and only if $\hat{\pi}_H(t) \in \hat{K} := [\alpha, \beta] \mathcal{L}[0, T] \otimes Q$ -a.e., i.e., the constraint on $\hat{\pi}_H$ is constant. Under the present assumptions, the wealth process satisfies

$$dV^{v_0,\pi}(t) = V^{v_0,\pi}(t) \left([r + \eta \sqrt{z(t)} \Sigma(z(t)) \pi(t)] dt + \pi(t) \Sigma(z(t)) dW(t) \right)$$

= $V^{v_0,\pi}(t) \left([r + \eta z(t) \hat{\pi}_H(t)] dt + \hat{\pi}_H(t) \sqrt{z(t)} dW(t) \right).$

Hence, maximizing $\mathbb{E}\left[U\left(V^{v_0,\pi}(T)\right)\right]$ over $\hat{\pi}_H$ subject to the constraint $\hat{\pi}_H(t) \in \hat{K} = [\alpha,\beta]$ $\mathcal{L}[0,T] \otimes Q$ -a.e. is equivalent to the constrained portfolio optimization problem in the Heston market \mathcal{M}_H , as considered in Section 5.2. As all requirements of Theorem 5.2.12 are satisfied, π^* (as defined in Theorem 5.2.12) maximises the expected utility over all admissible $\hat{\pi}_H$. Inverting the change of control through multiplication by $\sqrt{z(t)}\Sigma(z(t))^{-1}$ yields the claim. \Box

Proof of Lemma 5.4.1. We first verify that $G(t, v, z) = \frac{1}{b}v^b \exp(A_{\pi}(T-t) + B_{\pi}(T-t)z)$ is a solution to the Feynman-Kac PDE (5.20). The partial derivatives of G can be computed as

$$G_{t}(t,v,z) = -\left(A'_{\pi}(T-t) + B'_{\pi}(T-t)z\right)G(t,v,z), \quad G_{v}(t,v,z) = \frac{b}{v}G(t,v,z),$$

$$G_{vv}(t,v,z) = \frac{b(b-1)}{v^{2}}G(t,v,z), \quad G_{z}(t,v,z) = B_{\pi}(T-t)G(t,v,z),$$

$$G_{zz}(t,v,z) = (B_{\pi}(T-t))^{2}G(t,v,z), \quad \text{and} \quad G_{zv}(t,v,z) = \frac{bB_{\pi}(T-t)}{v}G(t,v,z).$$

Substituting these derivatives in (5.20), while omitting the arguments (t, v, z), yields

$$0 = G_t + (r + \eta \pi(t)z) vG_v + \kappa (\theta - z) G_z + \sigma \rho \pi(t) z vG_{zv} + \frac{1}{2} v^2 \pi(t)^2 z G_{vv} + \frac{1}{2} \sigma^2 z G_{zz}$$

$$= G \Big[- (A'_{\pi} + B'_{\pi}z) + (r + \eta \pi(t)z) b + \kappa (\theta - z) B_{\pi} + \sigma \rho \pi(t) b z B_{\pi} + \frac{1}{2} \pi(t)^2 z b (b - 1) + \frac{1}{2} \sigma^2 z B_{\pi}^2 \Big]$$

$$G \stackrel{G \neq 0}{\Leftrightarrow} \quad 0 = - (A'_{\pi} + B'_{\pi}z) + (r + \eta \pi(t)z) b + \kappa (\theta - z) B_{\pi} + \sigma \rho \pi(t) b z B_{\pi} + \frac{1}{2} \pi(t)^2 z b (b - 1) + \frac{1}{2} \sigma^2 z B_{\pi}^2$$

$$= \underbrace{-A'_{\pi} + rb + \kappa \theta B_{\pi}}_{=0 \text{ by } (5.18)} + \underbrace{\left[-B'_{\pi} + b \eta \pi(t) - \frac{1}{2} \pi(t)^2 b (1 - b) + (\sigma \rho b \pi(t) - \kappa) B_{\pi} + \frac{\sigma^2}{2} B_{\pi}^2 \right]}_{=0 \text{ by } (5.19)}$$

Moreover, G satisfies the boundary condition G(T, v, z) = U(v) because $A_{\pi}(0) = B_{\pi}(0) = 0$. Hence, G is a solution to the Feynman-Kac PDE (5.20). By assumption, such a solution is unique, and therefore

$$\frac{1}{b}v^{b}\exp\left(A_{\pi}(T-t) + B_{\pi}(T-t)z\right) = G(t,v,z) = J^{\pi}(t,v,z) \quad \forall (t,v,z) \in [0,T] \times (0,\infty) \times (0,\infty).$$

Proof of Corollary 5.4.2. $L^{\pi}(t, z)$ is defined as the solution to the equation

$$\begin{split} \Phi(t, v(1 - L^{\pi}(t, z)), z) &= \mathbb{E}\left[U\left(V^{v_0, \pi^*}(T)\right) \mid V^{v_0, \pi}(t) = v(1 - L^{\pi}(t, z)), \ z(t) = z\right] \\ &= \mathbb{E}\left[U\left(V^{v_0, \pi}(T)\right) \mid V^{v_0, \pi}(t) = v, \ z(t) = z\right] \\ &= J^{\pi}(t, v, z) \end{split}$$

From Lemma 5.2.3 we know that

$$\Phi(t, v(1 - L^{\pi}(t, z)), z) = \frac{1}{b} (v(1 - L^{\pi}(t, z)))^{b} \exp(A(T - t) + B(T - t)z).$$

Similarly, we have by Lemma 5.4.1

$$J^{\pi}(t, v, z) = \frac{1}{b} v^{b} \exp\left(A_{\pi}(T - t) + B_{\pi}(T - t)z\right).$$

Thus,

$$\Phi(t, v(1 - L^{\pi}(t, z)), z) = J^{\pi}(t, v, z)$$

$$\Rightarrow \frac{1}{b} (v(1 - L^{\pi}(t, z)))^{b} \exp(A(T - t) + B(T - t)z) = \frac{1}{b} v^{b} \exp(A_{\pi}(T - t) + B_{\pi}(T - t)z)$$

$$\Rightarrow \left(\frac{v(1 - L^{\pi}(t, z))}{v}\right)^{b} = \exp(A_{\pi}(T - t) - A(T - t) + [B_{\pi}(T - t) - B(T - t)]z)$$

$$\Rightarrow L^{\pi}(t, z) = 1 - \exp\left(\frac{1}{b} (A_{\pi}(T - t) - A(T - t) + [B_{\pi}(T - t) - B(T - t)]z)\right).$$

- Tom M. Apostol. Mathematical Analysis: A Modern Approach to Advanced Calculus. Addison-Wesley Publishing Company, Inc., 1957.
- Philippe Artzner et al. 'Coherent Measures of Risk'. In: Mathematical Finance 9.3 (1999), pp. 203–228.
- [3] Indrajit Bardhan. 'Consumption and investment under constraints'. In: Journal of Economic Dynamics and Control 18.5 (1994), pp. 909–929.
- [4] Suleyman Basak and Alexander Shapiro. 'Value-at-Risk-Based Risk Management: Optimal Policies and Asset Prices'. In: *The Review of Financial Studies* 14.2 (2001), pp. 371– 405.
- [5] Nicole Bäuerle and Zejing Li. 'Optimal Portfolios for Financial Markets with Wishart Volatility'. In: *Journal of Applied Probability* 50.4 (2013), pp. 1025–1043.
- [6] Baojun Bian, Sheng Miao and Harry Zheng. 'Smooth Value Functions for a Class of Nonsmooth Utility Maximization Problems'. In: SIAM Journal on Financial Mathematics 2.1 (2011), pp. 727–747.
- [7] Thomas Björk, Mriana Khapko and Agatha Murgoci. 'On time-inconsistent stochastic control in continuous time'. In: *Finance and Stochastics* 21 (2017), pp. 331–360.
- [8] Vladimir I. Bogachev. *Measure Theory*. Springer Berlin, Heidelberg, 2007.
- [9] Nicole Branger, Christian Schlag and Eva Schneider. 'Optimal Portfolios When Volatility Can Jump'. In: Journal of Banking & Finance 32.6 (2008), pp. 1087–1097.
- [10] An Chen, Thai Nguyen and Manuel Rach. 'A Collective Investment Problem in a Stochastic Volatility Environment: The Impact of Sharing Rules'. In: Annals of Operations Research 302.1 (2021), pp. 85–109.
- [11] An Chen, Thai Nguyen and Mitja Stadje. 'Risk management with multiple VaR constraints'. In: *Mathematical Methods of Operations Research* 88.2 (2013), pp. 297–337.
- [12] An Chen, Mitja Stadje and Fangyuan Zhang. On the equivalence between Value-at-Riskand Expected Shortfall-based risk measures in non-concave optimization. 2022. arXiv: 2002.02229 [math.OC].
- [13] Yuyang Cheng and Marcos Escobar-Anel. 'Optimal Investment Strategy in the Family of 4/2 Stochastic Volatility Models'. In: *Quantitative Finance* 21.10 (2021), pp. 1723–1751.
- [14] Rama Cont. 'Empirical Properties of Asset Returns: Stylized Facts and Statistical Issues'. In: *Quantitative Finance* 1.2 (2001), pp. 223–236.
- [15] John C. Cox and Chi-fu Huang. 'Optimal consumption and portfolio policies when asset prices follow a diffusion process'. In: *Journal of Economic Theory* 49.1 (1989), pp. 33–83.
- [16] Domenico Cuoco. 'Optimal Consumption and Equilibrium Prices with Portfolio Constraints and Stochastic Income'. In: *Journal of Economic Theory* 72.1 (1997), pp. 33– 73.
- [17] Jaksa Cvitanic and Ioannis Karatzas. 'Convex Duality in Constrained Portfolio Optimization'. In: Annals of Applied Probability 2.4 (1992), pp. 767–818.

- [18] Jaksa Cvitanic and Ioannis Karatzas. 'Hedging Contingent Claims with Constrained Portfolios'. In: Annals of Applied Probability 3.3 (1993), pp. 652–681.
- [19] Ashley Davey and Harry Zheng. 'Deep Learning for Constrained Utility Maximisation'. In: Methodology and Computing in Applied Probability 24 (2022), pp. 661–692.
- [20] Griselda Deelstra, Martino Grasselli and Pierre-François Koehl. 'Optimal Investment Strategies in a CIR Framework'. In: Journal of Applied Probability 37.4 (2000), pp. 936– 946.
- [21] Jérôme Detemple and Marcel Rindisbacher. 'Closed-Form Solutions for Optimal Portfolio Selection with Stochastic Interest Rate and Investment Constraints'. In: *Mathematical Finance* 15.4 (2005), pp. 539–568.
- [22] Emmanuele DiBenedetto. Partial Differential Equations. Birkhäuser Basel, 2009.
- [23] Yinghui Dong and Harry Zheng. 'Optimal Investment of DC Pension Plan under Short-Selling Constraints and Portfolio Insurance'. In: *Insurance: Mathematics and Economics* 85 (2019), pp. 47–59.
- [24] Yinghui Dong and Harry Zheng. 'Optimal Investment with S-shaped Utility and Trading and Value at Risk Constraints: An Application to Defined Contribution Pension Plan'. In: European Journal of Operational Research 281.2 (2020), pp. 341–356.
- [25] Catherine Donnelly et al. 'Less is more: Increasing retirement gains by using an upside terminal wealth constraint'. In: *Insurance: Mathematics and Economics* 64 (2015), pp. 259– 267.
- [26] Darrell Duffie and Rui Kan. 'A Yield-Factor Model of Interest Rates'. In: Mathematical Finance 6.4 (1996), pp. 379–406.
- [27] Rick Durrett. Probability: Theory and Examples. 5th Edition. Cambridge University Press, 2019.
- [28] Daniel Egloff, Markus Leippold and Liuren Wu. 'The Term Structure of Variance Swap Rates and Optimal Variance Swap Investments'. In: *The Journal of Financial and Quantitative Analysis* 45.5 (2010), pp. 1279–1310.
- [29] Ivar Ekeland and Roger Témam. *Convex Analysis and Variational Problems*. Society for Industrial and Applied Mathematics, 1999.
- [30] Marcos Escobar-Anel, Sebastian Ferrando and Alexey Rubtsov. 'Optimal Investment under Multi-Factor Stochastic Volatility'. In: *Quantitative Finance* 17.2 (2017), pp. 241– 260.
- [31] Marcos Escobar-Anel and Christoph Gschnaidtner. 'Parameters Recovery via Calibration in the Heston Model: A Comprehensive Review'. In: *Wilmott* 2016.86 (2016), pp. 60–81.
- [32] Marcos Escobar-Anel, Michel Kschonnek and Rudi Zagst. Mind the Cap! Constrained Portfolio Optimisation in Heston's Stochastic Volatility Model. 2023. arXiv: 2306.11158 [q-fin.PM].
- [33] Marcos Escobar-Anel, Michel Kschonnek and Rudi Zagst. Portfolio Optimization with Allocation Constraints and Stochastic Factor Market Dynamics. 2023. arXiv: 2303.09835 [q-fin.PM].
- [34] Marcos Escobar-Anel, Michel Kschonnek and Rudi Zagst. 'Portfolio Optimization: Not Necessarily Concave Utility and Constraints on Wealth and Allocation'. In: *Mathematical Methods of Operations Research* 95 (2022), pp. 101–140.
- [35] Marcos Escobar-Anel et al. 'Decrease of capital guarantees in life insurance products: Can reinsurance stop it?' In: *Insurance: Mathematics and Economics* 105 (2022), pp. 14–40.

- [36] Marcos Escobar-Anel et al. 'Portfolio optimization under Solvency II'. In: Annals of Operations Research 281.1 (2019), pp. 193–227.
- [37] Marcos Escobar-Anel et al. 'Pricing a CDO on stochastically correlated underlyings'. In: *Quantitative Finance* 10.3 (2010), pp. 265–277.
- [38] Damir Filipovic. Term-Structure Models. Springer Berlin, Heidelberg, 2009.
- [39] Wendell H. Fleming and Halil Mete Soner. Controlled Markov Processes and Viscosity Solutions. Springer, 2006.
- [40] Helyette Geman et al. 'Stochastic Volatility for Levy Processes'. In: Mathematical Finance 13.3 (2003), pp. 345–382.
- [41] Ilya Gikhman. 'A Short Remark on Feller's Square Root Condition'. In: SSRN Electronic Journal (2011).
- [42] Steven Heston. 'A Closed-Form Solution for Options with Stochastic Volatility with Applications to Bond and Currency Options'. In: *Review of Financial Studies* 6 (1993), pp. 327–343.
- [43] Jean-Baptiste Hiriart-Urruty and Claude Lemaréchal. Fundamentals of Convex Analysis. Springer Berlin, Heidelberg, 2001.
- [44] John Hull. 'VAR versus expected shortfall'. In: Risk (1st Mar. 2007). URL: https:// www.risk.net/risk-magazine/technical-paper/1506669/var-versus-expectedshortfall (visited on 26/04/2023).
- [45] Rufus Isaacs. *Differential Games*. Dover Publications, 1999.
- [46] Jan Kallsen and Johannes Muhle-Karbe. 'Utility Maximization in Affine Stochastic Volatility Models'. In: International Journal of Theoretical and Applied Finance 13.03 (2010), pp. 459–477.
- [47] Ioannis Karatzas, John P. Lehoczky and Steven E. Shreve. 'Optimal Portfolio and Consumption Decisions for a "Small Investor" on a Finite Horizon'. In: SIAM Journal on Control and Optimization 25.6 (1987), pp. 1557–1586.
- [48] Ioannis Karatzas and Steve Shreve. Brownian Motion and Stochastic Calculus. Springer New York, 1998.
- [49] Ioannis Karatzas et al. 'Martingale and Duality Methods for Utility Maximization in an Incomplete Market'. In: SIAM Journal on Control and Optimization 29.3 (1991), pp. 702– 730.
- [50] Ralph S.J. Koijen. 'The Cross-Section of Managerial Ability, Incentives, and Risk Preferences'. In: The Journal of Finance 69.3 (2014), pp. 1051–1098.
- [51] Ralf Korn. Moderne Finanzmathematik Theorie und praktische Anwendung. Springer Spektrum Wiesbaden, 2014.
- [52] Ralf Korn. 'Optimal portfolios with a positive lower bound on final wealth'. In: *Quantitative Finance* 5.3 (2005), pp. 315–321.
- [53] Ralf Korn and Holger Kraft. 'A Stochastic Control Approach to Portfolio Problems with Stochastic Interest Rates'. In: SIAM Journal on Control and Optimization 40.4 (2002), pp. 1250–1269.
- [54] Ralf Korn and Holger Kraft. 'On the Stability of Continuous-Time Portfolio Problems with Stochastic Opportunity Set'. In: *Mathematical Finance* 14.3 (2004), pp. 403–414.
- [55] Ralf Korn and Siegfried Trautmann. 'Continuous-time portfolio optimization under terminal wealth constraints'. In: Zeitschrift für Operations Research 42 (1995), pp. 69–92.

- [56] Holger Kraft. 'Optimal portfolios and Heston's Stochastic Volatility Model: An Explicit Solution for Power Utility'. In: *Quantitative Finance* 5.3 (2005), pp. 303–313.
- [57] Holger Kraft and Mogens Steffensen. 'A dynamic programming approach to constrained portfolios'. In: *European Journal of Operational Research* 229.2 (2013), pp. 453–461.
- [58] Michel Kschonnek et al. 'COVIX–An Index Allowing for the Assessment of the Pandemic Situation Based on Infections and Hospitalisation Data'. In: *Applied Sciences* 13.7 (2023).
- [59] Yue-Kuen Kwok. *Mathematical Models of Financial Derivatives*. Springer Berlin, Heidelberg, 2008.
- [60] Carl Lindberg. 'News-Generated Dependence and Optimal Portfolios for n Stocks in a Market of Barndorff-Nielsen and Shephard Type'. In: *Mathematical Finance* 16.3 (2006), pp. 549–568.
- [61] Jun Liu. 'Portfolio Selection in Stochastic Environments'. In: The Review of Financial Studies 20.1 (2006), pp. 1–39.
- [62] Jun Liu and Jun Pan. 'Dynamic Derivative Strategies'. In: Journal of Financial Economics 69.3 (2003), pp. 401–430.
- [63] Thomas Lux. 'CHAPTER 3 Stochastic Behavioral Asset-Pricing Models and the Stylized Facts'. In: Handbook of Financial Markets: Dynamics and Evolution. Handbooks in Finance. North-Holland, 2009, pp. 161–215.
- [64] Robert Merton. 'Lifetime Portfolio Selection under Uncertainty: The Continuous-Time Case'. In: *The Review of Economics and Statistics* 51 (1969), pp. 247–57.
- [65] Robert Merton. 'Optimum consumption and portfolio rules in a continuous-time model'. In: Journal of Economic Theory 4.3 (1971), pp. 373–413.
- [66] Mohamed Mnif. 'Portfolio Optimization with Stochastic Volatilities and Constraints: An Application in High Dimension'. In: Applied Mathematics and Optimization 56.2 (2007), pp. 243–264.
- [67] Santiago Moreno-Bromberg, Traian A. Pirvu and Anthony Reveillac. 'CRRA utility maximization under dynamic risk constraints'. In: *Communications on Stochastic Analysis* 07.02 (2013), pp. 179–198.
- [68] Jan Mossin. 'Optimal Multiperiod Portfolio Policies'. In: The Journal of Business 41.2 (1968), pp. 215–229.
- [69] Thibaut Moyaert and Mikael Petitjean. 'The Performance of Popular Stochastic Volatility Option Pricing Models During the Subprime Crisis'. In: Applied Financial Economics 21.14 (2011), pp. 1059–1068.
- [70] Marcel Nutz. 'Power Utility Maximization in Constrained Exponential Lévy Models'. In: Mathematical Finance 22.4 (2012), pp. 690–709.
- [71] Huyên Pham. 'Smooth Solutions to Optimal Investment Models with Stochastic Volatilities and Portfolio Constraints'. In: Applied Mathematics & Optimization 46 (2002), pp. 55– 78.
- [72] Traian A. Pirvu. 'Portfolio optimization under the Value-at-Risk constraint'. In: *Quant-itative Finance* 7.2 (2007), pp. 125–136.
- [73] Stanley R. Pliska. 'A Stochastic Calculus Model of Continuous Trading: Optimal Portfolios'. In: Mathematics of Operations Research 11.2 (1986), pp. 371–382.
- [74] Philip E. Protter. Stochastic Integration and Differential Equations. Springer Berlin Heidelberg New York, 2005.

- [75] Q.ai. 'The Average Bear Market Lasts 289 Days. How Long Do We Have Left?' In: Forbes (24th Aug. 2022). URL: https://www.forbes.com/sites/qai/2022/08/24/theaverage-bear-market-lasts-289-days-how-long-do-we-have-left/ (visited on 01/02/2023).
- [76] Ralph T. Rockafellar. *Conjugate Duality and Optimization*. Society for Industrial and Applied Mathematics, 1974.
- [77] Ralph T. Rockafellar and Roger J. -B. Wets. 'Variational systems, an introduction'. In: Multifunctions and Integrands. Springer Berlin Heidelberg, 1984, pp. 1–54.
- [78] Marc Romano and Nizar Touzi. 'Contingent Claims and Market Completeness in a Stochastic Volatility Model'. In: *Mathematical Finance* 7.4 (1997), pp. 399–412.
- [79] Paul A. Samuelson. 'Lifetime Portfolio Selection By Dynamic Stochastic Programming'. In: The Review of Economics and Statistics 51.3 (1969), pp. 239–246.
- [80] Carsten Sørensen. 'Dynamic Asset Allocation and Fixed Income Management'. In: *The Journal of Financial and Quantitative Analysis* 34.4 (1999), pp. 513–531.
- [81] Stephen J. Taylor. 'Modeling Stochastic Volatility: A Review and Comparative Study'. In: Mathematical Finance 4.2 (1994), pp. 183–204.
- [82] Lucie Tepla. 'Optimal Investment with Minimum Performance Constraints'. In: Journal of Economic Dynamics and Control 25 (2001), pp. 1629–1645.
- [83] Wolfgang Walter. Ordinary Differential Equations. Springer New York, NY, 1998.
- [84] James Owen Weatherall. 'The Peculiar Logic of the Black-Scholes Model'. In: Philosophy of Science 85.5 (2018), pp. 1152–1163.
- [85] Gan-Lin Xu. 'A duality method for optimal consumption and investment under shortselling prohibition'. PhD thesis. Carnegie Mellon University, 1990.
- [86] Yasuhiro Yamai and Toshinao Yoshiba. 'Value-at-risk versus expected shortfall: A practical perspective'. In: *Journal of Banking & Finance* 29.4 (2005). Risk Measurement, pp. 997–1015.
- [87] Rudi Zagst, Thomas Meyer and Hendrik Hagedorn. 'Integrated Modelling of Stock and Bond Markets'. In: International Journal of Finance and Accounting 19.1 (2007), pp. 4252– 4277.
- [88] Thaleia Zariphopoulou. 'A Solution Approach to Valuation with Unhedgeable Risks'. In: *Finance and Stochastics* 5.1 (2001), pp. 61–82.
- [89] Thaleia Zariphopoulou. 'Consumption-Investment Models with Constraints'. In: SIAM Journal on Control and Optimization 32.1 (1994), pp. 59–85.
- [90] Thaleia Zariphopoulou. 'Optimal asset allocation in a stochastic factor model an overview and open problems'. In: Advanced Financial Modelling. De Gruyter, 2009, pp. 427– 456.
- [91] Amina Bouzguenda Zeghal and Mohamed Mnif. 'Portfolio optimization with stochastic volatilities: A backward approach'. In: *Stochastic analysis and applications* 29.5 (2011), pp. 723–748.
- [92] Yichen Zhu and Marcos Escobar-Anel. 'Polynomial affine approach to HARA utility maximization with applications to Ornstein-Uhlenbeck 4/2 models.' In: *Applied Mathematics* and Computation 418 (2022), p. 126836.